# **POROUS ELASTICITY**

## Lectures On The Elasticity Of Porous Materials As An Application Of The Theory Of Mixtures

**Ray M. Bowen** Professor Emeritus of Mechanical Engineering President Emeritus Texas A&M University

Copyright Ray M. Bowen

## Preface

This work was originally planned as a textbook exploiting the structure of the Theory of Mixtures as the basis for the study of porous elasticity. The decision to write this book was made approximately thirty years ago! At that time, I was a faculty member in Mechanical Engineering at Rice University. It is an understatement to observe that it has taken awhile to complete, even partially, this project. I encountered a lot of diversions along the way. Not the least of which was an eight year period where I served as President of Texas A&M University. Prior to that time, I was a Dean of Engineering at Kentucky, an administrator at the National Science Foundation and a Provost and Interim President at Oklahoma State University. During my time as an administrator, I never lost my ambition to prepare this textbook. On occasion, during periods of relative calm or, at the other extreme, during periods of great stress, I would find comfort in returning to my manuscript. It would take someone not trained in Engineering to understand why I would find comfort thinking about this book when caught up in the tangles of university administration.

When I completed my term as President of Texas A&M University, I decided that I should spend the final years of my career back in the classroom teaching Mechanical Engineering and Mathematics. I have experienced the good fortune of two great and generous departments at Texas A&M willing to allow me to teach a variety of subjects in my areas of interests. It has been a joyous experience.

In addition to my teaching, I have revised my books on Vectors and Tensors, written with C.-C. Wang, and my Continuum Mechanics textbook. These books, of course, are out of print. This fact enabled me to arrange for their copyrights to be returned to the authors and to republish them informally by posting them on the web.<sup>1</sup> The reception to these free textbooks has encouraged me to complete, in some fashion, my textbook on porous elasticity. With this encouragement, I returned to a manuscript that has its roots in courses I taught at Rice University in the late 1970's.

One has to wonder if a book written over a thirty year period can make a contribution to knowledge in a subject studied by a large number of talented and knowledgeable investigators in the intervening years. I am not confident that I have made a contribution. The decision on that subject will need to be made by others.

Given the fact that it has not been used in the classroom and the fact that it has been typed by a less than competent proofreader (me!), it will come as a shock to no one that the manuscript is not free of a variety of errors. I can only hope they are minor. The readers of the posted versions of the textbooks mentioned above have assisted by emailing me about errors of all types in those works. This information has allowed me to make corrections and post updated versions several times over the last few years. For as long as body and mind will allow, I will follow the same practice with this work on porous elasticity.

<sup>&</sup>lt;sup>1</sup> These books can be found at <u>http://rbowen.tamu.edu/</u>.

As a part of my teaching, I have had the opportunity to utilize computational tools that were not available when this book was conceived. In addition to the availability of a modern word processor with the capability to type mathematical equations that was unimaginable in 1980, I have had the pleasure of learning how to use MATLAB and, to a lesser extent, MAPLE. My ability with these packages is still evolving, but they were utilized during the preparation of this book. In that regard, I wish to express my appreciation to Dr. Waqar Malik for his patience and assistance during the years we have been colleagues in Mechanical Engineering at Texas A&M University. This book has benefitted in many ways from his service as my teacher as I tried to exploit these essential computer tools in my teaching and my writing.

This textbook was initially posted in late 2008. The update posted in February, 2014 mainly simply corrects typos. There is not new material added. However, in the first posting of this book, I did not include an index. This deficiency has been corrected with the February, 2014 posted. The quality of the index could be better, and I do remind the reader that the book is posted in .pdf format and that format allows for efficient electronic searches.

Ray M. Bowen <u>rbowen@tamu.edu</u> *College Station, Texas* 

February, 2014

# **CONTENTS**

## **CHAPTER 1. Some Classical Porous Media Models**

	Page
1.1. Preliminary Definitions	1
1.2. Rigid Isentropic Solid Containing One Incompressible Fluid	4
1.3. Rigid Isotropic Solid Containing One Compressible Fluid	5
1.4. Rigid Isotropic Solid Containing N - 1 Incompressible Fluids	5
1.5. Linear Elastic Incompressible Isotropic Solid Containing An Incompressible Fluid	6
1.6. Linear Elastic Isotropic Solid Containing A Compressible Fluid	7
References	8

## **CHAPTER 2. Elements of The Theory of Mixtures**

2.1. Kinematics	9
2.2. Equations of Balance	12
2.3. Field Equations	13
2.4. Balance of Mass-Special Forms	15
2.5. Balance of Momentum-Special Forms	17
2.6. Entropy Inequality	19
2.7. Rigid Isotropic Solid Containing Incompressible Fluids	22
References	27

## **CHAPTER 3.** Porous Elasticity Models

3.1. Immiscible Mixtures	29
3.2. Immiscible Mixtures-Linearized Isotropic Models	31
3.3. Immiscible Mixtures-Field Equations for the Linearized Model	36
References	43

## **CHAPTER 4.** Porous Elasticity Models With Pore Pressures

4.1. Immiscible Mixtures-Definition of Pore Pressure	45
4.2. Binary Immiscible Porous Materials with Pore Pressure	54
4.3. Stability of Equilibrium-Classes of Initial-Boundary Value	57
4.4. Incompressible Immiscible Mixtures	60
4.5. Binary Incompressible Immiscible Mixtures	67
References	69

## **CHAPTER 5. Models Which Neglect Inertia**

5.1. Compressible Models	71
5.2. Incompressible Models	78
References	85

### **CHAPTER 6.** Further Transformations And Material Properties

87
91
97

## **CHAPTER 7. Singular Surfaces And Acceleration Waves**

7.1. Singular Surfaces	99
7.2. Acceleration Waves	102
References	124

### **CHAPTER 8. Plane Harmonic Waves**

8.1. One Dimensional Governing Equations	127
8.2. Plane Progressive Waves: Dispersion Relation	130
8.3. High Frequency Approximation	135
8.4. Low Frequency Approximation	140
8.5. High and Low Frequency Approximation for Nonconductors	142
8.6. Phase Velocities	143
8.7. Numerical Example	143
References	145

## **CHAPTER 9. Boundary Initial Value Problems: Inertia Neglected**

9.1. Governing Partial Differential Equations	149
9.2. Some Properties of the Space Part of the Operator	151
9.3. Boundary Initial Value Problems	154
9.4. A One Dimensional Example	155
9.5. Biot Problem	171
9.6. Modified Biot Problem: Time Dependent External Pressure	177
9.7. The Use of Green's Functions	180
References	186

## **CHAPTER 10. Boundary Initial Value Problems: Inertia Included**

10.1. Governing Partial Differential Equations	187
10.2. Boundary Initial Value Problems	188
10.3. A One Dimensional Example	189
10.4. Properties of the Roots of det $\left(\beta_n^2 \mathbf{M} + \beta_n \xi \mathbf{E} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q}\right) = 0$	195

· ·	,	
10.5. Inversion of $\overline{\mathbf{K}}_n(s)$		202
10.6. The Use of Green's Functions		204
10.7. Biot Problem with Inertia		209

10.8. Biot Problem with Inertia: Time Dependent External Pressure	215
References	227
Appendix A	229
Bibliography	237

# **Some Classical Porous Media Models**

The objective of this introductory chapter is to set the stage for the porous media models to be developed and utilized in the later chapters. The basic approach in this chapter is to simply list the essential features of five rather well known porous media models. In this way, the reader can relate later developments to specific cases. In rough terms a *porous material* is a solid material containing one or more fluid constituents. The nature of the solid material defines various types of porous materials, as does the nature of the fluid constituents. For example, many classical porous media models assume that the solid is a rigid material. A rigid solid containing one incompressible fluid is one of the simplest types of porous material models one can formulate. In more interesting cases the solid is allowed to be an elastic material. If it is taken to be a linear elastic material, then one has one of the key assumptions built into *porous elasticity*. Within a given class of porous material models defined by an assumption about the solid behavior, one can construct other models by various assumptions about the fluids within the solid. The fluid can be incompressible or compressible. There can be multiple phases of incompressible or compressible fluids. Throughout this work we shall see various examples of all of these cases.

This Chapter contains a discussion of five porous media models. They are essentially classical in nature, each being well known and most having a large body of literature associated with them. These five models will be classified as follows:

- 1. A rigid solid containing one incompressible fluid.
- 2. A rigid solid containing one compressible fluid.
- 3. A rigid solid containing N incompressible fluids.
- 4. A linear elastic incompressible solid containing one incompressible fluid.
- 5. A linear elastic solid containing one compressible fluid.

#### **1.1 Preliminary Definitions**

First, we must introduce the concepts of *volume fraction*, *true density* and *bulk density*. These concepts arise for a special porous material consisting of *one* solid and *one* fluid. From the simple minded idea that within the solid material one can identify various volumes occupied by the solid and by the fluid, one can imagine the following geometric configuration within an elementary volume of the porous material.



In the figure,  $V_f$  denotes the volume occupied by the fluid constituent and  $V_s$  denotes the corresponding quantity for the solid. One might call these quantities the true volumes. The quantity V defined by

$$V = V_f + V_s \tag{1.1.1}$$

is the total or *bulk* volume. Given this simple geometric arrangement, the quantity  $\varphi_f$  defined by

$$\varphi_f = \frac{V_f}{V_f + V_s} = \frac{V_f}{V} \tag{1.1.2}$$

is the volume fraction of the fluid, and  $\varphi_s$  defined by

$$\varphi_s = \frac{V_s}{V_f + V_s} = \frac{V_s}{V} \tag{1.1.3}$$

is the volume fraction of the solid. It is trivially true from (1.1.2) and (1.1.3) that

$$\varphi_f + \varphi_s = 1 \tag{1.1.4}$$

The *porosity* of a porous material is the fraction of the elementary volume occupied by the fluid. It is given the symbol f and is defined by

$$f = \varphi_f = 1 - \varphi_s \tag{1.1.5}$$

If the solid is rigid, then the porosity f is a prescribed function of position. If  $m_f$  denotes the mass of the fluid constituent in the elementary volume, then  $\gamma_f$  denotes the *true density* of the fluid defined by

$$\gamma_f = \frac{m_f}{V_f} \tag{1.1.6}$$

Likewise, the *true density* of the solid is given by

$$\gamma_s = \frac{m_s}{V_s} \tag{1.1.7}$$

The *bulk density* of the fluid is defined by

$$\rho_f = \frac{m_f}{V} = \frac{m_f}{V_f} \frac{V_f}{V} = \gamma_f \varphi_f \tag{1.1.8}$$

where (1.1.2) and (1.1.6) have been used. Likewise, the bulk density of the solid is defined by

$$\rho_s = \frac{m_s}{V} = \frac{m_s}{V_s} \frac{V_s}{V} = \gamma_s \varphi_s \tag{1.1.9}$$

The velocity of the fluid is denoted by  $\mathbf{v}_f$ , and the velocity of the solid is denoted by  $\mathbf{v}_s$ . The product  $\rho_f \mathbf{v}_f$  represents the mass flux passing a unit area. Its physical dimensions are mass per unit of time per unit of area. Therefore  $\mathbf{v}_f = \frac{\rho_f \mathbf{v}_f}{\rho_f}$  represents the *bulk volume flux*, i.e. the bulk volume per unit of time per unit of area. Since,  $\rho_f \mathbf{v}_f = \gamma_f \varphi_f \mathbf{v}_f$ , the product  $\varphi_f \mathbf{v}_f$  is the true volume per unit of time per unit of area. Likewise, the product  $\varphi_f (\mathbf{v}_f - \mathbf{v}_s)$  is the true volume per unit of time passing an area moving with the solid. This product is called the *true volume flux* and is given the symbol  $\mathbf{c}_f$ , i.e.

$$\mathbf{c}_f = \varphi_f(\mathbf{v}_f - \mathbf{v}_s) \tag{1.1.10}$$

The quantity  $\mathbf{c}_f$  has the physical dimensions of velocity and is called the *filtration velocity*. Also, sometimes it is called the *specific discharge*.

The simple ideas given above easily generalize to the case of a porous solid containing N-1 fluids. For each constituent, we introduce a volume fraction  $\varphi_a$ , where a = 1 corresponds to the solid and a = 2, ..., N corresponds to the fluids. As with equation (1.1.4),

$$\sum_{a=1}^{N} \varphi_a = 1$$
 (1.1.11)

As before, the *porosity* of a porous material is the fraction of the elementary volume occupied by all of the fluids. As a result,

3.7

$$f = \sum_{a=2}^{N} \varphi_a = 1 - \varphi_s$$
 (1.1.12)

Also as before, if the solid is rigid the porosity *f* is a prescribed function of position. The bulk density  $\rho_a$  of the  $a^{th}$  constituent is related to its true density by the formula

$$\rho_a = \varphi_a \gamma_a \tag{1.1.13}$$

If  $\mathbf{v}_a$  denotes the velocity of the  $a^{\text{th}}$  fluid, and  $\mathbf{v}_s = \mathbf{v}_1$  the velocity of the solid, then

$$\mathbf{c}_a = \varphi_a(\mathbf{v}_a - \mathbf{v}_s) \tag{1.1.14}$$

is the *filtration velocity* for the  $a^{th}$  fluid.

#### 1.2 Rigid Isotropic Solid Containing One Incompressible Fluid

As the first example of a classical porous media model, consider one consisting of a rigid porous solid containing one incompressible fluid. Since the fluid is incompressible, the true density  $\gamma_f$  can be taken to be a constant. The mathematical statement of the model requires that we give field equations and constitutive equations. Of course, a complete discussion of this model and the others in this chapter would also require a discussion of suitable boundary and, in some cases, initial data. Because of the special nature of this model, the only field equation is balance of mass and it takes the simple form

$$\operatorname{div} \mathbf{c}_f = 0 \tag{1.2.1}$$

The constitutive equation which completes the formulation of this model is known as *Darcy's law*. The formal statement is the following:

$$\mathbf{c}_f = -\frac{k}{\mu} \operatorname{grad} P_f \tag{1.2.2}$$

where  $\mathbf{c}_f$  is the filtration velocity of fluid,  $P_f$  is the *pore pressure* of the fluid, *k* is a positive number known as the *permeability* of the solid and  $\mu$  is a positive number known as the *viscosity* of the fluid. As will become clear later in this work, equation (1.2.2) is more than a pure constitutive equation. It involves constitutive equations which define the fluid and the solid, the equation of motion for the fluid and several other kinematic assumptions. In any case, equations (1.2.1) and (1.2.2) combine to yield a single partial differential equation for the pore pressure  $P_f$ . This equation is

$$\operatorname{div}(\frac{k}{\mu}\operatorname{grad} P_f) = 0 \tag{1.2.3}$$

If the ratio  $\frac{k}{\mu}$  is a constant, the pore pressure is a solution of

$$\operatorname{div}(\operatorname{grad} P_f) = \Delta P_f = 0 \tag{1.2.4}$$

As the reader can observe, this model has the mathematical structure of the linear theory of heat conduction in a rigid solid. Darcy's Law plays the same role in the theory of porous materials as does Fourier's Law in the theory of rigid heat conduction.

#### 1.3 Rigid Isotropic Solid Containing One Compressible Fluid

In this case, since the fluid is allowed to be compressible, the true density  $\gamma_f$  is not a constant. The field equation important to this model is balance of mass which takes the form

$$f\frac{\partial \gamma_f}{\partial t} + \operatorname{div}(\gamma_f \mathbf{c}_f) = 0 \tag{1.3.1}$$

The constitutive equations which complete this model are as follows:

$$\mathbf{c}_f = -\frac{k}{\mu} \operatorname{grad} P_f \tag{1.3.2}$$

$$P_f = P_f(\gamma_f) \tag{1.3.3}$$

and

$$f = f(\mathbf{x}) \tag{1.3.4}$$

where **x** is the position in the porous solid. Equation (1.3.2) is again Darcy's law with k representing the *permeability* of the solid and  $\mu$  the *viscosity* of the fluid. Equation (1.3.3) is the pressure density relationship which comes from the thermodynamics of the compressible fluid in the solid pores. Finally, equation (1.3.4) represents the porosity distribution in the rigid solid. This distribution is a property of the solid. For simplicity, most models of rigid porous solids take the porosity to be a constant.

If (1.3.1), (1.3.2), (1.3.3) and (1.3.4) are combined, the result is

$$f(\mathbf{x})\frac{\partial \gamma_f}{\partial t} = \operatorname{div}\left\{\frac{k}{\mu}\gamma_f \operatorname{grad} P_f(\gamma_f)\right\}$$
(1.3.5)

Equation (1.3.5) is a nonlinear partial differential equation for the true density  $\gamma_f$ . Solutions of this partial differential equation have some features in common with solutions of the equations of nonsteady heat conduction.

#### 1.4 Rigid Isotropic Solid Containing N - 1 Incompressible Fluids

If the model summarized in Section 1.2 is generalized to allow for *N*-1 incompressible fluids, we have a field equation representing balance of mass for each fluid of the form

$$\frac{\partial \phi_a}{\partial t} + \operatorname{div} \mathbf{c}_a = 0 \tag{1.4.1}$$

for a=2,3,...,N. Because the solid is taken to be rigid, equation (1.1.12) can be written

$$\sum_{a=2}^{N} \boldsymbol{\phi}_a = f(\mathbf{x}) \tag{1.4.2}$$

Because  $f(\mathbf{x})$  is a prescribed function for a rigid solid, equation (1.4.2) should be viewed as a constraint on the unknown volume fractions,  $\varphi_2, \varphi_3, ..., \varphi_N$ . The constitutive equations which complete this model are as follows:

$$\mathbf{c}_a = -\frac{k_a}{\mu_a} \operatorname{grad} P_a \tag{1.4.3}$$

for *a* = 2,3,..., *N*, and

$$P_a - P_2 = g_a(\varphi_3, \varphi_4, ..., \varphi_N)$$
(1.4.4)

for a = 3,...,N. In equations (1.4.3) and (1.4.4),  $P_a$  is the *pore pressure* for the  $a^{th}$  fluid,  $k_a$  is the *relative permeability* for the  $a^{th}$  fluid and  $\mu_a$  is the *viscosity* for the  $a^{th}$  fluid. Equation (1.4.3) is an extension of Darcy's law which sometimes is given the name *Muskat's* law. Equation (1.4.4) is a *capillary pressure-volume fraction* which arises from a thermodynamic study of this model. The difference  $P_a \cdot P_2$  is the *capillary pressure* between the  $a^{th}$  fluid and the 2nd fluid. As will be illustrated later when this model is developed from the fundamentals of mixture theory, the choice of the 2nd fluid as a reference is arbitrary. Given the properties  $f(\mathbf{x})$ ,  $\frac{k_a}{\mu_a}$ , for  $a = 2, 3, ..., N \cdot 1$ , and  $g_a$ , for a = 3, 4, ..., N, equations (1.4.1) through (1.4.4) represent  $3N \cdot 3$  equations which, hopefully, contain solutions for the  $3N \cdot 3$  unknowns  $\varphi_2, \varphi_3, ..., \varphi_N, \mathbf{c}_2, \mathbf{c}_3, ..., \mathbf{c}_N$  and  $P_2, P_3, ..., P_N$ .

#### 1.5 Linear Elastic Incompressible Isotropic Solid Containing An Incompressible Fluid

In this section, we see the first example where the solid is not rigid. It is allowed to be a deformable solid. In particular, it is allowed to undergo infinitesimal deformations as in the classical theory of linear elasticity. The assumption that the solid is incompressible means that the true density  $\gamma_s$  can be taken to be a constant in addition to our previous assumption that  $\gamma_f$  is constant. In this case one can show that the statement of balance of mass which replaces (1.2.1) is

$$\operatorname{div} \mathbf{c}_{f} = -\operatorname{div} \dot{\mathbf{w}}_{s} \tag{1.5.1}$$

where  $\mathbf{w}_s$  denotes the *displacement* of the solid and  $\dot{\mathbf{w}}_s$  denotes the time derivative of the displacement, i.e. the *velocity* of the solid. Because the solid is allowed to deform, one has an equation of motion of the form

$$\rho_s \ddot{\mathbf{w}}_s = -\operatorname{grad} P_f + (\lambda_s + \mu_s) \operatorname{grad}(\operatorname{div} \mathbf{w}_s) + \mu_s \Delta \mathbf{w}_s$$
(1.5.2)

where  $\ddot{\mathbf{w}}_s$  is the *acceleration* of the solid,  $P_f$  is the *pore pressure* of the fluid, and  $\lambda_s$  and  $\mu_s$  are *Lame' parameters* for the solid. As we shall see later, equation (1.5.2) involves constitutive assumptions which allow the equation of motion for the solid to be written in this special form. It turns out for this model that Darcy's law in the usual form

$$\mathbf{c}_f = -\frac{k}{\mu} \operatorname{grad} P_f \tag{1.5.3}$$

is valid.

#### 1.6 Linear Elastic Isotropic Solid Containing A Compressible Fluid

In this section, we list the defining equations for a linear elastic isotropic solid which is compressible and also contains a single compressible fluid. As in the last section, the solid is allowed to undergo infinitesimal deformations as in the classical theory of linear elasticity. The equation which replaces (1.5.1) can be shown to be

$$\frac{\partial f \gamma_f}{\partial t} + \operatorname{div}(\gamma_f \mathbf{c}_f) = -\operatorname{div}(f \gamma_f \dot{\mathbf{w}}_s)$$
(1.6.1)

where  $\mathbf{w}_s$  again denotes the *displacement of the solid* and  $\dot{\mathbf{w}}_s$  denotes the time derivative of the displacement, i.e. the *velocity of the solid*. Because we are dealing with a two constituent mixture, the porosity *f* in (1.6.1) equals the volume fraction of the fluid  $\phi_f$ . This result follows from equation (1.1.5). The equation of motion for the solid which replaces (1.5.2) turns out to be formally identical to (1.5.2), i.e.

$$\rho_s \ddot{\mathbf{w}}_s = -\operatorname{grad} P_f + (\lambda_s + \mu_s) \operatorname{grad}(\operatorname{div} \mathbf{w}_s) + \mu_s \Delta \mathbf{w}_s \tag{1.6.2}$$

As with equation (1.5.2), equation (1.6.2) involves constitutive assumptions which allow the equation of motion for the solid to be written in this special form. Again, it turns out for this model that Darcy's law in the usual form

$$\mathbf{c}_f = -\frac{k}{\mu} \operatorname{grad} P_f \tag{1.6.3}$$

remains valid. The additional constitutive equations which characterize this model are a *pressure density* relation of the form

$$P_f = P_f(\gamma_f) \tag{1.6.4}$$

and a porosity-pore pressure-strain relation of the form

$$f = f(P_f, \mathbf{E}_s) \tag{1.6.5}$$

where

$$\mathbf{E}_{s} = \frac{1}{2} (\operatorname{grad} \mathbf{w}_{s} + \operatorname{grad} \mathbf{w}_{s}^{T})$$
(1.6.6)

is the *infinitesimal strain tensor* of the solid.

#### **Bibliography**

- 1. BEAR, JACOB, Dynamics of Fluids in Porous Media, American Elsevier, New York (1972).
- 2. DEWIEST, ROGER J. M. (ed.), *Flow Through Porous Media*, Academic Press, New York (1969).
- 3. POLUBARINOVA-KOCHINA, P. YA., Theory of Ground Water Movement, Princeton Univ. Press, Princeton (1962)
- 4. BOWEN, R. M., Theory of Mixtures, in *Continuum Physics*, Vol. III (ed. A. C. Eringen), Academic Press, New York (1976).
- 5. DULLIEN, R. A. L., *Porous Media, Fluid Transport and Pore Structure*, Academic Press, New York (1979).
- 6. BEAR, JACOB and M. YAVUZ CORAPICIOGLU(ed.), *Fundamentals of Transport Phenomena in Porous Media*, Proceedings of the NATO Advanced Study Institute on Mechanics of Fluids in Porous Media, Newark, Delaware, Martinus Nijhoff, Dordrecht (1984).
- 7. BENNER, HOWARD, Rheology of Two-Phase Systems, *Rev. of Fluid Mechanics*, 2, 137-176 (1970).
- 8. HUBBERT, M. KING, The Theory of Ground-Water Motion, J. of Geology, XLVIII, 785-944 (1940).
- 9. MARBLE. FRANK E., Dynamics of Dusty Gases, Rev. of Fluid Mechanics, 2, 397-446 (1970).
- 10. PARI, GUNADHAR, Flow of Fluids through Porous Deformable Solids, *Appl. Mech. Reviews*, **16**, 421-423 (1963).
- 11. PHILIP, J. R., Flow in Porous Media, Rev. of Fluid Mechanics, 2, 177-204 (1970).

# **Elements Of The Theory Of Mixtures**

In order to fully understand the various assumptions build into the models of porous materials to be discussed in this work, it is helpful to understand some of the elements of the theory of mixtures. In this chapter, these elements will be summarized. The reader interested in an extensive discussion of this subject should consult references 1 through 4.

#### 2.1. Kinematics

In this section we formulate the kinematics and the field equations for a mixture in a form which was popularized by Truesdell [Ref. 5].

Consider N continuous bodies  $\mathscr{B}_1, \mathscr{B}_2, ..., \mathscr{B}_N$ , each of which is visualized by the region it occupies in physical space. The deformation of each body is characterized by a deformation function  $\chi_a$  such that

$$\mathbf{x} = \boldsymbol{\chi}_a(\mathbf{X}_a, t) \tag{2.1.1}$$

where **x** is the *spatial position* at the time *t* of the particle of the  $a^{th}$  constituent which occupies the position **X**<sub>*a*</sub> in a reference configuration. The *velocity* and *acceleration* of **X**<sub>*a*</sub> at the time *t* are defined, respectively, by

$$\dot{\mathbf{x}}_{a} = \frac{\partial \boldsymbol{\chi}_{a}(\mathbf{X}_{a}, t)}{\partial t}$$
(2.1.2)

and

$$\mathbf{\tilde{X}}_{a} = \frac{\partial^{2} \boldsymbol{\chi}_{a}(\mathbf{X}_{a}, t)}{\partial t^{2}}$$
(2.1.3)

The gradient of the deformation for  $X_a$  is a linear transformation defined by

$$\mathbf{F}_a = \mathrm{GRAD}\,\boldsymbol{\chi}_a(\mathbf{X}_a, t) \tag{2.1.4}$$

In (2.1.4), the notation GRAD denotes the gradient computed with respect to the position  $X_a$ . The linear transformation inverse to  $F_a$  is

$$\mathbf{F}_{a}^{-1} = \operatorname{grad} \boldsymbol{\chi}_{a}^{-1}(\mathbf{x}, t) \tag{2.1.5}$$

where  $\chi_a^{-1}$  is the function inverse to  $\chi_a$ . In (2.1.5) and later, the symbol grad denotes the gradient operation with respect to the spatial position **x**. The *velocity gradient* for  $\mathscr{B}_a$  at (**x**,*t*) is defined by

$$\mathbf{L}_{a} = \operatorname{grad} \, \mathbf{\dot{x}}_{a}(\mathbf{x}, t) \tag{2.1.6}$$

An alternate expression for  $L_a$  is

$$\mathbf{L}_{a} = \mathbf{\tilde{F}}_{a} \mathbf{F}_{a}^{-1} \tag{2.1.7}$$

where  $\mathbf{\dot{F}}_{a}$  is the time derivative of  $\mathbf{F}_{a}$ .

Each constituent is assigned a density or, in the language of Chapter 1, a *bulk density*. For the  $a^{th}$  constituent the density is denoted by  $\rho_a$ . It is regarded as a function of  $(\mathbf{x},t)$ . The *density of the mixture* at  $(\mathbf{x},t)$  is defined by

$$\rho(\mathbf{x},t) = \sum_{a=1}^{N} \rho_a(\mathbf{x},t)$$
(2.1.8)

The mass concentration of the  $a^{th}$  constituent at  $(\mathbf{x},t)$  is

$$c_a(\mathbf{x},t) = \frac{\rho_a(\mathbf{x},t)}{\rho(\mathbf{x},t)}$$
(2.1.9)

It follows from (2.1.8) and (2.1.9) that

$$\sum_{a=1}^{N} c_a(\mathbf{x}, t) = 1$$
 (2.1.10)

The *mean velocity*, or simply the *velocity of the mixture* at  $(\mathbf{x},t)$  is the mass-weighted average of the constituent velocities defined by

$$\dot{\mathbf{x}} = \frac{1}{\rho} \sum_{a=1}^{N} \rho_a \, \dot{\mathbf{x}}_a \tag{2.1.11}$$

If  $\psi$  is any function of  $(\mathbf{x}, t)$ , the derivatives of  $\psi$  following the motions generated by  $\mathbf{\hat{x}}_{a}$  and  $\mathbf{\dot{x}}$  are, respectively,

$$\dot{\boldsymbol{\psi}}_{a} = \frac{\partial \boldsymbol{\psi}(\mathbf{x},t)}{\partial t} + (\operatorname{grad} \boldsymbol{\psi}(\mathbf{x},t)) \, \dot{\mathbf{x}}_{a}$$
(2.1.12)

and

$$\dot{\psi} = \frac{\partial \psi(\mathbf{x}, t)}{\partial t} + (\operatorname{grad} \psi(\mathbf{x}, t))\dot{\mathbf{x}}$$
(2.1.13)

The *diffusion velocity* for the  $a^{th}$  constituent at  $(\mathbf{x}, t)$  is defined by

$$\mathbf{u}_a = \dot{\mathbf{x}}_a - \dot{\mathbf{x}} \tag{2.1.14}$$

It follows from (2.1.11), (2.1.14) and (2.1.8) that

$$\sum_{a=1}^{N} \rho_a \mathbf{u}_a = \mathbf{0} \tag{2.1.15}$$

The *true density* of the  $a^{th}$  constituent is denoted by  $\gamma_a$  and represents the mass of the  $a^{th}$  constituent per unit of volume of the  $a^{th}$  constituent. The quantity  $\varphi_a$  defined by

$$\varphi_a = \frac{\rho_a(\mathbf{x}, t)}{\gamma_a(\mathbf{x}, t)} \tag{2.1.16}$$

is the *volume fraction* of the  $a^{th}$  constituent. Physically  $\varphi_a$  represents the volume of the  $a^{th}$  constituent per unit volume of the mixture. Therefore,

$$\sum_{a=1}^{N} \varphi_a(\mathbf{x}, t) = 1$$
 (2.1.17)

From its definition, it should be clear that  $\varphi_a$ , a = 1,...,N, measures a local structure of the mixture. For certain mixtures, it is reasonable to expect that this local structure would have no effect on the response of the mixture. Such mixtures are usually called *miscible* mixtures. An *immiscible* mixture is one where locally one can distinguish between mixture volumes and constituent volumes. A model of an immiscible mixture would necessarily allow the volume fractions to affect the mixture response. In cases where the first constituent is taken to be a solid and the remaining N - 1 constituents are fluids in the pores of the solid, the *porosity* of the solid is defined as in Chapter 1 by

$$f = \sum_{a=2}^{N} \varphi_a(\mathbf{x}, t) = 1 - \varphi_1(\mathbf{x}, t)$$
(2.1.18)

In Chapter 1, the *filtration velocities* were introduced. In the notation used in this Chapter, they are defined by the formulas

$$\mathbf{c}_a = \varphi_a(\mathbf{\dot{x}}_a - \mathbf{\dot{x}}_1) \tag{2.1.19}$$

Because the first constituent is normally the solid constituent, the filtration velocities measure the velocity of the  $a^{th}$  constituent relative to that of the solid.

In a theory of mixtures, the meaning of incompressibility can be slightly complicated. If the  $a^{th}$  constituent is *incompressible* then by definition,  $\gamma_a$  is zero. Physically, this means that  $\gamma_a$ can only depend on  $\mathbf{X}_a$ . It usually assumed that when the  $a^{th}$  constituent is incompressible that  $\gamma_a$ is a constant. In any case, the *mixture is incompressible* when every constituent in the mixture is incompressible. In this case, (2.1.16) and (2.1.17) imply a constraining relationship between the *N* bulk densities,  $\rho_a$ , a=1, 2,..., N.

#### 2.2. Equations of Balance

In this section we shall briefly summarize the equations of balance and the resulting field equations which constitute the theory of mixtures [Ref. 1,2,3,4]. For simplicity, here the equations which will be given assume that chemical reactions are absent. If  $\mathscr{N}$  is a fixed spatial region with surface  $\partial \mathscr{N}$ , then the equations of balance for the  $a^{th}$  constituent are as follows:

Balance Of Mass:

$$\frac{\partial}{\partial t} \int_{\Theta} \rho_a dv = -\oint_{\partial \Theta} \rho_a \, \dot{\mathbf{x}}_a \cdot d\mathbf{s}$$
(2.2.1)

Balance Of Linear Momentum:

$$\frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_a \, \mathbf{\hat{x}}_a dv = - \oint_{\partial \mathscr{V}} \rho_a \, \mathbf{\hat{x}}_a (\, \mathbf{\hat{x}}_a \cdot d\mathbf{s}) + \oint_{\partial \mathscr{V}} \mathbf{T}_a d\mathbf{s} + \int_{\mathscr{V}} \rho_a \mathbf{b}_a dv + \int_{\mathscr{V}} \hat{\mathbf{p}}_a dv \tag{2.2.2}$$

where, for the  $a^{th}$  constituent,  $\mathbf{T}_a$  is the *partial stress tensor*,  $\mathbf{b}_a$  is the *external body force density* and  $\hat{\mathbf{p}}_a$  is the *momentum supply*. It is the momentum supply which characterizes the mechanical interaction of the various constituents during diffusion.

Balance Of Moment Of Momentum:

$$\frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_a(\mathbf{x} - \mathbf{0}) \times \dot{\mathbf{x}}_a dv = - \oint_{\partial \mathscr{V}} \rho_a((\mathbf{x} - \mathbf{0}) \times \dot{\mathbf{x}}_a) (\dot{\mathbf{x}}_a \cdot d\mathbf{s}) 
+ \oint_{\partial \mathscr{V}} (\mathbf{x} - \mathbf{0}) \times (\mathbf{T}_a d\mathbf{s}) + \int_{\mathscr{V}} \rho_a(\mathbf{x} - \mathbf{0}) \times \mathbf{b}_a dv + \int_{\mathscr{V}} (\mathbf{x} - \mathbf{0}) \times \hat{\mathbf{p}}_a dv + \int_{\mathscr{V}} \hat{\mathbf{m}}_a dv,$$
(2.2.3)

where  $\hat{\mathbf{m}}_{a}$  is the *momentum supply* for the  $a^{th}$  constituent.

Balance Of Energy:

$$\frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_a(\varepsilon_a + \frac{1}{2} \mathbf{\hat{x}}_a^2) dv = - \oint_{\partial \mathscr{V}} \rho_a(\varepsilon_a + \frac{1}{2} \mathbf{\hat{x}}_a^2) (\mathbf{\hat{x}}_a \cdot d\mathbf{s}) + \oint_{\partial \mathscr{V}} \mathbf{\hat{x}}_a \cdot (\mathbf{T}_a d\mathbf{s}) - \oint_{\partial \mathscr{V}} \mathbf{q}_a d\mathbf{s} + \int_{\mathscr{V}} \rho_a \mathbf{\hat{x}}_a \cdot \mathbf{b}_a dv + \int_{\mathscr{V}} \rho_a r_a dv + \int_{\mathscr{V}} (\hat{\varepsilon}_a + \mathbf{\hat{x}}_a \cdot \mathbf{\hat{p}}_a) dv,$$
(2.2.4)

where, for the  $a^{th}$  constituent  $\varepsilon_a$  is the *partial internal energy density*,  $\mathbf{q}_a$  is the *partial heat flux* vector,  $r_a$  is the heat supply density and  $\hat{\varepsilon}_a$  is the energy supply. For the mixture, the equations of balance take the following forms:

Balance Of Linear Momentum:

$$\sum_{a=1}^{N} \hat{\mathbf{p}}_a = \mathbf{0} \tag{2.2.5}$$

Balance Of Moment Of Momentum:

$$\sum_{a=1}^{N} \hat{\mathbf{m}}_a = \mathbf{0} \tag{2.2.6}$$

and

Balance of Energy:

$$\sum_{a=1}^{N} (\hat{\varepsilon}_a + \mathbf{\dot{x}}_a \cdot \mathbf{\hat{p}}_a) = 0$$
(2.2.7)

#### 2.3. Field Equations

Associated with the integral statements of balance given in the last section, one can derive field equations which reflect locally the statements of balance of mass, momentum, moment of momentum and energy. These equations take the following forms:

Balance Of Mass:

$$\frac{\partial \rho_a}{\partial t} + \operatorname{div}(\rho_a \, \mathbf{\hat{x}}_a) = 0 \tag{2.3.1}$$

Balance Of Linear Momentum:

$$\rho_a \mathbf{\tilde{x}}_a = \operatorname{div} \mathbf{T}_a + \rho_a \mathbf{b}_a + \hat{\mathbf{p}}_a \tag{2.3.2}$$

Balance Of Moment Of Momentum:

$$\mathbf{T}_a - \mathbf{T}_a^{\mathrm{T}} = \hat{\mathbf{M}}_a \tag{2.3.3}$$

where  $\hat{\mathbf{M}}_a$  is the skew-symmetric linear transformation constructed from the vector  $\hat{\mathbf{m}}_a$ .

Balance Of Energy:

$$\rho_a \dot{\boldsymbol{\varepsilon}}_a = \operatorname{tr}(\mathbf{T}_a^T \mathbf{L}_a) - \operatorname{div} \mathbf{q}_a + \rho_a r_a + \hat{\boldsymbol{\varepsilon}}_a$$
(2.3.4)

Equations (2.2.5), (2.2.6) and (2.2.7) are the field equations which govern the mixture as a whole. Equivalent forms of these equations, respectively, can be shown to be

$$\rho \ddot{\mathbf{x}} = \operatorname{div} \mathbf{T} + \rho \mathbf{b} \tag{2.3.5}$$

$$\mathbf{T} = \mathbf{T}^T \tag{2.3.6}$$

and

$$\rho \dot{\varepsilon}_{\mathrm{I}} = \mathrm{tr} \sum_{a=1}^{N} (\mathbf{T}_{a}^{T} \mathbf{L}_{a}) - \mathrm{div} \, \mathbf{k} - \sum_{a=1}^{N} \mathbf{u}_{a} \cdot \hat{\mathbf{p}}_{a} + \rho r \qquad (2.3.7)$$

where

$$\mathbf{T} = \sum_{a=1}^{N} (\mathbf{T}_a - \rho_a \mathbf{u}_a \otimes \mathbf{u}_a)$$
(2.3.8)

is the stress tensor for the mixture,

$$\mathbf{b} = \frac{1}{\rho} \sum_{a=1}^{N} \rho_a \mathbf{b}_a \tag{2.3.9}$$

is the external body force density for the mixture,

$$\varepsilon_{\rm I} = \frac{1}{\rho} \sum_{a=1}^{N} \rho_a \varepsilon_a \tag{2.3.10}$$

is the inner part of the internal energy density for the mixture,

$$\mathbf{k} = \sum_{a=1}^{N} (\mathbf{q}_a + \rho_a \varepsilon_a \mathbf{u}_a)$$
(2.3.11)

is one of the many possible heat fluxes one can define for the mixture and

$$r = \frac{1}{\rho} \sum_{a=1}^{N} \rho_a r_a$$
(2.3.12)

is the *heat supply density for the mixture*. For later use, it is useful to introduce the *inner part of the stress*  $T_I$  defined by

$$\mathbf{T}_{\mathrm{I}} = \sum_{a=1}^{N} \mathbf{T}_{a} \tag{2.3.13}$$

Note, from (2.2.6) and (2.3.3), that  $T_I$  is symmetric.

It is useful to note at this point that if the N equations (2.3.1) are summed the result takes the form

$$\frac{\partial \rho}{\partial t} + \operatorname{div}(\rho \dot{\mathbf{x}}) = 0 \tag{2.3.14}$$

after (2.1.8) and (2.1.11) are used.

#### 2.4. Balance of Mass-Special Forms

Given equation (2.2.1) and the corresponding field equation (2.3.1), it is useful to deduce special forms of the field equations governing balance of mass appropriate for some of the special cases which will be considered later. A simple rearrangement of (2.3.1) yields

$$\dot{\rho}_a + \rho_a \operatorname{div} \dot{\mathbf{x}}_a = 0 \tag{2.4.1}$$

If the  $a^{th}$  constituent is *incompressible*, then  $\dot{\gamma}_a = 0$ . In this special case, (2.4.1) can be reduced to

$$\dot{\varphi}_a + \varphi_a \operatorname{div} \dot{\mathbf{x}}_a = 0 \tag{2.4.2}$$

where (2.1.16) has been used. An alternate form of (2.4.2) is

$$\frac{\partial \varphi_a}{\partial t} + \operatorname{div}(\varphi_a \mathbf{\hat{x}}_a) = 0$$
(2.4.3)

It is a simple exercise to show that (2.4.3) contains (1.2.1), (1.4.1) and (1.5.1) as special cases. For example, if N = 2 and both constituents are incompressible, then (2.4.3) is equivalent to the following two equations:

$$\frac{\partial \varphi_1}{\partial t} + \operatorname{div}(\varphi_1 \, \mathbf{\hat{x}}_1) = 0 \tag{2.4.4}$$

and

$$\frac{\partial \varphi_2}{\partial t} + \operatorname{div}(\varphi_2 \, \mathbf{\hat{x}}_2) = 0 \tag{2.4.5}$$

If (2.1.17) is utilized, the sum of (2.4.4) and (2.4.5) can be written

$$\operatorname{div}(\varphi_2(\mathbf{\dot{x}}_2 - \mathbf{\dot{x}}_1) = -\operatorname{div}\mathbf{\dot{x}}_1$$
(2.4.6)

Given the definition of filtration velocity (2.1.19), it should be clear that (2.4.6) yields (1.5.1).

Next consider the case of a two constituent mixture where the first constituent is a rigid solid and the second constituent is a compressible fluid. Without loss of generality, the velocity of the rigid solid can be taken to be zero. It follows from (2.3.1) that for the rigid solid, the only implication of balance of mass is that  $\varphi_1$  can only depend upon position **x**. The exact form of this dependence is a property of the rigid body. For example, one might have a rigid body for which the porosity is a constant. It follows from (2.1.18) that the porosity *f* is given by a prescribed function of **x**. For the compressible fluid, (2.3.1) can be written

$$f(\mathbf{x})\frac{\partial\gamma_2}{\partial t} + \operatorname{div}(\varphi_2\gamma_2\mathbf{\hat{x}}_2) = 0$$
(2.4.7)

where (2.1.16) and (2.1.18) have been used. Because the velocity of the solid is zero, equation (2.1.19) allows (2.4.7) to be written

$$f(\mathbf{x})\frac{\partial \gamma_2}{\partial t} + \operatorname{div}(\gamma_2 \mathbf{c}_2) = 0$$
(2.4.8)

Equation (2.4.8) is the result (1.3.1).

The reader can easily verify that in the case where the solid is allowed to deform along with the fluid, the field equation which governs balance of mass for the fluid which replaces (2.4.7) can be written in the form of (1.6.1).

As the examples suggest, frequently we shall select the first constituent to be a solid and allow the remaining N - I constituents to be fluids. When one of the constituents is a solid, it is sometimes convenient to utilize its reference configuration to generate *material* forms of the field equations. The results of this argument for balance of mass will be briefly given here. In order to use a suggestive notation, the subscript I which designates the solid constituent will be replaced by the symbol s. Given, (2.4.1) for the solid, it is an elementary exercise to integrate the result to obtain

$$\rho_s \left| \det \mathbf{F}_s \right| = \rho_{s_R} \tag{2.4.9}$$

where  $\rho_{s_R}$  is the density of the solid in its reference configuration. For the fluid constituents, a = 2,...,N, define *N* - 1 contents by

$$m_a = \rho_a \left| \det \mathbf{F}_s \right| \tag{2.4.10}$$

Physically,  $m_a$  represents the mass of the  $a^{th}$  fluid per unit of solid reference configuration volume. The notation  $m_a^{\circ}$  will be used to denote material differentiation following the motion of the solid, i.e. from (2.1.13)

$$\ddot{m_a} = \frac{\partial m_a(\mathbf{x}, t)}{\partial t} + \operatorname{grad} m_a(\mathbf{x}, t) \cdot \dot{\mathbf{x}}_s$$
(2.4.11)

If  $\mathbf{v}_a$  is the velocity of the  $a^{th}$  fluid relative to the solid defined by  $\mathbf{v}_a = \mathbf{x}_a - \mathbf{x}_s$ , then the quantities  $\mathbf{j}_2, \mathbf{j}_3, ..., \mathbf{j}_N$  are defined by

$$\mathbf{j}_a = \mathbf{F}_a^{-1} \left( \mathbf{x}_a - \mathbf{x}_s \right) = \mathbf{F}_a^{-1} \mathbf{v}_a \tag{2.4.12}$$

Given the definitions (2.4.10), (2.4.11) and (2.4.12), it is now possible to start with equation (2.4.1) and derive

$$\overset{o}{m}_{a} + \operatorname{Div}(m_{a}\mathbf{j}_{a}) = 0 \tag{2.4.13}$$

for the material forms of balance of mass for the fluid constituents. In equation (2.4.13), Div denotes the divergence operator with respect to the material coordinates of the solid. It is possible to show that the product  $m_a \mathbf{j}_a$  represents the mass flux of the  $a^{th}$  fluid into a material element for the solid constituent. This mass flux is measured per unit of area of the reference configuration for the solid. Balance of mass statements of the form (2.4.13) first appear in the work of Biot [Ref. 6,7,8,9].

#### 2.5. Balance of Linear Momentum-Special Forms

It is useful at this point to illustrate how (2.3.2) can be specialized so as to recover some of the results listed in Chapter 1. Essentially, Darcy's law is a special case of the statement of balance of linear momentum. The various assumptions which must be made in order to go from (2.3.2) to Darcy's law are rather strong and need justification within the context of a acceptable constitutive theory for mixtures. Elements of this theory will be presented in Sections 2.6 and 2.7.

As the first example, consider a rigid solid containing an incompressible fluid. The equation of motion (2.3.2) for the fluid can be written

$$\rho_f \mathbf{\hat{x}}_f = \operatorname{div} \mathbf{T}_f + \rho_f \mathbf{b}_f + \mathbf{\hat{p}}_f$$
(2.5.1)

We shall make the following purely formal assumptions:

- *i*. Neglect the acceleration of the fluid.
- *ii*. Replace the partial stress  $\mathbf{T}_f$  by

$$\mathbf{T}_f = -\boldsymbol{\varphi}_f \boldsymbol{P}_f \mathbf{I} \tag{2.5.2}$$

where  $P_f$  is the fluid pore pressure and  $\varphi_f$  is the volume fraction of the fluid. Note that in this case  $\varphi_f$  is also the porosity of the solid.

*iii*. Replace the momentum supply  $\hat{\mathbf{p}}_{f}$  by

$$\hat{\mathbf{p}}_f = -\xi \, \mathbf{\hat{x}}_f + P_f \, \text{grad} \, \varphi_f \tag{2.5.3}$$

where  $\xi$  is a positive material constant. It is usually called the *drag coefficient*. Given these three assumptions, (2.5.1) can be written

$$\varphi_f \mathbf{\hat{x}}_f = -\frac{\varphi_f^2}{\xi} (\operatorname{grad} P_f - \rho_f \mathbf{b}_f)$$
(2.5.4)

If the external body force  $\mathbf{b}_f$  is zero, then (2.5.4) is Darcy's law (1.2.2). In the case under discussion, the filtration velocity is  $\varphi_f \mathbf{\hat{x}}_f$  (see (1.1.10)) and the ratio  $k/\mu$  is identified as the ratio

 $\left. \frac{\varphi_f^2}{\xi} \right|_{\xi}$ . Other assumptions will produce (2.5.4). Clearly, if any constant is added to the right side of (2.5.2), then (2.5.4) is again obtained. If the body force is not zero but is given by a gravitational potential  $\nu$  by the relationship,

$$\mathbf{b}_f = -\operatorname{grad} \mathbf{v} \tag{2.5.5}$$

then Darcy's law can be written

$$\varphi_f \mathbf{\hat{x}}_f = -g \, \frac{\varphi_f^2}{\xi} \operatorname{grad} \Phi \tag{2.5.6}$$

where g is the gravitational constant, and  $\Phi$  is the *pressure head* defined by

$$g\Phi = \frac{P_f}{\gamma_f} + v \tag{2.5.7}$$

While it is not necessarily clear at this point, the simplifying assumptions which produced (2.5.4) are also consistent with balance of linear momentum for the mixture. Because the solid has been assumed to be rigid, it is a material with *internal constraints*. Such materials, as we shall see later, are characterized by certain indeterminacies in their constitutive equations.

Equation (2.5.3) deserves some additional discussion. The term  $-\xi \mathbf{\hat{x}}_f$  represents a force on the fluid because of its relative motion between itself and the solid (which is at rest). The term  $P_f \operatorname{grad} \varphi_f$  is slightly more difficult to motivate at this point. The volume fraction  $\varphi_f$  is the porosity in this case of a two constituent mixture. Therefore,  $\operatorname{grad} \varphi_f$  results from a spatial distribution of porosity. The product  $P_f \operatorname{grad} \varphi_f$  is therefore a force on the fluid caused by the changes in porosity as the fluid moves through the pores of the solid. The origin of such a force is somewhat easier to motivate after one has available a constitutive theory for mixtures. In Section 2.7 it is shown how this force arises. In particular, we shall show that (2.5.3) is very special. Some motivation for the product  $P_f \operatorname{grad} \varphi_f$  can be found in reference 10.

#### 2.6. Entropy Inequality

In order to complete the elements of the theory of mixtures, one must have a formal statement of the entropy inequality appropriate for such theories. In the notation of Section 2.2, the entropy inequality is the statement that

$$\sum_{a=1}^{N} \left\{ \frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_a \eta_a dv + \oint_{\partial \mathscr{V}} \rho_a \eta_a (\mathbf{\dot{x}}_a \cdot d\mathbf{s}) + \oint_{\partial \mathscr{V}} \frac{1}{\theta_a} \mathbf{q}_a \cdot d\mathbf{s} - \int_{\mathscr{V}} (\rho_a r_a / \theta_a) dv \right\} \ge 0$$
(2.6.1)

where  $\eta_a$  is the *entropy density* for the  $a^{th}$  constituent and  $\theta_a$  is its *temperature*. The local version of (2.6.1) is

$$\sum_{a=1}^{N} \left( \rho_a \dot{\eta}_a + \operatorname{div}(\overset{\mathbf{q}_a}{/}_{\theta_a}) - \overset{\rho_a r_a}{/}_{\theta_a} \right) \ge 0$$
(2.6.2)

If (2.3.4) is used to eliminate each  $r_a$  from (2.6.2), the result can be written

$$\sum_{a=1}^{N} \frac{1}{\theta_{a}} \left\{ -\rho_{a} (\dot{\psi}_{a} + \eta_{a} \dot{\theta}_{a}) + \operatorname{tr}(\mathbf{T}_{a}^{T} \mathbf{L}_{a}) - \frac{\mathbf{q}_{a}}{\theta_{a}} \cdot \operatorname{grad} \theta_{a} + \hat{\varepsilon}_{a} \right\} \geq 0$$
(2.6.3)

where  $\psi_a$  is the *Helmholtz free energy density* for the  $a^{th}$  constituent defined by

$$\psi_a = \varepsilon_a - \theta_a \eta_a \tag{2.6.4}$$

Most models of mixtures are constrained in some fashion. One of the most common constraints is for the constituent temperatures to be constrained to have the same value. If we adopt this constraint here and we denote the common value by  $\theta$ , then (2.6.3) can be shown to reduce to

$$-\sum_{a=1}^{N} \dot{\Psi}_{a} - \rho \eta \dot{\theta} - \operatorname{tr} \sum_{a=1}^{N} (\rho_{a} \mathbf{K}_{a} \mathbf{L}_{a}) - \frac{\mathbf{h}}{\theta} \cdot \operatorname{grad} \theta - \sum_{a=1}^{N} \dot{\mathbf{x}}_{a} \cdot \hat{\mathbf{p}}_{a} \ge 0$$
(2.6.5)

where  $\theta$  is the *temperature of the mixture*,

$$\Psi_a = \rho_a \psi_a \tag{2.6.6}$$

is the *Helmholtz free energy* of the  $a^{th}$  constituent per unit of mixture volume,

$$\eta = \frac{1}{\rho} \sum_{a=1}^{N} \rho_a \eta_a \tag{2.6.7}$$

is the entropy density for the mixture,

$$\mathbf{K}_{a} = \frac{1}{\rho_{a}} (\Psi_{a} \mathbf{I} - \mathbf{T}_{a}^{T})$$
(2.6.8)

is the *chemical potential* for the  $a^{th}$  constituent and

$$\mathbf{h} = \sum_{a=1}^{N} (\mathbf{q}_a + \rho_a \eta_a \theta \mathbf{u}_a)$$
(2.6.9)

is another possible heat flux vector for the mixture. It is a consequence of the single temperature constraint that the energy supplies  $\hat{\varepsilon}_a$  do not appear in the entropy inequality (2.6.5). It turns out to be a consequence of this fact that these energy supplies are almost indeterminate in so far as the constitutive theory is concerned. The "almost" qualifier arises because the overall statement of balance of energy (2.2.7) must be satisfied. This requirement enters into the model formulation by the requirement that (2.2.7), or, equivalently, (2.3.7) control balance of energy for the mixture. In the additional special case where the mixture is *isothermal*, the entropy inequality (2.6.5) reduces to

$$-\sum_{a=1}^{N} \dot{\Psi}_{a} - \operatorname{tr} \sum_{a=1}^{N} (\rho_{a} \mathbf{K}_{a} \mathbf{L}_{a}) - \sum_{a=1}^{N} \dot{\mathbf{x}}_{a} \cdot \hat{\mathbf{p}}_{a} \ge 0$$
(2.6.10)

When the mixture is constrained to be isothermal, the entropy inequality does not involve the heat flux vector **h**. This quantity is indeterminate in so far as the entropy inequality is concerned. It is allowed to assume any value consistent with the energy equation (2.3.7). In effect, what we have concluded is that for an isothermal model one simply does not utilize the energy equation.

Two additional constraints which will arise in this work are the case of a rigid solid and the case of an incompressible mixture. For the case of a rigid solid, the first constituent is only allowed to undergo rigid body motions. Without loss of generality, one can simply assume the rigid solid is at rest. It is possible to conclude that for a rigid solid, the momentum supply  $\hat{\mathbf{p}}_s$  is unconstrained by the entropy inequality and can, thus, take on any value consistent with balance of linear momentum for the mixture in the form (2.2.5). A more interesting constraint arises when one assumes that all of the constituents in the mixture are incompressible [Ref. 11]. In this case, it turns out that the volume fractions, the constituent velocity gradients and the constituent velocities are no longer independent. This assertion follows from the following formulas:

$$\varphi_s \left| \det \mathbf{F}_s \right| = \varphi_{s_R} \tag{2.6.11}$$

and

$$\varphi_s + \sum_{a=2}^{N} \varphi_a = 1 \tag{2.6.12}$$

Equation (2.6.11) is simply (2.4.9) specialized for an incompressible solid. Of course, equation (2.6.12) is (2.1.17). Equations (2.6.11) and (2.6.12) establish a mathematical link between  $|\det \mathbf{F}_s|$  and the volume fractions of the *N* - *I* fluids. It is this relationship which causes the terms in (2.6.10) to be related. The exact form of this relationship follows by differentiation of (2.6.12) and use of (2.4.1). The result of this calculation is

$$\sum_{a=1}^{N} (\varphi_a \operatorname{tr} \mathbf{L}_a + \operatorname{grad} \varphi_a \cdot (\mathbf{\dot{x}}_a - \mathbf{\dot{x}}_s)) = 0$$
(2.6.13)

Given (2.6.13), it easily follows that (2.6.10) can be replaced by

$$-\sum_{a=1}^{N} \dot{\Psi}_{a} - \operatorname{tr} \sum_{a=1}^{N} (\rho_{a} \mathbf{K}_{a} - \varphi_{a} \lambda \mathbf{I}) \mathbf{L}_{a} - \sum_{a=1}^{N} (\dot{\mathbf{x}}_{a} - \dot{\mathbf{x}}_{s}) \cdot (\hat{\mathbf{p}}_{a} - \lambda \operatorname{grad} \varphi_{a}) \ge 0, \quad (2.6.14)$$

for any scalar multiplier  $\lambda$ . It should be mentioned that equation (2.2.5) was also used in the derivation of (2.6.14).

As a final result in this section, there is an inequality implied by (2.6.1) which will be useful later. If the *N* equations (2.2.4) are summed over the constituents and (2.2.7) is used the result can be written

$$\sum_{a=1}^{N} \frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_{a} (\varepsilon_{a} - \theta_{o} \eta_{a} + \frac{1}{2} \dot{\mathbf{x}}_{a}^{2}) dv = -\sum_{a=1}^{N} \oint_{\partial \mathscr{V}} \rho_{a} (\varepsilon_{a} - \theta_{o} \eta_{a} + \frac{1}{2} \dot{\mathbf{x}}_{a}^{2}) (\dot{\mathbf{x}}_{a} \cdot d\mathbf{s})$$
$$+ \sum_{a=1}^{N} \oint_{\partial \mathscr{V}} \left\{ \mathbf{T}_{a}^{\mathrm{T}} \dot{\mathbf{x}}_{a} - \mathbf{q}_{a} \cdot (1 - \frac{\theta_{o}}{\theta}) \right\} d\mathbf{s} + \sum_{a=1}^{N} \int_{\mathscr{V}} \left\{ \rho_{a} r_{a} (1 - \frac{\theta_{o}}{\theta}) + \rho_{a} \dot{\mathbf{x}}_{a} \cdot \mathbf{b}_{a} \right\} dv \qquad (2.6.15)$$
$$- \theta_{o} \left\{ \sum_{a=1}^{N} \frac{\partial}{\partial t} \int_{\mathscr{V}} \rho_{a} \eta_{a} dv + \sum_{a=1}^{N} \oint_{\partial \mathscr{V}} (\rho_{a} \eta_{a} \dot{\mathbf{x}}_{a} + \frac{1}{\theta} \mathbf{q}_{a}) \cdot d\mathbf{s}) - \sum_{a=1}^{N} \int_{\mathscr{V}} (\rho_{a} r_{a} / \theta) dv \right\}$$

where  $\theta_o$  is any positive constant. If (2.6.1) is used, then (2.6.15) yields the following inequality:

$$\sum_{a=1}^{N} \frac{\partial}{\partial t} \int_{\mathscr{A}} \rho_{a} (\varepsilon_{a} - \theta_{o} \eta_{a} + \frac{1}{2} \mathbf{\dot{x}}_{a}^{2}) dv \leq -\sum_{a=1}^{N} \oint_{\partial \mathscr{A}} \rho_{a} (\varepsilon_{a} - \theta_{o} \eta_{a} + \frac{1}{2} \mathbf{\dot{x}}_{a}^{2}) (\mathbf{\dot{x}}_{a} \cdot d\mathbf{s}) + \sum_{a=1}^{N} \oint_{\partial \mathscr{A}} \left\{ \mathbf{T}_{a}^{T} \mathbf{\dot{x}}_{a} - \mathbf{q}_{a} \cdot (1 - \frac{\theta_{o}}{\theta}) \right\} d\mathbf{s} + \sum_{a=1}^{N} \int_{\mathscr{A}} \left\{ \rho_{a} r_{a} (1 - \frac{\theta_{o}}{\theta}) + \rho_{a} \mathbf{\dot{x}}_{a} \cdot \mathbf{b}_{a} \right\} dv.$$

$$(2.6.16)$$

#### 2.7 Rigid Isotropic Solid Containing Incompressible Fluids

It is not the purpose of this work to make the reader an expert on the theory of mixtures. The introduction of this subject in this chapter is intended to give the reader an indication of the formal basis for the models to be discussed later. Toward this goal it is instructive to illustrate how one of the simplest porous media models can be viewed as a consequence of the theory of mixtures. This simple model is the one summarized in Section 1.4. It consists of a rigid isotropic solid containing N - I incompressible fluids. Of course, it is understood that N is an integer greater than or equal to two.

For simplicity, the porosity of the solid will be taken to be constant in this discussion. As a result, it follows from (2.1.18) that the *N* - *1* volume fractions obey

$$f = \sum_{a=2}^{N} \varphi_a(\mathbf{x}, t) = constant$$
(2.7.1)

Because of the relationship (2.7.1), it is often the case for this model that the volume fractions  $\varphi_a, a = 2, ..., N$ , are replaced as kinematic variables by the *saturations* defined by

$$s_a = \frac{\varphi_a}{f} \tag{2.7.2}$$

for a = 2,...,N. While some formal advantages arise by the introduction of these quantities, it is not necessary to complicate the notation by their introduction here. Because the solid is rigid, without loss of generality its velocity can be taken to be zero. This fact allows the entropy inequality (2.6.14) to be written

$$-\sum_{a=2}^{N} \dot{\Psi}_{a} - \operatorname{tr} \sum_{a=2}^{N} (\rho_{a} \mathbf{K}_{a} - \varphi_{a} \lambda \mathbf{I}) \mathbf{L}_{a} - \sum_{a=2}^{N} \dot{\mathbf{x}}_{a} \cdot (\hat{\mathbf{p}}_{a} - \lambda \operatorname{grad} \varphi_{a}) \ge 0, \qquad (2.7.3)$$

where, without loss of generality, the free energy  $\Psi_1 = \Psi_s$  is taken to be a constant. The constitutive equations which we shall take as defining the mixture are as follows:

$$\Psi_a = \Psi_a(\varphi_c) \tag{2.7.4}$$

$$\hat{\mathbf{p}}_a - \lambda \operatorname{grad} \varphi_a = \mathbf{e}_a(\varphi_c, \operatorname{grad} \varphi_c, \mathbf{\hat{x}}_b)$$
(2.7.5)

and

$$\rho_a \mathbf{K}_a - \varphi_a \lambda \mathbf{I} = \mathbf{R}_a(\varphi_c) \tag{2.7.6}$$

where a = 2,3,..., N. Our notational convention is such that the subscripts *a* and *b* range from 2,..., *N*. The subscripts *c*, *d* and *e* range from 3,..., *N*. As an arbitrary choice, equation (2.7.1) has been used to eliminate  $\varphi_2$  as a constitutive variable. Clearly, more general constitutive equations could have been written for this mixture. The reader interested in a somewhat more general discussion of this model should consult reference 11. The parameter  $\lambda$  appears in (2.7.5) and (2.7.6) because of the incompressible constraint. The special dependence reflects the arbitrariness which is intrinsic in the entropy inequality (2.7.3).

If (2.7.4) through (2.7.6) are substituted into the inequality (2.7.3), the result can be written

$$-\operatorname{tr}(\mathbf{R}_{2}(\varphi_{c})\mathbf{L}_{2}) - \operatorname{tr}\sum_{d=3}^{N}((\mathbf{R}_{d}(\varphi_{c}) - \varphi_{d}\frac{\partial\Psi_{1}(\varphi_{c})}{\partial\varphi_{d}}\mathbf{I})\mathbf{L}_{d})$$
  
$$- \dot{\mathbf{x}}_{2} \cdot (\mathbf{e}_{2}(\varphi_{c}, \operatorname{grad}\varphi_{c}, \dot{\mathbf{x}}_{b}) + \sum_{d=3}^{N}\frac{\partial\Psi_{2}(\varphi_{c})}{\partial\varphi_{d}}\operatorname{grad}\varphi_{d}) \qquad (2.7.7)$$
  
$$- \sum_{d=3}^{N} \dot{\mathbf{x}}_{d} \cdot (\mathbf{e}_{d}(\varphi_{c}, \operatorname{grad}\varphi_{c}, \dot{\mathbf{x}}_{b}) - \frac{\partial\Psi_{1}(\varphi_{c})}{\partial\varphi_{d}}\operatorname{grad}\varphi_{d} + \sum_{e=3}^{N}\frac{\partial\Psi_{d}(\varphi_{c})}{\partial\varphi_{e}}\operatorname{grad}\varphi_{e}) \geq 0,$$

where the quantity  $\Psi_I$  is the inner part of the Helmholtz Free Energy of the mixture defined by

$$\Psi_{\rm I} = \sum_{a=1}^{N} \Psi_a \tag{2.7.8}$$

As explained in reference 11, the inequality (2.7.7) implies the following restrictions:

$$\mathbf{R}_2(\varphi_c) = \mathbf{0} \tag{2.7.9}$$

$$\mathbf{R}_{d}(\varphi_{c}) = \varphi_{d} \frac{\partial \Psi_{I}(\varphi_{c})}{\partial \varphi_{d}} \mathbf{I}$$
(2.7.10)

for d = 3, ..., N, and

$$-\sum_{a=2}^{N} \mathbf{\dot{x}}_{a} \cdot \mathbf{\hat{f}}_{a}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{\dot{x}}_{b}) \ge 0$$
(2.7.11)

where

$$\hat{\mathbf{f}}_{2} = \mathbf{e}_{2}(\varphi_{c}, \operatorname{grad}\varphi_{c}, \mathbf{\dot{x}}_{b}) + \sum_{e=3}^{N} \frac{\partial \Psi_{2}(\varphi_{c})}{\partial \varphi_{e}} \operatorname{grad}\varphi_{e}$$
(2.7.12)

and

$$\hat{\mathbf{f}}_{d} = \mathbf{e}_{d}(\varphi_{c}, \operatorname{grad}\varphi_{c}, \dot{\mathbf{x}}_{b}) - \frac{\partial \Psi_{I}(\varphi_{c})}{\partial \varphi_{d}} \operatorname{grad}\varphi_{d} + \sum_{e=3}^{N} \frac{\partial \Psi_{d}(\varphi_{c})}{\partial \varphi_{e}} \operatorname{grad}\varphi_{e}$$
(2.7.13)

where d = 3, ..., N.

Given (2.7.9) and (2.7.10), it follows from (2.7.6) that

$$\rho_2 \mathbf{K}_2 = \varphi_2 \lambda \mathbf{I} \tag{2.7.14}$$

and

$$\rho_d \mathbf{K}_d = \varphi_d (\lambda + \frac{\partial \Psi_{\mathrm{I}}(\varphi_c)}{\partial \varphi_d}) \mathbf{I}$$
(2.7.15)

for d = 3,..., N. Given (2.7.12) and (2.7.13), it follows from (2.7.5) that

$$\hat{\mathbf{p}}_{2} = \lambda \operatorname{grad} \varphi_{2} - \sum_{e=3}^{N} \frac{\partial \Psi_{2}(\varphi_{c})}{\partial \phi_{e}} \operatorname{grad} \phi_{e} + \hat{\mathbf{f}}_{2}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{\dot{x}}_{b})$$
(2.7.16)

and

$$\hat{\mathbf{p}}_{d} = (\lambda + \frac{\partial \Psi_{I}(\varphi_{c})}{\partial \varphi_{d}}) \operatorname{grad} \varphi_{d} - \sum_{e=3}^{N} \frac{\partial \Psi_{d}(\varphi_{c})}{\partial \varphi_{e}} \operatorname{grad} \varphi_{e} + \hat{\mathbf{f}}_{d}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{\hat{x}}_{b})$$
(2.7.17)

for d = 3, ..., N. It is convenient to define fluid *pore pressures* by

$$P_2 = \lambda \tag{2.7.18}$$

and

$$P_d = \lambda + \frac{\partial \Psi_1(\varphi_c)}{\partial \varphi_d}$$
(2.7.19)

for d = 3,..., N. With these definitions, the *capillary pressures*, defined by  $P_d - P_2$ , are given by

$$P_d - P_2 = \frac{\partial \Psi_{\rm I}(\varphi_c)}{\partial \varphi_{\rm d}} \tag{2.7.20}$$

for d = 3,..., N. Equation (2.7.20) establishes within the thermodynamics of the theory of mixtures the special constitutive relation (1.4.4). Next we shall illustrate how one can find the result (1.4.3) within the results of this section.

First, it is helpful to extract a physical meaning for the N-1 quantities,  $\hat{\mathbf{f}}_a$ . These quantities are restricted by the inequality (2.7.11). When viewed as a function of the N-1 velocities,  $\hat{\mathbf{x}}_b$ , the quantity  $\Phi$  defined by

$$\Phi(\mathbf{\dot{x}}_{b}) = -\sum_{a=2}^{N} \mathbf{\dot{x}}_{a} \cdot \mathbf{\hat{f}}_{a}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{\dot{x}}_{b})$$
(2.7.21)

is a minimum at  $\mathbf{\hat{x}}_b = \mathbf{0}$ . It is elementary to use this observation to prove that

$$\hat{\mathbf{f}}_a(\phi_c, \operatorname{grad} \phi_c, \mathbf{0}) = \mathbf{0}$$
(2.7.22)

and

$$\sum_{a=2}^{N} \sum_{b=2}^{N} \mathbf{a}_{a} \cdot \left( \frac{\partial \mathbf{f}_{a}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{0})}{\partial \mathbf{\dot{x}}_{b}} \mathbf{b}_{b} \right) \leq 0$$
(2.7.23)

for all vectors  $\mathbf{a}_a, a = 2, 3, ..., N$  and  $\mathbf{b}_b, b = 2, 3, ..., N$ . Equation (2.7.22) shows that the quantities  $\hat{\mathbf{f}}_a$ , for a = 2, 3, ..., N, vanish when the fluid constituents have zero velocity. It is, therefore, natural to call these quantities, the *diffusion forces*. The definitions (2.7.18) and (2.7.19) allow (2.7.16) and (2.7.17) to be written

$$\hat{\mathbf{p}}_{b} = P_{b} \operatorname{grad} \varphi_{b} - \sum_{e=3}^{N} \frac{\partial \Psi_{b}(\varphi_{c})}{\partial \varphi_{e}} \operatorname{grad} \varphi_{e} + \hat{\mathbf{f}}_{b}(\varphi_{c}, \operatorname{grad} \varphi_{c}, \mathbf{\hat{x}}_{b})$$
(2.7.24)

In addition, equations (2.7.14), (2.7.15), (2.7.18) and (2.7.19) combine with (2.6.8) to yield

$$\mathbf{T}_{b} = -\varphi_{b}P_{b}\mathbf{I} + \Psi_{b}\mathbf{I} \tag{2.7.25}$$

If (2.7.24) and (2.7.25) are substituted into (2.3.2), balance of linear momentum for the  $b^{th}$  fluid takes the following form:

$$\rho_b \mathbf{\ddot{x}}_b = -\varphi_b \operatorname{grad} P_b + \rho_b \mathbf{b}_b + \mathbf{\hat{f}}_b$$
(2.7.26)

It is instructive at this point to attempt to find the results of Section 2.5 in the results obtained in this section. In Section 2.5, we assumed that the mixture consisted of a rigid solid and a single incompressible fluid. Therefore, N = 2. In this case, the constitutive equation (2.7.4) reduces to

$$\Psi_2 = constant \tag{2.7.27}$$

Without loss of generality, this constant can be taken to be zero. Therefore, (2.7.24) and (2.7.25) reduce to

$$\hat{\mathbf{p}}_2 = P_2 \operatorname{grad} \varphi_2 + \hat{\mathbf{f}}_2(\hat{\mathbf{x}}_2)$$
(2.7.28)

and

$$\mathbf{T}_2 = -\varphi_2 P_2 \mathbf{I} \tag{2.7.29}$$

Equation (2.7.29) is the result (2.5.2). Equation (2.7.28) reduces to (2.5.3) if the diffusion force  $\hat{\mathbf{f}}_2(\mathbf{\hat{x}}_2)$  takes the special form

$$\hat{\mathbf{f}}_{2}(\dot{\mathbf{x}}_{2}) = -\xi \dot{\mathbf{x}}_{2} \tag{2.7.30}$$

Clearly, (2.7.30) obeys (2.7.22). Given (2.7.30), it easily follows from (2.7.23) that the material coefficient  $\xi$ , the *drag coefficient*, cannot be negative. In fact, this result is sufficient to satisfy the residual inequality (2.7.11). The reader will recognize (2.7.30) as the linearized constitutive equation for the drag force which arises when one assumes small departures from the state  $\mathbf{\hat{x}}_2 = \mathbf{0}$  and that the rigid solid is isotropic.

Finally, it is instructive return to the case of an arbitrary number of incompressible fluids and to illustrate how the result (1.4.3) arises as a special case of the results given here. The reader should note in passing that (1.4.1) is a simple rearrangement of (2.4.3), and (1.4.4) has been established with the thermodynamic result (2.7.20). Equation (1.4.3) follows from (2.7.26) if one neglects the constituent accelerations, the external body forces and adopts the following constitutive equations for the diffusion forces  $\hat{\mathbf{f}}_{b}(\hat{\mathbf{x}}_{c})$ :

$$\hat{\mathbf{f}}_{b}(\mathbf{\dot{x}}_{c}) = -\xi_{b} \mathbf{\dot{x}}_{b}$$
(2.7.31)

With these specializations, (2.7.26) can be written

$$\varphi_b \mathbf{\hat{x}}_b = -\frac{\varphi_b^2}{\xi_b} \operatorname{grad} P_b$$
(2.7.32)

where the coefficient  $\xi_b$  has been assumed to be nonzero. Note that (2.7.23) shows that each  $\xi_b$  must obey the inequality

$$\xi_b \ge 0 \tag{2.7.33}$$

Because the filtration velocity equals  $\varphi_b \mathbf{\hat{x}}_b$  in this case, it is easily seen that (2.7.32) is equivalent to

the result (1.4.3). One simply has to identify the coefficient  $\frac{\varphi_b^2}{\xi_b}$  as the ratio  $\frac{k_b}{\mu_b}$ .

As indicated above, the reader interested in additional information about how one models incompressible porous materials by use of the theory of mixtures should consult reference 11.

#### References

- 1. TRUESDELL, C. and R. TOUPIN, *The Classical Field Theories*, Handbuch der Physik Vol III/1 (ed. S. Flugge), Springer Verlag, Berlin (1960).
- 2. BOWEN, R. M., Theory of Mixtures, in *Continuum Physics*, Vol III, (ed A. C. Eringen), Academic Press, New York (1976).
- 3. ATKIN, R. J., and R. E. CRAINE, Continuum Theories of Mixtures: Basic Theory and Historical Development, *Q. J. Mech. Appl. Math.*, **24**, 209-244 (1976).
- 4. ATKIN, R. J., and R. E. CRAINE, Continuum Theories of Mixtures: Applications, J. Inst. Maths Applics, 17, 153-207 (1976).
- 5. TRUESDELL, C., Sulle basi della thermomeccannica, *Rend. Lincei* (*8*), **22**, 33-38, 158-166 (1957).
- 6. BIOT, MAURICE A., Nonlinear and Semilinear Rheology of Porous Solids, *J. Geophysical Res.*, **78**, 4924-4937 (1973).
- 7. BIOT, MAURICE A., Variational-Lagrangian Irreversible Thermodynamics of Initially-Stressed Solids with Thermomolecular Diffusion and Chemical Reactions, *J. Mech. Phys. Solids*, **25**, 289-307 (1977).
- 8. BIOT, MAURICE A., Variational Lagrangian-Thermodynamics of Nonisothermal Finite Strain Mechanics of Porous Solids and Thermomolecular Diffusion, *Int. J. Solids Structures*, **13**, 579-597 (1977).
- 9. BIOT, MAURICE A., Theory of Finite Deformations of Porous Solids, *Indiana Univ. Math. Journal*, **21**, 597-620 (1972).
- 10. KARAKIN, A. V., and L. I. LOBKOVSKII, Mechanics of Porous Two-Phase Visco-Deformed Medium and its Geophysical Applications, *Lett. Appl. Engr. Sci.*, **17**, 797-805 (1979).
- 11. BOWEN, R. M., Incompressible Porous Media Models by Use of the Theory of Mixtures, *Intl. J. Engr. Sci.*, **18**, 1129-1148 (1980).

## **Porous Elasticity Models**

In this chapter the mathematical models which form the basis of the theory of porous elasticity will be given. For the most part, the basis of the models, as implied by the theory of mixtures, will not be given. References will be given where the reader can find this fundamental material.

#### 3.1. Immiscible Mixtures

As indicated in Section 2.1, a mixture is immiscible if volume fractions affect the response. In this section, an example of an immiscible mixture is briefly considered. The model is that of an immiscible compressible mixture whose constituents are a single solid and N-1 fluids. For the second, the immiscible mixture is taken to be incompressible. This example is discussed in detail in references 1.

An immiscible compressible mixture having a single temperature  $\theta$  is defined by the following constitutive equations:

$$\Psi_a = \Psi_a(\theta, \mathbf{F}_s, \rho_b, \varphi_b) \tag{3.1.1}$$

and

$$(\eta_a, \hat{\mathbf{p}}_a, \mathbf{K}_a, \mathbf{M}_a, \mathbf{h}, \dot{\boldsymbol{\varphi}}_a) = f_a(\theta, \operatorname{grad} \theta_a, \mathbf{F}_s, \operatorname{GRAD} \mathbf{F}_s, \rho_b, \operatorname{grad} \rho_b, \varphi_b, \operatorname{grad} \varphi_b, \mathbf{\hat{x}}_b - \mathbf{\hat{x}}_s).$$
(3.1.2)

In (3.1.1) and (3.1.2), the subscript *s* denotes the first constituent, the solid. In addition, it is understood in writing these equations that the index *a* ranges from 1 to *N* while the indices *b* and *c* range from 2 to *N*. Note that (3.1.2) shows that the volume fractions have been treated as internal state variables. This assumption is sufficiently general to allow one to treat, for example, the volume fractions as determined by the state of the mixture or as given constants. Both special cases occur in the classical literature on porous elasticity. If (3.1.1) and (3.1.2) are required to be consistent with the entropy inequality (2.6.5), the following results are obtained [Ref. 1]:

$$\rho\eta = -\frac{\partial \Psi_{\rm I}}{\partial \theta} \tag{3.1.3}$$

$$\rho_s \mathbf{K}_s = -\mathbf{F}_s \frac{\partial \Psi_1}{\partial \mathbf{F}_s}^T \tag{3.1.4}$$
(3.1.5)

and

$$-\mathbf{m} \cdot \operatorname{grad} \theta / \theta - \sum_{c=2}^{N} (\mathbf{\dot{x}}_{c} - \mathbf{\dot{x}}_{s}) \cdot \mathbf{\hat{f}}_{c} - \sum_{c=2}^{N} \sigma_{c} \omega_{c} \ge 0$$
(3.1.6)

 $\mathbf{K}_{c} = \boldsymbol{\mu}_{c} \mathbf{I}$ 

where

$$\mu_c = \frac{\partial \Psi_1}{\partial \rho_c} \tag{3.1.7}$$

$$\sigma_c = -\frac{\partial \Psi_{\rm I}}{\partial \varphi_c} \tag{3.1.8}$$

$$\hat{\mathbf{f}}_{c} = \hat{\mathbf{p}}_{c} - \mu_{c} \operatorname{grad} \rho_{c} + \sigma_{c} \operatorname{grad} \varphi_{c} + (\mathbf{F}_{s}^{-1})^{\mathrm{T}} \left\{ \frac{\partial \Psi_{c}}{\partial \mathbf{F}_{s}} [\operatorname{GRAD} \mathbf{F}_{s}] \right\}$$

$$+ \sum_{b=2}^{N} \frac{\partial \Psi_{c}}{\partial \rho_{b}} \operatorname{grad} \rho_{b} + \sum_{b=2}^{N} \frac{\partial \Psi_{c}}{\partial \varphi_{b}} \operatorname{grad} \varphi_{b}$$
(3.1.9)

and

$$\mathbf{m} = \mathbf{h} + \theta \sum_{a=1}^{N} \frac{\partial \Psi_a}{\partial \theta} \mathbf{u}_a$$
(3.1.10)

Of course, the heat flux vector **h** is defined by (2.6.9). In (3.1.9) a special notation has been used in the term involving  $\text{GRAD} \mathbf{F}_s$ . In components, this notation means the following:

$$\frac{\partial \Psi_c}{\partial \mathbf{F}_s} \left[ \text{GRAD} \, \mathbf{F}_s \right] = \frac{\partial \Psi_c}{\partial F_{s_{jk}}} \frac{\partial F_{s_{jk}}}{\partial X_L} \mathbf{i}_L \tag{3.1.11}$$

where  $\mathbf{i}_{L}$  denotes the  $L^{th}$  basis vector in a rectangular Cartesian coordinate system. Finally, as follows from (3.1.2), the quantity  $\omega_{c}$  in (3.1.6) is the function whose value is  $\dot{\varphi}_{c}$ , *i.e.*,

$$\dot{\varphi}_{c} = \omega_{c}(\theta, \operatorname{grad} \theta, \mathbf{F}_{s}, \operatorname{GRAD} \mathbf{F}_{s}, \rho_{b}, \operatorname{grad} \rho_{b}, \varphi_{b}, \operatorname{grad} \varphi_{b}, \dot{\mathbf{x}}_{b} - \dot{\mathbf{x}}_{s})$$
(3.1.12)

Next, certain thermodynamic equilibrium results, which follow from (3.1.6), will be recorded. For a model employing internal state variables, there are several natural definitions of equilibrium. The reader interested in a discussion of these possibilities can consult Bowen [Ref. 1, 2, 3]. Here, equilibrium is a state in which the temperature gradient is zero, the constituent

velocities are all equal and the *N* - 1 quantities  $\sigma_c$ , c = 2,...,N, defined by (3.1.8) all vanish. It shall be assumed that the *N* - 1 equations (3.1.8) can be inverted to obtain

$$\varphi_b = \varphi_b(\theta, \mathbf{F}_s, \rho_c, \sigma_c) \tag{3.1.13}$$

The assumed inversion implies that the  $(N-1) \times (N-1)$  symmetric matrix

$$\frac{\partial \sigma_c}{\partial \varphi_b} = -\frac{\partial^2 \Psi_{\rm I}}{\partial \varphi_b \partial \varphi_c} \tag{3.1.14}$$

is nonsingular for all values of its argument. Given (3.1.13), it follows that in the equilibrium state the volume fractions are determined by the temperature and the deformation of the solid. By the kind of argument mentioned in Section 2.7, it is possible to show from (3.1.6) that, in the equilibrium state defined here,  $\mathbf{m}$ ,  $\hat{\mathbf{f}}_c$  and  $\omega_c$  each vanish [Ref. 1].

#### 3.2 Immiscible Mixtures-Linearized Isotropic Models

In order to further simplify the formalities, it will be assumed that the solid is *isotropic*. With this assumption, the constitutive equations will be linearized about a thermodynamic equilibrium state of uniform temperature  $\theta^+$ , uniform deformation  $\mathbf{F}_s = \mathbf{I}$  for the solid, uniform bulk densities  $\rho_c = \rho_c^+$  and uniform volume fractions calculated from (3.1.13) by the formulas

$$\varphi_b^+ = \varphi_b(\theta^+, \mathbf{I}, \rho_c^+, 0) \tag{3.2.1}$$

It is readily established that the state defined in this fashion is a solution of the field equations which govern the immiscible mixture if  $\mathbf{b}_a = \mathbf{0}$ , for a = 1,...,N, and r = 0. It is convenient to refer to this state as the *static solution* of the field equations.

The equations which govern the immiscible mixture are linearized by requiring the temperature changes, the displacement gradient of the solid, the density changes for the fluids, and the departure from thermodynamic equilibrium to be small. As is customary, the *displacement* of the  $a^{th}$  constituent is defined by

$$\mathbf{w}_{a}(\mathbf{X}_{a},t) = \boldsymbol{\chi}_{a}(\mathbf{X}_{a},t) - \mathbf{X}_{a}$$
(3.2.2)

Given (3.2.2), the displacement gradient of the  $a^{th}$  constituent is defined by

$$\mathbf{H}_{a} = \operatorname{GRAD} \mathbf{w}_{a}(\mathbf{X}_{a}, t) \tag{3.2.3}$$

and, by (3.2.2) and (2.1.4),

$$\mathbf{H}_a = \mathbf{F}_a - \mathbf{I} \tag{3.2.4}$$

A formal linearization of the constitutive equations and the field equations involves expansions about the static solution

$$\begin{pmatrix} \theta, \operatorname{grad} \theta, \mathbf{F}_{s}, \operatorname{GRAD} \mathbf{F}_{s}, \rho_{b}, \operatorname{grad} \rho_{b}, \varphi_{b}, \operatorname{grad} \varphi_{b}, \mathbf{\dot{x}}_{b} - \mathbf{\dot{x}}_{s} \end{pmatrix}$$

$$= \left(\theta^{+}, \mathbf{0}, \mathbf{I}, \mathbf{0}, \rho_{b}^{+}, \mathbf{0}, \phi_{b}^{+}, \mathbf{0}, \mathbf{0}\right).$$

$$(3.2.5)$$

Departures from this state are measured by a positive number  $\varepsilon$  defined by

$$\varepsilon^{2} = (\theta - \theta^{+})^{2} + \operatorname{grad} \theta \cdot \operatorname{grad} \theta + \operatorname{tr}(\mathbf{H}_{s}\mathbf{H}_{s}^{\mathrm{T}}) + \operatorname{GRAD}\mathbf{F}_{s} \cdot \operatorname{GRAD}\mathbf{F}_{s} + \sum_{b=2}^{N} (\rho_{b} - \rho_{b}^{+})^{2} / \rho_{b}^{+2} + \sum_{b=2}^{N} \operatorname{grad} \rho_{b} \cdot \operatorname{grad} \rho_{b} \cdot \operatorname{grad} \rho_{b} + \sum_{b=2}^{N} (\varphi_{b} - \varphi_{b}^{+})^{2} + \sum_{b=2}^{N} \operatorname{grad} \varphi_{b} \cdot \operatorname{grad} \varphi_{b} + \sum_{b=2}^{N} (\mathbf{x}_{b} - \mathbf{x}_{s}) \cdot (\mathbf{x}_{b} - \mathbf{x}_{s}).$$

$$(3.2.6)$$

Given the parameter  $\varepsilon$ , the formal expansion of  $\Psi_a$  which follows from (3.1.1) can be written

$$\begin{split} \Psi_{a} &= \Psi_{a}^{+} + \alpha_{a}(\theta - \theta^{+}) + \iota_{as}(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \iota_{ab}(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \sum_{b=2}^{N} \left( \frac{\partial \Psi_{a}}{\partial \varphi_{b}} \right)^{+} (\varphi_{b} - \varphi_{b}^{+}) \\ &- \frac{1}{2} v_{a}(\theta - \theta^{+})^{2} - \tau_{as}(\theta - \theta^{+})(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \tau_{ab}(\theta - \theta^{+})(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} \\ &+ \sum_{b=2}^{N} \Theta_{ab}(\theta - \theta^{+})(\varphi_{b} - \varphi_{b}^{+}) + \frac{1}{2} \lambda_{ass}(\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \lambda_{asb}(\operatorname{tr} \mathbf{E}_{s})(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} \end{split}$$
(3.2.7)  
$$&+ \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \lambda_{abc}(\rho_{b} - \rho_{b}^{+})(\rho_{c} - \rho_{c}^{+}) / \rho_{b}^{+} \rho_{c}^{+} + \mu_{ass}\operatorname{tr}(\mathbf{E}_{s}\mathbf{E}_{s}) + \sum_{b=2}^{N} \Gamma_{asb}(\operatorname{tr} \mathbf{E}_{s})(\varphi_{b} - \varphi_{b}^{+}) \\ &- \sum_{b=2}^{N} \sum_{c=2}^{N} \Gamma_{abc}((\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \Phi_{abc}(\varphi_{b} - \varphi_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + O(\varepsilon^{3}), \end{split}$$

where the coefficients in (3.2.7) are material constants. These constants can be identified as partial derivatives of  $\Psi_a$  evaluated at the static solution. The complicated result (3.2.7) reflects our assumption that the solid is isotropic. It also reflects the various restrictions one would have developed from the concept of material frame indifference. This concept is explained in textbooks on Continuum Mechanics [Ref. 4]. In (3.2.7), the quantity  $\mathbf{E}_s$  is the *infinitesimal strain tensor* for the solid defined by

$$2\mathbf{E}_s = \mathbf{H}_s + \mathbf{H}_s^T \tag{3.2.8}$$

By use of (2.7.8), it follows from (3.2.7) that  $\Psi_I$  is given by

$$\Psi_{I} = \Psi_{I}^{+} + \alpha(\theta - \theta^{+}) + \iota_{s}(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \iota_{b}(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} - \frac{1}{2} \nu(\theta - \theta^{+})^{2} - \tau_{s}(\theta - \theta^{+})(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \tau_{b}(\theta - \theta^{+})(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \sum_{b=2}^{N} \Theta_{b}(\theta - \theta^{+})(\varphi_{b} - \varphi_{b}^{+}) + \frac{1}{2} \lambda_{ss}(\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \lambda_{sb}(\operatorname{tr} \mathbf{E}_{s})(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \lambda_{bc}(\rho_{b} - \rho_{b}^{+}) (\rho_{c} - \rho_{c}^{+}) / \rho_{b}^{+} \rho_{c}^{+}$$
(3.2.9)  
$$+ \mu_{ss} \operatorname{tr}(\mathbf{E}_{s}\mathbf{E}_{s}) + \sum_{b=2}^{N} \Gamma_{sb}(\operatorname{tr} \mathbf{E}_{s})(\varphi_{b} - \varphi_{b}^{+}) - \sum_{b=2}^{N} \sum_{c=2}^{N} \Gamma_{bc}((\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \Phi_{bc}(\varphi_{b} - \varphi_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + O(\varepsilon^{3}).$$

In writing (3.2.9), the result (3.1.8) and the fact that the  $\sigma_a$  vanish at the static solution has been used. Also, the various material coefficients in (3.2.9) are defined in terms of those in (3.2.7) by a set of formulas like

$$\Phi_{bc} = \sum_{a=2}^{N} \Phi_{abc} \tag{3.2.10}$$

It follows from (3.2.9) and the thermodynamic results (3.1.3), (3.1.4), (3.1.7) and (3.1.8) that

$$\rho\eta = -\alpha + \nu(\theta - \theta^{+}) + \tau_{s}(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \tau_{b} (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} - \sum_{b=2}^{N} \Theta_{b} (\varphi_{b} - \varphi_{b}^{+}) + O(\varepsilon^{2}), \quad (3.2.11)$$

$$\rho_{s_{R}}\mathbf{K}_{s} = -\iota_{s}\mathbf{I} - (\lambda_{ss} + \iota_{s})(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2(\mu_{ss} + \iota_{s})\mathbf{E}_{s} + \sum_{b=2}^{N}\lambda_{sb}((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+}))\mathbf{I} + \tau_{s}(\theta - \theta^{+})\mathbf{I} - \sum_{b=2}^{N}\Gamma_{sb}(\varphi_{b} - \varphi_{b}^{+})\mathbf{I} + O(\varepsilon^{2}),$$
(3.2.12)

$$\rho_{c}^{+}\mu_{c} = -\iota_{c} - \lambda_{sc}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \lambda_{cb} \left(\rho_{b} - \rho_{b}^{+}\right) / \rho_{b}^{+} + \tau_{c} \left(\theta - \theta^{+}\right) - \sum_{b=2}^{N} \Gamma_{cb} \left(\varphi_{b} - \varphi_{b}^{+}\right) + O(\varepsilon^{2}) \quad (3.2.13)$$

and

$$\sigma_{c} = -\Theta_{c}(\theta - \theta^{+}) - \Gamma_{sc}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \Gamma_{bc}(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} - \sum_{b=2}^{N} \Phi_{cb}(\phi_{b} - \phi_{b}^{+}) + O(\varepsilon^{2}). \quad (3.2.14)$$

In writing (3.2.12), equation (2.4.9) has been used to introduce the reference density for the solid  $\rho_{s_{R}}$ . In addition, the quantity det  $\mathbf{F}_{s}$  has been replaced by the approximate expression

$$\det \mathbf{F}_{s} = 1 + \operatorname{tr} \mathbf{E}_{s} + O(\varepsilon^{2}) \tag{3.2.15}$$

It follows from (2.3.13), (2.6.8), (2.7.8) that

$$\mathbf{T}_{\mathrm{I}} = -\sum_{a=1}^{N} \rho_a \mathbf{K}_a + \Psi_{\mathrm{I}} \mathbf{I}$$
(3.2.16)

Given (3.1.5), this result becomes

$$\mathbf{T}_{\mathrm{I}} = -\rho_{s}\mathbf{K}_{s} - \sum_{b=2}^{N} \rho_{b}\mu_{b}\mathbf{I} + \Psi_{\mathrm{I}}\mathbf{I}$$
(3.2.17)

By use of the same approximations that yielded (3.2.9), (3.2.11), (3.2.12) and (3.2.13), it follows from (3.2.17) that

$$\mathbf{T}_{I} = (\Psi_{I}^{+} + \iota_{s} + \sum_{c=2}^{N} \iota_{c})\mathbf{I} + (\lambda_{ss} + \sum_{c=2}^{N} \lambda_{sc} + \iota_{s})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2(\mu_{ss} + \iota_{s})\mathbf{E}_{s} - \sum_{b=2}^{N} (\lambda_{sb} + \sum_{c=2}^{N} \lambda_{cb})((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})\mathbf{I} - (\tau_{s} + \sum_{c=2}^{N} \tau_{c} - \alpha)(\theta - \theta^{+})\mathbf{I} + \sum_{b=2}^{N} (\Gamma_{sb} + \sum_{c=2}^{N} \Gamma_{cb})(\varphi_{b} - \varphi_{b}^{+}))\mathbf{I} + O(\varepsilon^{2})$$
(3.2.18)

Because we are pursuing a linearization about the static solution described at the beginning of this section, in this uniform state everything is a constant. It is convenient, at this point, to simplify matters by making the formal assumption that the constant coefficients  $\Psi_a^+$ ,  $t_s$  and  $t_b$ , for b = 2,...,N, are zero. The result is that the static solution has zero partial free energy  $\Psi_a$ , zero inner part of the free energy for the mixture  $\Psi_1$ , zero chemical potentials  $\mathbf{K}_s$  and  $\mu_c$ . It follow from (3.2.18) that  $\mathbf{T}_I$  vanishes at the static solution. In other words, there is no residual stress for the static solution. As a result of this simplification, equations (3.2.9), (3.2.12), (3.2.13) and (3.2.18) reduce to

$$\Psi_{I} = \alpha(\theta - \theta^{+}) - \frac{1}{2}\nu(\theta - \theta^{+})^{2} - \tau_{s}(\theta - \theta^{+})(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \tau_{b}(\theta - \theta^{+})(\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+} + \sum_{b=2}^{N} \Theta_{b}(\theta - \theta^{+})(\varphi_{b} - \varphi_{b}^{+}) + \frac{1}{2}\lambda_{ss}(\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N}\lambda_{sb}(\operatorname{tr} \mathbf{E}_{s})(\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+} + \frac{1}{2}\sum_{b=2}^{N}\sum_{c=2}^{N}\lambda_{bc}(\rho_{b} - \rho_{b}^{+})(\rho_{c} - \rho_{c}^{+})/\rho_{b}^{+}\rho_{c}^{+} + \mu_{ss}\operatorname{tr}(\mathbf{E}_{s}\mathbf{E}_{s}) + \sum_{b=2}^{N}\Gamma_{sb}(\operatorname{tr} \mathbf{E}_{s})(\varphi_{b} - \varphi_{b}^{+}) - \sum_{b=2}^{N}\sum_{c=2}^{N}\Gamma_{bc}((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + \frac{1}{2}\sum_{b=2}^{N}\sum_{c=2}^{N}\Phi_{bc}(\varphi_{b} - \varphi_{b}^{+})(\varphi_{c} - \varphi_{c}^{+}) + O(\varepsilon^{3})$$

$$(3.2.19)$$

$$\rho_{s_{R}}\mathbf{K}_{s} = -\frac{\partial\Psi_{I}}{\partial\mathbf{E}_{s}} = -\lambda_{ss}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{b=2}^{N}\lambda_{sb}((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})\mathbf{I} + \tau_{s}(\theta - \theta^{+})\mathbf{I} - \sum_{b=2}^{N}\Gamma_{sb}(\varphi_{b} - \varphi_{b}^{+})\mathbf{I} + O(\varepsilon^{2}),$$
(3.2.20)

$$\rho_{c}^{+}\mu_{c} = \rho_{c}^{+} \frac{\partial \Psi_{I}}{\partial \rho_{c}} = -\lambda_{sc}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \lambda_{cb} (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \tau_{c} (\theta - \theta^{+}) - \sum_{b=2}^{N} \Gamma_{cb} (\varphi_{b} - \varphi_{b}^{+}) + O(\varepsilon^{2}).$$
(3.2.21)

$$\mathbf{T}_{\mathbf{I}} = (\lambda_{ss} + \sum_{c=2}^{N} \lambda_{sc})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{b=2}^{N} (\lambda_{sb} + \sum_{c=2}^{N} \lambda_{cb})((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})\mathbf{I} - (\tau_{s} + \sum_{c=2}^{N} \tau_{c} - \alpha)(\theta - \theta^{+})\mathbf{I} + \sum_{b=2}^{N} (\Gamma_{sb} + \sum_{c=2}^{N} \Gamma_{cb})(\varphi_{b} - \varphi_{b}^{+}))\mathbf{I} + O(\varepsilon^{2})$$

$$(3.2.22)$$

The corresponding linearized expressions for **m**,  $\hat{\mathbf{f}}_c$  and  $\omega_c$  turn out to be

$$\mathbf{m} = -\kappa \operatorname{GRAD} \theta - \sum_{b=2}^{N} \varsigma_{b} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) + O(\varepsilon^{2}), \qquad (3.2.23)$$

$$\hat{\mathbf{f}}_{c} = -\gamma_{c} \operatorname{GRAD} \theta - \sum_{b=2}^{N} \xi_{cb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) + O(\varepsilon^{2})$$
(3.2.24)

and

Chapter 3

$$\omega_{c} = \sum_{b=2}^{N} \sum_{d=2}^{N} \Lambda_{cb} \Phi_{bd}^{-1} \sigma_{d} + O(\varepsilon^{2})$$
(3.2.25)

where the  $\Phi_{bd}^{-1}$  are the inverse elements of the coefficients  $\Phi_{bd}$  introduced in (3.2.10). It follows from (3.2.9) that the  $\Phi_{bd}$  are given by

$$\Phi_{bd} = -\frac{\partial \sigma_b}{\partial \varphi_d} = \frac{\partial^2 \Psi_1}{\partial \varphi_b \partial \varphi_d}.$$
(3.2.26)

The material constant  $\kappa$  in (3.2.23) is the *thermal conductivity* of the mixture, and the coefficients  $\xi_{cb}$  in (3.2.24) are the *drag coefficients*. As indicated in Section 3.1, the assumed inversion of (3.1.8) insures that the inverse elements  $\Phi_{bd}^{-1}$  exist. Given (3.2.23), (3.2.24) and (3.2.25), the residual entropy inequality (3.1.6) yields

$$\frac{1}{\theta^{+}} \kappa \mathbf{a} \cdot \mathbf{a} + \mathbf{a} \cdot \sum_{b=2}^{N} (\varsigma_{b} / \theta^{+} + \gamma_{b}) \mathbf{d}_{b} + \sum_{b=2}^{N} \sum_{c=2}^{N} \xi_{bc} \mathbf{d}_{b} \cdot \mathbf{d}_{c} \ge 0$$
(3.2.27)

and

$$\sum_{c=2}^{N} \sum_{b=2}^{N} \sum_{d=2}^{N} \sigma_{c} \Lambda_{cb} \Phi_{bd}^{-1} \sigma_{d} \ge 0$$
(3.2.28)

Equation (3.2.27) must hold for all vectors **a** and **d**<sub>2</sub>,..., **d**<sub>N</sub>. Equation (3.2.28) must hold for all scalars  $\sigma_2, ..., \sigma_N$ .

## 3.3 Immiscible Mixtures-Field Equations for the Linearized Model

When one simply drops the terms  $O(\varepsilon^3)$  in (3.2.19) and the terms  $O(\varepsilon^2)$  in (3.2.11), (3.2.14), (3.2.23), (3.2.24), (3.2.25), (3.2.20) and (3.2.21), one has the constitutive equations appropriate for the linearized model. The field equations which govern motions near the static solution turn out to be

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\lambda_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \sum_{b=2}^{N} \lambda_{sb} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{b}) + (\alpha_{s} - \tau_{s} - \gamma_{s}) \operatorname{GRAD} \theta$$
(3.3.1)
$$+ \sum_{b=2}^{N} \Gamma_{sb} \operatorname{GRAD} \varphi_{b} - \sum_{b=2}^{N} \xi_{sb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$

$$\rho_{c}^{+} \frac{\partial^{2} \mathbf{w}_{c}}{\partial t^{2}} = \sum_{b=2}^{N} \lambda_{cb} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{b}) + \lambda_{sc} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + (\alpha_{c} - \tau_{c} - \gamma_{c}) \operatorname{GRAD} \theta + \sum_{b=2}^{N} \Gamma_{cb} \operatorname{GRAD} \varphi_{b} - \sum_{b=2}^{N} \xi_{cb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(3.3.2)

$$c_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left( (\tau_{s} - \alpha_{s} - \varsigma_{s}/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} + \sum_{b=2}^{N} (\tau_{b} - \alpha_{b} - \varsigma_{b}/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{b}}{\partial t} - \sum_{b=2}^{N} \Theta_{b}\frac{\partial\varphi_{b}}{\partial t} \right) \quad (3.3.3)$$
$$= \kappa \operatorname{Div}(\operatorname{GRAD}\theta)$$

$$\frac{\partial \phi_c}{\partial t} = -\sum_{b=2}^N \Lambda_{cb} \left( \phi_b - \phi_b^+ + \sum_{d=2}^N \Phi_{bd}^{-1} \Gamma_{sd} \operatorname{Div} \mathbf{w}_s + \sum_{d=2}^N \sum_{e=2}^N \Phi_{bd}^{-1} \Gamma_{ed} \operatorname{Div} \mathbf{w}_e + \sum_{d=2}^N \Phi_{bd}^{-1} \Theta_d (\theta - \theta^+) \right) (3.3.4)$$

where

$$\xi_{sb} = -\sum_{c=2}^{N} \xi_{cb}$$
(3.3.5)

$$\gamma_s = -\sum_{c=2}^N \gamma_c \tag{3.3.6}$$

and

$$\varsigma_s = -\sum_{c=2}^N \varsigma_c. \tag{3.3.7}$$

In addition, the coefficient  $c_v$  in (3.3.3) is the *specific heat at constant volume* for the mixture. This quantity is related to the coefficient v in (3.2.19) by the formula

$$c_{v} = \theta^{+} v \tag{3.3.8}$$

Among the many formulas which were used to deduce (3.3.1) through (3.3.4) are the expressions

$$\left(\rho_{b}-\rho_{b}^{+}\right)/\rho_{b}^{+}=-\operatorname{Div}\mathbf{w}_{b}+O(\varepsilon^{2}),$$
(3.3.9)

for b = 2, ..., N. These results follow by integration of (2.4.1) to obtain

$$\rho_b \left| \det \mathbf{F}_b \right| = \rho_b^+ \tag{3.3.10}$$

and the utilization of the following approximation for  $|\det \mathbf{F}_b|$ 

$$\left|\det \mathbf{F}_{b}\right| = 1 + \operatorname{Div} \mathbf{w}_{b} + O(\varepsilon^{2})$$
 (3.3.11)

Equations (3.3.1), (3.3.2), (3.3.3) and (3.3.4) can be shown to have unique solutions for various initial-boundary value problems if the quadratic form  $\Sigma$ , defined by

$$\Sigma = \Psi_{I} + \rho \eta (\theta - \theta^{+}) = \frac{1}{2} \frac{c_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} + \frac{1}{2} \lambda_{ss} (tr \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \lambda_{sb} (tr \mathbf{E}_{s}) (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}$$

$$+ \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \lambda_{bc} (\rho_{b} - \rho_{b}^{+}) (\rho_{c} - \rho_{c}^{+}) / \rho_{b}^{+} \rho_{c}^{+} + \mu_{ss} tr (\mathbf{E}_{s} \mathbf{E}_{s})$$

$$+ \sum_{b=2}^{N} \Gamma_{sb} (tr \mathbf{E}_{s}) (\varphi_{b} - \varphi_{b}^{+}) - \sum_{b=2}^{N} \sum_{c=2}^{N} \Gamma_{bc} ((\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}) (\varphi_{c} - \varphi_{c}^{+})$$

$$+ \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \Phi_{bc} (\varphi_{b} - \varphi_{b}^{+}) (\varphi_{c} - \varphi_{c}^{+})$$
(3.3.12)

is positive definite. Given (3.3.12), one can establish that

$$\frac{\partial}{\partial t} \int_{\Theta} \left( \Sigma + \frac{1}{2} \rho_{s_{R}} \frac{\partial \mathbf{w}_{s}}{\partial t} \cdot \frac{\partial \mathbf{w}_{s}}{\partial t} + \frac{1}{2} \sum_{b=2}^{N} \rho_{b}^{+} \frac{\partial \mathbf{w}_{b}}{\partial t} \cdot \frac{\partial \mathbf{w}_{b}}{\partial t} \right) dv = \oint_{\partial \Theta} (\mathbf{t} - \frac{\theta - \theta^{+}}{\theta^{+}} \mathbf{m}) \cdot d\mathbf{s}$$

$$-\int_{\Theta} \left( \sum_{c=2}^{N} \sum_{b=2}^{N} \sum_{d=2}^{N} \sigma_{c} \Lambda_{cb} \Phi_{bd}^{-1} \sigma_{c} \right) dv \qquad (3.3.13)$$

$$-\int_{\Theta} \left( \frac{1}{\theta^{+}} \kappa \operatorname{GRAD} \theta \cdot \operatorname{GRAD} \theta + \operatorname{GRAD} \theta \cdot \sum_{b=2}^{N} (\varsigma_{b} / \theta^{+} + \gamma_{b}) \frac{\partial \mathbf{w}_{b}}{\partial t} + \sum_{b=2}^{N} \sum_{c=2}^{N} \xi_{cb} \frac{\partial \mathbf{w}_{c}}{\partial t} \cdot \frac{\partial \mathbf{w}_{b}}{\partial t} \right) dv$$

where **t** is a vector defined by

$$\mathbf{t} = -(\rho_{s_{R}}\mathbf{K}_{s} - \alpha_{s}(\theta - \theta^{+})\mathbf{I})\frac{\partial \mathbf{w}_{s}}{\partial t} - \sum_{b=2}^{N}(\rho_{b}^{+}\mu_{b} - \alpha_{b}(\theta - \theta^{+}))\frac{\partial \mathbf{w}_{b}}{\partial t}$$
(3.3.14)

It is convenient for use later on to rearrange (3.3.14) and write the result as

$$\mathbf{t} = -\left(\rho_{s_{R}}\mathbf{K}_{s} - \alpha_{s}(\theta - \theta^{+})\mathbf{I} + \sum_{b=2}^{N}(\rho_{b}^{+}\mu_{b} - \alpha_{b}(\theta - \theta^{+}))\mathbf{I}\right)\frac{\partial\mathbf{w}_{s}}{\partial t}$$
  
$$-\sum_{b=2}^{N}(\rho_{b}^{+}\mu_{b} - \alpha_{b}(\theta - \theta^{+}))\left(\frac{\partial\mathbf{w}_{b}}{\partial t} - \frac{\partial\mathbf{w}_{s}}{\partial t}\right)$$
(3.3.15)

where the symmetry of  $T_I$  has been used. It is easily seen from (3.2.19), (3.2.20) and (3.2.21) that (3.2.17) can be written

$$\mathbf{T}_{\mathbf{I}} = -\left(\rho_{s_{R}}\mathbf{K}_{s} - \alpha_{s}(\theta - \theta^{+})\mathbf{I} + \sum_{b=2}^{N}(\rho_{b}^{+}\mu_{b} - \alpha_{b}(\theta - \theta^{+}))\mathbf{I}\right) + O(\varepsilon^{2})$$
(3.3.16)

Therefore, consistent with the linearization adopted in this section, (3.3.15) can be replaced by

$$\mathbf{t} = \mathbf{T}_{\mathrm{I}} \frac{\partial \mathbf{w}_{s}}{\partial t} - \sum_{b=2}^{N} (\rho_{b}^{+} \mu_{b} - \alpha_{b} (\theta - \theta^{+})) \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(3.3.17)

Given (3.2.27) and (3.2.28), it follows from (3.3.13) that the following inequality must hold

$$\frac{\partial}{\partial t} \int_{\mathscr{H}} \left( \Sigma + \frac{1}{2} \rho_{s_{R}} \frac{\partial \mathbf{w}_{s}}{\partial t} \cdot \frac{\partial \mathbf{w}_{s}}{\partial t} + \frac{1}{2} \sum_{b=2}^{N} \rho_{b}^{+} \frac{\partial \mathbf{w}_{b}}{\partial t} \cdot \frac{\partial \mathbf{w}_{b}}{\partial t} \right) dv \leq \bigoplus_{\partial \mathscr{H}} (\mathbf{t} - \frac{\theta - \theta^{+}}{\theta^{+}} \mathbf{m}) \cdot d\mathbf{s}$$
(3.3.18)

The reader should note that (3.3.18) could have been derived by specializing (2.6.16) to the linearized case under discussion here.

The reader with experience in the classical theory of elasticity will recognize how (3.3.18) can be used to construct a uniqueness theorem if  $\Sigma$  is positive definite [Ref. 5,6]. If  $\Sigma$  is assumed to be positive definite, it is necessary and sufficient that the following restrictions hold for the material constants:

$$c_{\rm v} > 0,$$
 (3.3.19)

$$\mu_s > 0 \tag{3.3.20}$$

and the  $(2N-1) \times (2N-1)$  symmetric matrix

is positive definite. Among the implications of the positive definite nature of this matrix is that the  $(N-1) \times (N-1)$  matrix  $[\Phi_{bc}]$  is positive definite. In the following, we shall assume that  $\Sigma$  is positive definite and utilize the necessary restrictions on the properties of the mixture which appear in (3.3.12)

Equation (3.3.4) contains two special cases which need to be mentioned here. If the volume fractions  $\varphi_2, \varphi_3, ..., \varphi_N$  are taken to be constants, then (3.3.4) yields this special case by taking  $\Lambda_{cb}$  to be zero for c, b = 2, ..., N. It is convenient to refer to this case as the *frozen* one. The most interesting case arises when the volume fractions adjust instantly to their equilibrium values defined by the vanishing of the affinities  $\sigma_2, \sigma_3, ..., \sigma_N$ . It is useful to view this case as one which also follows from (3.3.4). The coefficients  $\Lambda_{cb}$  have the physical dimension of inverse time. As a result, they can be thought of as defining characteristic times of relaxation for the volume fractions to their equilibrium values. If we formally allow these characteristic times to approach zero as a limit, (3.3.4) yields

$$\varphi_{b} - \varphi_{b}^{+} = -\sum_{d=2}^{N} \Phi_{bd}^{-1} \Gamma_{sd} \operatorname{tr} \mathbf{E}_{s} + \sum_{d=2}^{N} \sum_{e=2}^{N} \Phi_{bd}^{-1} \Gamma_{ed} \left( (\rho_{e} - \rho_{e}^{+}) / \rho_{e}^{+} \right) - \sum_{d=2}^{N} \Phi_{bd}^{-1} \Theta_{d} \left( \theta - \theta^{+} \right)$$
(3.3.21)

where (3.3.9) and (3.2.8) have been used. Equation (3.3.21) gives the volume fractions of the fluid constituents in terms of the stress on the solid, the densities of the fluids and the temperature. When one assumes the validity of (3.3.21), the mixture is said to be in *shifting equilibrium*. Equation (3.3.21) can be used to eliminate the volume fractions from the other constitutive equations for this model. The results of this elimination are as follows:

$$\Psi_{I} = \alpha(\theta - \theta^{+}) - \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} - \overline{\tau}_{s} (\theta - \theta^{+}) (\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \overline{\tau}_{b} (\theta - \theta^{+}) (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \overline{\lambda}_{sb} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{bc} (\rho_{b} - \rho_{b}^{+}) (\rho_{c} - \rho_{c}^{+}) / \rho_{b}^{+} \rho_{c}^{+} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$
(3.3.22)

$$\rho\eta = -\frac{\partial \Psi_{I}(\theta, \mathbf{E}_{s}, \rho_{b})}{\partial \theta} = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + \overline{\tau}_{s}(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \overline{\tau}_{b}(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}$$
(3.3.23)

$$\rho_{s_{R}}\mathbf{K}_{s} = -\frac{\partial\Psi_{I}(\theta, \mathbf{E}_{s}, \rho_{b})}{\partial\mathbf{E}_{s}} = -\overline{\lambda}_{ss}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{b=2}^{N}\overline{\lambda}_{sb}((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})\mathbf{I} + \overline{\tau}_{s}(\theta - \theta^{+})\mathbf{I} (3.3.24)$$

$$\rho_{c}^{+}\mu_{c} = \rho_{c}^{+} \frac{\partial \Psi_{I}(\theta, \mathbf{E}_{s}, \rho_{b})}{\partial \rho_{c}} = \overline{\tau}_{c}(\theta - \theta^{+}) - \overline{\lambda}_{sc}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \overline{\lambda}_{cb}(\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}$$
(3.3.25)

$$\mathbf{T}_{\mathbf{I}} = (\overline{\lambda}_{ss} + \sum_{c=2}^{N} \overline{\lambda}_{sc})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{b=2}^{N} (\overline{\lambda}_{sb} + \sum_{c=2}^{N} \overline{\lambda}_{cb})((\rho_{b} - \rho_{b}^{+})/\rho_{b}^{+})\mathbf{I} - (\overline{\tau}_{s} + \sum_{c=2}^{N} \overline{\tau}_{c} - \alpha)(\theta - \theta^{+})\mathbf{I}$$
(3.3.26)

where

$$\overline{c}_{v} = c_{v} + \theta^{+} \sum_{b=2}^{N} \sum_{d=2}^{N} \Theta_{b} \Phi_{bd}^{-1} \Theta_{d}$$
(3.3.27)

$$\overline{\tau}_{s} = \tau_{s} + \sum_{b=2}^{N} \sum_{d=2}^{N} \Theta_{b} \Phi_{bd}^{-1} \Gamma_{sd}$$
(3.3.28)

$$\overline{\tau}_c = \tau_c + \sum_{b=2}^N \sum_{d=2}^N \Theta_b \Phi_{bd}^{-1} \Gamma_{cd}$$
(3.3.29)

$$\overline{\lambda}_{ss} = \lambda_{ss} - \sum_{b=2}^{N} \sum_{d=2}^{N} \Gamma_{sb} \Phi_{bd}^{-1} \Gamma_{sd}$$
(3.3.30)

$$\overline{\lambda}_{sc} = \lambda_{sc} - \sum_{b=2}^{N} \sum_{d=2}^{N} \Gamma_{sb} \Phi_{bd}^{-1} \Gamma_{cd}, \qquad (3.3.31)$$

and

$$\overline{\lambda}_{cb} = \lambda_{cb} - \sum_{e=2}^{N} \sum_{d=2}^{N} \Gamma_{ce} \Phi_{ed}^{-1} \Gamma_{bd}.$$
(3.3.32)

The coefficients (3.3.27) through (3.3.32) are the *shifting equilibrium* properties of the mixture. It is sometimes convenient to refer to the material coefficients introduced in (3.2.9) as the *frozen equilibrium* properties of the mixture.

The field equations in the shifting equilibrium case which replace (3.3.1) through (3.3.3) are easily shown to be

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \sum_{b=2}^{N} \overline{\lambda}_{sb} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{b}) + (\alpha_{s} - \overline{\tau}_{s} - \gamma_{s}) \operatorname{GRAD} \theta - \sum_{b=2}^{N} \xi_{sb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(3.3.33)

$$\rho_{c}^{+} \frac{\partial^{2} \mathbf{w}_{c}}{\partial t^{2}} = \sum_{b=2}^{N} \overline{\lambda}_{cb} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{b}) + \overline{\lambda}_{sc} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + (\alpha_{c} - \overline{\tau}_{c} - \gamma_{c}) \operatorname{GRAD} \theta$$
$$-\sum_{b=2}^{N} \xi_{cb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(3.3.34)

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left( (\overline{\tau}_{s} - \alpha_{s} - \varsigma_{s}/\theta^{+}) \operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} + \sum_{b=2}^{N} (\overline{\tau}_{b} - \alpha_{b} - \varsigma_{b}/\theta^{+}) \operatorname{Div}\frac{\partial\mathbf{w}_{b}}{\partial t} \right)$$

$$= \kappa \operatorname{Div}(\operatorname{GRAD}\theta)$$
(3.3.35)

Solutions of (3.3.33) through (3.3.35) must also obey the inequality (3.3.18), where  $\Sigma$  is again defined by  $(3.3.12)_1$ . Given (3.3.22) and (3.3.23), the explicit for of the quadratic form  $\Sigma$  turns out to be

$$\Sigma = \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} + \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \overline{\lambda}_{sb} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{bc} (\rho_{b} - \rho_{b}^{+}) (\rho_{c} - \rho_{c}^{+}) / \rho_{b}^{+} \rho_{c}^{+} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$
(3.3.36)

The fact that  $\Sigma$  is assumed to be positive definite yields

$$\overline{c}_{\rm v} > 0 \tag{3.3.37}$$

$$\mu_{ss} > 0$$
 (3.3.38)

and the  $N \times N$  symmetric matrix

$$\begin{bmatrix} \overline{\lambda}_{ss} + \frac{2}{3} \overline{\mu}_{ss} & \overline{\lambda}_{s2} & \dots & \overline{\lambda}_{sN} \\ \overline{\lambda}_{s2} & \overline{\lambda}_{22} & \dots & \overline{\lambda}_{2N} \\ \vdots & \vdots & \vdots & \vdots \\ \vdots & \vdots & \vdots & \vdots \\ \overline{\lambda}_{sN} & \overline{\lambda}_{2N} & \dots & \overline{\lambda}_{NN} \end{bmatrix}$$

is positive definite.

Frequently, it is physically acceptable to assume that the mixture is *isothermal*. This special case can be justified in cases where the mixture is a perfect conductor. This case is approached

when the thermal conductivity  $\kappa$  is large. As equation (3.3.3) illustrates, if  $\kappa$  is allowed to grow without bound, then the temperature must obey Div(GRAD $\theta$ ) = 0. With suitable boundary conditions, the only solution of this equation is  $\theta = \theta^+$ . A *nonconductor* is a mixture for which the constitutive equations are independent of the temperature gradient. In this case, it follows from (3.2.23) and (3.2.24) that  $\kappa = \gamma_c = 0$  for c = 2,..., N. For a nonconductor, (3.3.3) can be used to eliminate the temperature from the remaining field equations. As in the shifting equilibrium case, one can rewrite the governing constitutive and field equations in terms of a new set of material constants. These constants, which are defined by complicated formulas, are the mixture versions of the isotropic material constants one encounters in the classical theory of elasticity. It is also interesting to point out that our constitutive and field equations contain the degenerate case where there is no relative motion between the constituents. This case can be reached formally by allowing the drag coefficients,  $\xi_{cb}$ , to become large. It is a simple exercise to show that in the limit of infinite drag coefficients, equations (3.3.1) and (3.3.2) force the constituent velocities to all be equal.

In closing this section it is useful to note that (3.3.16) can be rearranged to yield

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\mathbf{T}_{I} + \sum_{b=2}^{N} \rho_{b}^{+} \mu_{b}\mathbf{I}) + \alpha(\theta - \theta^{+})\mathbf{I}, \qquad (3.3.39)$$

where the order term has been dropped. As we shall see later, equation (3.3.39) is frequently used in the applications to eliminate the formal appearance of the solid's chemical potential in favor of the stress  $T_{I}$ .

# References

- 1. BOWEN, R. M., Compressible Porous Media Models by Use of the Theory of Mixtures, *Intl. J. Engr. Sci.*, **20**, 697-736 (1982).
- 2. BOWEN, R. M., Thermochemistry of a Reacting Mixture of Elastic Materials with Diffusion, *Arch. Rational Mech. Anal.*, **34**, 97-127 (1969).
- 3. BOWEN, R. M., Thermochemistry of a Reacting Materials, J. Chem. Phys., 49, 1625-1637 (1968).
- 4. BOWEN, R. M., Erratum: Thermochemistry of a Reacting Materials, J. Chem. Phys., 50, 4601-4602 (1969).
- 5. BOWEN, R. M., Introduction to Continuum Mechanics for Engineers, Plenum Press, New York (1989).
- 6. GURTIN. M. E., *The Linear Theory of Elasticity*, Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), Springer Verlag, Berlin (1972).
- 7. CARLSON, D. E., *Linear Thermoelasticity*, Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), Springer Verlag, Berlin (1972).

# **Porous Elasticity Models With Pore Pressure**

In this chapter the model developed in Chapter 3 will be specialized so as to yield the concept of pore pressures. This special case is further specialized to the case of a binary mixture consisting of a fluid and a solid. As a final specialization, the case where each constituent is incompressible is discussed.

#### 4.1 Immiscible Mixtures-Definition of Pore Pressures

In Chapters 1 and 2 examples were discussed which involved the pore pressures of the fluids in the porous solid. In Chapter 1, these quantities were introduced as fundamental physical variables. In Section 7 of Chapter 2 the pore pressures were defined in terms of other parameters associated with the model of a rigid isotropic solid containing incompressible fluids. In this section the pore pressures will be defined in a somewhat more general fashion. For the most part the discussion in this section represents a generalization of a similar discussion in reference 1.

For the  $b^{th}$  fluid in a mixture consisting of a solid and N - 1 fluids, the *pore pressure* of the fluid is defined by

$$P_b = \int \gamma_b d\,\mu_b + \text{constant.} \tag{4.1.1}$$

In order for (4.1.1) to have meaning, the true density  $\gamma_b = \rho_b/\varphi_b$  must be a function of the chemical potential  $\mu_b$ . A trivial functional relationship arises when the  $b^{th}$  fluid is incompressible. In this case,  $\gamma_b$  is a constant and (4.1.1) integrates directly. If the results (2.7.14) and (2.7.15) are combined with the definitions (2.7.18) and (2.7.19), one can see that (4.1.1) is implicit in the results of Section 2.7. When the  $b^{th}$  fluid is not incompressible, things get somewhat more complicated.

The relationship  $\gamma_b = \gamma_b(\mu_b)$  required by (4.1.1) is an extra constitutive equation which must be consistent with our previous constitutive assumptions. Therefore, we must investigate the restrictions that are implied by the assumed existence of the function  $\gamma_b = \gamma_b(\mu_b)$ . We shall always assume that this function can be inverted and written

$$\mu_b = \tilde{\mu}_b(\gamma_b). \tag{4.1.2}$$

As indicated above, equation (4.1.2) must be consistent with the constitutive assumptions made in Section 3.1. As (3.1.1, (2.7.8) and (3.1.7) show, the chemical potential  $\mu_b$  is given by

$$\mu_b = \mu_b(\theta, \mathbf{F}_s, \rho_d, \varphi_d). \tag{4.1.3}$$

One case where (4.1.3) and (4.1.2) are consistent is when (4.1.3) degenerates trivially into (4.1.2). If we exclude this possibility, then we must look for a more complicated reconciliation of (4.1.2) and (4.1.3). If (4.1.2) and (4.1.3) are consistent, then it must be true that

$$\gamma_b = \tilde{\mu}_b^{-1} \circ \mu_b(\theta, \mathbf{F}_s, \rho_d, \varphi_d) \tag{4.1.4}$$

or, with the definition (2.1.16),

$$\rho_b / \varphi_b = \tilde{\mu}_b^{-1} \circ \mu_b(\theta, \mathbf{F}_s, \rho_d, \varphi_d).$$
(4.1.5)

Equation (4.1.5) can be viewed as a set of *N* - 1 equations for the determination of  $\rho_b$ , b = 2,...,N, in terms of  $\theta$ ,  $\mathbf{F}_s, \varphi_2, ..., \varphi_N$ . Relationships such as (4.1.5) are not generally consistent with our constitutive assumptions. However, if we assume that the relaxation process governed by (3.1.12) occurs instantaneously, then (3.1.12) is replaced by

$$\sigma_b(\theta, \mathbf{F}_s, \rho_d, \varphi_d) = 0 \tag{4.1.6}$$

Equation (3.4.6) does have the potential of being solved to obtain (4.1.5). If we now restrict the discussion to the linearized model discussed in Sections 3.2 and 3.3, it is possible to construct the restrictions on the constitutive functions which follow from equation (4.1.1).

In the linearized approximation, it is readily established that (4.1.1) and (4.1.2) become

$$P_b = \gamma_b^+ \mu_b \tag{4.1.7}$$

and

$$\rho_b^+ \mu_b = \Xi_b \left( \frac{\gamma_b - \gamma_b^+}{\gamma_b^+} \right) \tag{4.1.8}$$

respectively, where from (1.1.13),

$$\frac{\gamma_{b} - \gamma_{b}^{+}}{\gamma_{b}^{+}} = \frac{\rho_{b} - \rho_{b}^{+}}{\rho_{b}^{+}} - \frac{\varphi_{b} - \varphi_{b}^{+}}{\varphi_{b}^{+}}$$
(4.1.9)

In equation (4.1.7), the assumption has been made that the pore pressure vanishes in the state where the chemical potential vanishes. The quantity  $\Xi_b$  in (4.1.8) is a constant material property which is to be determined. The inversion which was assumed in order to reach (4.1.2) implies that  $\Xi_b \neq 0$ .

Equation (4.1.8) must be consistent with the fact that  $\rho_b^+ \mu_b$  is given by (3.3.25) and  $\varphi_b - \varphi_b^+$  is given by the linearized version of (4.1.6), i.e. by (3.3.21). If (4.1.9) is substituted into (4.1.8) and if (3.3.25) and (3.3.21) are substituted into the result, the resulting formula must hold identically in the variables  $tr\mathbf{E}_s$ ,  $\theta$  and  $\rho_d$ , for d = 2,...,N. As a result, the following results can easily be obtained

$$\varphi_b^+ \overline{\lambda}_{sb} = -\Xi_b \sum_{d=2}^N \Phi_{bd}^{-1} \Gamma_{sd}$$
(4.1.10)

$$\varphi_b^+ \overline{\tau}_b = \Xi_b \sum_{d=2}^N \Phi_{bd}^{-1} \Theta_d \tag{4.1.11}$$

and

$$\varphi_b^+ \overline{\lambda}_{be} = \Xi_b \left[ \varphi_b^+ \delta_{be} - \sum_{d=2}^N \Phi_{bd}^{-1} \Gamma_{ed} \right]$$
(4.1.12)

for fixed *b* and for e = 2,...,N. In (4.1.12), the quantity  $\delta_{be}$  denotes the usual Kronecker delta. If every fluid has a pore pressure defined by (4.1.1), then equations (4.1.10) through (4.1.12) represent a system of  $2(N-1) + (N-1)^2$  equations which relate the various material constants. If the assumption is made that  $\overline{\lambda}_{se} \neq 0$ , for e = 2,...,N, it is a straight forward manipulation to derive from these equations the following results:

$$\Xi_{b} = -\frac{\varphi_{b}^{+}\overline{\lambda}_{sb}}{\sum_{d=2}^{N} \Phi_{bd}^{-1}\Gamma_{sd}}$$
(4.1.13)

$$\Theta_b = -\sum_{e=2}^N \left( \frac{\overline{\tau}_e}{\overline{\lambda}_{se}} \Phi_{be} \left( \sum_{d=2}^N \Phi_{ed}^{-1} \Gamma_{sd} \right) \right)$$
(4.1.14)

and

$$\Gamma_{bc} = \varphi_b^+ \Phi_{bc} + \sum_{e=2}^N \left( \frac{\overline{\lambda}_{be}}{\overline{\lambda}_{se}} \Phi_{ce} \left( \sum_{d=2}^N \Phi_{ed}^{-1} \Gamma_{sd} \right) \right)$$
(4.1.15)

for b, c = 2,..., N. Equations (4.1.14) and (4.1.15) show that the pore pressure definition (4.1.1), when applied to every fluid in the mixture, forces the material constants  $\Theta_b$  and  $\Gamma_{bc}$ , which appear in (3.3.21) to be determined by the other material constants in the model. One would use equations (4.1.14) and (4.1.15) to eliminate  $\Theta_b$  and  $\Gamma_{bc}$ , for b, c = 2,..., N, from equation (3.3.21). Equation (4.1.12) enables us to construct explicitly the relationship (4.1.2). In particular, if (4.1.13), is substituted into (4.1.9) and the result is substituted into (4.1.7), the result can be written

$$P_b = K_b \left[ \frac{\gamma_b - \gamma_b^+}{\gamma_b^+} \right]$$
(4.1.16)

where  $K_b$  is the *bulk modulus* for the  $b^{th}$  fluid defined by

$$K_{b} = -\frac{\overline{\lambda}_{sb}}{\sum_{d=2}^{N} \Phi_{bd}^{-1} \Gamma_{sd}}$$
(4.1.17)

Because a properly defined bulk modulus is positive, it follows from (4.1.17) that we must require

$$-\frac{\overline{\lambda}_{sb}}{\sum_{d=2}^{N}\Phi_{bd}^{-1}\Gamma_{sd}} > 0 \tag{4.1.18}$$

for b = 2,...,N. It is possible to use (4.1.12) to establish that

$$-\frac{\overline{\lambda}_{sb}}{\sum_{d=2}^{N}\Phi_{bd}^{-1}\Gamma_{sd}} = \frac{\overline{\lambda}_{bb}}{\varphi_{b}^{+}} \frac{1}{\left(1 - \frac{1}{\varphi_{b}^{+}}\sum_{d=2}^{N}\Phi_{bd}^{-1}\Gamma_{bd}\right)}$$
(4.1.19)

Equation (4.1.19) provides an alternate formula for the bulk modulus for the  $b^{th}$  fluid. It also follows from (4.1.18) and (4.1.19) that

$$\frac{1}{\varphi_b^+} \sum_{d=2}^N \Phi_{bd}^{-1} \Gamma_{bd} < 1 \tag{4.1.20}$$

If we utilize the definition (4.1.17), equations (4.1.13), (4.1.14) and (4.1.15) can be written

$$\Xi_{b} = -\frac{\varphi_{b}^{+} \overline{\lambda}_{sb}}{\sum_{d=2}^{N} \Phi_{bd}^{-1} \Gamma_{sd}} = \frac{\varphi_{b}^{+}}{K_{b}}$$
(4.1.21)

$$\Theta_b = -\sum_{e=2}^N \left( \frac{\overline{\tau}_e}{\overline{\lambda}_{se}} \Phi_{be} \left( \sum_{d=2}^N \Phi_{ed}^{-1} \Gamma_{sd} \right) \right) = \sum_{e=2}^N \frac{\overline{\tau}_e \Phi_{be}}{K_e}$$
(4.1.22)

and

$$\Gamma_{bc} = \varphi_b^+ \Phi_{bc} + \sum_{e=2}^N \left( \frac{\overline{\lambda}_{be}}{\overline{\lambda}_{se}} \Phi_{ce} \left( \sum_{d=2}^N \Phi_{ed}^{-1} \Gamma_{sd} \right) \right) = \varphi_b^+ \Phi_{bc} - \sum_{e=2}^N \frac{\overline{\lambda}_{be} \Phi_{ec}}{K_e}$$
(4.1.23)

These results, allow (3.3.21) to be written

$$\varphi_{b} - \varphi_{b}^{+} = \frac{\overline{\lambda}_{sb}}{K_{b}} \operatorname{tr} \mathbf{E}_{s} + \varphi_{b}^{+} (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} - \frac{1}{K_{b}} \sum_{e=2}^{N} \overline{\lambda}_{be} (\rho_{e} - \rho_{e}^{+}) / \rho_{e}^{+} - \frac{\overline{\tau}_{b}}{K_{b}} (\theta - \theta^{+}) \qquad (4.1.24)$$

Given the result (4.1.7), the chemical potentials can be eliminated from (3.3.25) to obtain

$$\varphi_c^+ P_c = \overline{\tau}_c (\theta - \theta^+) - \overline{\lambda}_{sc} (\operatorname{tr} \mathbf{E}_s) + \sum_{b=2}^N \overline{\lambda}_{cb} (\rho_b - \rho_b^+) / \rho_b^+$$
(4.1.25)

Of course, (4.1.25) also follows from (4.1.24), (4.1.9) and (4.1.16). As in Section 2.7, the *capillary* pressures are defined by  $P_c - P_2$  for c = 3,..., N. With this definition, (4.1.25) can be used to obtain the capillary pressures in terms of  $\theta - \theta^+$ ,  $tr \mathbf{E}_s$  and  $(\rho_b - \rho_b^+)/\rho_b^+$ .

If equation (4.1.7) is used in (3.3.39), the result is

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\mathbf{T}_{I} + \sum_{b=2}^{N} \varphi_{b}^{+} P_{b}\mathbf{I}) + \alpha(\theta - \theta^{+})\mathbf{I}$$
(4.1.26)

It follows from (4.1.26) and (3.3.24) that

$$\mathbf{T}_{\mathrm{I}} + \sum_{b=2}^{N} \varphi_{b}^{+} P_{b} \mathbf{I} = \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s}) \mathbf{I} + 2\mu_{ss} \mathbf{E}_{s} - \sum_{b=2}^{N} \overline{\lambda}_{sb} ((\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}) \mathbf{I} - (\overline{\tau}_{s} - \alpha)(\theta - \theta^{+}) \mathbf{I}$$
(4.1.27)

Equation (3.3.26) can be used to derive an expression for the stress  $\mathbf{T}_{\rm I}$  in terms of the pore pressures. A first step in this derivation involves solving (4.1.25) for the fluid densities in terms of the pore pressures, the temperature and the strain of the solid. Because the quantity  $\Sigma$ , defined by (3.3.36), is positive definite, it is possible to show that the  $(N-1) \times (N-1)$  symmetric matrix  $\left[\overline{\lambda}_{cb}\right]$  is positive definite and, thus, invertible. Therefore, equation (4.1.25) yields

$$(\rho_b - \rho_b^+) / \rho_b^+ = \sum_{c=2}^N \overline{\lambda}_{bc}^{-1} \Big[ \varphi_c^+ P_c - \overline{\tau}_c (\theta - \theta^+) + \overline{\lambda}_{sc} (\operatorname{tr} \mathbf{E}_s) \Big]$$
(4.1.28)

If this result is substituted into (3.3,36) and rearranged, the result turns out to be

$$\mathbf{T}_{\mathbf{I}} = (\overline{\lambda}_{ss} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc}) (\operatorname{tr} \mathbf{E}_{s}) \mathbf{I} + 2\mu_{ss} \mathbf{E}_{s} - \sum_{c=2}^{N} \left( 1 + \sum_{b=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \right) \varphi_{c}^{+} P_{c} \mathbf{I}$$

$$- (\overline{\tau}_{s} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\tau}_{c} - \alpha) (\theta - \theta^{+}) \mathbf{I}$$

$$(4.1.29)$$

As in classical elasticity, it is possible to solve (4.1.29) for the stress  $\mathbf{T}_{1}$ . The result of this inversion is

$$2\mu_{ss}\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{\overline{\lambda}_{ss} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc}}{\overline{\lambda}_{sc} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc}} + \frac{2}{3} \mu_{ss}} \frac{1}{3} (\operatorname{tr} \mathbf{T}_{I}) + \frac{2\mu_{ss}}{3 \left( \overline{\lambda}_{ss} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc} \right) + 2\mu_{ss}} \sum_{c=2}^{N} \left( 1 + \sum_{b=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \right) \varphi_{c}^{+} P_{c} \mathbf{I}$$

$$+ \frac{2\mu_{ss}}{3 \left( \overline{\lambda}_{ss} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc} \right) + 2\mu_{ss}} (\overline{\tau}_{s} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\tau}_{c} - \alpha) (\theta - \theta^{+}) \mathbf{I}$$

$$(4.1.30)$$

As we shall see later, it is sometimes convenient to utilize the pore pressures as independent variables in the constitutive equations. Equation (4.1.28) is the key equation which allows for the introduction of pore pressures in place of fluid densities. Equation (4.1.29) is one result of this elimination. Likewise, one can use (4.1.28) to eliminate the fluid densities from the entropy formula (3.3.23). The result of this elimination is

$$\rho\eta = -\alpha + (\overline{c}_{v} + \theta^{+} \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\tau}_{b} \overline{\lambda}_{bc}^{-1} \overline{\tau}_{c}) (\theta - \theta^{+}) / \theta^{+}$$

$$+ (\overline{\tau}_{s} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\tau}_{b} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc}) (\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\tau}_{b} \overline{\lambda}_{bc}^{-1} \varphi_{c}^{+} P_{c}$$

$$(4.1.31)$$

A similar elimination can be made from (4.1.28). The result of this calculation is

$$(\varphi_b - \varphi_b^+) / \varphi_b^+ = \left(\sum_{c=2}^N \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc}\right) (\operatorname{tr} \mathbf{E}_s) + \sum_{c=2}^N \overline{\lambda}_{bc}^{-1} (\varphi_c^+ P_c) - \frac{P_b}{K_b} - \sum_{c=2}^N \overline{\lambda}_{bc}^{-1} \overline{\tau}_c (\theta - \theta^+)$$
(4.1.32)

Equations (4.1.29), (4.1.31) and (4.1.32) reflect a symmetry which is worthy of mention. It is best revealed if we use (4.1.26) to rewrite (4.1.29) as an equation for  $\rho_{s_R} \mathbf{K}_s$ . The result is

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\overline{\lambda}_{ss} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\lambda}_{sc})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{c=2}^{N} \left(\sum_{b=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1}\right) \varphi_{c}^{+} P_{c}\mathbf{I} + (\overline{\tau}_{s} - \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{sb} \overline{\lambda}_{bc}^{-1} \overline{\tau}_{c})(\theta - \theta^{+})\mathbf{I}$$

$$(4.1.33)$$

An examination of (4.1.31), (4.1.32) and (4.1.33) shows that

$$\frac{\partial \varphi_b}{\partial \mathbf{E}_s} = \rho_{s_R} \frac{\partial \mathbf{K}_s}{\partial P_b}, \qquad (4.1.34)$$

$$\rho_{s_{R}} \frac{\partial \mathbf{K}_{s}}{\partial \theta} = \frac{\partial \rho \eta}{\partial \mathbf{E}_{s}}$$
(4.1.35)

$$\frac{\partial \varphi_b}{\partial \theta} = \frac{\partial \rho \eta}{\partial P_b} \tag{4.1.36}$$

These special relationships arise from the definition (4.1.1) and the fact that  $\Psi_1$ , expressed as a function of  $(\theta, \mathbf{E}_s, \rho_b)$ , is a thermodynamic potential for the determination of  $\rho\eta, \mathbf{K}_s$  and  $\mu_b$ , for b=2,...,N. It is elementary to show that the function  $\Upsilon$ , defined by

$$\Upsilon = \Psi_{\mathrm{I}} - \sum_{b=2}^{N} \rho_b \mu_b \tag{4.1.37}$$

is a thermodynamic potential for  $\rho\eta$ , **K**<sub>s</sub>, and  $\rho_b$ , for b=2,...,N, in the sense that

$$\rho \eta = -\frac{\partial \Upsilon(\theta, \mathbf{E}_s, \mu_b)}{\partial \theta}$$
(4.1.38)

$$\rho_{s_R} \mathbf{K}_s = -\frac{\partial \Upsilon(\theta, \mathbf{E}_s, \mu_b)}{\partial \mathbf{E}_s}$$
(4.1.39)

and

$$\rho_c = -\frac{\partial \Upsilon(\theta, \mathbf{E}_s, \mu_b)}{\partial \mu_c} \tag{4.1.40}$$

If we represent the solution of (4.1.1) and (4.1.2) by  $\mu_b = \mu_b(P_b)$ , then one can define a function  $\tilde{\Upsilon}$  by

$$\tilde{\Upsilon}(\theta, \mathbf{E}_s, P_b) = \Upsilon(\theta, \mathbf{E}_s, \mu_b(P_b))$$
(4.1.41)

Given (4.1.38) through (4.1.41) and (4.1.1), it is possible to show that

$$\rho \eta = -\frac{\partial \tilde{\Upsilon}(\theta, \mathbf{E}_s, P_b)}{\partial \theta}$$
(4.1.42)

$$\rho_{s_{R}}\mathbf{K}_{s} = -\frac{\partial \tilde{\Upsilon}(\boldsymbol{\theta}, \mathbf{E}_{s}, P_{b})}{\partial \mathbf{E}_{s}}$$
(4.1.43)

$$\varphi_c = -\frac{\partial \Upsilon(\theta, \mathbf{E}_s, P_b)}{\partial P_c} \tag{4.1.44}$$

The essentials of the argument which produced (4.1.42), (4.1.43) and (4.1.44) can be found in the work of BIOT [Ref. 2, 3, 4].

As mentioned in Section 2.4, the *content* of the  $b^{th}$  fluid is defined by

$$m_b = \rho_b \left| \det \mathbf{F}_s \right| \tag{4.1.45}$$

Consistent with the linearized assumptions we have adopted, (4.1.45) can be replaced by

$$(m_b - m_b^+) / m_b^+ = (\rho_b - \rho_b^+) / \rho_b^+ + \operatorname{tr} \mathbf{E}_s$$
(4.1.46)

where the approximation (3.3.11) for  $|\det \mathbf{F}_s|$  has been used. Just as  $m_b$  represents the mass of the  $b^{th}$  fluid per unit of undeformed solid volume, the volume of the  $b^{th}$  fluid per unit of undeformed solid volume is  $v_b$  defined by

$$v_b = \varphi_b \left| \det \mathbf{F}_s \right| \tag{4.1.47}$$

The linearized form of (4.1.47) is

$$(v_b - v_b^+)/v_b^+ = (\varphi_b - \varphi_b^+)/\varphi_b^+ + \operatorname{tr} \mathbf{E}_s$$
 (4.1.48)

where, for notational convenience, we have not utilized  $v_b^+ = \varphi_b^+$ .

It is convenient in some applications to use (4.1.46) and (4.1.48) to eliminate fluid densities and volume fractions in favor of the quantities  $m_b$  and  $v_b$ . If (4.1.46) is used, equation (4.1.25) is replaced by

$$\varphi_c^+ P_c = \overline{\tau}_c (\theta - \theta^+) - (\overline{\lambda}_{sc} + \sum_{b=2}^N \overline{\lambda}_{cb})(\operatorname{tr} \mathbf{E}_s) + \sum_{b=2}^N \overline{\lambda}_{cb} (m_b - m_b^+) / m_b^+$$
(4.1.49)

Likewise, (3.3.23) is replaced by

$$\rho\eta = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + (\overline{\tau}_{s} + \sum_{b=2}^{N} \overline{\tau}_{b})(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \overline{\tau}_{b} (m_{b} - m_{b}^{+}) / m_{b}^{+}$$
(4.1.50)

and (3.3.24) is replaced by

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\overline{\lambda}_{ss} + \sum_{b=2}^{N} \overline{\lambda}_{sb})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{b=2}^{N} \overline{\lambda}_{sb}((m_{b} - m_{b}^{+})/m_{b}^{+})\mathbf{I} + \overline{\tau}_{s}(\theta - \theta^{+})\mathbf{I}$$
(4.1.51)

A similar rearrangement allows (3.3.26) to be written

$$\mathbf{T}_{\mathrm{I}} = (\overline{\lambda}_{ss} + 2\sum_{c=2}^{N} \overline{\lambda}_{sc} + \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{cb})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{b=2}^{N} (\overline{\lambda}_{sb} + \sum_{c=2}^{N} \overline{\lambda}_{cb})((m_{b} - m_{b}^{+})/m_{b}^{+})\mathbf{I} - (\overline{\tau}_{s} + \sum_{c=2}^{N} \overline{\tau}_{c} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.1.52)$$

Likewise, (4.1.24) can be written

$$\frac{\varphi_{b}-\varphi_{b}^{+}}{\varphi_{b}^{+}} = \left(\frac{\overline{\lambda}_{sb}+\sum_{e=2}^{N}\overline{\lambda}_{be}}{\varphi_{b}^{+}K_{b}}-1\right) \operatorname{tr} \mathbf{E}_{s} + (m_{b}-m_{b}^{+})/m_{b}^{+} - \frac{1}{\varphi_{b}^{+}K_{b}}\sum_{e=2}^{N}\overline{\lambda}_{be}(m_{e}-m_{e}^{+})/m_{e}^{+} - \frac{\overline{\tau}_{b}}{\varphi_{b}^{+}K_{b}}(\theta-\theta^{+}) (4.1.53)$$

A more convenient version of (4.1.53) follows by utilizing (4.1.48) to obtain

$$\frac{v_{b} - v_{b}^{+}}{v_{b}^{+}} = \left(\frac{\overline{\lambda}_{sb} + \sum_{e=2}^{N} \overline{\lambda}_{be}}{\varphi_{b}^{+} K_{b}}\right) \operatorname{tr} \mathbf{E}_{s} + (m_{b} - m_{b}^{+})/m_{b}^{+} - \frac{1}{\varphi_{b}^{+} K_{b}} \sum_{e=2}^{N} \overline{\lambda}_{be} (m_{e} - m_{e}^{+})/m_{e}^{+} - \frac{\overline{\tau}_{b}}{\varphi_{b}^{+} K_{b}} (\theta - \theta^{+}) (4.1.54)$$

When the independent variables are selected to be  $(\theta, \mathbf{E}_s, m_b)$ , it is possible to introduce a potential whose derivatives yield (4.1.49) and (4.1.52) [Ref. 3]. The quantity W defined by

$$W(\theta, \mathbf{E}_{s}, m_{b}) = \left|\det \mathbf{F}_{s}\right| \Psi_{I}(\theta, \mathbf{E}_{s}, m_{b})$$
(4.1.55)

is the inner part of the free energy of the mixture *per unit of undeformed solid volume*. If we now use (3.2.15) and (3.3.21), it follows that

$$W(\theta, \mathbf{E}_{s}, m_{b}) = \alpha(\theta - \theta^{+}) - \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} - (\overline{\tau}_{s} + \sum_{b=2}^{N} \overline{\tau}_{b} - \alpha)(\theta - \theta^{+})(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \overline{\tau}_{b} (\theta - \theta^{+}) \Big( (m_{b} - m_{b}^{+}) / m_{b}^{+} \Big) + \frac{1}{2} (\overline{\lambda}_{ss} + 2\sum_{b=2}^{N} \overline{\lambda}_{sb} + \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{bc}) (\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{b=2}^{N} \Big( \overline{\lambda}_{sb} + 2\sum_{c=2}^{N} \overline{\lambda}_{cb} \Big) (\operatorname{tr} \mathbf{E}_{s}) \Big( (m_{b} - m_{b}^{+}) / m_{b}^{+} \Big) + \frac{1}{2} \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{bc} \Big( (m_{b} - m_{b}^{+}) / m_{b}^{+} \Big) \Big( (m_{c} - m_{c}^{+}) / m_{c}^{+} \Big) + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$

$$(4.1.56)$$

It easily follows from (4.1.56) and (4.1.49) that

$$\varphi_{c}^{+}P_{c} = \frac{\partial W}{\partial m_{c}} = \overline{\tau}_{c}(\theta - \theta^{+}) - \left(\overline{\lambda}_{sc} + \sum_{b=2}^{N} \overline{\lambda}_{cb}\right)(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \overline{\lambda}_{cb}\left((m_{b} - m_{b}^{+})/m_{b}^{+}\right)$$
(4.1.57)

In a like fashion (4.1.56) and (4.1.52) yield

$$\mathbf{T}_{\mathbf{I}} = \frac{\partial W}{\partial \mathbf{E}_{s}} = (\overline{\lambda}_{ss} + 2\sum_{c=2}^{N} \overline{\lambda}_{sc} + \sum_{b=2}^{N} \sum_{c=2}^{N} \overline{\lambda}_{cb})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{b=2}^{N} (\overline{\lambda}_{sb} + \sum_{c=2}^{N} \overline{\lambda}_{cb})((m_{b} - m_{b}^{+})/m_{b}^{+})\mathbf{I} - (\overline{\tau}_{s} + \sum_{c=2}^{N} \overline{\tau}_{b} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.1.58)$$

The relationship of the result (4.1.50) to the quantity  $W(\theta, \mathbf{E}_s, m_b)$  is slightly more complicated. First, we introduce the entropy per unit of undeformed solid volume by the definition

$$\mathbf{H} = \left| \det \mathbf{F}_s \right| \rho \eta \tag{4.1.59}$$

Given (3.2.15) and (4.1.50), it easily follows that

$$\mathbf{H} = -\frac{\partial W}{\partial \theta} = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + (\overline{\tau}_{s} + \sum_{b=2}^{N} \overline{\tau}_{b} - \alpha)(\operatorname{tr} \mathbf{E}_{s}) - \sum_{b=2}^{N} \overline{\tau}_{b} (m_{b} - m_{b}^{+}) / m_{b}^{+} \qquad (4.1.60)$$

#### 4.2 Binary Immiscible Porous Materials With Pore Pressure

The special case of a single fluid contained in a solid is of special importance. Because of the complexity of the governing equations, very few problems have been examined with more than one fluid. For this reason, in this section the case where N = 2 will be examined.

For the case where the solid and fluid are compressible, equations (4.1.24), (3.3.22), (3.3.23), (3.3.24) and (4.1.25) reduce to

$$\frac{\varphi_f - \varphi_f^+}{\varphi_f^+} = \frac{\lambda_{sb}}{\varphi_f^+ K_f} \operatorname{tr} \mathbf{E}_s + \left(1 - \frac{\lambda_{ff}}{\varphi_f^+ K_f}\right) ((\rho_f - \rho_f^+) / \rho_f^+) - \frac{\overline{\tau}_b}{\varphi_f^+ K_f} (\theta - \theta^+)$$
(4.2.1)

$$\Psi_{\mathrm{I}} = \alpha (\theta - \theta^{+}) - \frac{1}{2} \frac{\overline{c}_{\mathrm{v}}}{\theta^{+}} (\theta - \theta^{+})^{2} - \overline{\tau}_{s} (\theta - \theta^{+}) (\operatorname{tr} \mathbf{E}_{s}) + \overline{\tau}_{f} (\theta - \theta^{+}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \overline{\lambda}_{sf} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ff} \left( (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} \right)^{2} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s}),$$

$$(4.2.2)$$

$$\rho\eta = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + \overline{\tau}_{s}(\operatorname{tr} \mathbf{E}_{s}) - \overline{\tau}_{f}(\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+}$$
(4.2.3)

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\mathbf{T}_{I} + \varphi_{f}^{+}P_{f}\mathbf{I}) + \alpha(\theta - \theta^{+})\mathbf{I} = -\overline{\lambda}_{ss}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \overline{\lambda}_{sf}((\rho_{f} - \rho_{f}^{+})/\rho_{f}^{+})\mathbf{I} + \overline{\tau}_{s}(\theta - \theta^{+})\mathbf{I}$$

$$(4.2.4)$$

$$\varphi_{f}^{+}P_{f} = \overline{\tau}_{f}(\theta - \theta^{+}) - \overline{\lambda}_{sf}(\operatorname{tr} \mathbf{E}_{s}) + \overline{\lambda}_{ff}(\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+}$$
(4.2.5)

where the subscript f denotes the single fluid constituent. From equations (4.1.17), the bulk modulus for the fluid is given by

$$K_f = -\frac{\overline{\lambda}_{sf} \Phi_{ff}}{\Gamma_{sf}}$$
(4.2.6)

The inequality (4.1.18) reduces to

$$-\bar{\lambda}_{sf} \frac{\Phi_{ff}}{\Gamma_{sf}} > 0 \tag{4.2.7}$$

Since the coefficient  $\Phi_{ff}$  must be positive, (4.2.7) shows that

$$\lambda_{sf} / \Gamma_{sf} < 0 \tag{4.2.8}$$

The single fluid versions of (4.1.27), (3.3.26), (4.1.29), (4.1.31) and (4.1.32) take the forms

$$\mathbf{T}_{\mathrm{I}} + \varphi_{f}^{+} P_{f} \mathbf{I} = \overline{\lambda}_{ss}(\operatorname{tr} \mathbf{E}_{s}) \mathbf{I} + 2\mu_{ss} \mathbf{E}_{s} - \overline{\lambda}_{sf} \left( (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} \right) \mathbf{I} - (\overline{\tau}_{s} - \alpha) (\theta - \theta^{+}) \mathbf{I}$$
(4.2.9)

$$\mathbf{T}_{\mathrm{I}} = (\overline{\lambda}_{ss} + \overline{\lambda}_{sf})(\mathrm{tr}\,\mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - (\overline{\lambda}_{sf} + \overline{\lambda}_{ff})((\rho_{f} - \rho_{f}^{+})/\rho_{f}^{+})\mathbf{I} - (\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)(\theta - \theta^{+})\mathbf{I}$$
(4.2.10)

$$\mathbf{T}_{\mathrm{I}} = (\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - (1 + \overline{\lambda}_{sf} / \overline{\lambda}_{ff})\varphi_{f}^{+}P_{f}\mathbf{I} - (\overline{\tau}_{s} - \overline{\lambda}_{sf}\overline{\tau}_{f} / \overline{\lambda}_{ff} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.2.11)$$

$$\rho\eta = -\alpha + (\overline{c}_{v} + \theta^{+} \overline{\tau}_{f}^{2} / \overline{\lambda}_{ff})(\theta - \theta^{+}) / \theta^{+} + (\overline{\tau}_{s} - \overline{\tau}_{f} \overline{\lambda}_{sf} / \overline{\lambda}_{ff})(\operatorname{tr} \mathbf{E}_{s}) - (\overline{\tau}_{f} / \overline{\lambda}_{ff})\varphi_{f}^{+}P_{f} \quad (4.2.12)$$

$$(\varphi_f - \varphi_f^+) / \varphi_f^+ = (\overline{\lambda}_{sf} / \overline{\lambda}_{ff}) (\operatorname{tr} \mathbf{E}_s) + (1/\overline{\lambda}_{ff}) \left( 1 - \frac{\overline{\lambda}_{ff}}{\varphi_f^+ K_f} \right) \phi_f^+ P_f - (\overline{\tau}_f / \overline{\lambda}_{ff}) (\theta - \theta^+)$$
(4.2.13)

The field equations which follow from (3.3.33), (3.3.34) and (3.3.35) are

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{s} - \overline{\tau}_{s} + \gamma) \operatorname{GRAD} \theta - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right)$$
(4.2.14)

$$\rho_{f}^{+} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} = \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \operatorname{GRAD} \theta$$

$$-\xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$

$$(4.2.15)$$

and

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left( (\overline{\tau}_{s} - \alpha_{s} + \varsigma/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} + (\overline{\tau}_{f} - \alpha_{f} - \varsigma/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{f}}{\partial t} \right)$$

$$= \kappa \operatorname{Div}(\operatorname{GRAD}\theta)$$

$$(4.2.16)$$

where

$$\xi = \xi_{ff} = -\xi_{sf} \tag{4.2.17}$$

$$\gamma = \gamma_f = -\gamma_s \tag{4.2.18}$$

and

$$\varsigma = \varsigma_f = -\varsigma_s \tag{4.2.19}$$

Given (4.2.17), (4.2.18) and (4.2.19), the inequality (3.2.27) shows that

$$\kappa \ge 0 \tag{4.2.20}$$

and

$$\kappa \xi / \theta^+ \ge \frac{1}{4} (\gamma + \varsigma / \theta^+)^2 \tag{4.2.21}$$

It follows from (4.2.21) that

$$\xi \ge 0 \tag{4.2.22}$$

As mentioned earlier, the coefficient  $\kappa$  is the *thermal conductivity*. The coefficient  $\xi$  is called the *drag coefficient*.

### 4.3 Stability of Equilibrium: Classes of Initial - Boundary Value Problems

In this section some comments will be given about the types of initial-boundary value problems one can formulate with a linearized porous media model of the type being discussed in this Chapter. For simplicity, we shall continue to assume that we are dealing with a porous elastic solid containing one fluid. While much of which is said generalizes to the case of N - 1 fluids, it is not necessary to complicate the discussion here.

At several points in the discussion, mention has been made of the inequality (3.3.18). For the case of a solid containing a single fluid, it takes the form

$$\frac{\partial}{\partial t} \int_{\mathscr{V}} \left( \Sigma + \frac{1}{2} \rho_{s_{R}} \frac{\partial \mathbf{w}_{s}}{\partial t} \cdot \frac{\partial \mathbf{w}_{s}}{\partial t} + \frac{1}{2} \rho_{f}^{+} \frac{\partial \mathbf{w}_{f}}{\partial t} \cdot \frac{\partial \mathbf{w}_{f}}{\partial t} \right) dv$$

$$\leq \oint_{\partial \mathscr{V}} \left( \mathbf{t} - \frac{\theta - \theta^{+}}{\theta^{+}} \mathbf{m} \right) \cdot d\mathbf{s}$$
(4.3.1)

where

$$\mathbf{t} = \mathbf{T}_{\mathrm{I}} \frac{\partial \mathbf{w}_{s}}{\partial t} - (\varphi_{f}^{+} P_{f} - \alpha_{f} (\theta - \theta^{+})) \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(4.3.2)

Equation (4.3.2) is equation (3.3.17) specialized to the case where N = 2 and where the fluid has pore pressure. Of course, equation (4.1.7) has been used to introduce the pore pressure  $P_f$ . By use of (3.3.36), in the compressible case  $\Sigma$  is given by

$$\Sigma = \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} + \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \overline{\lambda}_{sf} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ff} \left( (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} \right)^{2} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$

$$(4.3.3)$$

The fact that  $\Sigma$  is positive definite yields

$$\overline{c}_{v} > 0 \tag{4.3.4}$$

$$\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} > 0 \tag{4.3.5}$$

$$(\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss})\overline{\lambda}_{ff} > \overline{\lambda}_{sf}^2$$
(4.3.6)

and

$$\mu_{ss} > 0$$
 (4.3.7)

$$\overline{\lambda}_{\rm ff} > 0 \tag{4.3.8}$$

The essential idea of the stability of equilibrium is to show that the inequality (4.3.1), combined with certain other assumptions, allows one to prove that states that are in equilibrium initially remain so indefinitely. Once this simple physical fact is formally established, it is rather simple to establish a uniqueness theorem for the governing field equations. If the compressible case is considered first, for initial conditions it is required that the initial displacements of the solid and the fluid are such that

$$\mathbf{E}_{s}(\mathbf{X},0) = \mathbf{0} \tag{4.3.9}$$

$$\frac{\partial \mathbf{w}_s(\mathbf{X},0)}{\partial t} = \mathbf{0} \tag{4.3.10}$$

$$\operatorname{tr} \mathbf{E}_{f}(\mathbf{X}, 0) \equiv \operatorname{Div} \mathbf{w}_{f}(\mathbf{X}, 0) = 0$$
(4.3.11)

and

$$\frac{\partial \mathbf{w}_f(\mathbf{X}, 0)}{\partial t} = \mathbf{0} \tag{4.3.12}$$

Therefore, we have assumed that the fluid and the solid have zero initial velocity. In addition, we have assumed that the strain of the solid is initially zero, and, from (3.3.9), that the bulk density of the fluid is initially  $\rho_f^+$ . For the temperature, we shall assume that initially

$$\Theta(\mathbf{X},0) = \theta^+ \tag{4.3.13}$$

Next boundary conditions must be prescribed on the surface  $\partial \mathscr{V}$ . At this point in the discussion, we shall simply say that boundary conditions are prescribed which cause the surface integral in (4.3.1) to vanish. As a result, (4.3.1) reduces to

$$\frac{\partial}{\partial t} \int_{\mathscr{Y}} \left( \Sigma + \frac{1}{2} \rho_{s_R} \frac{\partial \mathbf{w}_s}{\partial t} \cdot \frac{\partial \mathbf{w}_s}{\partial t} + \frac{1}{2} \rho_f^+ \frac{\partial \mathbf{w}_f}{\partial t} \cdot \frac{\partial \mathbf{w}_f}{\partial t} \right) dv \le 0$$
(4.3.14)

Given our initial conditions (4.3.9) through (4.3.12), equation (4.3.14) shows that

$$\int_{\mathscr{Y}} \left( \Sigma(\mathbf{X},t) + \frac{1}{2} \rho_{s_R} \frac{\partial \mathbf{w}_s(\mathbf{X},t)}{\partial t} \cdot \frac{\partial \mathbf{w}_s(\mathbf{X},t)}{\partial t} + \frac{1}{2} \rho_f^+ \frac{\partial \mathbf{w}_f(\mathbf{X},t)}{\partial t} \cdot \frac{\partial \mathbf{w}_f(\mathbf{X},t)}{\partial t} \right) dv = 0 \quad (4.3.15)$$

for all *t*. If we presume that we are dealing with continuous solutions of the field equations, it follows from (4.3.15) that the integrand must vanish for all  $(\mathbf{X}, t)$ . This fact and the positive definite nature of  $\Sigma$  yield

$$\mathbf{E}_{s}(\mathbf{X},t) = \mathbf{0} \tag{4.3.16}$$

$$\frac{\partial \mathbf{w}_s(\mathbf{X}, t)}{\partial t} = \mathbf{0} \tag{4.3.17}$$

$$\operatorname{tr} \mathbf{E}_{f}(\mathbf{X}, t) \equiv \operatorname{Div} \mathbf{w}_{f} = 0 \tag{4.3.18}$$

$$\frac{\partial \mathbf{w}_f(\mathbf{X}, t)}{\partial t} = \mathbf{0} \tag{4.3.19}$$

and

$$\theta(\mathbf{X},t) = \theta^+ \tag{4.3.20}$$

for all  $(\mathbf{X}, t)$ . Therefore, with suitable boundary conditions, solutions of the field equations which start out in the equilibrium state defined by (4.3.9) through (4.3.13), remain in that state for future time.

A uniqueness theorem follows from the above results if one considers two possible solutions of the field equations which obey the same initial and boundary conditions. The difference in these two solutions will have zero initial and boundary data. If these initial conditions for the difference solution obey (4.3.9) through (4.3.13), and if the boundary conditions for the difference solution causes the surface integral in (4.3.1) to vanish, then the difference solution must obey (4.3.6) through (4.3.20). It is in the sense of these three equations that the solution of the field equations is unique.

It is not difficult to characterize many cases where the surface integral in (4.3.1) vanishes. If **n** denotes the outward drawn unit normal to the surface  $\partial \mathcal{N}$ , some of these cases are as follows:

- *i)* The fluid and solid displacements vanish on  $\partial \mathcal{N}$ , and the temperature equals  $\theta^+$ .
- *ii)* The stress vector  $\mathbf{T}_{\mathbf{i}}\mathbf{n}$  vanishes on  $\partial \mathcal{O} / \mathbf{i}$ , the fluid and solid displacements are equal and the temperature equals  $\theta^+$ .
- *iii)* The stress vector  $\mathbf{T}_{\mathbf{I}}\mathbf{n}$  vanishes on  $\partial \mathscr{V}$ , the pore pressure  $P_f$  vanishes on  $\partial \mathscr{V}$  and the temperature equals  $\theta^+$ .
- *iv*) The solid displacement vanishes on  $\partial \mathscr{N}$ , the pore pressure  $P_f$  vanishes on  $\partial \mathscr{N}$  and the temperature equals  $\theta^+$ .
- *v*) Cases identical to *i*) and *ii*) except that the temperature condition is replaced by the requirement that  $\mathbf{m} \cdot \mathbf{n}$  vanishes on  $\partial \mathcal{N}$ .
- *vi*) Combinations of *i*) through *v*), where the cases are prescribed on different parts of  $\partial \mathcal{O}$ .

The reader will recognize in the above list the usual traction and displacement boundary conditions from elasticity. The conditions on the temperature field correspond to cases familiar from conduction heat transfer, i.e. when the boundary is not insulated one prescribes the surface temperature. The conditions on the pore pressure and the relative displacement of the fluid and solid correspond to whether or not the boundary is *pervious* or *impervious* to the flow of fluid across the boundary. If one prescribes the relative displacement to be zero, as in *ii*), then the fluid is not allowed to flow across the boundary. If instead, one prescribes the pore pressure, relative motion is allowed and, thus, fluid can flow across the boundary. In a rough sense the temperature and pore pressure are analogous quantities as far as the prescription of the boundary conditions are concerned. As we shall see later, in a certain special case there is a formal analogy between temperature and pore pressure.

### 4.4 Incompressible Immiscible Mixtures

In Section 2.7 it was shown how for a mixture consisting of a rigid solid and N - 1 incompressible fluids one could use the constraint of incompressibility to formulate a mathematical model. When the solid is allowed to be a deformable incompressible solid, a similar formulation can be presented [Ref. 5, 6]. However, it is instructive in this Section to take an equivalent approach and obtain the incompressible results as a limit of the results in Section 3.4.

#### **Porous Elasticity Models With Pore Pressure**

From the results of the Section 4.2, when the mixture is assumed to be in shifting equilibrium and each fluid is assumed to have a pore pressure, the volume fractions are given by (4.1.24). If we make the additional assumption that one of the fluids is incompressible, then by further restrictions on the coefficients in (4.1.24) we can accommodate this special case. Equation (4.1.24) repeated is

$$\varphi_{b} - \varphi_{b}^{+} = \frac{\overline{\lambda}_{sb}}{K_{b}} \operatorname{tr} \mathbf{E}_{s} + \varphi_{b}^{+} (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+} - \frac{1}{K_{b}} \sum_{e=2}^{N} \overline{\lambda}_{be} (\rho_{e} - \rho_{e}^{+}) / \rho_{e}^{+} - \frac{\overline{\tau}_{b}}{K_{b}} (\theta - \theta^{+})$$
(4.4.1)

If the  $b^{th}$  fluid is incompressible, then  $\gamma_b = \gamma_b^+$  and equation (4.1.9) reduces to

$$\frac{\rho_b - \rho_b^+}{\rho_b^+} = \frac{\varphi_b - \varphi_b^+}{\varphi_b^+}$$
(4.4.2)

In order for equation (4.4.1) to reduce to (4.4.2), it is necessary for

$$\frac{\overline{\lambda}_{sb}}{K_b} \to 0 \tag{4.4.3}$$

$$\frac{\overline{\lambda}_{be}}{K_b} \to 0 \tag{4.4.4}$$

and

$$\frac{\overline{\tau}_b}{K_b} \to 0 \tag{4.4.5}$$

If every fluid is incompressible, then (4.4.3), (4.4.4) and (4.4.5) hold for b = 2, ..., N. It then follows from (4.1.17), (4.1.23) and (4.1.22) that, in the incompressible limit,

$$\Gamma_{sd} = 0 \tag{4.4.6}$$

$$\Gamma_{dc} = \varphi_b^+ \Phi_{dc} \tag{4.4.7}$$

and

$$\Theta_d = 0 \tag{4.4.8}$$

for c, d = 2, ..., N. These results reduce (3.3.27) through (3.3.32) to

$$\overline{c}_{v} = c_{v} \tag{4.4.9}$$

$$\overline{\tau}_{\rm s} = \tau_{\rm s} \tag{4.4.10}$$

$$\overline{\tau}_c = \tau_c \tag{4.4.11}$$

$$\overline{\lambda}_{ss} = \lambda_{ss} \tag{4.4.12}$$

$$\overline{\lambda}_{sc} = \lambda_{sc} \tag{4.4.13}$$

$$\bar{\lambda}_{cb} = \lambda_{cb} \tag{4.4.14}$$

As a consequence of (4.4.9) through (4.4.14), in those cases where all of the fluids are incompressible, we need not distinguish between the shifting equilibrium material coefficients and those originally introduced in Section 3.2. This distinction will be ignored without comment throughout the remainder of this chapter.

If the *solid* is also incompressible then we must satisfy the additional condition that  $\gamma_s$  is a constant. Because  $\gamma_s = \rho_s / \phi_s$ , equations (2.4.9) and (3.2.15) show that

$$\frac{\varphi_s - \varphi_s^+}{\varphi_s^+} = -\operatorname{tr} \mathbf{E}_s \tag{4.4.15}$$

must be true when the solid is incompressible. Equation (4.4.15) also follows from (2.6.11) and (3.2.15). As a result of the relationship (2.6.12), the volume fraction of the solid can be eliminated from (4.4.15) to obtain

$$\sum_{b=2}^{N} (\varphi_b - \varphi_b^+) = \varphi_s^+ \operatorname{tr} \mathbf{E}_s$$
(4.4.16)

Equation (4.4.16) is the incompressibility constraint for an incompressible linear elastic solid containing N - 1 incompressible fluids. It is possible that the reader will recognize (4.4.16) as the linearized special case of the constraint (2.6.13). Equation (4.4.16) is an extra relationship between the variables which must be accommodated by the constitutive equations. One method of achieving this accommodation is to allow the pore pressure for the  $N^{th}$  fluid to become indeterminant. It follows from (4.1.25) that

$$\phi_{N}^{+}P_{N} = \tau_{N}(\theta - \theta^{+}) - \lambda_{sN}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \lambda_{Nb} (\rho_{b} - \rho_{b}^{+}) / \rho_{b}^{+}$$
(4.4.17)

Since the fluids are incompressible, equation (4.4.2) can be used to replace (4.4.17) by

$$\phi_{N}^{+}P_{N} = \tau_{N}(\theta - \theta^{+}) - \lambda_{sN}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N} \lambda_{Nb}(\phi_{b} - \phi_{b}^{+}) / \phi_{b}^{+}$$
(4.4.18)

By appropriate choice of the coefficients in (4.4.18), it can be made to reduce to (4.4.16). The details of this choice are revealed by writing (4.4.18) as

$$\phi_{N}^{+}P_{N}/\lambda_{NN} = (\tau_{N}/\lambda_{NN})(\theta - \theta^{+}) - (\lambda_{NN}/\lambda_{NN})(\operatorname{tr} \mathbf{E}_{s}) + \sum_{b=2}^{N-1} (\lambda_{Nb}/\lambda_{NN})(\varphi_{b} - \varphi_{b}^{+})/\varphi_{b}^{+} + (\varphi_{N} - \varphi_{N}^{+})/\varphi_{N}^{+}$$

$$(4.4.19)$$

Clearly, if we assume that

$$\frac{1}{\lambda_{_{NN}}} \to 0 \tag{4.4.20}$$

$$\frac{\tau_N}{\lambda_{NN}} \to 0 \tag{4.4.21}$$

$$\frac{\lambda_{Nb}}{\lambda_{NN}} \to \frac{\varphi_b^+}{\varphi_N^+} \tag{4.4.22}$$

for *b* = 2,...,*N* - 1, and

$$\frac{\lambda_{sN}}{\lambda_{NN}} \to \frac{\varphi_s^+}{\varphi_N^+} \tag{4.4.23}$$

where  $P_N$  remains bounded, then (4.4.19) reduces to the desired result (4.4.16). Equations (4.4.20), (4.4.21), (4.4.22) and (4.4.23) are additional restrictions on the material constants resulting from the assumption of incompressibility. Unlike (4.4.6), (4.4.7) and (4.4.8), equations (4.4.20), (4.4.21) (4.4.22) and (4.4.23) cause the pore pressure  $P_N$  not to appear in the constitutive equations for the model. In short, the pore pressure  $P_N$  becomes indeterminate in so far as the constitutive equations are concerned. Of course, this indeterminacy is exactly like the one which appeared in Section 2.7 where the constraint of incompressibility was used during the derivation of the thermodynamic restrictions.

The limits (4.4.20) through (4.4.23) have an impact on the constitutive equations for the pore pressures  $P_{\beta}$ , for  $\beta = 2,...,N - 1$ . Because of (4.4.20) and (4.4.22), the coefficient  $\lambda_{cN} = \lambda_{Nc}$ , which appears in (4.1.25) is unbounded in the incompressible limit. However, if one forms the difference  $P_{\beta} - P_N$ , for  $\beta = 2,...,N$ , the result can be seen to have a well defined limit. It follows from (4.1.25) that this difference is given by

$$\varphi_{\beta}^{+}(P_{\beta}-P_{N}) = (\tau_{\beta} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}}\tau_{N})(\theta - \theta^{+}) - (\lambda_{s\beta} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}}\lambda_{sN})(\operatorname{tr} \mathbf{E}_{s})$$

$$+ \sum_{\alpha=2}^{N-1} (\lambda_{\beta\alpha} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}}\lambda_{N\alpha})(\varphi_{\alpha} - \varphi_{\alpha}^{+})/\varphi_{\alpha}^{+} + (\lambda_{\beta N} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}}\lambda_{NN})(\varphi_{N} - \varphi_{N}^{+})/\varphi_{N}^{+},$$

$$(4.4.24)$$

where (4.4.2) has been used. It follows from (4.4.19) and (4.4.22) that in the incompressible limit the coefficient of  $(\varphi_N - \varphi_N^+)/\varphi_N^+$  in (4.4.24) vanishes. If the limits of the remaining coefficients in (4.4.24) are written

$$\tau_{\beta} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}} \tau_{N} \to \tau_{\beta}^{*}$$
(4.4.25)

$$\lambda_{s\beta} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}} \lambda_{sN} \to \lambda_{s\beta}^{*}$$
(4.4.26)

and

$$\lambda_{\beta\alpha} - \frac{\varphi_{\beta}^{+}}{\varphi_{N}^{+}} \lambda_{N\alpha} \to \lambda_{\beta\alpha}^{*}$$
(4.4.27)

then (4.4.24) becomes

$$\varphi_{\beta}^{+}(P_{\beta}-P_{N}) = \tau_{\beta}^{*}(\theta-\theta^{+}) - \lambda_{s\beta}^{*}(\operatorname{tr} \mathbf{E}_{s}) + \sum_{\alpha=2}^{N-1} \lambda_{\beta\alpha}^{*}(\varphi_{\alpha}-\varphi_{\alpha}^{+})/\varphi_{\alpha}^{+}$$
(4.4.28)

for  $\beta = 2,..., N - 1$ . Equations (4.4.28) are the constitutive equation for the capillary pressures within the incompressible mixture. A manipulation similar to the one which produced (4.4.24) can be applied to (3.3.24) to obtain

$$\rho_{s_{R}}\mathbf{K}_{s} - \varphi_{s}^{+}P_{N}\mathbf{I} = -(\lambda_{ss} - \frac{\varphi_{s}^{+}}{\varphi_{N}^{+}}\lambda_{sN})(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{\beta=2}^{N-1}(\lambda_{s\beta} - \frac{\varphi_{s}^{+}}{\varphi_{N}^{+}}\lambda_{N\beta})((\varphi_{\beta} - \varphi_{\beta}^{+})/\varphi_{\beta}^{+})\mathbf{I} + (\lambda_{sN} - \frac{\varphi_{s}^{+}}{\varphi_{N}^{+}}\lambda_{NN})((\varphi_{N} - \varphi_{N}^{+})/\varphi_{N}^{+})\mathbf{I} + (\tau_{s} - \frac{\varphi_{s}^{+}}{\varphi_{N}^{+}}\tau_{N})(\theta - \theta^{+})\mathbf{I}$$

$$(4.4.29)$$

As with (4.4.24), the coefficient of  $(\varphi_N - \varphi_N^+) / \varphi_N^+$  vanishes in the incompressible limit. If the limits of the coefficients in (4.4.29) are written

$$\lambda_{ss} - \frac{\varphi_s^+}{\varphi_N^+} \lambda_{sN} \to \lambda_{ss}^* \tag{4.4.30}$$

$$\lambda_{s\beta} - \frac{\varphi_s^+}{\varphi_N^+} \lambda_{N\beta} \to \lambda_{s\beta}^* \tag{4.4.31}$$

$$\tau_s - \frac{\varphi_s^+}{\varphi_N^+} \tau_N \to \tau_s^* \tag{4.4.32}$$

then (4.4.29) becomes

$$\rho_{s_{R}}\mathbf{K}_{s} - \varphi_{s}^{+}P_{N}\mathbf{I} = -\lambda_{ss}^{*}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \sum_{\beta=2}^{N-1}\lambda_{s\beta}^{*}((\varphi_{\beta} - \varphi_{\beta}^{+})/\varphi_{\beta}^{+})\mathbf{I} + \tau_{s}^{*}(\theta - \theta^{+})\mathbf{I}.$$
(4.4.33)

Note that (4.4.26) and (4.4.31) provide two apparently different definitions of the coefficients  $\lambda_{s\beta}^*$  for  $\beta = 2,..., N - 1$ . It is readily shown by use of (4.4.22) and (4.4.23) that the coefficients  $\lambda_{s\beta} - (\varphi_{\beta}^+ / \varphi_N^+) \lambda_{sN}$  and  $\lambda_{s\beta} - (\varphi_s^+ / \varphi_N^+) \lambda_{N\beta}$  have the same incompressible limit.

The incompressible limit of the constitutive equation for the entropy density, equation (3.3.23), is

$$\rho\eta = -\alpha + c_{v} \left(\theta - \theta^{+}\right) / \theta^{+} + \tau_{s}^{*} \operatorname{tr} \mathbf{E}_{s} - \sum_{\beta=2}^{N-1} \tau_{\beta}^{*} \left(\left(\varphi_{\beta} - \varphi_{\beta}^{+}\right) / \varphi_{\beta}^{+}\right)$$
(4.4.34)

Equation (4.4.34) is a consequence of (3.3.23), (4.4.2), (4.4.25), (4.4.32) and the result (4.4.9). Just as equations (4.4.28), (4.4.33) and (4.4.34) have been obtained as a limit of the corresponding compressible result, it is possible to derive an incompressible version of the expression for  $\Psi_{I}$ , equation (3.3.22). The result turns out to be

$$\Psi_{\rm I} = \alpha (\theta - \theta^{+}) - \frac{1}{2} \frac{c_{\rm v}}{\theta^{+}} (\theta - \theta^{+})^{2} - \tau_{s}^{*} (\theta - \theta^{+}) (\operatorname{tr} \mathbf{E}_{s})$$
  
+ 
$$\sum_{\beta=2}^{N-1} \tau_{\beta}^{*} (\theta - \theta^{+}) (\varphi_{\beta} - \varphi_{\beta}^{+}) / \varphi_{\beta}^{+} + \frac{1}{2} \lambda_{ss}^{*} (\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{\beta=2}^{N-1} \lambda_{s\beta}^{*} (\operatorname{tr} \mathbf{E}_{s}) (\varphi_{\beta} - \varphi_{\beta}^{+}) / \varphi_{\beta}^{+}$$
(4.4.35)  
+ 
$$\frac{1}{2} \sum_{\beta=2}^{N-1} \sum_{\alpha=2}^{N-1} \lambda_{\beta\alpha}^{*} (\varphi_{\beta} - \varphi_{\beta}^{+}) (\varphi_{\alpha} - \varphi_{\alpha}^{+}) / \varphi_{\beta}^{+} \varphi_{\alpha}^{+} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s}).$$

Given (4.4.35), it is readily established that

$$\rho\eta = -\frac{\partial \Psi_{\rm I}}{\partial \theta} \tag{4.4.36}$$
$$P_{\beta} - P_{N} = \frac{\partial \Psi_{I}}{\partial \phi_{\beta}} \tag{4.4.37}$$

and

$$\rho_{s_R} \mathbf{K}_s - \varphi_s^+ P_N \mathbf{I} = -\frac{\partial \Psi_{\mathbf{I}}}{\partial \mathbf{E}_s}$$
(4.4.38)

Given (4.4.35) and (4.4.34), it follows that

$$\Sigma = \Psi_{\rm I} + \rho \eta (\theta - \theta^{+}) = \frac{1}{2} \frac{c_{\rm v}}{\theta^{+}} (\theta - \theta^{+})^{2} + \frac{1}{2} \lambda_{ss}^{*} (\operatorname{tr} \mathbf{E}_{s})^{2} - \sum_{\beta=2}^{N-1} \lambda_{s\beta}^{*} (\operatorname{tr} \mathbf{E}_{s}) (\varphi_{\beta} - \varphi_{\beta}^{+}) / \varphi_{\beta}^{+} + \frac{1}{2} \sum_{\beta=2}^{N-1} \sum_{\alpha=2}^{N-1} \lambda_{\beta\alpha}^{*} (\varphi_{\beta} - \varphi_{\beta}^{+}) (\varphi_{\alpha} - \varphi_{\alpha}^{+}) / \varphi_{\beta}^{+} \varphi_{\alpha}^{+} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$
(4.4.39)

As before, the fact that  $\Sigma$  is assumed to be positive definite yields

$$c_{\rm v} > 0$$
 (4.4.40)

$$\mu_{ss} > 0$$
 (4.4.41)

and the  $(N-1) \times (N-1)$  symmetric matrix

$\left[\lambda_{ss}^* + \frac{2}{3}\mu_{ss}^*\right]$	$\lambda_{s2}^{*}$		•	•	$\lambda^*_{s,N-1}$
$\lambda_{s2}^{*}$	$\lambda_{\scriptscriptstyle 22}^{*}$		•	•	$\lambda^*_{2,N-1}$
•	•				
					•
$\lambda^*_{s,N-1}$	$\lambda^*_{2,N-1}$				$\lambda^*_{N-1,N-1}$

is positive definite.

It is possible to show that the incompressible versions of (4.1.27), (4.4.23) and (4.1.29) are

$$\mathbf{T}_{\mathbf{I}} + P_{N}\mathbf{I} + \sum_{\beta=2}^{N-1} \varphi_{\beta}^{+} (P_{\beta} - P_{N})\mathbf{I} = \lambda_{ss}^{*} (\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{\beta=2}^{N-1} \lambda_{s\beta}^{*} ((\varphi_{\beta} - \varphi_{\beta}^{+})/\varphi_{\beta}^{+})\mathbf{I} - (\tau_{s}^{*} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.4.42)$$

$$\mathbf{T}_{\mathbf{I}} = -P_{N}\mathbf{I} + (\lambda_{ss}^{*} + \sum_{\beta=2}^{N-1} \lambda_{s\beta}^{*})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - \sum_{\beta=2}^{N-1} (\lambda_{s\beta}^{*} + \sum_{\alpha=2}^{N-1} \lambda_{\alpha\beta}^{*})((\varphi_{\beta} - \varphi_{\beta}^{+})/\varphi_{\beta}^{+})\mathbf{I} - (\tau_{s}^{*} + \sum_{\beta=2}^{N-1} \tau_{\beta}^{*} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.4.43)$$

and

$$\mathbf{T}_{\mathbf{I}} = -P_{N}\mathbf{I} + (\lambda_{ss}^{*} - \sum_{\beta=2}^{N-1}\sum_{\alpha=2}^{N-1}\lambda_{s\beta}^{*}\lambda_{\beta\alpha}^{*-1}\lambda_{s\alpha}^{*})(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} -\sum_{\alpha=2}^{N-1} \left(1 + \sum_{\beta=2}^{N-1}\lambda_{s\beta}^{*}\lambda_{\beta\alpha}^{*-1}\right) \varphi_{\alpha}^{+}(P_{\alpha} - P_{N})\mathbf{I} - (\tau_{s}^{*} - \sum_{\beta=2}^{N-1}\sum_{\alpha=2}^{N-1}\lambda_{s\beta}^{*}\lambda_{\beta\alpha}^{*-1}\tau_{\alpha}^{*} - \alpha)(\theta - \theta^{+})\mathbf{I}$$

$$(4.4.44)$$

where the positive definite character of  $\Sigma$  insures that the matrix  $\left[\lambda_{\beta\alpha}^*\right]$  has an inverse.

The field equations which replace (3.3.33), (3.3.34) and (3.3.35) in the incompressible case turn out to be

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = -\varphi_{s}^{+} \operatorname{GRAD} P_{N} + (\lambda_{ss}^{*} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \sum_{\beta=2}^{N-1} \lambda_{s\beta}^{*} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{\beta}) + (\alpha_{s} - \tau_{s}^{*} - \gamma_{s}) \operatorname{GRAD} \theta - \sum_{b=2}^{N} \xi_{sb} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$

$$\rho_{\beta}^{+} \frac{\partial^{2} \mathbf{w}_{\beta}}{\partial t^{2}} = -\varphi_{\beta}^{+} \operatorname{GRAD} P_{N} + \lambda_{s\beta}^{*} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \sum_{\beta=2}^{N-1} \lambda_{\beta\alpha}^{*} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{\alpha}) + (\alpha_{\beta} - \tau_{\beta}^{*} - \gamma_{\beta}) \operatorname{GRAD} \theta - \sum_{b=2}^{N} \xi_{\beta b} \left( \frac{\partial \mathbf{w}_{b}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$

$$(4.4.46)$$

for  $\beta = 2, ..., N-1$ ,

$$\rho_N^+ \frac{\partial^2 \mathbf{w}_N}{\partial t^2} = -\phi_N^+ \operatorname{GRAD} P_N + (\alpha_N - \gamma_N) \operatorname{GRAD} \theta - \sum_{b=2}^N \xi_{Nb} \left( \frac{\partial \mathbf{w}_b}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t} \right)$$
(4.4.47)

and

$$c_{v}\frac{\partial\theta}{\partial t} + \theta^{+}(\tau_{s}^{*} - \alpha_{s} - \zeta_{s}/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} + \theta^{+}\sum_{\beta=2}^{N-1}(\tau_{\beta}^{*} - \alpha_{\beta} - \zeta_{\beta}/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{\beta}}{\partial t} - \theta^{+}(\alpha_{N} + \zeta_{N}/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{N}}{\partial t} = \kappa\operatorname{Div}(\operatorname{GRAD}\theta)$$

$$(4.4.48)$$

In addition, these field equations must be supplemented with the constraint equation (4.4.16) written in terms of displacements. This equation is

$$\varphi_s^+ \operatorname{Div} \mathbf{w}_s + \sum_{b=2}^N \varphi_b^+ \operatorname{Div} \mathbf{w}_b = 0$$
(4.4.49)

## 4.5 Binary Incompressible Immiscible Mixtures

If both the fluid and solid are incompressible, we need to specialize the results of Section 4.4 to the binary case or simply view the incompressible case as the limit of the compressible one. In any case, equations (4.4.35) (4.4.34), (4.4.33) and (4.4.43) take the forms:

$$\Psi_{\mathrm{I}} = \alpha (\theta - \theta^{+}) - \frac{1}{2} \frac{c_{\mathrm{v}}}{\theta^{+}} (\theta - \theta^{+})^{2} - \tau_{s}^{*} (\theta - \theta^{+}) (\operatorname{tr} \mathbf{E}_{s}) + \frac{1}{2} \lambda_{ss}^{*} (\operatorname{tr} \mathbf{E}_{s})^{2} + \mu_{ss} \operatorname{tr} (\mathbf{E}_{s} \mathbf{E}_{s})$$

$$(4.5.1)$$

$$\rho \eta = -\alpha + c_v \left(\theta - \theta^+\right) / \theta^+ + \tau_s^* \operatorname{tr} \mathbf{E}_s$$
(4.5.2)

$$\rho_{s_{R}}\mathbf{K}_{s} - \varphi_{s}^{*}P_{f}\mathbf{I} = -\lambda_{ss}^{*}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \tau_{s}^{*}(\theta - \theta^{+})\mathbf{I}$$

$$(4.5.3)$$

and

$$\mathbf{T}_{\mathbf{I}} = -P_f \mathbf{I} + \lambda_{ss}^* (\operatorname{tr} \mathbf{E}_s) \mathbf{I} + 2\mu_{ss} \mathbf{E}_s - (\tau_s^* - \alpha)(\theta - \theta^+) \mathbf{I}$$
(4.5.4)

The field equations in this case are special cases of (4.4.45) through (4.4.49). These special cases are

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = -\varphi_{s}^{+} \operatorname{GRAD} P_{f} + (\lambda_{ss}^{*} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + (\alpha_{s} - \tau_{s}^{*} + \gamma) \operatorname{GRAD} \theta - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right)$$

$$(4.5.5)$$

$$\rho_f^+ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} = -\varphi_f^+ \operatorname{GRAD} P_f + (\alpha_f - \gamma) \operatorname{GRAD} \theta - \xi \left( \frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t} \right)$$
(4.5.6)

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+}\left(\left(\tau_{s}^{*} - \alpha_{s} + \varsigma/\theta^{+}\right)\operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} - \left(\alpha_{f} + \varsigma/\theta^{+}\right)\operatorname{Div}\frac{\partial\mathbf{w}_{f}}{\partial t}\right) = \kappa\operatorname{Div}(\operatorname{GRAD}\theta) \quad (4.5.7)$$

and

$$\varphi_s^+ \operatorname{Div} \mathbf{w}_s + \varphi_f^+ \operatorname{Div} \mathbf{w}_f = 0 \tag{4.5.8}$$

In the binary incompressible case, it follows from (4.4.39) that

$$\Sigma = \frac{1}{2} \frac{c_{\rm v}}{\theta^+} (\theta - \theta^+)^2 + \frac{1}{2} \lambda_{ss}^* (\operatorname{tr} \mathbf{E}_s)^2 + \mu_{ss} \operatorname{tr}(\mathbf{E}_s \mathbf{E}_s)$$
(4.5.9)

The result (4.5.9) also follows from (4.5.1), (4.5.2) and the formula,  $\Sigma = \Psi_{I} + \rho \eta (\theta - \theta^{+})$ . The requirement that  $\Sigma$  be positive definite yields

$$c_{\rm v} > 0$$
 (4.5.10)

$$\lambda_{ss}^* + \frac{2}{3}\mu_{ss} > 0 \tag{4.5.11}$$

and

$$\mu_{ss} > 0 \tag{4.5.12}$$

The arguments in Section 4.3 carry over with minor modification for the incompressible case. Because  $\Sigma$  in (4.5.9) does not depend on Div  $\mathbf{w}_f$ , it is not necessary to prescribe the initial condition (4.3.11). However, because the constraint (4.5.8) must be obeyed, (4.3.11) is a consequence of (4.3.9). Likewise, (4.3.18) is not a consequence of the positive definite character of  $\Sigma$ . It is a consequence of the constraint (4.5.8) and (4.3.16).

## References

- 1. BOWEN, R. M., Compressible Porous Media Models by Use of the Theory of Mixtures, *Intl. J. Engr. Sci.*, **20**, 697-736 (1982).
- 2. BIOT, MAURICE A., General Theory of Three-Dimensional Consolidation, *J. Appl. Phys.*, **12**, 155-164 (1941).
- 3. BIOT, MAURICE A., Theory of Finite Deformations of Porous Solids, *Indiana Univ. Math. Journal*, **21**, 597-620 (1972).
- 4. BIOT, MAURICE A., Nonlinear and Semilinear Rheology of Porous Solids, *J. Geophysical Res.* ,**78** ,4924-4937 (1973).
- 5. BOWEN, R. M., Incompressible Porous Media Models by Use of the Theory of Mixtures, *Intl. J. Engr. Sci.*, **18**, 1129-1148 (1980).
- 6. BOWEN, R. M., Diffusion Models Implied by the Theory of Mixtures, in *Rational Thermodynamics*, by C. TRUESDELL, 2nd Ed., Springer-Verlag, New York (1984).

## **Models Which Neglect Inertia**

In order to reach the results summarized in Sections 4.2 and 4.5, many specializing assumptions were necessary. In this Chapter, additional assumptions which one often finds in the porous elasticity literature are discussed. The two special cases which we shall focus on are the special models one gets when the fluid inertia is neglected, and the special model one gets when both the fluid and solid inertia are neglected. For the moment, we shall not worry about the validity of these approximations.

#### 5.1 Compressible Models

The first special case which shall be considered is the compressible binary porous material defined by equations 4.2.1 through (4.2.22). If the inertia of the fluid constituent is simply neglected, the only formal change is that (4.2.16) is replaced by

$$\xi\left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) = \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \operatorname{GRAD} \theta \quad (5.1.1)$$

Because the highest time derivative of  $\mathbf{w}_f$  which now appears in the field equations is one, the number of initial conditions which can be satisfied by the governing field equations is reduced by one. Therefore, the approximate field equations are a *singular perturbation* of those which retain the fluid inertia. Typical of singular perturbation problems, the solutions of the approximate equations cannot be expected to approximate those which retain the fluid inertia in the time neighborhood of the lost initial condition. Usually this fact is described by stating that the approximate equations are not valid within a *boundary layer* of the boundary t = 0. It turns out that characteristic times in porous elasticity are rather small. The effect of this fact is that in a nondimensional sense the time boundary layer is usually small. In any case, if we proceed with an investigation of the field equations without fluid inertia, there are certain interesting results which can be obtained.

The first formal manipulation is to eliminate the fluid displacements in favor of the fluid pore pressure. Equations (4.2.11) and (4.2.12) give the stress,  $T_I$ , and the entropy,  $\eta$ , in terms of the fluid pore pressure. It is convenient to write (4.2.5) in the form

$$\varphi_f^+ P_f = \overline{\tau}_f (\theta - \theta^+) - \overline{\lambda}_{sf} \operatorname{Div} \mathbf{w}_s - \overline{\lambda}_{ff} \operatorname{Div} \mathbf{w}_f$$
(5.1.2)

where (3.39), with the order term omitted, (3.2.3) and (3.28) have been used. Given(5.1.2), equation (5.1.1) can be rewritten in the form

#### **Chapter 5**

$$\xi \left( \frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t} \right) = -\varphi_f^+ \operatorname{GRAD} P_f + (\alpha_f - \gamma) \operatorname{GRAD} \theta$$
(5.1.3)

Equation (5.1.3) allows for the elimination of the combination  $\left(\frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t}\right)$  from the constitutive

equation for the heat flux m. The binary version of (3.2.23), with the order term omitted, is

$$\mathbf{m} = -\kappa \operatorname{GRAD} \theta - \varsigma \left( \frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t} \right)$$
(5.1.4)

Therefore, when (5.1.3) is used to eliminate the term  $\left(\frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t}\right)$  from (5.1.4) the result is

$$\mathbf{m} = -(\kappa + (\zeta/\xi)(\alpha_f - \gamma)) \operatorname{GRAD} \theta + (\zeta/\xi)\varphi_f^+ \operatorname{GRAD} P_f$$
(5.1.5)

Note that in the derivation of (5.1.5) it has been assumed that  $\xi$  is nonzero. This assumption, along with the requirement (4.2.22), shows that we have assumed

$$\xi > 0 \tag{5.1.6}$$

Next we shall eliminate the displacement of the fluid,  $\mathbf{w}_f$ , between (5.1.2) and (5.1.3). If we compute the time derivative of (5.1.2) and the divergence of (5.1.3) the results are

$$\varphi_{f}^{+} \frac{\partial P_{f}}{\partial t} = \overline{\tau}_{f} \frac{\partial \theta}{\partial t} - \overline{\lambda}_{sf} \operatorname{Div} \frac{\partial \mathbf{w}_{s}}{\partial t} - \overline{\lambda}_{ff} \operatorname{Div} \frac{\partial \mathbf{w}_{f}}{\partial t}$$
(5.1.7)

and

$$\xi \left( \operatorname{Div} \frac{\partial \mathbf{w}_f}{\partial t} - \operatorname{Div} \frac{\partial \mathbf{w}_s}{\partial t} \right) = -\varphi_f^+ \operatorname{Div}(\operatorname{GRAD} P_f) + (\alpha_f - \gamma) \operatorname{Div}(\operatorname{GRAD} \theta)$$
(5.1.8)

If  $\operatorname{Div} \frac{\partial \mathbf{w}_f}{\partial t}$  is eliminated from these two equations, the result is

$$(1/\overline{\lambda}_{ff})\varphi_{f}^{+}\frac{\partial P_{f}}{\partial t} - (1/\overline{\lambda}_{ff})\overline{\tau}_{f}\frac{\partial \theta}{\partial t} + (1 + \overline{\lambda}_{sf}/\overline{\lambda}_{ff})\operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t}$$

$$= (\varphi_{f}^{+}/\xi)\operatorname{Div}(\operatorname{GRAD} P_{f}) - ((\alpha_{f} - \gamma)/\xi)\operatorname{Div}(\operatorname{GRAD} \theta)$$
(5.1.9)

Equation (5.1.9) replaces the equation of motion for the fluid in the special case where the inertia of the fluid can be neglected. The equation of motion of the solid, (4.2.14), can also be simplified in this case. Equation (5.1.1) can be used to eliminate the term  $\xi \left( \frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t} \right)$  from (4.2.14), and

(5.1.2) can be used to eliminate  $\text{Div} \mathbf{w}_f$ . The result of this elimination is

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = -\varphi_{f}^{+} (1 + \overline{\lambda}_{sf} / \overline{\lambda}_{ff}) \operatorname{GRAD} P_{f} + (\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + (\alpha - \overline{\tau}_{s} + \overline{\tau}_{f} \overline{\lambda}_{sf} / \overline{\lambda}_{ff}) \operatorname{GRAD} \theta.$$
(5.1.10)

The energy equation appropriate to this special case follows from (4.2.16). Equation (4.2.16) can be rearranged to the form

$$\overline{c}_{v} \frac{\partial \theta}{\partial t} + \theta^{+} \left( (\overline{\tau}_{s} - \alpha) \operatorname{Div} \frac{\partial \mathbf{w}_{s}}{\partial t} + \overline{\tau}_{f} \operatorname{Div} \frac{\partial \mathbf{w}_{f}}{\partial t} \right)$$

$$= \operatorname{Div} \left( \kappa \operatorname{GRAD} \theta + \theta^{+} (\varsigma/\theta^{+} + \alpha_{f}) \left( \frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t} \right) \right)$$
(5.1.11)

The desired result can be obtained by use of (5.1.7) to eliminate  $\text{Div}(\partial \mathbf{w}_f / \partial t)$  from the left side of (5.1.11) and (5.1.8) to eliminate  $\text{Div}\left(\frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t}\right)$  from the right side of (5.1.11). The result of this elimination is

$$(\overline{c}_{v} + \theta^{+} \overline{\tau}_{f}^{2} / \overline{\lambda}_{ff}) \frac{\partial \theta}{\partial t} - \theta^{+} (1 / \overline{\lambda}_{ff}) \overline{\tau}_{f} \varphi_{f}^{+} \frac{\partial P_{f}}{\partial t} + \theta^{+} (\overline{\tau}_{s} - \overline{\tau}_{f} \overline{\lambda}_{sf} / \overline{\lambda}_{ff} - \alpha) \operatorname{Div} \frac{\partial \mathbf{w}_{s}}{\partial t}$$

$$= (\kappa + \theta^{+} (\alpha_{f} + \varsigma / \theta^{+}) (\alpha_{f} - \gamma) / \xi) \operatorname{Div}(\operatorname{GRAD} \theta)$$

$$- \theta^{+} ((\alpha_{f} + \varsigma / \theta^{+}) / \xi) \varphi_{f}^{+} \operatorname{Div}(\operatorname{GRAD} P_{f})$$
(5.1.12)

By rearrangement of the inequality (4.2.21), it is possible to show that

$$(1/\xi)(\kappa + \theta^{+}(\alpha_{f} + \varsigma/\theta^{+})(\alpha_{f} - \gamma)/\xi) \geq \frac{1}{4} \{(\alpha_{f} + \varsigma/\theta^{+}) + (\alpha_{f} - \gamma)\}^{2}$$

$$(5.1.13)$$

The algebra necessary to derive (5.1.13) from (4.2.21) comes from the following sequence of rearrangements

$$\frac{1}{4}(\gamma + \varsigma/\theta^{+})^{2} = \frac{1}{4}\left((\varsigma/\theta^{+} + \alpha_{f}) - (\alpha_{f} - \gamma)\right)^{2}$$

$$= \frac{1}{4}\left((\varsigma/\theta^{+} + \alpha_{f}) + (\alpha_{f} - \gamma)\right)^{2} - (\varsigma/\theta^{+} + \alpha_{f})(\alpha_{f} - \gamma)$$
(5.1.14)

and the substitution of the result, (5.1.14), into (4.2.21). It follows from (5.1.13) and (5.1.6) that

$$(\kappa + \theta^+ (\alpha_f + \varsigma / \theta^+) (\alpha_f - \gamma) / \xi) \ge 0$$
(5.1.15)

Equations (5.1.9), (5.1.10) and (5.1.12) represent the field equations appropriate to the model where fluid inertia has been neglected. These equations, which are intended to yield the solid displacement,  $\mathbf{w}_s$ , the pore pressure,  $P_f$  and the temperature,  $\theta$ , must be supplemented by boundary and initial conditions. The results of Section 4.3 provide guidance as to the types of initial boundary data which are sufficient to imply uniqueness. If the fluid inertia is neglected, equation (4.3.1) reduces to

$$\frac{\partial}{\partial t} \int_{\mathscr{V}} \left( \Sigma + \frac{1}{2} \rho_{s_R} \frac{\partial \mathbf{w}_s}{\partial t} \cdot \frac{\partial \mathbf{w}_s}{\partial t} \right) dv \leq \oint_{\partial \mathscr{V}} \left( \mathbf{t} - \frac{\theta - \theta^+}{\theta^+} \mathbf{m} \right) \cdot d\mathbf{s}.$$
(5.1.16)

where

$$\Sigma = \frac{1}{2\theta^{+}} (\overline{c}_{v} + \theta^{+} \overline{\tau}_{f}^{2} / \overline{\lambda}_{ff}) (\theta - \theta^{+})^{2} + \frac{1}{2} (\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff}) (\operatorname{tr} \mathbf{E}_{s})^{2} - (\overline{\tau}_{f} / \overline{\lambda}_{ff}) (\varphi_{f}^{+} P_{f}) (\theta - \theta^{+}) + \frac{1}{2} (1 / \overline{\lambda}_{ff}) (\varphi_{f}^{+} P_{f})^{2} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$
(5.1.17)

and

$$\mathbf{t} = \mathbf{T}_{\mathrm{I}} \frac{\partial \mathbf{w}_{s}}{\partial t} - (\varphi_{f}^{+} P_{f} - \alpha_{f} (\theta - \theta^{+})) \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(5.1.18)

Equation (5.1.17) follows from (4.3.3) after (5.1.2) and the approximation of (3.3.9)

$$\left(\rho_{f}-\rho_{f}^{+}\right)/\rho_{f}^{+}=-\operatorname{Div}\mathbf{w}_{f}$$
(5.1.19)

have been used to eliminate  $\left(\rho_f - \rho_f^+\right) / \rho_f^+$ . If the result (5.1.3) is used, (5.1.18) can be replaced by

$$\mathbf{t} = \mathbf{T}_{\mathrm{I}} \frac{\partial \mathbf{w}_{s}}{\partial t} - (\varphi_{f}^{+} P_{f} - \alpha_{f} (\theta - \theta^{+})) \left( -\frac{\varphi_{f}^{+}}{\xi} GRADP_{f} + \frac{(\alpha_{f} - \gamma)}{\xi} GRAD\theta \right)$$
(5.1.20)

An examination of (5.1.20) and (5.1.5) illustrates that, for this approximate theory, unique solutions will result by placing conditions on the stress vector,  $\mathbf{T}_{i}\mathbf{n}$ , or the solid displacement,  $\mathbf{w}_{s}$ , the pore

pressure or its gradient and the temperature or its gradient. The precise nature of these conditions can be deduced from (5.1.17), (5.1.20) and (5.1.5).

If in addition to neglecting the fluid inertia, the assumption is made that the temperature is a constant, then an interesting analogy is obtained between the isothermal poroelasticity of compressible constituents and dynamic thermoelasticity. As mentioned earlier, the isothermal case is obtained is the thermal conductivity  $\kappa$  is assumed to be very large. In the limit as  $\kappa \to \infty$ , equation (5.1.12) reduces to Div(GRAD $\theta$ ) = 0. With appropriate boundary data, the solution to this equation is  $\theta = \theta^+$ . This result reduces (5.1.9) and (5.1.10) to

$$(1/\overline{\lambda}_{ff})\varphi_{f}^{+}\frac{\partial P_{f}}{\partial t} + (1+\overline{\lambda}_{sf}/\overline{\lambda}_{ff})\operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t} = (\varphi_{f}^{+}/\xi)\operatorname{Div}(\operatorname{GRAD} P_{f})$$
(5.1.21)

and

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = -\phi_{f}^{+} (1 + \overline{\lambda}_{sf} / \overline{\lambda}_{ff}) \operatorname{GRAD} P_{f} + (\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s})$$
(5.1.22)

An examination of these equations reveals that they are formally identical to the equations of dynamic thermoelasticity [Ref. 1]. In order to establish this analogy, one has to make the following identifications:

Isothermal Poroelasticity	Dynamic Thermoelasticity			
$arphi_f^+ P_f$	The temperature differential			
$1/\overline{\lambda}_{ff}$	The specific heat per unit of volume divided by the reference temperature			
$\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^2 / \overline{\lambda}_{ff} + \frac{2}{3} \mu_{ss}$	The isothermal bulk modulus of elasticity			
$\mu_{ss}$	The isothermal shear modulus of elasticity			
$1/\xi$	The thermal conductivity divided by the reference temperature			
$1 + \overline{\lambda}_{sf} / \overline{\lambda}_{ff}$	The stress-temperature modulus			

In thermoelasticity, the stress-temperature modulus is usually written as three times the produce of the coefficient of thermal expansion and the isothermal bulk modulus. The analogy between porous elasticity with fluid inertia neglected and dynamic thermoelasticity was first mentioned by Lubinski [Ref. 2]. References 3, 4 and 5 should also be consulted for readers interested in additional information about this analogy.

If the additional approximation is made that the inertia of the solid can also be neglected, then we can obtain yet another set of approximate field equations. This approximation, as we shall see, is frequently adopted as the appropriate one for porous elasticity applications. If the inertia of the solid is neglected, then the field equations (5.1.9) and (5.1.11) are not formally changed. The equation of motion (5.1.10) is replaced by

$$(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) - \varphi_{f}^{+} (1 + \overline{\lambda}_{sf} / \overline{\lambda}_{ff}) \operatorname{GRAD} P_{f} + (\alpha - \overline{\tau}_{s} + \overline{\tau}_{f} \ \overline{\lambda}_{sf} / \overline{\lambda}_{ff}) \operatorname{GRAD} \theta = \mathbf{0}$$
(5.1.23)

Given (4.2.11), a more compact version of (5.1.23) is

$$\operatorname{Div} \mathbf{T}_{\mathrm{I}} = \mathbf{0} \tag{5.1.24}$$

Equation (4.2.11) can be used to eliminate the displacement of the solid from (5.1.9) in favor of the stress  $T_{I}$ . Some of this elimination has already been performed. It follows from (4.1.30) that

$$2\mu_{ss}\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}}{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} \frac{1}{3} (\operatorname{tr} \mathbf{T}_{I})\mathbf{I} + \frac{2\mu_{ss}(\mathbf{I} + \overline{\lambda}_{sf}/\overline{\lambda}_{ff})}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}} \varphi_{f}^{+}P_{f}\mathbf{I} + \frac{2\mu_{ss}(\overline{\tau}_{s} - \overline{\lambda}_{sf}\overline{\tau}_{f}/\overline{\lambda}_{ff} - \alpha)}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}} (\theta - \theta^{+})\mathbf{I}$$
(5.1.25)

It follows from (5.1.25) that

$$\operatorname{tr} \mathbf{E}_{s} = \operatorname{Div} \mathbf{w}_{s} = \frac{1}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}} \operatorname{tr} \mathbf{T}_{\mathbf{I}} + \frac{3(1 + \overline{\lambda}_{sf}/\overline{\lambda}_{ff})}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}} \varphi_{f}^{+} P_{f} + \frac{3(\overline{\tau}_{s} - \overline{\lambda}_{sf}\overline{\tau}_{f}/\overline{\lambda}_{ff} - \alpha)}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}} (\theta - \theta^{+})$$

$$(5.1.26)$$

If the time derivative of equation (5.1.26) is used to eliminate the term  $\text{Div}(\partial \mathbf{w}_s/\partial t)$  from (5.1.9) and (5.1.12), the results are

$$\frac{\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \varphi_{f}^{+} \frac{\partial P_{f}}{\partial t} + \frac{(\overline{\tau}_{s} - \alpha)(\overline{\lambda}_{ff} + \overline{\lambda}_{sf}) - \overline{\tau}_{f}(\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} + \overline{\lambda}_{sf})}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \frac{\partial \theta}{\partial t} + \frac{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \frac{1}{3} \frac{\partial \operatorname{tr} \mathbf{T}_{I}}{\partial t} + \frac{(\varphi_{f}^{+}/\xi)\operatorname{Div}(\operatorname{GRAD} P_{f}) - ((\alpha_{f} - \gamma)/\xi)\operatorname{Div}(\operatorname{GRAD} \theta)}$$
(5.1.27)

and

$$\theta^{+} \frac{(\overline{\tau}_{s} - \alpha)(\overline{\lambda}_{ff} + \overline{\lambda}_{sf}) - \overline{\tau}_{f}(\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} + \overline{\lambda}_{sf})}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \varphi_{f}^{+} \frac{\partial P_{f}}{\partial t} \\ + \left(\overline{c}_{v} + \theta^{+} \frac{\overline{\tau}_{f}^{2}(\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss}) - 2(\overline{\tau}_{s} - \alpha)\overline{\tau}_{f}\overline{\lambda}_{sf} + (\overline{\tau}_{s} - \alpha)^{2}\overline{\lambda}_{ff}}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})}\right) \frac{\partial \theta}{\partial t} \\ + \theta^{+} \frac{(\overline{\tau}_{s} - \alpha)\overline{\lambda}_{ff} - \overline{\tau}_{f}\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \frac{1}{3} \frac{\partial \operatorname{tr} \mathbf{T}_{I}}{\partial t}$$

$$= (\kappa + \theta^{+}(\alpha_{f} + \varsigma/\theta^{+})((\alpha_{f} - \gamma)/\zeta)) \operatorname{Div}(\operatorname{GRAD} \theta) \\ - \theta^{+}((\alpha_{f} + \varsigma/\theta^{+})/\zeta)\varphi_{f}^{+} \operatorname{Div}(\operatorname{GRAD} P_{f})$$

$$(5.1.28)$$

In the isothermal case, obtained by allowing the thermal conductivity  $\kappa$  to become unbounded and by imposing sufficient boundary conditions to yield the solution  $\Theta(\mathbf{X}, t) = \theta^+$ , (5.1.27) can be written

$$\frac{\partial (P_f + \frac{B}{3} \operatorname{tr} \mathbf{T}_I)}{\partial t} = c \operatorname{Div}(\operatorname{GRAD} P_f)$$
(5.1.29)

where c is a positive number called the *compressible consolidation coefficient* defined by

#### **Chapter 5**

$$c = \frac{1}{\xi} \frac{\overline{\lambda}_{ff} (\overline{\lambda}_{ss} + \frac{2}{3} \mu_{ss}) - \overline{\lambda}_{sf}^2}{\overline{\lambda}_{ss} + \frac{2}{3} \mu_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}$$
(5.1.30)

and *B* is defined by

$$B = \frac{1}{\varphi_f^+} \frac{\overline{\lambda}_{ff} + \overline{\lambda}_{sf}}{\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}$$
(5.1.31)

The coefficient *B* is usually called the Skempton coefficient [Ref, 6].

Returning to the case that is not necessarily isothermal, but still making the assumption that the inertia of the fluid and solid can be neglected, we have learned that given a stress field  $\mathbf{T}_{I} = \mathbf{T}_{I}(\mathbf{X},t)$  which obeys (5.1.24), equations (5.1.27) and (5.1.28) can be utilized to determine the pore pressure and the temperature. Given this information, equation (5.1.25) can be used to calculate the strain field for the solid,  $\mathbf{E}_{s} = \mathbf{E}_{s}(\mathbf{X},t)$ . Unfortunately this strain field does not necessarily correspond to a strain derived from a displacement field  $\mathbf{w}_{s} = \mathbf{w}_{s}(\mathbf{X},t)$ . As can be seen from (3.2.3) and (3.2.8), given  $\mathbf{E}_{s} = \mathbf{E}_{s}(\mathbf{X},t)$  the displacement field is a solution of the partial differential equation

$$2\mathbf{E}_{s}(\mathbf{X},t) = \mathrm{GRAD}\,\mathbf{w}_{s}(\mathbf{X},t) + \left(\mathrm{GRAD}\,\mathbf{w}_{s}(\mathbf{X},t)\right)^{t}$$
(5.1.32)

As is well known from classical elasticity, the integrability or compatibility condition on this partial differential equation is

$$\operatorname{Div}(\operatorname{GRAD} \mathbf{E}_{s}) + \operatorname{GRAD}(\operatorname{GRAD}(\operatorname{tr} \mathbf{E}_{s})) - (\operatorname{GRAD}(\operatorname{Div}(\mathbf{E}_{s})))^{T} = \mathbf{0}$$
(5.1.33)

Equation (5.1.33) is a necessary condition for the existence of a displacement field  $\mathbf{w}_s = \mathbf{w}_s(\mathbf{X}, t)$ . A discussion of (5.1.33) within the context of classical elasticity can be found in any elasticity book. Reference 7 contains an excellent discussion of the compatibility equations of elasticity.

Equation (5.1.33) imposes additional conditions on the stress, pore pressure and temperature fields. The explicit form of these conditions is obtained by and substituting (5.1.25) into (5.1.33). The result turns out to be

Div GRAD 
$$\begin{pmatrix} \frac{3(\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + 2\mu_{ss}}{2(\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + 2\mu_{ss}} \mathbf{T}_{I} - \frac{\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}}{2(\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + 2\mu_{ss}} (\operatorname{tr} \mathbf{T}_{I})\mathbf{I} \end{pmatrix}$$
  
+ GRAD(GRAD(tr T<sub>1</sub>))  
+  $\frac{\mu_{ss}(1 + \bar{\lambda}_{sf}/\bar{\lambda}_{ff})}{(\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + \mu_{ss}}} \varphi_{f}^{+} ((\operatorname{Div}(\operatorname{GRAD} P_{f})))\mathbf{I} + \operatorname{GRAD}(\operatorname{GRAD} P_{f}))$   
+  $\frac{\mu_{ss}(\bar{\tau}_{s} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + \mu_{ss}}{(\bar{\lambda}_{ss} - \bar{\lambda}_{sf}^{2}/\bar{\lambda}_{ff}) + \mu_{ss}}} ((\operatorname{Div}(\operatorname{GRAD} \theta))\mathbf{I} + \operatorname{GRAD}(\operatorname{GRAD} \theta)) = \mathbf{0}$  (5.1.34)

where equation (5.1.24) has been used. Equation (5.1.34) is an extra condition which must be obeyed by the stress, the pore pressure and the temperature. The isothermal version of (5.1.34) is equivalent to a result first obtained by Rice & Cleary [Ref. 8].

### 5.2 Incompressible Models

If the binary fluid solid mixture is incompressible, then the appropriate field and constitutive equations are given in Section 4.5. The constitutive equations (4.5.1) through (4.5.4) remain valid in the case where the fluid inertia is neglected. Equations (5.1.3) and (5.1.5) are also valid with the understanding that the pore pressure,  $P_f$ , is the indeterminant pressure. These equations can be used to repeat the calculations of Section 5.1 in the incompressible case. An equivalent approach is to utilize the results of Section 4.4 to calculate the incompressible limits of formulas derived in Section 5.1. We will adopt this approach in this section.

The results we shall need from Section 4.4, specialized to the case of a mixture of a fluid and a solid are (4.4.20) (4.4.21), (4.4.23), (4.4.30, and (4.4.32). These results are

$$\frac{1}{\lambda_{ff}} \to 0 \tag{5.2.1}$$

$$\frac{\tau_f}{\lambda_{ff}} \to 0$$
 (5.2.2)

$$\frac{\lambda_{sf}}{\lambda_{ff}} \to \frac{\varphi_s^+}{\varphi_f^+} \tag{5.2.3}$$

$$\lambda_{ss} - \frac{\varphi_s^+}{\varphi_f^+} \lambda_{sf} \to \lambda_{ss}^* \tag{5.2.4}$$

and

$$\tau_s - \frac{\varphi_s^+}{\varphi_f^+} \tau_f \to \tau_s^* \tag{5.2.5}$$

In the manipulations below, we will also need to use the result (2.1.17) in the form

$$\varphi_s^+ + \varphi_f^+ = 1 \tag{5.2.6}$$

A result which follows from (5.2.3) and (5.2.6) is

$$1 + \frac{\lambda_{sf}}{\lambda_{ff}} \to \frac{1}{\varphi_f^+} \tag{5.2.7}$$

It is also true from (5.2.3) and (5.2.4) that

$$\lambda_{ss} - \frac{\lambda_{sf}^2}{\lambda_{ff}} = \lambda_{ss} - \frac{\lambda_{sf}}{\lambda_{ff}} \lambda_{sf} \to \lambda_{ss}^*$$
(5.2.8)

and from (5.2.3) and (5.2.5) that

$$\tau_s - \frac{\lambda_{sf}}{\lambda_{ff}} \tau_f \to \tau_s^* \tag{5.2.9}$$

Given (5.2.2) and (5.2.7), equation (5.1.9) reduces, in the incompressible case, to

$$\operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t} = \left(\varphi_{f}^{+2} / \xi\right)\operatorname{Div}(\operatorname{GRAD} P_{f}) - \varphi_{f}^{+}\left(\left(\alpha_{f} - \gamma\right) / \xi\right)\operatorname{Div}(\operatorname{GRAD} \theta)$$
(5.2.10)

It follows from (5.2.7), (5.2.8) and (5.2.9) that the incompressible limit of (5.1.10) is

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = -\operatorname{GRAD} P_{f} + (\lambda_{ss}^{*} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) - (\tau_{s}^{*} - \alpha) \operatorname{GRAD} \theta$$
(5.2.11)

Likewise, it follows from (5.2.2) and (5.2.9) that the incompressible limit of (5.1.12) is

$$c_{v} \frac{\partial \theta}{\partial t} + \theta^{+} (\tau_{s}^{*} - \alpha) \operatorname{Div} \frac{\partial \mathbf{w}_{s}}{\partial t}$$
  
=  $(\kappa + \theta^{+} (\alpha_{f} + \varsigma/\theta^{+}) (\alpha_{f} - \gamma)/\xi) \operatorname{Div}(\operatorname{GRAD} \theta)$   
 $- ((\alpha_{f} + \varsigma/\theta^{+})/\xi) \varphi_{f}^{+} \operatorname{Div}(\operatorname{GRAD} P_{f})$  (5.2.12)

Equations (5.2.10), (5.2.11) and (5.2.12) are the field equations one would use, along with appropriate initial and boundary data, for the determination of the solid displacement,  $\mathbf{w}_s$ , the pore

pressure,  $P_f$  and the temperature,  $\theta$ , in the case where the fluid and solid are incompressible and the inertia of the fluid is neglected.

If we make the additional assumption, as in Section 5.1, that the inertia of the solid can also be neglected, we can derive incompressible versions of equations (5.1.26) through (5.1.28). (5.1.28). First, consider equation (5.1.26). If we utilize (5.2.8), (5.2.7) and (5.2.9), the incompressible version of (5.1.26) is

$$\operatorname{tr} \mathbf{E}_{s} = \operatorname{Div} \mathbf{w}_{s} = \frac{1}{3\lambda_{ss}^{*} + 2\mu_{ss}} \operatorname{tr} \mathbf{T}_{\mathbf{I}} + \frac{3}{3\lambda_{ss}^{*} + 2\mu_{ss}} P_{f} + \frac{3(\tau_{s}^{*} - \alpha)}{3\lambda_{ss}^{*} + 2\mu_{ss}} (\theta - \theta^{+})$$
(5.2.13)

Likewise, the incompressible version of (5.1.25) is

$$2\mu_{ss}\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{\lambda_{ss}^{*}}{3\lambda_{ss}^{*} + 2\mu_{ss}} (\operatorname{tr} \mathbf{T}_{I})\mathbf{I} + \frac{2\mu_{ss}}{3\lambda_{ss}^{*} + 2\mu_{ss}} P_{f}\mathbf{I} + \frac{2\mu_{ss}(\tau_{s}^{*} - \alpha)}{3\lambda_{ss}^{*} + 2\mu_{ss}} (\theta - \theta^{+})\mathbf{I}$$
(5.2.14)

In order to obtain the incompressible version of (5.1.27), we need, first, the incompressible limit of the coefficient

$$\frac{\lambda_{ss} + \frac{2}{3}\mu_{ss} + 2\lambda_{sf} + \lambda_{ff}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})}$$

A rearrangement yields

$$\frac{\lambda_{ss} + \frac{2}{3}\mu_{ss} + 2\lambda_{sf} + \lambda_{ff}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})} = \frac{\lambda_{ss} - \frac{\lambda_{sf}^2}{\lambda_{ff}} + \frac{2}{3}\mu_{ss} + \lambda_{ff} (1 + \frac{\lambda_{sf}}{\lambda_{ff}})^2}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})} = \frac{\frac{1}{\lambda_{ff}} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})}$$
(5.2.15)
$$= \frac{\frac{1}{\lambda_{ff}} (\lambda_{ss} - (\frac{\lambda_{sf}}{\lambda_{ff}})\lambda_{sf}) + \frac{2}{3}\frac{\mu_{ss}}{\lambda_{ff}} + (1 + \frac{\lambda_{sf}}{\lambda_{ff}})^2}{\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss}}$$

As a result,

$$\frac{\lambda_{ss} + \frac{2}{3}\mu_{ss} + 2\lambda_{sf} + \lambda_{ff}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})} \rightarrow \frac{1}{\varphi_f^{+2}} \frac{1}{\lambda_{ss}^* + \frac{2}{3}\mu_{ss}}$$
(5.2.16)

#### Chapter 5

where (5.2.2), (5.2.3), (5.2.7) and (5.2.8) have been used. The coefficient of  $\frac{\partial \theta}{\partial t}$  in (5.1.27) can be written

$$\frac{(\tau_{s}-\alpha)(\lambda_{ff}+\lambda_{sf})-\tau_{f}(\lambda_{ss}+\frac{2}{3}\mu_{ss}+\lambda_{sf})}{\lambda_{ff}(\lambda_{ss}-\lambda_{sf}^{2}/\lambda_{ff}+\frac{2}{3}\mu_{ss})} = \frac{(\tau_{s}-\alpha)(1+\frac{\lambda_{sf}}{\lambda_{ff}})-\frac{\tau_{f}}{\lambda_{ff}}(\lambda_{ss}+\frac{2}{3}\mu_{ss}+\lambda_{sf})}{\lambda_{ss}-\lambda_{sf}^{2}/\lambda_{ff}+\frac{2}{3}\mu_{ss}}$$
$$= \frac{(\tau_{s}-\frac{\lambda_{sf}}{\lambda_{ff}}\tau_{f}-\alpha)(1+\frac{\lambda_{sf}}{\lambda_{ff}})-\frac{\tau_{f}}{\lambda_{ff}}(\lambda_{ss}-\lambda_{sf}^{2}/\lambda_{ff}+\frac{2}{3}\mu_{ss})}{\lambda_{ss}-\lambda_{sf}^{2}/\lambda_{ff}+\frac{2}{3}\mu_{ss}}$$
(5.2.17)
$$\rightarrow \frac{1}{\varphi_{f}^{+}}\frac{(\tau_{s}^{*}-\alpha)}{\lambda_{ss}^{*}+\frac{2}{3}\mu_{ss}}$$

where (5.2.2), (5.2.7) (5.2.8) and (5.2.9) have been used.

The coefficient of  $\frac{1}{3} \frac{\partial \operatorname{tr} \mathbf{T}_{\mathrm{I}}}{\partial t}$  in (5.1.27) has the incompressible limit

$$\frac{\lambda_{ff} + \lambda_{sf}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3} \mu_{ss})} \rightarrow \frac{1}{\varphi_f^+} \frac{1}{\lambda_{ss}^* + \frac{2}{3} \mu_{ss}}$$
(5.2.18)

Equations (5.2.16), (5.2.17) and (5.2.18) yield, from (5.1.27),

$$\frac{1}{\varphi_{f}^{+}(\lambda_{ss}^{*}+\frac{2}{3}\mu_{ss})}\frac{\partial(P_{f}+\frac{1}{3}\operatorname{tr}\mathbf{T}_{I})}{\partial t}+\frac{\tau_{s}^{*}-\alpha}{\varphi_{f}^{+}(\lambda_{ss}^{*}+\frac{2}{3}\mu_{ss})}\frac{\partial\theta}{\partial t}$$

$$=(\varphi_{f}^{+}/\xi)\operatorname{Div}(\operatorname{GRAD} P_{f})-((\alpha_{f}-\gamma)/\xi)\operatorname{Div}(\operatorname{GRAD} \theta)$$
(5.2.19)

Similar arguments yield, for the coefficients of (5.1.28),

$$c_{v} + \theta^{+} \frac{\tau_{f}^{2} (\lambda_{ss} + \frac{2}{3} \mu_{ss}) - 2(\tau_{s} - \alpha)\tau_{f}\lambda_{sf} + (\tau_{s} - \alpha)^{2}\lambda_{ff}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^{2} / \lambda_{ff} + \frac{2}{3} \mu_{ss})}$$

$$= c_{v} + \theta^{+} \frac{\tau_{f}^{2} (\lambda_{ss} - \lambda_{sf}^{2} / \lambda_{ff} + \frac{2}{3} \mu_{ss}) + (\tau_{s} - \frac{\lambda_{sf}}{\lambda_{ff}} \tau_{f} - \alpha)^{2}\lambda_{ff}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^{2} / \lambda_{ff} + \frac{2}{3} \mu_{ss})}$$

$$\Rightarrow c_{v} + \theta^{+} \frac{(\tau_{s}^{*} - \alpha)^{2}}{(\lambda_{ss}^{*} + \frac{2}{3} \mu_{ss})}$$
(5.2.20)

and

$$\frac{(\tau_s - \alpha)\lambda_{ff} - \tau_f \lambda_{sf}}{\lambda_{ff} (\lambda_{ss} - \lambda_{sf}^2 / \lambda_{ff} + \frac{2}{3}\mu_{ss})} \rightarrow \frac{\tau_s^* - \alpha}{\lambda_{ss}^* + \frac{2}{3}\mu_{ss})}$$
(5.2.21)

Therefore, the incompressible version of (5.1.28) is

$$\theta^{+} \frac{\tau_{s}^{*} - \alpha}{(\lambda_{ss}^{*} + \frac{2}{3}\mu_{ss})} \frac{\partial(P_{f} + \frac{1}{3}\operatorname{tr}\mathbf{T}_{I})}{\partial t} + \left(c_{v} + \theta^{+} \frac{(\tau_{s}^{*} - \alpha)^{2}}{(\lambda_{ss}^{*} + \frac{2}{3}\mu_{ss})}\right) \frac{\partial\theta}{\partial t}$$
$$= (\kappa + \theta^{+} (\alpha_{f} + \varsigma/\theta^{+})(\alpha_{f} - \gamma)/\xi))\operatorname{Div}(\operatorname{GRAD}\theta)$$
$$-\theta^{+} ((\alpha_{f} + \varsigma/\theta^{+})/\xi)\varphi_{f}^{+}\operatorname{Div}(\operatorname{GRAD}P_{f})$$
(5.2.22)

In the isothermal limit, i.e. in the case where  $\kappa \to \infty$  in (5.2.22) and sufficient boundary conditions are provided to yield the solution  $\Theta(\mathbf{X},t) = \theta^+$ , equation (5.2.19) reduces to the incompressible limit of (5.1.29), i.e.

$$\frac{\partial (P_f + \frac{1}{3} \operatorname{tr} \mathbf{T}_I)}{\partial t} = c^* \operatorname{Div}(\operatorname{GRAD} P_f)$$
(5.2.23)

where  $c^*$  is the *incompressible consolidation coefficient*. This coefficient is the incompressible limit of *c* in (5.1.30) and has the explicit form

$$c^* = \frac{\varphi_f^{+2}}{\xi} (\lambda_{ss}^* + \frac{2}{3}\mu_{ss})$$
(5.2.24)

Notice, in passing, that in the incompressible limit, B, defined by (5.1.31), takes the value

$$B = 1$$
 (5.2.25)

Returning to the case that is not necessarily isothermal, but still making the assumption that the inertia of the fluid and solid can be neglected, we have learned that given a stress field  $\mathbf{T}_{I} = \mathbf{T}_{I}(\mathbf{X},t)$  which obeys (5.1.24), equations (5.2.19) and (5.2.22) can be utilized to determine the pore pressure and the temperature. As in Section 5.1, this information can be used to calculate the strain field for the solid  $\mathbf{E}_{s}(\mathbf{X},t)$ . Also, as in Section 5.1, this strain field does not necessarily correspond to a strain derived from a displacement field  $\mathbf{w}_{s} = \mathbf{w}_{s}(\mathbf{X},t)$ . The additional condition one must place on the stress, pore pressure and temperature fields is the incompressible version of (5.1.34). This result is

Div GRAD 
$$\left(\frac{3\lambda_{ss}^{*}+2\mu_{ss}}{2\lambda_{ss}^{*}+2\mu_{ss}}\mathbf{T}_{I}-\frac{\lambda_{ss}^{*}}{2\lambda_{ss}^{*}+2\mu_{ss}}(\operatorname{tr}\mathbf{T}_{I})\mathbf{I}\right)$$
  
+ GRAD(GRAD(tr  $\mathbf{T}_{I}$ ))  
+  $\frac{\mu_{ss}}{\lambda_{ss}^{*}+\mu_{ss}}\left((\operatorname{Div}(\operatorname{GRAD} P_{f}))\mathbf{I}+\operatorname{GRAD}(\operatorname{GRAD} P_{f})\right)$   
+  $\frac{\mu_{ss}(\tau_{s}^{*}-\alpha)}{\lambda_{ss}^{*}+\mu_{ss}}\left((\operatorname{Div}(\operatorname{GRAD} \theta))\mathbf{I}+\operatorname{GRAD}(\operatorname{GRAD} \theta)\right)=\mathbf{0}$  (5.2.26)

For incompressible materials, the case where both the fluid and solid inertias can be neglected, there is another convenient form of the governing partial differential equations. These equations involve the elimination of the pore pressure and the stress in favor of the quantity Div  $\mathbf{w}_s$ . This elimination is possible because the pore pressure and the stress enter (5.2.19) and (5.2.22) in the combination  $P_f + \frac{1}{3} \operatorname{tr} \mathbf{T}_I$ . The manipulations leading to the desired results are somewhat simplified if we, first, solve (5.2.13) for  $P_f + \frac{1}{3} \operatorname{tr} \mathbf{T}_I$  to obtain

$$P_{f} + \frac{1}{3} \operatorname{tr} \mathbf{T}_{I} = (\lambda_{ss}^{*} + \frac{2}{3} \mu_{ss}) \operatorname{Div} \mathbf{w}_{s} - (\tau_{s}^{*} - \alpha)(\theta - \theta^{+})$$
(5.2.27)

Given the fact that we are neglecting the inertia of the fluid and the solid, it follows from (5.2.11) that

$$\operatorname{GRAD} P_f = (\lambda_{ss}^* + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_s) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_s) - (\tau_s^* - \alpha_s) \operatorname{GRAD} \theta \quad (5.2.28)$$

The divergence of this equation yields

$$\operatorname{Div}(\operatorname{GRAD} P_f) = (\lambda_{ss}^* + 2\mu_{ss})\operatorname{Div}(\operatorname{GRAD}(\operatorname{Div} \mathbf{w}_s)) - (\tau_s^* - \alpha)\operatorname{Div}(\operatorname{GRAD} \theta)$$
(5.2.29)

The two results we seek are obtained by utilizing (5.2.27) and (5.2.29) to eliminate  $P_f + \frac{1}{3} \text{tr } \mathbf{T}_I$  and Div(GRAD  $P_f$ ), respectively, from (5.2.19) and (5.2.22). The results of this manipulation are

$$\frac{\partial \operatorname{Div} \mathbf{w}_{s}}{\partial t} = \frac{\varphi_{f}^{+2}}{\xi} (\lambda_{ss}^{*} + 2\mu_{ss}) \operatorname{Div}(\operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s})) - \frac{\varphi_{f}^{+}}{\xi} (\varphi_{f}^{+}(\tau_{s}^{*} - \alpha) + (\alpha_{f} - \gamma)) \operatorname{Div}(\operatorname{GRAD}\theta)$$
(5.2.30)

and

$$\theta^{+}(\tau_{s}^{*}-\alpha)\frac{\partial\operatorname{Div}\mathbf{w}_{s}}{\partial t}+c_{v}\frac{\partial\theta}{\partial t}$$

$$=\left(\kappa+\theta^{+}\left(\alpha_{f}+\varsigma/\theta^{+}\right)\left(\varphi_{f}^{+}(\tau_{s}^{*}-\alpha)+(\alpha_{f}-\gamma)/\xi\right)\right)\operatorname{Div}(\operatorname{GRAD}\theta) \qquad (5.2.31)$$

$$-\theta^{+}\left(\varphi_{f}^{+}/\xi\right)\left(\alpha_{f}+\varsigma/\theta^{+}\right)\left(\lambda_{ss}^{*}+2\mu_{ss}\right)\operatorname{Div}(\operatorname{GRAD}(\operatorname{Div}\mathbf{w}_{s}))$$

An interesting conclusion from (5.2.30) is that in the isothermal case,  $\text{Div} \mathbf{w}_s$  obeys a classical diffusion equation.

## References

- 1. CARLSON, D. E., *Linear Thermoelasticity*, Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), Springer Verlag, Berlin (1972).
- 2. LUBINSKI, A., The Theory of Elasticity for Porous Bodies Displaying a Strong Pore Structure, *Proc. 2nd U. S. Nat. Congr. Appl. Mech.*, 247-256 (1954).
- 3. BIOT, MAURICE A., Thermoelasticity and Irreversible Thermodynamics, *J. Appl. Phys.*, 27, 240-253 (1956).
- 4. BIOT, MAURICE A., Mechanics of Deformation and Acoustic Propagation in Porous Media, *J. Appl. Phys.*, **33**, 1482-1498 (1962).
- 5. GEERTSMA, J., A Remark on the Analogy Between Thermoelasticity and the Elasticity of Saturated Porous Media, *J. Mech. Phys. Solids*, **6**, 13-16 (1957).
- 6. SKEMPTON, A. W., The Pore-Pressure Coefficients A and B, Geotechnique, 4, 143-147 (1954).
- 7. GURTIN. M. E., *The Linear Theory of Elasticity*, Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), Springer Verlag, Berlin (1972).
- 8. RICE, J. R., and M. P. CLEARY, Some Basic Solutions for Fluid-Saturated Porous Media With Compressible Constituents, *Review of Geophysics and Space Physics*, **14**, 227-241, (1976).

# **Further Transformations and Material Properties**

In spite of the many forms of the governing equations presented in Chapters 4 and 5, there are still further forms which we need to record. These expressions will be given in this Chapter. In addition, an effort will be made to make connections with notations and formalisms one can find in other, very important, references on the subject of porous elasticity. In this chapter, we continue to restrict the discussion to that of a binary mixture of a compressible fluid and a compressible solid with pore pressure. When we discuss the case of an incompressible mixture, we shall follow the technique of Section 5.2 where the incompressible case is derived as a limit of the compressible case.

#### 6.1 Constitutive Equations-Alternate Forms

If we continue to consider a binary mixture of a compressible fluid and a compressible solid with pore pressure, then the following four equations summarize the constitutive equations for the pore pressure, the entropy, the chemical potential for the solid, the inner part of the stress and the volume of the fluid per unit volume of undeformed solid.

$$\varphi_f^+ P_f = \overline{\tau}_f (\theta - \theta^+) - (\overline{\lambda}_{sf} + \overline{\lambda}_{ff}) (\operatorname{tr} \mathbf{E}_s) + \overline{\lambda}_{ff} (m_b - m_b^+) / m_b^+$$
(6.1.1)

$$\rho \eta = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+}) + (\overline{\tau}_{s} + \overline{\tau}_{f}) (\operatorname{tr} \mathbf{E}_{s}) - \overline{\tau}_{f} (m_{f} - m_{f}^{+}) / m_{f}^{+}$$
(6.1.2)

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\overline{\lambda}_{ss} + \overline{\lambda}_{sf})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \overline{\lambda}_{sf}((m_{f} - m_{f}^{+})/m_{f}^{+})\mathbf{I} + \overline{\tau}_{s}(\theta - \theta^{+})\mathbf{I}$$
(6.1.3)

$$\mathbf{T}_{\mathrm{I}} = (\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff})(\mathrm{tr}\,\mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - (\overline{\lambda}_{sf} + \overline{\lambda}_{ff})((m_{f} - m_{f}^{+})/m_{f}^{+})\mathbf{I} -(\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)(\theta - \theta^{+})\mathbf{I}$$
(6.1.4)

and

$$\frac{\upsilon_f - \upsilon_f^+}{\upsilon_f^+} = \left(\frac{\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}{\varphi_f^+ K_f}\right) \operatorname{tr} \mathbf{E}_s + \left(1 - \frac{\overline{\lambda}_{ff}}{\varphi_f^+ K_f}\right) (m_f - m_f^+) / m_f^+ - \frac{\overline{\tau}_f}{\varphi_f^+ K_f} (\theta - \theta^+)$$
(6.1.5)

Equation (6.1.1) through (6.1.4) are binary versions of (4.1.49) through (4.1.52), respectively. Equation (6.1.5) is the binary version of equation (4.1.54).

In Section 4.1, we introduced the thermodynamic potential  $W(\theta, \mathbf{E}_s, m_b)$  and showed that it was given by equation (4.1.56). The binary mixture version of (4.1.56) is

$$W(\theta, \mathbf{E}_{s}, m_{b}) = \alpha(\theta - \theta^{+}) - \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} - (\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)(\theta - \theta^{+})(\operatorname{tr} \mathbf{E}_{s}) + \overline{\tau}_{f} (\theta - \theta^{+})(m_{f} - m_{f}^{+}) / m_{f}^{+} + \frac{1}{2} (\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff})(\operatorname{tr} \mathbf{E}_{s})^{2} - (\overline{\lambda}_{sf} + 2\overline{\lambda}_{ff})(\operatorname{tr} \mathbf{E}_{s}) ((m_{f} - m_{f}^{+}) / m_{f}^{+}) + \frac{1}{2} \overline{\lambda}_{ff} ((m_{f} - m_{f}^{+}) / m_{f}^{+}) ((m_{f} - m_{f}^{+}) / m_{f}^{+}) + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s})$$

$$(6.1.6)$$

This result, in turn, yields the binary mixture versions of (4.1.57), (4.1.58) and (4.1.59). These results are

$$\varphi_{f}^{+}P_{f} = \frac{\partial W}{\partial m_{f}} = \overline{\tau}_{f}(\theta - \theta^{+}) - \left(\overline{\lambda}_{sf} + \overline{\lambda}_{ff}\right)(\operatorname{tr} \mathbf{E}_{s}) + \overline{\lambda}_{ff}(m_{f} - m_{f}^{+})/m_{f}^{+}$$
(6.1.7)

$$\mathbf{T}_{\mathrm{I}} = \frac{\partial W}{\partial \mathbf{E}_{s}} = (\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - (\overline{\lambda}_{sf} + \overline{\lambda}_{ff})(m_{f} - m_{f}^{+})/m_{f}^{+}\mathbf{I} - (\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)(\theta - \theta^{+})\mathbf{I}$$
(6.1.8)

and

$$\mathbf{H} = -\frac{\partial W}{\partial \theta} = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + (\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)(\operatorname{tr} \mathbf{E}_{s}) - \overline{\tau}_{f}(m_{f} - m_{f}^{+}) / m_{f}^{+}$$
(6.1.9)

In the following, the entropy of the mixture per unit of undeformed solid volume, defined by (4.1.59), will be used as the constitutive equation for the entropy.

In several cases in Chapter 4 we wrote the constitutive equations with temperature, the strain of the solid and the fluid pore pressure as the independent variables. The results with this choice of independent variables are as follows:

$$(m_f - m_f^+) / m_f^+ = \frac{1}{\overline{\lambda}_{ff}} \varphi_f^+ P_f - \frac{\overline{\tau}_f}{\overline{\lambda}_{ff}} (\theta - \theta^+) + (1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}) (\operatorname{tr} \mathbf{E}_s)$$
(6.1.10)

$$\mathbf{H} = -\alpha + (\overline{c}_{v} + \theta^{+} \frac{\overline{\tau}_{f}^{2}}{\overline{\lambda}_{ff}})(\theta - \theta^{+}) / \theta^{+} + (\overline{\tau}_{s} - \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}} \overline{\tau}_{f} - \alpha)(\operatorname{tr} \mathbf{E}_{s}) - \frac{\overline{\tau}_{f}}{\overline{\lambda}_{ff}} \varphi_{f}^{+} P_{f}$$
(6.1.11)

$$\mathbf{T}_{\mathrm{I}} = (\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}})(\operatorname{tr} \mathbf{E}_{s})\mathbf{I} + 2\mu_{ss}\mathbf{E}_{s} - (1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}})\varphi_{f}^{+}P_{f}\mathbf{I} - (\overline{\tau}_{s} - \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}\tau_{c} - \alpha)(\theta - \theta^{+})\mathbf{I}$$
(6.1.12)

#### **Porous Elasticity Models**

and

$$(\upsilon_f - \upsilon_f^+) / \upsilon_f^+ = (1 + \frac{\overline{\lambda}_{sc}}{\overline{\lambda}_{ff}})(\operatorname{tr} \mathbf{E}_s) + \frac{1}{\overline{\lambda}_{ff}}(1 - \frac{\overline{\lambda}_{ff}}{\varphi_f^+ K_f})\varphi_f^+ P_f - \frac{\overline{\tau}_f}{\overline{\lambda}_{ff}}(\theta - \theta^+)$$
(6.1.13)

Equation (6.1.10) follows from equation (6.1.7), equations (6.1.11) and (6.1.12) are the result of using (6.1.10) to eliminate the quantity  $(m_f - m_f^+)/m_f^+$  from (6.1.9) and (6.1.8). Equation (6.1.13) follows by the same elimination applied to (6.1.5).

In Sections 4.1 and 5.1 we found it convenient to eliminate the strain of the solid in favor of the inner part of the stress. Next, we shall record versions of the two sets of constitutive equations above where this elimination has been performed. First, we can solve (6.1.8) to obtain

$$2\mu_{ss}\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}} \frac{1}{3} (\operatorname{tr} \mathbf{T}_{I})\mathbf{I}$$
$$+ \frac{2\mu_{ss}}{3} \frac{(\overline{\lambda}_{ff} + \overline{\lambda}_{sf})}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf}} + \overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}} (m_{f} - m_{f}^{+})/m_{f}^{+}\mathbf{I}$$
$$+ \frac{2\mu_{ss}}{3} \frac{(\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf}} + \overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}} (\theta - \theta^{+})\mathbf{I}$$

This expression can be used to eliminate  $\mathbf{E}_s$  in favor of  $\mathbf{T}_I$  in equations (6.1.7), (6.1.8) and (6.1.5). The results turn out to be

$$\varphi_{f}^{+}P_{f} = -\frac{(\overline{\lambda}_{ff} + \overline{\lambda}_{sf})}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T}_{I} + \overline{\lambda}_{ff} \frac{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} (m_{b} - m_{b}^{+})/m_{b}^{+} + \frac{\overline{\tau}_{f}(\overline{\lambda}_{ss} + \overline{\lambda}_{sf} + \frac{2}{3}\mu_{ss}) - (\overline{\lambda}_{ff} + \overline{\lambda}_{sf})(\overline{\tau}_{s} - \alpha)}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} (\theta - \theta^{+})$$

$$(6.1.15)$$

## Chapter 6

$$H = -\alpha + \left(\overline{c_{v}} + \theta^{+} \frac{(\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)^{2}}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}\right) \frac{(\theta - \theta^{+})}{\theta^{+}} + \frac{(\overline{\tau}_{s} + \overline{\tau}_{f} - \alpha)}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T}_{I} + \frac{(\overline{\tau}_{s} - \alpha)(\overline{\lambda}_{sf} + \overline{\lambda}_{ff}) - \overline{\tau}_{f}(\overline{\lambda}_{ss} + \overline{\lambda}_{sf} + \frac{2}{3}\mu_{ss})}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} (m_{f} - m_{f}^{+})/m_{f}^{+}$$

$$(6.1.16)$$

and

$$\frac{\nu_{f} - \nu_{f}^{+}}{\nu_{f}^{+}} = \frac{\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}{\varphi_{f}^{+} K_{f}} \frac{1}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T}_{\mathrm{I}}$$

$$+ \left( \frac{1}{\varphi_{f}^{+} K_{f}} \frac{(\overline{\lambda}_{sf} + \overline{\lambda}_{ff})^{2}}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} + \left(1 - \frac{\overline{\lambda}_{ff}}{\varphi_{f}^{+} K_{f}}\right)\right) (m_{f} - m_{f}^{+}) / m_{f}^{+} \qquad (6.1.17)$$

$$+ \frac{1}{\varphi_{f}^{+} K_{f}} \frac{(\overline{\lambda}_{sf} + \overline{\lambda}_{ff})(\overline{\tau}_{s} - \alpha) - \overline{\tau}_{f}(\overline{\lambda}_{ss} + \overline{\lambda}_{sf} + \frac{2}{3}\mu_{ss})}{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} (\theta - \theta^{+})$$

A similar set of calculations starting with (6.1.10) yields

$$2\mu_{ss}\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}}{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}} \frac{1}{3}(\operatorname{tr}\mathbf{T}_{I})\mathbf{I} + \frac{2\mu_{ss}(1 + \overline{\lambda}_{sf}/\overline{\lambda}_{ff})}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}}}\varphi_{f}^{+}P_{f}\mathbf{I}$$

$$+ \frac{2\mu_{ss}(\overline{\tau}_{s} - \overline{\lambda}_{sf}\overline{\tau}_{f}/\overline{\lambda}_{ff} - \alpha)}{3(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + 2\mu_{ss}}}(\theta - \theta^{+})\mathbf{I}$$

$$(6.1.18)$$

$$(m_{f} - m_{f}^{+})/m_{f}^{+} = \frac{(1 + \overline{\lambda}_{sf}/\overline{\lambda}_{ff})}{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T}_{I} + \frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} \varphi_{f}^{+}P_{f} + \frac{(\overline{\lambda}_{ff} + \overline{\lambda}_{sf})(\tau_{s} - \alpha) - (\overline{\lambda}_{ss} + \overline{\lambda}_{sf} + \frac{2}{3}\mu_{ss})\tau_{f}}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss})} (\theta - \theta^{+})$$

$$(6.1.19)$$

$$\mathbf{H} = -\alpha + \left( \overline{c_{v}} + \theta^{+} \frac{\overline{c_{f}}^{2} ((\overline{\lambda_{ss}} - \overline{\lambda_{sf}}^{2} / \overline{\lambda_{ff}}) + \frac{2}{3} \mu_{ss}) + (\overline{c_{s}} - \frac{\overline{\lambda_{sf}}}{\overline{\lambda_{ff}}} \overline{c_{f}} - \alpha)^{2} \overline{\lambda_{ff}}}{\overline{\lambda_{ff}} \left( (\overline{\lambda_{ss}} - \overline{\lambda_{sf}}^{2} / \overline{\lambda_{ff}}) + \frac{2}{3} \mu_{ss} \right)} \right) (\theta - \theta^{+}) / \theta^{+}$$

$$+ \frac{(\overline{c_{s}} - \frac{\overline{\lambda_{sf}}}{\overline{\lambda_{ff}}} \overline{c_{f}} - \alpha)}{(\overline{\lambda_{ss}} - \overline{\lambda_{sf}}^{2} / \overline{\lambda_{ff}}) + \frac{2}{3} \mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T_{I}} + \frac{(\overline{c_{s}} - \alpha)(\overline{\lambda_{ff}} + \overline{\lambda_{sf}}) - \overline{c_{f}} \left(\overline{\lambda_{ss}} + \overline{\lambda_{sf}} + \frac{2}{3} \mu_{ss}\right)}{\overline{\lambda_{ff}} \left( (\overline{\lambda_{ss}} - \overline{\lambda_{sf}}^{2} / \overline{\lambda_{ff}}) + \frac{2}{3} \mu_{ss} \right)} \varphi_{f}^{+} P_{f}$$

$$(6.1.20)$$

and

$$(\upsilon_{f} - \upsilon_{f}^{+})/\upsilon_{f}^{+} = \frac{\left(1 + \overline{\lambda}_{sf}/\overline{\lambda}_{ff}\right)}{\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}} \frac{1}{3} \operatorname{tr} \mathbf{T}_{\mathbf{I}} + \left(\frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}{\overline{\lambda}_{ff}((\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}) + \frac{2}{3}\mu_{ss})} - \frac{1}{\varphi_{f}^{+}K_{f}}\right) \varphi_{f}^{+}P_{f}$$

$$+ \frac{\left(\overline{\lambda}_{ff} + \overline{\lambda}_{sf}\right)(\overline{\tau}_{s} - \alpha) - \overline{\tau}_{f}(\lambda_{ss} + \overline{\lambda}_{sf} + \frac{2}{3}\mu_{ss})}{\overline{\lambda}_{ff}(\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2}/\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss})} (\theta - \theta^{+})$$

$$(6.1.21)$$

Equation (6.1.18) has been given earlier as (5.1.25)

### 6.2 Connections With Other Formulations

The literature on porous elasticity is quite old and there is a large body of literature available to study. It is important to be able to make connections between the formulation given here, which is built on the formalism of the Theory of Mixtures, with these earlier works. In this section we shall relate the notation we have been using to that found in the pioneering papers of Biot and an important formulation by Rice and Cleary.

The first Biot formulation was published in 1941 in Reference 1. This work was an extension and improvement of an earlier formulation of Terzaghi [Ref. 2]. In the notation adopted here, Biot formulated an isothermal porous elasticity model and proposed constitutive equations as follows:

$$\mathbf{E}_{s} = \frac{1+\nu_{p}}{E_{p}} \mathbf{T}_{\mathrm{I}} - \frac{\nu_{p}}{E_{p}} (\operatorname{tr} \mathbf{T}_{\mathrm{I}}) \mathbf{I} + \frac{1}{3H} P_{f} \mathbf{I}$$
(6.2.1)

and

$$v_f - v_f^+ = \frac{1}{3H} \operatorname{tr} \mathbf{T}_{\mathrm{I}} + \frac{1}{R} P_f$$
 (6.2.2)

where  $v_p$ ,  $E_p$ , H and R are material constants. In analogy with classical elasticity, the coefficient  $v_p$  can be identified as the (isothermal) *Poisson ratio at constant pore pressure*. Likewise,  $E_p$  is the (isothermal) *Young's modulus at constant pore pressure*. If we compare (6.2.1) and (6.2.2) with isothermal versions of (6.1.18) and (6.1.21), respectively, we can relate the material coefficients introduced by Biot to those used here. The results of this comparison are

$$\nu_{P} = \frac{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}}}{2(\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}} + \mu_{ss})}$$
(6.2.3)

$$E_{P} = \frac{3\mu_{ss}(\bar{\lambda}_{ss} - \frac{\bar{\lambda}_{sf}^{2}}{\bar{\lambda}_{ff}} + \frac{2}{3}\mu_{ss})}{\bar{\lambda}_{ss} - \frac{\bar{\lambda}_{sf}^{2}}{\bar{\lambda}_{ff}} + \mu_{ss}}$$
(6.2.4)

$$\varphi_{f}^{+}H = \frac{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}}{1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}}$$
(6.2.5)

and

$$\frac{1}{R} = \varphi_f^+ \frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}{\overline{\lambda}_{ff}((\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^2 / \overline{\lambda}_{ff}) + \frac{2}{3}\mu_{ss})} - \frac{1}{K_f}$$
(6.2.6)

In an isothermal theory, the only other material property is the drag coefficient  $\xi$ . This coefficient is related to the permeability k and the viscosity of the fluid  $\mu_f$  by the formula

$$\frac{k}{\mu_f} = \frac{\varphi_f^{+2}}{\xi} \Longrightarrow \xi = \varphi_f^{+2} \frac{\mu_f}{k}$$
(6.2.7)

Equations like (6.2.7) have appeared, earlier, in Section 2.7.

In 1976, Rice and Cleary published an important paper where the Biot model defined by (6.2.1) and (6.2.2) was adopted [Ref. 3]. They redefined certain of the material constants so as to illustrate various points of importance. Rather than the material constants H and R, they introduced constants  $K_s$ ' and  $K_s$ ", where

$$\frac{1}{K_{s}} = \frac{1}{K} - \frac{1}{H}$$
(6.2.8)

and

$$\frac{\varphi_f^+}{K_s''} = \frac{1}{H} - \frac{1}{R}$$
(6.2.9)

The symbol K in (6.2.8) is, in the notation used here, defined by

$$K = \overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^2}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}$$
(6.2.10)

The symbols H and R are given by (6.2.5) and (6.2.6), respectively. Given the definitions (6.2.8) through (6.2.10) and the results (6.2.5) and (6.2.6), it follows that

$$\frac{1}{K_{s'}} = \frac{1}{K} - \frac{1}{H} = \frac{1 - \varphi_{f}^{+} \left(1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}\right)}{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}}$$
(6.2.11)

and

$$\frac{1}{K_{s}"} = \frac{1}{\varphi_{f}^{+}H} - \frac{1}{\varphi_{f}^{+}R} = \frac{\overline{\lambda}_{ff}\left(1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}\right) - \varphi_{f}^{+}\left(\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}\right)}{\overline{\lambda}_{ff}\left((\overline{\lambda}_{ss} - \overline{\lambda}_{sf}^{2} / \overline{\lambda}_{ff}) + \frac{2}{3}\mu_{ss}} - \frac{1}{K_{f}}$$
(6.2.12)

Rice and Cleary wrote the constitutive equations (6.2.1) and (6.2.2) in the forms

$$2G\mathbf{E}_{s} = \mathbf{T}_{I} + P_{f}\mathbf{I} - \frac{v_{p}}{1 + v_{p}} \left(\operatorname{tr} \mathbf{T}_{I} + 3P_{f}\right)\mathbf{I} + \frac{2G}{3} \left(\frac{1}{H} - \frac{1}{K}\right)P_{f}\mathbf{I}$$
(6.2.13)

and

$$\upsilon_f - \upsilon_f^+ = \frac{1}{3H} (\operatorname{tr} \mathbf{T}_{\mathrm{I}} + 3P_f) - \frac{\varphi_f^+}{K_s} P_f$$
(6.2.14)

where their symbol G is what we have called  $\mu_{ss}$ . They also wrote their constitutive equations in the form

#### **Chapter 6**

$$2G\mathbf{E}_{s} = \mathbf{T}_{I} - \frac{v_{p}}{1 + v_{p}} (\operatorname{tr} \mathbf{T}_{I}) \mathbf{I} + \frac{3(v_{m} - v_{p})}{B(1 + v_{p})(1 + v_{m})} P_{f} \mathbf{I}$$
(6.2.15)

and

$$(m_f - m_f^+) / m_f^+ = \frac{3(\nu_m - \nu_p)}{2GB(1 + \nu_m)(1 + \nu_p)} \left( \operatorname{tr} \mathbf{T}_{\mathrm{I}} + \frac{3}{B} P_f \right)$$
(6.2.16)

where  $v_m$  is the (isothermal) Poisson's ratio at constant  $m_f$ . It is defined in terms of the symbols used here by the formula

$$\nu_{m} = \frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}}{2(\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \mu_{ss})}$$
(6.2.17)

The symbol *B* in (6.2.16) is the Skempton coefficient defined by (5.1.31).<sup>1</sup> Equations (6.2.15) and (6.2.16) are easily shown to be the isothermal versions of (6.1.18) and (6.1.19), respectively.

Biot and Willis, wrote their constitutive equations in yet a different form [Ref. 4]. Their formulation adopted a notation used earlier by Biot [Ref. 5]. In any case, they wrote their isothermal constitutive equations as

$$\mathbf{T}_{I} + \varphi_{f}^{+} P_{f} \mathbf{I} = A(\text{tr} \, \mathbf{E}_{s}) + 2N \mathbf{E}_{s} - Q((\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+}) \mathbf{I}$$
(6.2.18)

and

$$-\varphi_{f}^{+}P_{f} = Q \operatorname{tr} \mathbf{E}_{s} - R((\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+})$$
(6.2.19)

The reader is cautioned that the property R in (6.2.19) is not he same as the R introduced in (6.2.2). In any case, if (6.2.18) and (6.2.19) are compared with the isothermal versions of (4.2.9) and (4.2.5), respectively, we see that

$$A = \overline{\lambda}_{ss} \tag{6.2.20}$$

$$N = \mu_{ss} \tag{6.2.21}$$

$$Q = \overline{\lambda}_{sf} \tag{6.2.22}$$

<sup>&</sup>lt;sup>1</sup> Rice and Cleary actually used the symbol  $\nu$  for what we are calling  $\nu_p$  and the symbol  $\nu_u$  for what we are calling  $\nu_m$ .

## **Porous Elasticity Models**

and

$$R = \overline{\lambda}_{ff} \tag{6.2.23}$$

A fundamental part of reference 4 is to relate the four constants A, N, Q and R to four constants the authors label  $\mu, \kappa, \delta$  and  $\gamma$ . The advantage of the constants  $\mu, \kappa, \delta$  and  $\gamma$ , as explained by the authors, is that they are measurable coefficients. In terms of the notation being used here, the coefficients  $\mu, \kappa, \delta$  and  $\gamma$  are given by

$$\mu = \mu_{ss} \tag{6.2.24}$$

$$\kappa = \frac{1}{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^2}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}}$$
(6.2.25)

$$\delta = \frac{1 - \varphi_f^+ \left(1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}\right)}{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^2}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}}$$
(6.2.26)

and

$$\gamma = \varphi_{f}^{+2} \frac{\overline{\lambda}_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff} + \frac{2}{3}\mu_{ss}}{\overline{\lambda}_{ff} \left(\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}\right)} - \varphi_{f}^{+} \frac{1 + \frac{\overline{\lambda}_{sf}}{\overline{\lambda}_{ff}}}{\overline{\lambda}_{ss} - \frac{\overline{\lambda}_{sf}^{2}}{\overline{\lambda}_{ff}} + \frac{2}{3}\mu_{ss}}$$
(6.2.27)

It follows from (6.2.11) and (6.2.26) that

$$\delta = \frac{1}{K_s}$$
(6.2.28)

and

$$\gamma = -\varphi_f^+ \left( \frac{1}{K_s''} + \frac{1}{K_f} \right)$$
(6.2.29)

Rice and Cleary have summarized material properties for six rock like porous materials [Ref. 3]. Their Table 1 contains the following numerical values

Property	Ruhr Sandstone	Tennessee Marble	Charcoal Granite	Berea Sandstone	Westerly Granite	Weber Sandstone
$\mu_{ss}$ (kbar)	133	240	187	60	160	122
V <sub>P</sub>	.12	.25	.27	.20	.25	.15
V <sub>m</sub>	.31	.27	.30	.33	.34	.29
В	.88	.51	.55	.62	.85	.73
$arphi_f^+$	.02	.02	.02	.19	.01	.06
$\kappa_w$ (cm/sec)	$2(10)^{-7}$	$(10)^{-10}$	$(10)^{-10}$	$2(10)^{-4}$	$4(10)^{-10}$	$(10)^{-6}$

 Table 1 Typical Rock Properties

The quantity  $\kappa_w$  in Table 1 is a permeability measure for water which relates to the permeability in (6.2.7) by the formula

$$\kappa_w = \frac{k}{\mu_w} \gamma_w g = \frac{\varphi_f^{+2}}{\xi} \gamma_w g \tag{6.2.30}$$

where,  $\gamma_w$  is the true density of water,  $\mu_w$  is the viscosity of water and g is the gravitational constant. Given the information in this table, we can first calculate  $\overline{\lambda}_{ff}, \overline{\lambda}_{sf}$  and  $\overline{\lambda}_{ss}$  by solving equations (5.131), (6.2.3) and (6.2.17). The result of this elimination is

$$\frac{\overline{\lambda}_{ff}}{2\mu_{ss}} = B^2 \varphi_f^{+2} \frac{\left(\frac{\nu_m}{1 - 2\nu_m} + \frac{1}{3}\right)^2}{\left(\frac{\nu_m}{1 - 2\nu_m} - \frac{\nu_P}{1 - 2\nu_P}\right)}$$
(6.2.31)

$$\frac{\overline{\lambda}_{fs}}{2\mu_{ss}} = B\varphi_{f}^{+}\left(\frac{\nu_{m}}{1-2\nu_{m}} + \frac{1}{3}\right) \left(1 - \frac{B\varphi_{f}^{+}\left(\frac{\nu_{m}}{1-2\nu_{m}} + \frac{1}{3}\right)}{\left(\frac{\nu_{m}}{1-2\nu_{m}} - \frac{\nu_{P}}{1-2\nu_{P}}\right)}\right)$$
(6.2.32)

and

$$\frac{\overline{\lambda}_{ss}}{2\mu_{ss}} = \left(\frac{\nu_m}{1 - 2\nu_m} - \frac{\nu_p}{1 - 2\nu_p}\right) \left(1 - \frac{B\varphi_f^+ \left(\frac{\nu_m}{1 - 2\nu_m} + \frac{1}{3}\right)}{\left(\frac{\nu_m}{1 - 2\nu_m} - \frac{\nu_p}{1 - 2\nu_p}\right)}\right)^2 + \frac{\nu_p}{1 - 2\nu_p}$$
(6.2.33)

## **Porous Elasticity Models**

Next, we can calculate the drag coefficient  $\xi$  from (6.2.30). In this last calculation we use  $\mu_w = .01$  poise and  $\gamma_w = 1$  gm/cm<sup>3</sup> for the properties of water. These formulas, and the tabulated values above, yield the following numerical values

Property	Ruhr	Tennessee	Charcoal	Berea	Westerly	Weber
	Sandstone	Marble	Granite	Sandstone	Granite	Sandstone
$\overline{\lambda}_{ff}$ (kbar)	.165	.486	.326	4.44	.075	1.03
$\overline{\lambda}_{sf}$ (kbar)	5.21	4.02	4.13	13.99	3.48	9.91
$\overline{\lambda}_{ss}$ (kbar)	206.41	273.21	271.91	84.05	311.71	147.62
$\xi$ (kbar-sec/cm <sup>2</sup> )	$1.9(10)^{-3}$	3.92	3.92	$1.77(10)^{-4}$	.245	.353(10) <sup>-3</sup>

Table 2 Calculated Rock Properties-Fluid Mixture Properties

In the remainder of this work, we shall adopt the numerical values above as typical and utilize them for the purposes of illustration. It needs to be stressed that the numerical values for the drag coefficient do presume the fluid is water with the properties given above. The drag coefficients for other fluids can be derived from the above formulas if needed so long as one knows the true density and the viscosity of the fluids. The numerical values adopted from the Rice and Cleary article and their use to calculate the information in Table 2 follows the work of Bowen and Lockett [Ref. 6].

The literature on the poroelastic properties of porous materials is extensive. It is too extensive to be treated properly here. References 7 through 11 might be of interest to the reader. Proelastic properties can also be found in references 12 and 13.

## References

- 1. BIOT, MAURICE, General Theory of Three Dimensional Consolidation, J. Appl. Phys., **12**, 155-164 (1941).
- 2. TERZAGHI, K., Erdbaumechanik auf Bodenphysikalischer Grundlage, *Leipzig F. Deuticke*. (1925).
- 3. RICE, J. R., and M. P. CLEARY, Some Basic Solutions for Fluid-Saturated Porous Media With Compressible Constituents, *Review of Geophysics and Space Physics*, **14**, 227-241, (1976).
- 4. BIOT, MAURICE, and D. G. WILLIS, The Elastic Coefficients of the Theory of Consolidation, *J. Appl. Mech.*, **24**, 594-601 (1957).
- 5. BIOT, MAURICE, Theory of Propagation of Elastic Waves in a Fluid-Saturated Porous Solid. I. Low-Frequency Range, *J. Acoust. Soc. of America*, **28**, 168-178 (1956).
- 6. BOWEN, R. M., and R. R. LOCKETT, Inertial Effects in Poroelasticity, *J. of Applied Mechanics*, **50**, 334-342 (1983).
- 7. YEW, C. H. and P. N. JOGI, The Determination of Biot's Parameters for Sandstones Part 1: Static Tests, *Experimental Mechanics*, **18**, 167-172 (1978).
- 8. YEW, C. H., P. N. JOGI and K. E. GRAY, Estimation of the Mechanical Properties of Fluid-Saturated Rocks Using the Measured Wave Motions, *J. of Energy Resources Technology*, **101**, 112-116 (1979).

- 9. KIM, Y. K., and H. B. KINGSBURY, Dynamic Characterization of Poroelastic Materials, *Experimental Mechanics*, **19**, 252-258 (1979).
- 10. YAVARI, B., and A. BEDFORD, Comparison of Numerical Calculations of Two Biot Coefficients with Analytical Solutions, *J. Acoust. Soc. of America*, **90**, 985-990 (1991).
- 11. FLATT, I., The Biot-Willis Elastic Coefficients for a Sandstone, *J. of Applied Mechanics*, **26**, 296-297 (1959).
- 12. WANG, HERBERT F., Theory of Linear Poroelasticity with Applications to Geomechanics and Hydrogeology, Princeton University Press, Princeton and Oxford (2000).
- 13. DETOURNAY, EMMANUEL, and ALEXANDER H.-D. CHENG, Fundamentals of Poroelasticity, in Comprehensive Rock Engineering: Principles, Practice and Projects, Vol. II, Analysis and Design Method., ed. D. FAIRHURST, Pergman Press, 1993.

## **Singular Surfaces and Acceleration Waves**

In this chapter we shall look at an elementary application of certain of the partial differential equations introduced in Chapter 6. The objective is to create a better understanding of the physical phenomena characterized by the linear poroelasticity models we have been considering. The application in this chapter is the examination of the propagation of acceleration waves. This information is useful when one tries to understand the properties of the underlying partial differential equations. It is also useful when studying more complicated applications.

#### 7.1 Singular Surfaces

It is possible to gain insight into the equations of linear poroelasticity by examining whether or not they will support solutions with various types of discontinuities such as shock waves and, as we shall define below, acceleration waves. The formalism required to study these kinds of discontinuities can be somewhat complicated. Here, we shall keep the formulism to a minimum and make generous use of reference material. The approach in this section depends heavily on the work of Truesdell and Toupin [Ref. 1] and to a certain extent on the work of Friedlander [Ref. 2].

Our first task is to introduce some of the kinematics of singular surfaces. If  $\Sigma(t)$  is a three dimensional surface, it can be represented by the equation

$$f\left(\mathbf{x},t\right) = 0 \tag{7.1.1}$$

then it is true that

$$\mathbf{n} = \operatorname{grad} f(\mathbf{x}, t) / \left\| \operatorname{grad} f(\mathbf{x}, t) \right\|$$
(7.1.2)

is the *unit normal* to  $\Sigma(t)$ , and

$$u_{n} = -\frac{\partial f\left(\mathbf{x},t\right)}{\partial t} / \left\| \operatorname{grad} f\left(\mathbf{x},t\right) \right\|$$
(7.1.3)

is the *normal speed* of  $\Sigma(t)$ .

It is convenient to consider a family of surfaces defined by

$$f\left(\mathbf{x},t\right) = \alpha \tag{7.1.4}$$

where  $\alpha = 0$  defines the surface  $\sum(t)$ . Without loss of generality we can regard **n** to be directed from the surface  $\alpha = 0$  towards surfaces with  $\alpha > 0$ . We assume that (7.1.4) can be inverted to yield

$$t = \hat{t}\left(\mathbf{x}, \alpha\right) \tag{7.1.5}$$

Given (7.1.5), a function  $\varphi$  of  $(\mathbf{x}, t)$  can be replaced by a function  $\hat{\varphi}$  of  $(\mathbf{x}, \alpha)$  by the rule

$$\hat{\varphi}(\mathbf{x},\alpha) = \varphi(\mathbf{x},\hat{t}(\mathbf{x},\alpha))$$
(7.1.6)

It follows from (7.1.6) that

$$\operatorname{grad} \hat{\varphi}(\mathbf{x}, \alpha) = \operatorname{grad} \varphi(\mathbf{x}, t) + \frac{\partial \varphi(\mathbf{x}, t)}{\partial t} \operatorname{grad} \hat{t}(\mathbf{x}, \alpha)$$
(7.1.7)

Likewise it follows from (7.1.4) and (7.1.5) that

grad 
$$\hat{t}(\mathbf{x}, \alpha) = -\text{grad } f(\mathbf{x}, t) / \frac{\partial f(\mathbf{x}, t)}{\partial t}$$
 (7.1.8)

By formulas like (7.1.2) and (7.1.3), (7.1.8) can be written

$$\operatorname{grad} \hat{t}\left(\mathbf{x}, \alpha\right) = \frac{\mathbf{n}_{(\alpha)}}{u_{n(\alpha)}}$$
(7.1.9)

where  $\mathbf{n}_{(\alpha)}$  is the unit normal to the surface (7.1.4) and  $u_{n(\alpha)}$  is its normal speed. The special surface  $\alpha = 0$  has its normal and normal speed denoted by  $\mathbf{n}$  and  $u_n$ , respectively. Given (7.1.9), equation (7.1.7) becomes

grad 
$$\hat{\varphi}(\mathbf{x}, \alpha) = \operatorname{grad} \varphi(\mathbf{x}, t) + \frac{\mathbf{n}_{(\alpha)}}{u_{n(\alpha)}} \frac{\partial \varphi(\mathbf{x}, t)}{\partial t}$$
 (7.1.10)

The displacement derivative of  $\varphi(\mathbf{x},t)$ , following a particular surface  $\alpha = 0$ , is defined by

$$\frac{\delta\varphi(\mathbf{x},t)}{\delta t} = \left(\operatorname{grad}\hat{\varphi}(\mathbf{x},\alpha)\big|_{\alpha=0}\right) \cdot \mathbf{n}u_n \tag{7.1.11}$$

Physically, the displacement derivative represents the time rate of change of the quantity  $\varphi(\mathbf{x},t)$  seen by an observer moving with speed  $u_n$  along a curve whose tangent is **n**. This curve is called the *normal trajectory*. It follows from (7.1.10) that the displacement derivative can be written

$$\frac{\delta\varphi(\mathbf{x},t)}{\delta t} = \frac{\partial\varphi(\mathbf{x},t)}{\partial t} + \operatorname{grad}\varphi(\mathbf{x},t) \cdot \mathbf{n}u_n$$
(7.1.12)

If we continue to single out a particular moving surface,  $\alpha = 0$ , at the time *t*, this surface can be regarded as the common boundary between two regions  $\mathscr{V}^+$  and  $\mathscr{V}^-$ . Let  $\varphi(\mathbf{x}, t)$ , for each *t*, be continuous in  $\mathscr{V}^+$  and  $\mathscr{V}^-$  and approach limits  $\varphi^+$  and  $\varphi^-$  as **x** approaches a point  $\mathbf{x}_0$  on the surface while remaining in  $\mathscr{V}^+$  and  $\mathscr{V}^-$ , respectively. The *jump* of  $\varphi$  across the surface  $\alpha = 0$ is defined by

$$\left[\varphi\right] = \varphi^{-} - \varphi^{+} \tag{7.1.13}$$

where, formally,

$$\varphi^{-} = \lim_{\alpha \uparrow 0^{-}} \hat{\varphi}(\mathbf{x}, \alpha) \tag{7.1.14}$$

and

$$\varphi^{+} = \lim_{\alpha \downarrow 0^{+}} \hat{\varphi}(\mathbf{x}, \alpha)$$
(7.1.15)

The jump  $[\varphi]$  is a function of position on the surface as well as the time. As a matter of convention, we regard the unit normal **n** as directed into  $\mathscr{V}^+$ .

Because differentiation at constant  $\alpha$  commutes with the jump operation, the jump of (7.1.10) yields

$$\operatorname{grad}[\varphi] = \left[\operatorname{grad}\varphi(\mathbf{x},t)\right] + \left[\frac{\partial\varphi(\mathbf{x},t)}{\partial t}\right]\frac{\mathbf{n}}{u_n}$$
(7.1.16)

The projection of (7.1.16) tangential to the surface is a result known as the geometrical condition of compatibility [Ref. 1, Sec. 174]. If, as a special case, the jump of  $[\varphi]$  is zero, (7.1.16) yields

$$\left[\operatorname{grad}\varphi(\mathbf{x},t)\right] = -\left[\frac{\partial\varphi(\mathbf{x},t)}{\partial t}\right]\frac{\mathbf{n}}{u_n}$$
(7.1.17)
Equation (7.1.17) is known as *Maxwell's theorem* [Ref. 1, Sec. 175]. It yields the interesting kinematic result that the jump [grad  $\varphi(\mathbf{x}, t)$ ] is necessarily normal to the surface in the case where  $[\varphi]$  is zero. Returning to the case where  $[\varphi] \neq 0$ , it follows from (7.1.11) that

$$\left[\frac{\delta\varphi(\mathbf{x},t)}{\delta t}\right] = \frac{\delta[\varphi(\mathbf{x},t)]}{\delta t}$$
(7.1.18)

and, from (7.1.12) and (7.1.18),

$$\frac{\delta[\varphi(\mathbf{x},t)]}{\delta t} = \left[\frac{\partial \varphi(\mathbf{x},t)}{\partial t}\right] + \left[\operatorname{grad}\varphi(\mathbf{x},t)\right] \cdot \mathbf{n}u_n \tag{7.1.19}$$

Equation (7.1.19) is known as the *kinematic condition of compatibility* [Ref. 1, Sec. 180]. In the special case where  $[\phi]$  is zero, (7.1.19) yields

$$\left[\frac{\partial \varphi(\mathbf{x},t)}{\partial t}\right] = -\left[\operatorname{grad}\varphi(\mathbf{x},t)\right] \cdot \mathbf{n}u_n \tag{7.1.20}$$

which also follows from (7.1.17).

### **7.2 Acceleration Waves**

Next, we shall utilize the above kinematic results to study various types of singular surfaces propagating in porous materials defined by equations (4.2.14) through (4.2.22). In particular, we are interested in a binary mixture with pore pressure consisting of a compressible isotropic linear elastic solid and a compressible linear elastic fluid. As in Chapter 4, the mixture is assumed to have pore pressure. In particular, the porous material is governed by the following partial differential equations:

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{s} - \overline{\tau}_{s} + \gamma) \operatorname{GRAD} \theta - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right)$$
(7.2.1)

$$\rho_{f}^{+} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} = \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \operatorname{GRAD} \theta$$

$$-\xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(7.2.2)

and

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left( (\overline{\tau}_{s} - \alpha_{s} + \varsigma/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{s}}{\partial t} + (\overline{\tau}_{f} - \alpha_{f} - \varsigma/\theta^{+})\operatorname{Div}\frac{\partial\mathbf{w}_{f}}{\partial t} \right)$$

$$= \kappa \operatorname{Div}(\operatorname{GRAD}\theta)$$
(7.2.3)

where

$$\kappa \ge 0 \tag{7.2.4}$$

and

$$\kappa \xi / \theta^+ \ge \frac{1}{4} (\gamma + \zeta / \theta^+)^2$$
(7.2.5)

As indicated in Chapter 4, it follows from (7.2.5) that

$$\xi \ge 0 \tag{7.2.6}$$

Also as indicated in Chapter 4, we shall require that the inequalities (4.3.4) through (4.3.8) be satisfied. These formulas, repeated here, are

$$\overline{c}_{v} > 0 \tag{7.2.7}$$

$$\overline{\lambda}_{ss} + \frac{2}{3}\mu_{ss} > 0 \tag{7.2.8}$$

$$(\bar{\lambda}_{ss} + \frac{2}{3}\mu_{ss})\bar{\lambda}_{ff} > \bar{\lambda}_{sf}^2$$
(7.2.9)

$$\mu_{ss} > 0$$
 (7.2.10)

and

$$\overline{\lambda}_{ff} > 0 \tag{7.2.11}$$

In addition to the field equations (7.2.1), (7.2.2) and (7.2.3), there are jump balance equations which hold across the surface of discontinuity. We have not developed these equations in this textbook. A careful development of these equations at this point would take us too deeply into the details of Continuum Mechanics. A quick and somewhat inelegant way of deriving the jump balance equations involves the integration of (7.2.1), (7.2.2) and (7.2.3) over a volume that moves with and contains the singular surface. The results of this calculation turn out to be<sup>1</sup>

<sup>&</sup>lt;sup>1</sup> The basic material that would allow for the derivation of (7.2.12) through (7.2.14) from the foundations of Continuum Mechanics can be found reference 1, pages 525-529. Reference 4 also contains sufficient material to be helpful in understanding the jump balance equations. A careful examination of the analysis leading to equations

$$\rho_{s_{R}} u_{n} \left[ \frac{\partial \mathbf{w}_{s}}{\partial t} \right] + \left[ \overline{\lambda}_{ss} \left( \text{Div} \, \mathbf{w}_{s} \right) \mathbf{I} + \mu_{ss} \left( \text{GRAD} \, \mathbf{w}_{s} + \left( \text{GRAD} \, \mathbf{w}_{s} \right)^{T} \right) + \overline{\lambda}_{sf} \left( \text{Div} \, \mathbf{w}_{f} \right) \mathbf{I} \right] \mathbf{n}$$
(7.2.12)
$$+ (\alpha_{s} - \overline{\tau}_{s} + \gamma) \left[ \boldsymbol{\theta} \right] \mathbf{n} = \mathbf{0}$$

$$\rho_{f}^{+}u_{n}\left[\frac{\partial \mathbf{w}_{f}}{\partial t}\right] + \left[\overline{\lambda}_{sf}\left(\operatorname{Div}\mathbf{w}_{s}\right) + \overline{\lambda}_{ff}\left(\operatorname{Div}\mathbf{w}_{f}\right)\right]\mathbf{n} + (\alpha_{f} - \overline{\tau}_{f} - \gamma)\left[\boldsymbol{\theta}\right]\mathbf{n} = \mathbf{0}$$
(7.2.13)

\_

$$u_{n} \left[ \overline{c}_{v} \theta + \theta^{+} \left( \overline{\tau}_{s} - \alpha_{s} \right) \operatorname{Div} \mathbf{w}_{s} + \theta^{+} \left( \overline{\tau}_{f} - \alpha_{f} \right) \operatorname{Div} \mathbf{w}_{f} \right]$$
  
=  $\left[ -\kappa \operatorname{GRAD} \theta - \varsigma \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) \right] \cdot \mathbf{n}$  (7.2.14)

By definition, an *acceleration wave* in the porous material defined by (7.2.1) through (7.2.6) is a moving singular surface across which<sup>2</sup>

$$\begin{bmatrix} \mathbf{w}_f \end{bmatrix} = \begin{bmatrix} \mathbf{w}_s \end{bmatrix} = \mathbf{0} \tag{7.2.15}$$

$$\left[\frac{\partial \mathbf{w}_{f}}{\partial t}\right] = \left[\frac{\partial \mathbf{w}_{s}}{\partial t}\right] = \mathbf{0}$$
(7.2.16)

$$\left[\theta\right] = 0 \tag{7.2.17}$$

but 
$$\begin{bmatrix} \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \end{bmatrix}$$
,  $\begin{bmatrix} \frac{\partial^2 \mathbf{w}_s}{\partial t^2} \end{bmatrix}$  and  $\begin{bmatrix} \frac{\partial \theta}{\partial t} \end{bmatrix}$  are not necessarily zero. If we select  $\varphi$  in (7.1.17) to be  $\mathbf{w}_f$ ,  $\mathbf{w}_s$ ,  $\theta$ ,  $\frac{\partial \mathbf{w}_f}{\partial t}$  and  $\frac{\partial \mathbf{w}_s}{\partial t}$ , respectfully, then equations (7.2.15), (7.2.16) and (7.2.17) yield

<sup>(7.2.1)</sup> through (7.2.14) will reveal that the jump balance equations <u>do</u> include terms arising from the momentum supplies introduced in equation (2.2.2). When the constitutive equations for the momentum supplies depend upon density gradients, deformation gradients, temperature gradients and other gradients, one cannot automatically assume they are sufficiently smooth to yield zero contribution to the jump balance of linear momentum equation. Indeed, they do contribute a terms to the jump balance of linear momentum. These terms are in (7.2.12) and (7.2.13). The contribution from the momentum supplies has been treated incorrectly in some of the historical literature. Examples can be found in the articles written by Bowen [Ref. 5, 6].

 $<sup>^{2}</sup>$  If we had included a proof of the results (7.2.12), (7.2.13) and (7.2.14), it would have been seen that we had already assumed (7.2.15).

$$\left[\operatorname{GRAD}\mathbf{w}_{f}\right] = \mathbf{0} \tag{7.2.18}$$

$$\left[\operatorname{GRAD}\mathbf{w}_{s}\right] = \mathbf{0} \tag{7.2.19}$$

$$\left[\operatorname{GRAD}\left(\frac{\partial \mathbf{w}_f}{\partial t}\right)\right] = -\left[\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right] \otimes \frac{\mathbf{n}}{u_n}$$
(7.2.20)

$$\left[\operatorname{GRAD}\left(\frac{\partial \mathbf{w}_{s}}{\partial t}\right)\right] = -\left[\frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}}\right] \otimes \frac{\mathbf{n}}{u_{n}}$$
(7.2.21)

$$\left[\operatorname{GRAD}\theta\right] = -\left[\frac{\partial\theta}{\partial t}\right]\frac{\mathbf{n}}{u_n} \tag{7.2.22}$$

Because of (7.2.18), we can take  $\varphi$  in (7.1.20) to be grad  $\mathbf{w}_f$  and the result can be written

$$u_n^2 \left[ \text{GRAD} \left( \text{GRAD} \, \mathbf{w}_f \right) \right] = \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \otimes \mathbf{n} \otimes \mathbf{n}$$
(7.2.23)

after (7.2.21) is used. An identical calculation yields

$$u_n^2 \left[ \text{GRAD} \left( \text{GRAD} \, \mathbf{w}_s \right) \right] = \left[ \frac{\partial^2 \mathbf{w}_s}{\partial t^2} \right] \otimes \mathbf{n} \otimes \mathbf{n}$$
(7.2.24)

It follows from (7.2.23) that

$$u_n^2 \left[ \text{GRAD} \left( \text{Div} \, \mathbf{w}_f \right) \right] = \left( \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \cdot \mathbf{n} \right) \mathbf{n}$$
(7.2.25)

Likewise, it follows from (7.2.24) that

$$u_n^2 [\text{GRAD}(\text{Div}\,\mathbf{w}_s)] = \left( \left[ \frac{\partial^2 \mathbf{w}_s}{\partial t^2} \right] \cdot \mathbf{n} \right) \mathbf{n}$$
(7.2.26)

and

$$u_n^2 \left[ \text{Div} \left( \text{GRAD} \, \mathbf{w}_s \right) \right] = \left[ \frac{\partial^2 \mathbf{w}_s}{\partial t^2} \right]$$
(7.2.27)

Given the definition of an acceleration wave, equations (7.2.12) and (7.2.13) are identically satisfied. Equation (7.2.14), which represents the jump balance of energy across the acceleration wave, reduces to

$$\kappa [\text{GRAD}\,\theta] \cdot \mathbf{n} = 0 \tag{7.2.28}$$

Equation (7.2.28) provides a point of departure for two important special cases. The first case is that of a *conductor*, where we rule out the case  $\kappa = 0$ . In this case, (7.2.4) is replaced by

$$\kappa > 0 \tag{7.2.29}$$

and (7.2.28) forces

$$[\operatorname{GRAD}\theta] \cdot \mathbf{n} = 0 \tag{7.2.30}$$

which, in turn, from (7.1.20) yields

$$\begin{bmatrix} \frac{\partial \theta}{\partial t} \end{bmatrix} = 0 \tag{7.2.31}$$

Finally, this last result and (7.1.17) yield

$$[\operatorname{GRAD}\theta] = \mathbf{0} \tag{7.2.32}$$

The second case we shall examine is for a *nonconductor*, i.e., when  $\kappa = 0$ . In this case, which we shall examine below, we cannot utilize the results (7.2.31) and (7.2.32).

Consider first, the case of a conductor. On the assumption that an acceleration wave, as we have defined it, exists, the balance equations (7.2.1), (7.2.2) and (7.2.3) can be used to computer necessary conditions that must be satisfied. If we form the jump of these equations across the acceleration wave, it follows from (7.2.16), (7.2.25), (7.2.26), (7.2.27) and (7.2.32) that (7.2.1) reduces to

$$\rho_{s_{R}}\left[\frac{\partial^{2}\mathbf{w}_{s}}{\partial t^{2}}\right] = (\overline{\lambda}_{ss} + \mu_{ss})\left(\left[\frac{\partial^{2}\mathbf{w}_{s}}{\partial t^{2}}\right] \cdot \mathbf{n}\right)\frac{\mathbf{n}}{u_{n}^{2}} + \mu_{ss}u_{n}^{2}\left[\frac{\partial^{2}\mathbf{w}_{s}}{\partial t^{2}}\right]\frac{1}{u_{n}^{2}} + \overline{\lambda}_{sf}\left(\left[\frac{\partial^{2}\mathbf{w}_{f}}{\partial t^{2}}\right] \cdot \mathbf{n}\right)\frac{\mathbf{n}}{u_{n}^{2}}$$
(7.2.33)

A similar calculation yields from (7.2.2)

$$\rho_{f}^{+} \left[ \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} \right] = \overline{\lambda}_{sf} \left( \left[ \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} \right] \cdot \mathbf{n} \right) \frac{\mathbf{n}}{u_{n}^{2}} + \overline{\lambda}_{ff} \left( \left[ \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} \right] \cdot \mathbf{n} \right) \frac{\mathbf{n}}{u_{n}^{2}}$$
(7.2.34)

An entirely similar calculation yields, from (7.2.3),

$$\left[\operatorname{Div}(\operatorname{GRAD}\theta)\right] = -\frac{\theta^{+}}{\kappa} \left( (\overline{\tau}_{s} - \alpha_{s} + \varsigma/\theta^{+}) \left[ \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} \right] \cdot \frac{\mathbf{n}}{u_{n}} + (\overline{\tau}_{f} - \alpha_{f} - \varsigma/\theta^{+}) \left[ \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} \right] \cdot \frac{\mathbf{n}}{u_{n}} \right) (7.2.35)$$

Equations (7.2.33) and (7.2.34), that do not involve the jumps in derivatives of the temperature, can be rearranged to yield

$$(\overline{\lambda}_{ss} + \mu_{ss})(\mathbf{a}_{s} \cdot \mathbf{n})\mathbf{n} + \mu_{ss}\mathbf{a}_{s} + \overline{\lambda}_{sf}(\mathbf{a}_{f} \cdot \mathbf{n})\mathbf{n} = \rho_{s_{R}}u_{n}^{2}\mathbf{a}_{s}$$
(7.2.36)

and

$$\overline{\lambda}_{sf} \left( \mathbf{a}_{s} \cdot \mathbf{n} \right) \mathbf{n} + \overline{\lambda}_{ff} \left( \mathbf{a}_{f} \cdot \mathbf{n} \right) \mathbf{n} = \rho_{f}^{+} u_{n}^{2} \mathbf{a}_{f}$$
(7.2.37)

where

$$\mathbf{a}_{f} = \begin{bmatrix} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} \end{bmatrix}$$
(7.2.38)

and

$$\mathbf{a}_{s} = \left[\frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}}\right] \tag{7.2.39}$$

are the respective *amplitudes* of the acceleration wave. For a given direction **n**, equations (7.2.36) and (7.2.37) determine the possible squared speeds and the direction of the two amplitudes,  $\mathbf{a}_f$  and  $\mathbf{a}_s$ .

The analysis of (7.2.36) and (7.2.37) is straightforward. If we form the cross product of each equation with the unit normal **n**, the two equations yield

 $\left(u_n^2 - \frac{\mu_{ss}}{\rho_{s_R}}\right) \mathbf{a}_s \times \mathbf{n} = \mathbf{0}$ (7.2.40)

and

$$\mathbf{a}_f \times \mathbf{n} = \mathbf{0} \tag{7.2.41}$$

Equation (7.2.41) shows that the amplitude for the fluid is necessarily parallel to the unit normal. In other words the acceleration wave in the fluid is *longitudinal*. Equation (7.2.40) shows that when  $u_n^2 = \frac{\mu_{ss}}{\rho_{s_R}}$ , the amplitude of the acceleration wave in the solid has components perpendicular to the unit normal. This *transverse wave* has multiplicity of two since the plane perpendicular to the unit normal is two dimensional. In the following, we shall designate the squared speed for the transverse wave by  $u_{n(3)}^2$ . Equation (7.2.40) shows that

$$u_{n(3)}^2 = \frac{\mu_{ss}}{\rho_{s_R}}$$
(7.2.42)

Equation (7.2.10) shows that the squared wave speed  $u_{n(3)}^2$  is, indeed, positive.

If we next project (7.2.36) and (7.2.37) in the direction of the unit normal, the results are

$$(\overline{\lambda}_{ss} + 2\mu_{ss} - \rho_{s_R} u_n^2) (\mathbf{a}_s \cdot \mathbf{n}) + \overline{\lambda}_{sf} (\mathbf{a}_f \cdot \mathbf{n}) = 0$$
(7.2.43)

and

$$\overline{\lambda}_{sf} \left( \mathbf{a}_{s} \cdot \mathbf{n} \right) + \left( \overline{\lambda}_{ff} - \rho_{f}^{+} u_{n}^{2} \right) \left( \mathbf{a}_{f} \cdot \mathbf{n} \right) = 0$$
(7.2.44)

It is useful to rewrite (7.2.43) and (7.2.44) as the matrix equation

S

$$\left(\mathbf{Q} - u_n^2 \mathbf{M}\right) \mathbf{a} = \mathbf{0} \tag{7.2.45}$$

where

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix}$$
(7.2.46)

$$\mathbf{M} = \begin{bmatrix} \rho_f^+ & 0\\ 0 & \rho_{s_R} \end{bmatrix}$$
(7.2.47)

and

$$\mathbf{a} = \begin{bmatrix} \mathbf{a}_f \cdot \mathbf{n} \\ \mathbf{a}_s \cdot \mathbf{n} \end{bmatrix}$$
(7.2.48)

Equation (7.2.45) shows that in order that the normal components of the two amplitudes be nonzero, the normal wave speeds must be solutions of

$$\det\left(\mathbf{Q}-u_n^2\mathbf{M}\right)=0\tag{7.2.49}$$

If the determinant (7.2.49), which is a quadratic in the squared normal speed, is expanded and solved for these squared speeds, the results are

$$\frac{u_{n(1)}^{2}}{u_{n(2)}^{2}} = \frac{1}{2} \left( \frac{\overline{\lambda}_{ff}}{\rho_{f}^{+}} + \frac{\overline{\lambda}_{ss} + 2\mu_{ss}}{\rho_{s_{R}}} \pm \sqrt{\left( \frac{\overline{\lambda}_{ff}}{\rho_{f}^{+}} - \frac{\overline{\lambda}_{ss} + 2\mu_{ss}}{\rho_{s_{R}}} \right)^{2} + \frac{4\overline{\lambda}_{sf}^{2}}{\rho_{f}^{+}} \rho_{s_{R}}} \right)$$
(7.2.50)

Because of the inequality (7.2.9), the symmetric matrix  $\mathbf{Q}$  is positive definite. The diagonal matrix  $\mathbf{M}$  is also positive definite. Therefore, the two squared wave speeds given by (7.2.50) are the positive eigenvalues of the matrix  $\mathbf{Q}$  with respect to the inner product induced by  $\mathbf{M}$ . The matrix  $\mathbf{a}$ , defined by (7.2.48), is the eigenvector for the eigenvalue problem (7.2.45).

The numerical values in Table 2 of Section 6.2 can be used to calculate the wave speeds from equations (7.2.50) and (7.2.42). In order to proceed with these calculations, we need numerical values for the true densities of the solid and the fluid. As with the tables in Section 6.2, we shall take the fluid to be water and use  $\gamma_f = \gamma_w = 1 \text{ gm/cm}^3$ . The true densities for the solids are taken from Farmer [Ref. 7]. These values and the resulting acceleration wave speeds are shown in the table below.

Property	Ruhr	Tennessee	Charcoal	Berea	Westerly	Weber
	Sandstone	Marble	Granite	Sandstone	Granite	Sandstone
$\gamma_{s_R}$ (g/cm <sup>3</sup> )	2.6	2.7	2.7	2.6	2.7	2.6
$u_{n(1)}$ (m/sec)	4340	5346	4955	3210	4805	4059
$u_{n(2)}$ (m/sec)	728	1522	1220	1313	740	1124
$u_{n(3)}$ (m/sec)	2285	3012	2658	1688	2369	2234

Table 1 Calculated Acceleration Wave Speeds

As reflected in this table, in the following we shall always assume that the eigenvalues  $u_{n(1)}^2$  and  $u_{n(2)}^2$  are distinct.

Since the three wave speeds  $u_{n(1)}^2$ ,  $u_{n(2)}^2$  and  $u_{n(3)}^2$  are constant, a wave of a given initial shape retains that shape as the wave propagates, i.e., the waves form a family of parallel surfaces. For example, an initially plane wave remains plane. As an illustration of the effect of diffusion on the growth and decay of acceleration, we shall next investigate the growth and decay of *plane* acceleration waves. As a consequence of this assumption the unit normal, **n**, is a constant. This calculation requires that we utilize (7.1.19) with the choice  $\varphi = \frac{\partial^2 \mathbf{w}_f}{\partial t^2}$ . Therefore,

$$\frac{\delta}{\delta t} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] = \left[ \frac{\partial^3 \mathbf{w}_f}{\partial t^3} \right] + \left[ \text{GRAD} \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \cdot \mathbf{n} u_n$$
(7.2.51)

A entirely similar calculation with  $\varphi = \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t}$  yields

$$\frac{\delta}{\delta t} \left[ \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right] = \left[ \text{GRAD} \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] + \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right) \right] \cdot \mathbf{n} u_n$$
(7.2.52)

The term  $\left[\text{GRAD}\frac{\partial \mathbf{w}_f}{\partial t}\right]$  in (7.2.52) can be eliminated in favor of  $\left[\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right]$  if we use (7.2.20). The result of this elimination can be used to eliminate the term  $\left[\text{GRAD}\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right]$  from (7.2.51). The first step in this elimination yields

$$2\frac{\delta}{\delta t}\left[\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right] + \left[\left[\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right] \otimes \frac{\delta}{\delta t}\left(\frac{\mathbf{n}}{u_n}\right)\right] \mathbf{n} u_n = \left[\frac{\partial^3 \mathbf{w}_f}{\partial t^3}\right] - u_n^2 \left[\left[\mathrm{GRAD}\left(\mathrm{GRAD}\left(\frac{\partial \mathbf{w}_f}{\partial t}\right)\right]\right] (\mathbf{n}, \mathbf{n}) (7.2.53)$$

An expansion of the second term yields,

$$2\frac{\delta}{\delta t} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] + \left( \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \frac{\delta \mathbf{n}}{\delta t} \cdot \mathbf{n} - \frac{1}{u_n} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \frac{\delta u_n}{\delta t} \right)$$

$$= \left[ \frac{\partial^3 \mathbf{w}_f}{\partial t^3} \right] - u_n^2 \left( \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right) \right] \right) (\mathbf{n}, \mathbf{n})$$
(7.2.54)

Because  $\mathbf{n} \cdot \mathbf{n} = 1$ ,  $\frac{\delta \mathbf{n}}{\delta t} \cdot \mathbf{n} = 0$ . Therefore, (7.2.54) reduces to

$$2\frac{\delta}{\delta t} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] - \frac{1}{u_n} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] \frac{\delta u_n}{\delta t} = \left[ \frac{\partial^3 \mathbf{w}_f}{\partial t^3} \right] - u_n^2 \left[ \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right) \right] \right] (\mathbf{n}, \mathbf{n})$$
(7.2.55)

In our case, since **n** and  $u_n$  are constants, (7.2.54) and (7.2.55) both reduce to

$$2\frac{\delta}{\delta t} \left[ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \right] = \left[ \frac{\partial^3 \mathbf{w}_f}{\partial t^3} \right] - u_n^2 \left[ \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right) \right] \right] (\mathbf{n}, \mathbf{n})$$
(7.2.56)

An identical calculation yields

$$2\frac{\delta}{\delta t} \left[\frac{\partial^2 \mathbf{w}_s}{\partial t^2}\right] = \left[\frac{\partial^3 \mathbf{w}_s}{\partial t^3}\right] - u_n^2 \left(\left[\text{GRAD}\left(\text{GRAD}\frac{\partial \mathbf{w}_s}{\partial t}\right)\right]\right) (\mathbf{n}, \mathbf{n})$$
(7.2.57)

In equations (7.2.56) and (7.2.57) the last terms are, in Cartesian components,

$$\left( \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_f}{\partial t} \right) \right] \right) (\mathbf{n}, \mathbf{n}) = \left[ \frac{\partial^3 \mathbf{w}_f}{\partial X_k \partial X_j \partial t} \right] n_k n_j$$
(7.2.58)

and

$$\left( \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_s}{\partial t} \right) \right] \right) (\mathbf{n}, \mathbf{n}) = \left[ \frac{\partial^3 \mathbf{w}_s}{\partial X_k \partial X_j \partial t} \right] n_k n_j$$
(7.2.59)

respectively.

Equations (7.2.56) and (7.2.57) provide ordinary differential equations for the amplitudes (7.2.38) and (7.2.39) after we have evaluated their right hand sides. The time derivatives of (7.2.2) and (7.2.1) yield the jumps

$$\rho_{f}^{+} \left[ \frac{\partial^{3} \mathbf{w}_{f}}{\partial t^{3}} \right] = \overline{\lambda}_{sf} \left[ \text{GRAD}(\text{Div} \frac{\partial \mathbf{w}_{s}}{\partial t}) \right] + \overline{\lambda}_{ff} \left[ \text{GRAD}(\text{Div} \frac{\partial \mathbf{w}_{f}}{\partial t}) \right] + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \left[ \text{GRAD} \frac{\partial \theta}{\partial t} \right] - \xi \left[ \left[ \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} \right] - \left[ \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} \right] \right]$$
(7.2.60)

and

$$\rho_{s_{R}}\left[\frac{\partial^{3}\mathbf{w}_{s}}{\partial t^{3}}\right] = (\overline{\lambda}_{ss} + \mu_{ss})\left[\operatorname{GRAD}(\operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t})\right] + \mu_{ss}\left[\operatorname{Div}(\operatorname{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t})\right] + \overline{\lambda}_{sf}\left[\operatorname{GRAD}(\operatorname{Div}\frac{\partial \mathbf{w}_{f}}{\partial t})\right] + (\alpha_{s} - \overline{\tau}_{s} + \gamma)\left[\operatorname{GRAD}\frac{\partial \theta}{\partial t}\right] - \xi\left[\left[\frac{\partial^{2}\mathbf{w}_{s}}{\partial t^{2}}\right] - \left[\frac{\partial^{2}\mathbf{w}_{f}}{\partial t^{2}}\right]\right]$$
(7.2.61)

The next formal step is to substitute (7.2.60) and (7.2.61) into (7.2.56) and (7.2.57), respectively. The results turn out to be

$$2\rho_{f}^{+}\frac{\delta\mathbf{a}_{f}}{\delta t} = \overline{\lambda}_{sf}\left[\mathrm{GRAD}(\mathrm{Div}\frac{\partial\mathbf{w}_{s}}{\partial t})\right] + \overline{\lambda}_{ff}\left[\mathrm{GRAD}(\mathrm{Div}\frac{\partial\mathbf{w}_{f}}{\partial t})\right]$$
$$-\rho_{f}^{+}u_{n}^{2}\left(\left[\mathrm{GRAD}\left(\mathrm{GRAD}\frac{\partial\mathbf{w}_{f}}{\partial t}\right)\right]\right)\left(\mathbf{n},\mathbf{n}\right)$$
$$-\xi\left(\mathbf{a}_{f}-\mathbf{a}_{s}\right) + (\alpha_{f}-\overline{\tau}_{f}-\gamma)\left[\mathrm{GRAD}\frac{\partial\theta}{\partial t}\right]$$
(7.2.62)

$$2\rho_{s_{R}}\frac{\delta \mathbf{a}_{s}}{\delta t} = (\overline{\lambda}_{ss} + \mu_{ss}) \left[ \text{GRAD}(\text{Div}\frac{\partial \mathbf{w}_{s}}{\partial t}) \right] + \mu_{ss} \left[ \text{Div}(\text{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t}) \right] \\ + \overline{\lambda}_{sf} \left[ \text{GRAD}(\text{Div}\frac{\partial \mathbf{w}_{f}}{\partial t}) \right] - \rho_{s_{R}} u_{n}^{2} \left( \left[ \text{GRAD} \left( \text{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t} \right) \right] \right) \right) (\mathbf{n}, \mathbf{n}) \\ - \xi \left( \mathbf{a}_{s} - \mathbf{a}_{f} \right) \\ + (\alpha_{s} - \overline{\tau}_{s} + \gamma) \left[ \text{GRAD}\frac{\partial \theta}{\partial t} \right]$$
(7.2.63)

These two complicated looking equations are simplified somewhat if we use the kinematic formulas

$$\left[\operatorname{GRAD}(\operatorname{Div}\frac{\partial \mathbf{w}_{f}}{\partial t})\right] = \left(\left(\left[\operatorname{GRAD}\left(\operatorname{GRAD}\frac{\partial \mathbf{w}_{f}}{\partial t}\right)\right]\right)(\mathbf{n},\mathbf{n})\right) \cdot \mathbf{n}$$
(7.2.64)

$$\left[\operatorname{GRAD}(\operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t})\right] = \left(\left(\left[\operatorname{GRAD}\left(\operatorname{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t}\right)\right]\right)(\mathbf{n},\mathbf{n})\right) \cdot \mathbf{n}$$
(7.2.65)

and

$$\left[\operatorname{Div}(\operatorname{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t})\right] = \left(\left[\operatorname{GRAD}\left(\operatorname{GRAD}\frac{\partial \mathbf{w}_{s}}{\partial t}\right)\right]\right)(\mathbf{n},\mathbf{n})$$
(7.2.66)

These formulas can be derived from (7.1.16). Rather than include their derivations here, it is convenient to point out that they are special cases of Equation (176.8) of Ref. 1. The reduction of Equation (178.8) of Ref. 1 to the results (7.2.64) through (7.2.66), does make use of (7.2.15), (7.2.16), the plane wave assumption, the fact that the speeds are constants, and the fact that all of the amplitudes are assumed not to vary tangent to the wave front. It simplifies the notation somewhat if we introduce the symbols

$$\mathbf{d}_{f} = \left( \left[ \text{GRAD} \left( \text{GRAD} \frac{\partial \mathbf{w}_{f}}{\partial t} \right) \right] \right) (\mathbf{n}, \mathbf{n})$$
(7.2.67)

and

$$\mathbf{d}_{s} = \left( \left[ \mathbf{GRAD} \left( \mathbf{GRAD} \frac{\partial \mathbf{w}_{f}}{\partial t} \right) \right] \right) (\mathbf{n}, \mathbf{n})$$
(7.2.68)

These symbols along with (7.2.64) through (7.2.66) allow (7.2.62) and (7.2.63) to be written

$$2\rho_{f}^{+}\frac{\delta \mathbf{a}_{f}}{\delta t} = \overline{\lambda}_{ff} \left(\mathbf{d}_{f} \cdot \mathbf{n}\right) \mathbf{n} + \overline{\lambda}_{sf} \left(\mathbf{d}_{s} \cdot \mathbf{n}\right) \mathbf{n} - \rho_{f}^{+} u_{n}^{2} \mathbf{d}_{f} -\xi \left(\mathbf{a}_{f} - \mathbf{a}_{s}\right) + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \left[\text{GRAD}\frac{\partial \theta}{\partial t}\right]$$
(7.2.69)

$$2\rho_{s_{R}}\frac{\delta\mathbf{a}_{s}}{\delta t} = \overline{\lambda}_{sf}\left(\mathbf{d}_{f}\cdot\mathbf{n}\right)\mathbf{n} + (\overline{\lambda}_{ss} + \mu_{ss})\left(\mathbf{d}_{s}\cdot\mathbf{n}\right)\mathbf{n} + \mu_{ss}\mathbf{d}_{s} - \rho_{s_{R}}u_{n}^{2}\mathbf{d}_{s} -\xi\left(\mathbf{a}_{s}-\mathbf{a}_{f}\right) + (\alpha_{s}-\overline{\tau}_{s}+\gamma)\left[\mathrm{GRAD}\frac{\partial\theta}{\partial t}\right]$$
(7.2.70)

The term  $\left[\text{GRAD}\frac{\partial \theta}{\partial t}\right]$ , which appears in both (7.2.69) and (7.2.70), can be related to the two amplitudes if we utilize (7.2.35) and certain of our kinematic formulas. Equation (7.2.35), rewritten in terms of the amplitudes (7.2.38) and (7.2.39) is

$$\left[\operatorname{Div}(\operatorname{GRAD}\theta)\right] = -\frac{\theta^{+}}{\kappa} \left( (\overline{\tau}_{s} - \alpha_{s} + \varsigma/\theta^{+}) \mathbf{a}_{s} \cdot \frac{\mathbf{n}}{u_{n}} + (\overline{\tau}_{f} - \alpha_{f} - \varsigma/\theta^{+}) \mathbf{a}_{f} \cdot \frac{\mathbf{n}}{u_{n}} \right)$$
(7.2.71)

The kinematic formula which links  $\left[\text{GRAD}\frac{\partial\theta}{\partial t}\right]$  to  $\left[\text{Div}(\text{GRAD}\theta)\right]$  is

$$\left[\operatorname{GRAD}\frac{\partial\theta}{\partial t}\right] = -u_n \left[\operatorname{Div}(\operatorname{GRAD}\theta)\right]\mathbf{n}$$
(7.2.72)

The derivation of this equation is similar to the derivation of some of the other kinematic formulas used thus far. One begins with (7.1.20) and makes the choice  $\varphi = \text{GRAD}\theta$ . The result can be written

$$\left[\operatorname{GRAD}\frac{\partial\theta}{\partial t}\right] = -\left[\operatorname{GRAD}\left(\operatorname{GRAD}\theta\right)\right]\mathbf{n}u_n \tag{7.2.73}$$

Given (7.2.32) and the choice  $\varphi = \text{GRAD}\theta$ , it follows from (7.1.17) and (7.1.20) that

$$[\operatorname{GRAD}(\operatorname{GRAD}\theta)] = (\mathbf{n} \cdot [\operatorname{GRAD}(\operatorname{GRAD}\theta)]\mathbf{n})\mathbf{n} \otimes \mathbf{n}$$
(7.2.74)

This result yields

$$[\operatorname{GRAD}(\operatorname{GRAD}\theta)]\mathbf{n} = (\mathbf{n} \cdot [\operatorname{GRAD}(\operatorname{GRAD}\theta)]\mathbf{n})\mathbf{n}$$
(7.2.75)

Also, since  $\text{Div}(\text{GRAD}\theta) = \text{tr}(\text{GRAD}(\text{GRAD}\theta))$ , it follows from (7.2.74) that

$$\left[\operatorname{Div}(\operatorname{GRAD}\theta)\right] = \operatorname{tr}\left(\left[\operatorname{GRAD}(\operatorname{GRAD}\theta)\right]\right) = \mathbf{n} \cdot \left[\operatorname{GRAD}(\operatorname{GRAD}\theta)\right]\mathbf{n}$$
(7.2.76)

Equations (7.2.75) and (7.2.76) allow (7.2.73) to be written in the form (7.2.72). With the result (7.2.72), it follow from (7.2.71) that

$$\left[\operatorname{GRAD}\frac{\partial\theta}{\partial t}\right] = \frac{\theta^{+}}{\kappa} \left( \left(\overline{\tau}_{s} - \alpha_{s} + \zeta/\theta^{+}\right) \mathbf{a}_{s} \cdot \mathbf{n} + \left(\overline{\tau}_{f} - \alpha_{f} - \zeta/\theta^{+}\right) \mathbf{a}_{f} \cdot \mathbf{n} \right) \mathbf{n}$$
(7.2.77)

When (7.2.77) is substituted into (7.2.62) and (7.2.63) the results are

$$2\rho_{f}^{+}\frac{\partial \mathbf{a}_{f}}{\partial t} = \overline{\lambda}_{ff}\left(\mathbf{d}_{f}\cdot\mathbf{n}\right)\mathbf{n} + \overline{\lambda}_{sf}\left(\mathbf{d}_{s}\cdot\mathbf{n}\right)\mathbf{n} - \rho_{f}^{+}u_{n}^{2}\mathbf{d}_{f} - \xi\left(\mathbf{a}_{f}-\mathbf{a}_{s}\right)$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{f}-\alpha_{f}+\gamma)(\overline{\tau}_{f}-\alpha_{f}-\zeta/\theta^{+})\left(\mathbf{a}_{f}\cdot\mathbf{n}\right)\mathbf{n}$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{f}-\alpha_{f}+\gamma)(\overline{\tau}_{s}-\alpha_{s}+\zeta/\theta^{+})\left(\mathbf{a}_{s}\cdot\mathbf{n}\right)\mathbf{n}$$
(7.2.78)

and

$$2\rho_{s_{R}}\frac{\delta\mathbf{a}_{s}}{\delta t} = \overline{\lambda}_{sf}\left(\mathbf{d}_{f}\cdot\mathbf{n}\right)\mathbf{n} + (\overline{\lambda}_{ss} + \mu_{ss})\left(\mathbf{d}_{s}\cdot\mathbf{n}\right)\mathbf{n} + \mu_{ss}\mathbf{d}_{s} - \rho_{s_{R}}u_{n}^{2}\mathbf{d}_{s} - \xi\left(\mathbf{a}_{s}-\mathbf{a}_{f}\right)$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{s}-\alpha_{s}-\gamma)(\overline{\tau}_{f}-\alpha_{f}-\zeta/\theta^{+})\left(\mathbf{a}_{f}\cdot\mathbf{n}\right)\mathbf{n}$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{s}-\alpha_{s}-\gamma)(\overline{\tau}_{s}-\alpha_{s}+\zeta/\theta^{+})\left(\mathbf{a}_{s}\cdot\mathbf{n}\right)\mathbf{n}$$
(7.2.79)

Equations (7.2.78) and (7.2.79) are used to determine the magnitudes of the amplitudes  $\mathbf{a}_{f}$  and  $\mathbf{a}_{s}$ . The directions of these amplitudes are determined by the propagation condition (7.2.36) and (7.2.37). Utilizing these directions, the approach to the solutions involves projecting (7.2.78) and (7.2.79) in such a fashion that the jumps  $\mathbf{d}_{f}$  and  $\mathbf{d}_{s}$  are eliminated. This approach is best illustrated by first studying the transverse acceleration wave. In this case, the normal speed squared is given by (7.2.42) and the amplitude of the acceleration wave in the fluid obeys (7.2.41). Given these results, the projection of (7.2.78) into the transverse plane is obtained by forming its cross product with the unit normal,  $\mathbf{n}$ . The result of this calculation is

$$\mathbf{n} \times \mathbf{d}_{f} = \frac{\xi}{\rho_{f}^{+} u_{n(3)}^{2}} \mathbf{n} \times \mathbf{a}_{s}$$
(7.2.80)

Likewise, the cross product of (7.2.79) with the unit normal yields

$$\frac{\delta \mathbf{n} \times \mathbf{a}_s}{\delta t} = -\frac{\xi}{2\rho_{s_R}} \mathbf{n} \times \mathbf{a}_s \tag{7.2.81}$$

Equation (7.2.81) yields the amplitude of the transverse wave as a function of time to be

$$\mathbf{n} \times \mathbf{a}_{s}(t) = \mathbf{n} \times \mathbf{a}_{s}(0) e^{-\frac{\xi}{2\rho_{s_{R}}t}}$$
(7.2.82)

Because of the thermodynamic result (7.2.6), the amplitude of the transverse wave is either constant or it decays exponentially according to the formula (7.2.82). Given (7.2.82), the tangential projection  $\mathbf{n} \times \mathbf{d}_{f}$  is, from (7.2.80) given by

$$\mathbf{n} \times \mathbf{d}_{f} = \frac{\xi}{\rho_{f}^{+} u_{n(3)}^{2}} \mathbf{n} \times \mathbf{a}_{s} \left(0\right) e^{-\frac{\xi}{2\rho_{s_{R}}}t}$$
(7.2.83)

For either of the longitudinal waves, we can project (7.2.78) and (7.2.79) in the normal direction and obtain

$$2\rho_{f}^{+}\frac{\delta\mathbf{a}_{f}\cdot\mathbf{n}}{\delta t} = \overline{\lambda}_{ff}\mathbf{d}_{f}\cdot\mathbf{n} + \overline{\lambda}_{sf}\mathbf{d}_{s}\cdot\mathbf{n} - \rho_{f}^{+}u_{n}^{2}\mathbf{d}_{f}\cdot\mathbf{n} - \xi\left(\mathbf{a}_{f}\cdot\mathbf{n} - \mathbf{a}_{s}\cdot\mathbf{n}\right)$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{f} - \alpha_{f} + \gamma)(\overline{\tau}_{f} - \alpha_{f} - \zeta/\theta^{+})\mathbf{a}_{f}\cdot\mathbf{n}$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{f} - \alpha_{f} + \gamma)(\overline{\tau}_{s} - \alpha_{s} + \zeta/\theta^{+})\mathbf{a}_{s}\cdot\mathbf{n}$$
(7.2.84)

and

$$2\rho_{s_{R}}\frac{\delta\mathbf{a}_{s}\cdot\mathbf{n}}{\delta t} = \overline{\lambda}_{sf}\mathbf{d}_{f}\cdot\mathbf{n} + (\overline{\lambda}_{ss} + 2\mu_{ss})\mathbf{d}_{s}\cdot\mathbf{n} - \rho_{s_{R}}u_{n}^{2}\mathbf{d}_{s}\cdot\mathbf{n} - \xi\left(\mathbf{a}_{s}\cdot\mathbf{n} - \mathbf{a}_{f}\cdot\mathbf{n}\right)$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{s} - \alpha_{s} - \gamma)(\overline{\tau}_{f} - \alpha_{f} - \zeta/\theta^{+})\left(\mathbf{a}_{f}\cdot\mathbf{n}\right)$$
$$-\frac{\theta^{+}}{\kappa}(\overline{\tau}_{s} - \alpha_{s} - \gamma)(\overline{\tau}_{s} - \alpha_{s} + \zeta/\theta^{+})\left(\mathbf{a}_{s}\cdot\mathbf{n}\right)$$
(7.2.85)

These two equations are best analyzed by writing them as the single matrix equation

$$2\mathbf{M}\frac{\delta \mathbf{a}}{\delta t} = \left(\mathbf{Q} - u_n^2 \mathbf{M}\right)\mathbf{d} - \xi \mathbf{E}\mathbf{a} - \frac{\theta^+}{\kappa} \left(\mathbf{A} - \mathbf{B}\right) \left(\mathbf{A}^T - \mathbf{D}^T\right)\mathbf{a}$$
(7.2.86)

where  $\mathbf{a}, \mathbf{M}$  and  $\mathbf{Q}$  are given be (7.2.48), (7.2.47) and (7.2.45), respectively. In addition, the matrices  $\mathbf{E}$  and  $\mathbf{H}$  are given by

$$\mathbf{E} = \begin{bmatrix} 1 & -1 \\ -1 & 1 \end{bmatrix}$$
(7.2.87)

and

$$\mathbf{A} = \begin{bmatrix} \overline{\tau}_f - \alpha_f \\ \overline{\tau}_s - \alpha_s \end{bmatrix}$$
(7.2.88)

$$\mathbf{B} = \gamma \begin{bmatrix} -1\\1 \end{bmatrix} \tag{7.2.89}$$

$$\mathbf{D} = \frac{\varsigma}{\theta^+} \begin{bmatrix} 1\\ -1 \end{bmatrix}$$
(7.2.90)

From (7.2.45), the column matrix **a** is the eigenvector of **Q** associated with the eigenvalue  $u^2$  with respect to the inner product induced by **M**. Because we have assumed these eigenvalues are distinct, we can write each eigenvector **a** in the form

$$\mathbf{a} = a\mathbf{r} \tag{7.2.91}$$

where a is a scalar and  $\mathbf{r}$  is a normalized column matrix with the properties

$$\left(\mathbf{Q} - u^2 \mathbf{M}\right)\mathbf{r} = \mathbf{0} \tag{7.2.92}$$

and

$$\mathbf{r}^T \mathbf{M} \mathbf{r} = 1 \tag{7.2.93}$$

The matrix equation (7.2.86) can be projected in the direction  $\mathbf{r}$  to yield

$$2\frac{\delta a}{\delta t} = \mathbf{r}^{T} \left( \mathbf{Q} - u_{n}^{2} \mathbf{M} \right) \mathbf{d} - \left( \xi \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \mathbf{r}^{T} \left( \mathbf{A} - \mathbf{B} \right) \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \mathbf{r} \right) a$$
(7.2.94)

after (7.2.93) is utilized. The symmetry of the matrices **Q** and **M** allow the term  $\mathbf{r}^T (\mathbf{Q} - u_n^2 \mathbf{M}) \mathbf{d}$ to be written  $\mathbf{d}^T (\mathbf{Q} - u_n^2 \mathbf{M}) \mathbf{r}$  which by (7.2.92) is zero. Therefore, the magnitude of the acceleration wave with the speed  $u_n$  is determined by solving the ordinary differential equation

$$2\frac{\delta a}{\delta t} = -\left(\xi \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \mathbf{r}^{T} \left(\mathbf{A} - \mathbf{B}\right) \left(\mathbf{A}^{T} - \mathbf{D}^{T}\right) \mathbf{r}\right) a$$
(7.2.95)

The behavior of the solution depends upon the sign of the sign of the term

 $\xi \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \mathbf{r}^{T} (\mathbf{A} - \mathbf{B}) (\mathbf{A}^{T} - \mathbf{D}^{T}) \mathbf{r}$ . This coefficient, as we shall see, cannot be negative. As a result, the amplitude of the longitudinal acceleration wave is either a constant of decays exponentially.

The proof that the term  $\xi \mathbf{r}^T \mathbf{E} \mathbf{r} + \frac{\theta^+}{\kappa} \mathbf{r}^T (\mathbf{A} - \mathbf{B}) (\mathbf{A}^T - \mathbf{D}^T) \mathbf{r}$  cannot be negative is straight forward but somewhat complicated. The first step is to note the identity

$$(\mathbf{A} - \mathbf{B})(\mathbf{A}^{T} - \mathbf{D}^{T}) = \left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D}) - \frac{1}{2}(\mathbf{B} - \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D}) + \frac{1}{2}(\mathbf{B} - \mathbf{D})\right)^{T}$$
$$= \left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T} - \frac{1}{4}(\mathbf{B} - \mathbf{D})(\mathbf{B} - \mathbf{D})^{T} \qquad (7.2.96)$$
$$+ \frac{1}{2}\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)(\mathbf{B} - \mathbf{D})^{T} - \frac{1}{2}(\mathbf{B} - \mathbf{D})\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T}$$

It follows from this expression that

$$\mathbf{r}^{T} (\mathbf{A} - \mathbf{B}) (\mathbf{A}^{T} - \mathbf{D}^{T}) \mathbf{r} = \mathbf{r}^{T} \begin{pmatrix} \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right) \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right)^{T} - \frac{1}{4} (\mathbf{B} - \mathbf{D}) (\mathbf{B} - \mathbf{D})^{T} \\ + \frac{1}{2} \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right) (\mathbf{B} - \mathbf{D})^{T} - \frac{1}{2} (\mathbf{B} - \mathbf{D}) \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right)^{T} \end{pmatrix}^{T} \\ = \mathbf{r}^{T} \left( \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right) \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right)^{T}\right) \mathbf{r} - \frac{1}{4} \mathbf{r}^{T} (\mathbf{B} - \mathbf{D}) (\mathbf{B} - \mathbf{D})^{T} \mathbf{r} (7.2.97) \\ + \frac{1}{2} \mathbf{r}^{T} \left( \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right) (\mathbf{B} - \mathbf{D})^{T} - (\mathbf{B} - \mathbf{D}) \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right)^{T} \right) \mathbf{r} \\ = \left( \left(\mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D})\right)^{T} \mathbf{r} \right)^{2} - \frac{1}{4} \mathbf{r}^{T} (\mathbf{B} - \mathbf{D}) (\mathbf{B} - \mathbf{D})^{T} \mathbf{r} \\ \end{pmatrix}$$

The definitions (7.2.89) and (7.2.90) show that

$$(\mathbf{B} - \mathbf{D}) (\mathbf{B} - \mathbf{D})^{T} = \left(\gamma + \frac{\varsigma}{\theta^{+}}\right)^{2} \begin{bmatrix} -1\\1 \end{bmatrix} \begin{bmatrix} -1 & 1 \end{bmatrix} = \left(\gamma + \frac{\varsigma}{\theta^{+}}\right)^{2} \mathbf{E}$$
(7.2.98)

where the definition (7.2.87) has been used. Equations (7.2.97) and (7.2.98) show that

$$\xi \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \mathbf{r}^{T} \left( \mathbf{A} - \mathbf{B} \right) \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \mathbf{r} = \left( \xi - \frac{1}{4} \frac{\theta^{+}}{\kappa} \left( \gamma + \frac{\zeta}{\theta^{+}} \right)^{2} \right) \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \left( \left( \mathbf{A} - \frac{1}{2} \left( \mathbf{B} + \mathbf{D} \right) \right)^{T} \mathbf{r} \right)^{2} (7.2.99)$$

Because we are working with the assumption that  $\kappa > 0$ , equation (7.2.5) tells us that

 $\xi - \frac{1}{4} \frac{\theta^+}{\kappa} \left( \gamma + \frac{\zeta}{\theta^+} \right)^2 \ge 0.$  Also, the definition (7.2.87) shows that  $\mathbf{r}^T \mathbf{E} \mathbf{r} = \left( r_g - r_s \right)^2 \ge 0.$  Because the last term in (7.2.99) is non-negative, we have the result

$$\xi \mathbf{r}^{T} \mathbf{E} \mathbf{r} + \frac{\theta^{+}}{\kappa} \mathbf{r}^{T} \left( \mathbf{A} - \mathbf{B} \right) \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \mathbf{r} \ge 0$$
(7.2.100)

Equation (7.2.100) gives the result stated above, i.e., that the amplitude of each longitudinal acceleration wave cannot increase.

There are further simplifications the formula (7.2.99) that are useful. If we look at the two cases  $\mathbf{r} = \mathbf{r}_1$  and  $\mathbf{r} = \mathbf{r}_2$ , it is possible to show that

$$\mathbf{r}_{1}^{T}\mathbf{E}\mathbf{r}_{1} = \operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right)\frac{u_{n(1)}^{2} - u_{0}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} = \left(\frac{1}{\rho_{f}^{+}} + \frac{1}{\rho_{s_{R}}}\right)\frac{u_{n(1)}^{2} - u_{0}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.101)

and

$$\mathbf{r}_{2}^{T}\mathbf{E}\mathbf{r}_{2} = \operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right)\frac{u_{0}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} = \left(\frac{1}{\rho_{f}^{+}} + \frac{1}{\rho_{s_{R}}}\right)\frac{u_{0}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.102)

where

$$u_0^2 = \frac{\operatorname{tr}\left(\operatorname{adj}(\mathbf{Q})\mathbf{E}\right)}{\operatorname{tr}\left(\operatorname{adj}(\mathbf{M})\mathbf{E}\right)} = \frac{\overline{\lambda}_{ff} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ss} + 2\mu_{ss}}{\rho_f^+ + \rho_{s_R}}$$
(7.2.103)

The derivation of (7.2.101) and (7.2.102) involves a rather straight forward use of the spectral representation of **Q**. Given the eigenvalue problem (7.2.45), it is possible to show that the spectral representation of **Q** is

$$\mathbf{Q} = u_{n(1)}^{2} \mathbf{M} \mathbf{r}_{1}^{T} \mathbf{r}_{1} \mathbf{M} + u_{n(2)}^{2} \mathbf{M} \mathbf{r}_{2}^{T} \mathbf{r}_{2} \mathbf{M}$$
(7.2.104)

where

$$\mathbf{M} = \mathbf{M}\mathbf{r}_1^T \mathbf{r}_1 \mathbf{M} + \mathbf{M}\mathbf{r}_2^T \mathbf{r}_2 \mathbf{M}$$
(7.2.105)

These two equations yield

$$\mathbf{M}\mathbf{r}_{1}^{T}\mathbf{r}_{1}\mathbf{M} = \frac{\mathbf{Q} - u_{n(2)}^{2}\mathbf{M}}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.106)

and

$$\mathbf{M}\mathbf{r}_{2}^{T}\mathbf{r}_{2}\mathbf{M} = \frac{\mathbf{Q} - u_{n(1)}^{2}\mathbf{M}}{u_{n(2)}^{2} - u_{n(1)}^{2}}$$
(7.2.107)

Given (7.2.106), it follows that

$$\mathbf{r}_{1}^{T}\mathbf{E}\mathbf{r}_{1} = \operatorname{tr}\left(\mathbf{r}_{1}\mathbf{r}_{1}^{T}\mathbf{E}\right) = \operatorname{tr}\left(\mathbf{M}\mathbf{r}_{1}\mathbf{r}_{1}^{T}\mathbf{M}\mathbf{M}^{-1}\mathbf{E}\mathbf{M}^{-1}\right)$$
$$= \operatorname{tr}\left(\left(\frac{\mathbf{Q}-u_{n(2)}^{2}\mathbf{M}}{u_{n(1)}^{2}-u_{n(2)}^{2}}\right)\mathbf{M}^{-1}\mathbf{E}\mathbf{M}^{-1}\right) = \frac{\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{Q}\mathbf{M}^{-1}\mathbf{E}\right)-u_{n(2)}^{2}\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right)}{u_{n(1)}^{2}-u_{n(2)}^{2}}$$
(7.2.108)

Because the matrix  $\mathbf{Q}$  obeys its own characteristic equation, it is true that

$$\left(\mathbf{Q}\mathbf{M}^{-1}\right)^{2} - \mathrm{tr}\left(\mathbf{Q}\mathbf{M}^{-1}\right)\mathbf{Q}\mathbf{M}^{-1} + \mathrm{det}\left(\mathbf{Q}\mathbf{M}^{-1}\right)\mathbf{I} = \mathbf{0}$$
(7.2.109)

This equation can be rearranged into the form

$$\mathbf{M}^{-1}\mathbf{Q}\mathbf{M}^{-1} - \mathrm{tr}\left(\mathbf{Q}\mathbf{M}^{-1}\right)\mathbf{M}^{-1} + \mathrm{det}\left(\mathbf{Q}\mathbf{M}^{-1}\right)\mathbf{Q}^{-1} = \mathbf{0}$$
(7.2.110)

Therefore,

$$tr(\mathbf{M}^{-1}\mathbf{Q}\mathbf{M}^{-1}\mathbf{E}) = tr(\mathbf{Q}\mathbf{M}^{-1})tr(\mathbf{M}^{-1}\mathbf{E}) - det(\mathbf{Q}\mathbf{M}^{-1})tr(\mathbf{Q}^{-1}\mathbf{E})$$
  
= tr(\mathbf{Q}\mathbf{M}^{-1})tr(\mathbf{M}^{-1}\mathbf{E}) - det\mathbf{M}^{-1}tr((adj\mathbf{Q})\mathbf{E}) (7.2.111)

where adj**Q** denotes the adjoint of the matrix **Q**. Because  $tr(\mathbf{Q}\mathbf{M}^{-1}) = u_{n(1)}^2 + u_{n(2)}^2$  and  $det(\mathbf{Q}\mathbf{M}^{-1}) = u_{n(1)}^2 u_{n(2)}^2$ , (7.2.111) can be rewritten

$$\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{Q}\mathbf{M}^{-1}\mathbf{E}\right) = \left(u_{n(1)}^{2} + u_{n(2)}^{2}\right)\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right) - \det\mathbf{M}^{-1}\operatorname{tr}\left(\left(\operatorname{adj}\mathbf{Q}\right)\mathbf{E}\right)$$
(7.2.112)

Equation (7.2.112) allows (7.2.108) to be written

$$\mathbf{r}_{1}^{T}\mathbf{E}\mathbf{r}_{1} = \frac{\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{Q}\mathbf{M}^{-1}\mathbf{E}\right) - u_{n(2)}^{2}\operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right)}{u_{n(1)}^{2} - u_{n(2)}^{2}} = \operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right) \frac{u_{n(1)}^{2} - \operatorname{det}\mathbf{M}^{-1}\operatorname{tr}\left(\left(\operatorname{adj}\mathbf{Q}\right)\mathbf{E}\right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.113)  
$$= \operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right) \frac{u_{n(1)}^{2} - \frac{\operatorname{tr}\left(\left(\operatorname{adj}\mathbf{Q}\right)\mathbf{E}\right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}}{u_{n(1)}^{2} - u_{n(2)}^{2}} = \operatorname{tr}\left(\mathbf{M}^{-1}\mathbf{E}\right) \frac{u_{n(1)}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.113)

where the definition (7.2.103) has been used. The derivation of (7.2.102) follows by an identical argument.

Clearly the quantity  $u_0^2$  has the physical dimensions of velocity squared. One can show from (7.2.1) through (7.2.3) that  $u_0^2$  is the squared longitudinal wave speed in the limiting case where  $\xi \to \infty$  and  $\kappa \to \infty$ . The case  $\xi \to \infty$  corresponds to the case where there is no diffusion, i.e., the drag is so great that the two constituents move at the same velocity. The case  $\kappa \to \infty$ corresponds, with appropriate boundary conditions, to the case where the temperature is a constant. For this reason,  $u_0$  is called the *frozen isothermal wave speed*. This speed first appeared in the porous media article by Biot [Ref. 8]. If we adopt the numerical values shown in Table 2 of Section 6.2 and the true density values shown in Table 1 of this Section, the following values for the frozen isothermal wave speed are obtained.

Property	Ruhr	Tennessee	Charcoal	Berea	Westerly	Weber
	Sandstone	Marble	Granite	Sandstone	Granite	Sandstone
$\gamma_{s_R}$ (g/cm <sup>3</sup> )	2.6	2.7	2.7	2.6	2.7	2.6
$u_0$ (m/sec)	4337	5345	4954	3209	4802	4059

Table 2 Calculated Frozen Isothermal Wave Speeds

This table and Table 1 of this Section show that the frozen isothermal wave speeds are close to the acceleration wave speeds  $u_{n(1)}$ . It is a theoretical result that one can establish from the spectral representation (7.2.104) that

$$u_{n(1)}^2 \ge u_0^2 \ge u_{n(2)}^2 \tag{7.2.114}$$

Tables 1 and 2 show that, to the numerical approximation used here,  $u_0 = u_{n(1)}$ , for Weber Sandstone.

A similar calculation allows the term 
$$\frac{\theta^+}{\kappa} \left( \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^T \mathbf{r} \right)^2$$
 in (7.2.99) to be written

$$\frac{\theta^{+}}{\kappa} \left( \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^{T} \mathbf{r}_{1} \right)^{2} = \frac{\theta^{+}}{\kappa} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^{T} \mathbf{r}_{1} \mathbf{r}_{1}^{T} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)$$

$$= \frac{\overline{c}_{v}}{\kappa} \frac{\left( p_{n(1)}^{2} - u_{n(1)}^{2} \right) \left( u_{n(1)}^{2} - p_{n(2)}^{2} \right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.115)

$$\frac{\theta^{+}}{\kappa} \left( \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^{T} \mathbf{r}_{2} \right)^{2} = \frac{\theta^{+}}{\kappa} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^{T} \mathbf{r}_{2} \mathbf{r}_{2}^{T} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)$$

$$= \frac{\overline{c}_{v}}{\kappa} \frac{\left( p_{n(1)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(2)}^{2} - u_{n(2)}^{2} \right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.116)

for the two cases  $\mathbf{r} = \mathbf{r}_1$  and  $\mathbf{r} = \mathbf{r}_2$ , respectively. The quantities  $p_{n(1)}^2$  and  $p_{n(2)}^2$  are the two roots of the equation

$$\det\left(\mathbf{Q} + \frac{\theta}{c_{v}}\left(A - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(A - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T} - p_{n}^{2}\mathbf{M}\right) = 0$$
(7.2.117)

These quantities are positive because the symmetric matrix

 $\mathbf{Q} + \frac{\theta}{c_v} \left( A - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right) \left( A - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^T$  is positive definite. It is possible to use the spectral form of this matrix to establish that

$$p_{n(1)}^2 \ge u_{n(1)}^2 \ge p_{n(2)}^2 \ge u_{n(2)}^2$$
(7.2.118)

If we adopt the formulas (7.2.101) and (7.2.115), it follows from (7.2.99) that

$$\xi \mathbf{r}_{1}^{T} \mathbf{E} \mathbf{r}_{1} + \frac{\theta^{+}}{\kappa} \mathbf{r}_{1}^{T} \left( \mathbf{A} - \mathbf{B} \right) \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \mathbf{r}_{1}$$

$$= \left( \xi - \frac{1}{4} \frac{\theta^{+}}{\kappa} \left( \gamma + \frac{\zeta}{\theta^{+}} \right)^{2} \right) \left( \frac{1}{\rho_{f}^{+}} + \frac{1}{\rho_{s_{R}}} \right) \frac{u_{n(1)}^{2} - u_{0}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} + \frac{\overline{c}_{v}}{\kappa} \frac{\left( p_{n(1)}^{2} - u_{n(1)}^{2} \right) \left( u_{n(1)}^{2} - p_{n(2)}^{2} \right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.119)

for the choice  $\mathbf{r} = \mathbf{r}_1$ . By an entirely similar calculation,

$$\xi \mathbf{r}_{2}^{T} \mathbf{E} \mathbf{r}_{2} + \frac{\theta^{+}}{\kappa} \mathbf{r}_{2}^{T} (\mathbf{A} - \mathbf{B}) (\mathbf{A}^{T} - \mathbf{D}^{T}) \mathbf{r}_{2}$$

$$= \left( \xi - \frac{1}{4} \frac{\theta^{+}}{\kappa} \left( \gamma + \frac{\zeta}{\theta^{+}} \right)^{2} \right) \left( \frac{1}{\rho_{f}^{+}} + \frac{1}{\rho_{s_{R}}} \right) \frac{u_{0}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} + \frac{\overline{c}_{v}}{\kappa} \frac{\left( p_{n(1)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(2)}^{2} - u_{n(2)}^{2} \right)}{u_{n(1)}^{2} - u_{n(2)}^{2}}$$
(7.2.120)

for the choice  $\mathbf{r} = \mathbf{r}_2$ .

Equations (7.2.119) and (7.2.120) have the advantage of allowing the calculation of the coefficient in the amplitude ordinary differential equation, equation (7.2.95), without explicit knowledge of the eigenvectors  $\mathbf{r}_1$  and  $\mathbf{r}_2$ . In the next chapter we shall investigate the propagation of plane harmonic waves in the poroelastic mixture adopted in this chapter. We shall see the expressions (7.2.119) and (7.2.120) arise in one of the limiting cases examined for these types of waves.

In closing this chapter, it is useful to briefly examine a case omitted in the derivation of the above results. This case is the one ruled out by the assumption (7.2.29), i.e.  $\kappa = 0$ . If we allow for this possibility, then (7.2.28) is satisfied without necessarily having [GRAD $\theta$ ]  $\cdot \mathbf{n} = 0$ . In this case, the inequality (7.2.5) forces

$$\gamma = -\frac{\varsigma}{\theta^+} \tag{7.2.121}$$

or, from the definitions (7.2.89) and (7.2.90),

$$\mathbf{B} = \mathbf{D} \tag{7.2.122}$$

Logically, if one reaches the result  $\kappa = 0$  with the argument that the constitutive equations do not have a dependence on the temperature gradient, one could conclude from (3.2.24) that  $\gamma$  and, from (7.2.121),  $\varsigma$  are zero. For our purposes here, the result (7.2.121) is a sufficient specialization.

Rather than repeat the calculations we carried out the case for a conductor, it is convenient to simply state the key results. It can be shown from (7.2.3), with  $\kappa = 0$ , that

$$u_n^2 \overline{c}_v [\text{GRAD}\,\theta] = -\theta^+ \left(\overline{\tau}_f - \alpha_f - \gamma\right) \left(\mathbf{a}_f \cdot \mathbf{n}\right) \mathbf{n} - \theta^+ \left(\overline{\tau}_s - \alpha_s + \gamma\right) \left(\mathbf{a}_s \cdot \mathbf{n}\right) \mathbf{n}$$
(7.2.123)

Another jump that arises in the derivation of the differential equation for the amplitude is  $[\text{GRAD}\frac{\partial\theta}{\partial t}]$ . This quantity is also determined by the energy equation and can be shown to take the form

$$\overline{c}_{\nu} \left[ \operatorname{GRAD} \frac{\partial \theta}{\partial t} \right] = -\theta^{+} \left( \overline{\tau}_{f} - \alpha_{f} - \gamma \right) \left( \mathbf{d}_{f} \cdot \mathbf{n} \right) \mathbf{n} - \theta^{+} \left( \overline{\tau}_{s} - \alpha_{s} + \gamma \right) \left( \mathbf{d}_{s} \cdot \mathbf{n} \right) \mathbf{n}$$
(7.2.124)

The results for the nonconductor shall be given without proof. One can formally adopt (7.2.36) through (7.2.50) with the following changes:

a) Replace 
$$\overline{\lambda}_{ff}$$
 with  $\overline{\lambda}_{ff} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{f} - \alpha_{f} - \gamma)^{2}$ .  
b) Replace  $\overline{\lambda}_{sf}$  with  $\overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{f} - \alpha_{f} - \gamma) (\overline{\tau}_{s} - \alpha_{s} + \gamma)$   
c) Replace  $\overline{\lambda}_{ss}$  with  $\overline{\lambda}_{ss} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{s} - \alpha_{s} + \gamma)^{2}$ .

The amplitude equation for transverse waves, equation (7.2.82), can be adopted without formal change. For the longitudinal waves, the amplitude is given by

$$2\frac{\delta a}{\delta t} = -\left(\xi \mathbf{r}^T \mathbf{E} \mathbf{r}\right) a \tag{7.2.125}$$

rather than (7.2.95). However, in this case the eigenvectors  $\mathbf{r}_1$  and  $\mathbf{r}_2$  are calculated from (7.2.92) with  $\mathbf{Q}$  given by

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{f} - \alpha_{f} - \gamma)^{2} & \overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{f} - \alpha_{f} - \gamma) (\overline{\tau}_{s} - \alpha_{s} + \gamma) \\ \overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{f} - \alpha_{f} - \gamma) (\overline{\tau}_{s} - \alpha_{s} + \gamma) & \overline{\lambda}_{ss} + \frac{\theta^{+}}{\overline{c}_{v}} (\overline{\tau}_{s} - \alpha_{s} + \gamma)^{2} + 2\mu_{ss} \end{bmatrix}$$
(7.2.126)

rather than by (7.2.46).

The nonconductor case just discussed also provides a framework for the study of the growth and decay of shock waves in porous materials defined by (7.2.1) through (7.2.3). By definition, a *shock wave* in the porous material defined by (7.2.1) through (7.2.6) is a moving singular surface across which

$$\begin{bmatrix} \mathbf{w}_f \end{bmatrix} = \begin{bmatrix} \mathbf{w}_s \end{bmatrix} = \mathbf{0} \tag{7.2.127}$$

but  $\left[\frac{\partial^2 \mathbf{w}_f}{\partial t^2}\right]$ ,  $\left[\frac{\partial^2 \mathbf{w}_s}{\partial t^2}\right]$  and  $\left[\frac{\partial \theta}{\partial t}\right]$  are not necessarily zero. The amplitudes of the shock wave are defined by

$$\mathbf{a}_{f} = \left[\frac{\partial \mathbf{w}_{f}}{\partial t}\right] \tag{7.2.128}$$

and

$$\mathbf{a}_{s} = \left[\frac{\partial \mathbf{w}_{s}}{\partial t}\right] \tag{7.2.129}$$

The jump balance equations (7.2.12) through (7.2.14), with  $\kappa = 0$ , yield the results

$$\rho_{s_{R}}u_{n}^{2}\mathbf{a}_{s} - (\overline{\lambda}_{ss} + \mu_{ss})(\mathbf{a}_{s} \cdot \mathbf{n})\mathbf{n} - \mu_{ss}\mathbf{a}_{s} - \overline{\lambda}_{sf}(\mathbf{a}_{f} \cdot \mathbf{n})\mathbf{n} + (\alpha_{s} - \overline{\tau}_{s} + \gamma)u_{n}[\theta]\mathbf{n} = \mathbf{0} \quad (7.2.130)$$

$$\rho_f^+ u_n^2 \mathbf{a}_f - \overline{\lambda}_{sf} \left( \mathbf{a}_s \cdot \mathbf{n} \right) \mathbf{n} - \overline{\lambda}_{ff} \left( \mathbf{a}_f \cdot \mathbf{n} \right) \mathbf{n} + (\alpha_f^- - \overline{\tau}_f^- - \gamma) u_n \left[ \boldsymbol{\theta} \right] \mathbf{n} = \mathbf{0}$$
(7.2.131)

and

$$u_{n}\overline{c}_{v}[\theta] + \theta^{+}(\alpha_{s} - \overline{\tau}_{s} + \gamma)\mathbf{a}_{s} \cdot \mathbf{n} + \theta^{+}(\alpha_{f} - \overline{\tau}_{f} - \gamma)\mathbf{a}_{f} \cdot \mathbf{n} = 0$$
(7.2.132)

Equation (7.2.132) gives the jump in the temperature across the shock wave in terms of the two shock amplitudes. If this result is substituted into (7.2.130) and (7.2.131) the result is

$$\left(\overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{s} - \overline{\tau}_{s} + \gamma) (\alpha_{f} - \overline{\tau}_{f} - \gamma)\right) (\mathbf{a}_{f} \cdot \mathbf{n}) \mathbf{n} + \left(\overline{\lambda}_{ss} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{s} - \overline{\tau}_{s} + \gamma) (\alpha_{s} - \overline{\tau}_{s} + \gamma) + \mu_{ss}\right) (\mathbf{a}_{s} \cdot \mathbf{n}) \mathbf{n} + \mu_{ss} \mathbf{a}_{s} = \rho_{s_{R}} u_{n}^{2} \mathbf{a}_{s}$$

$$\left(\overline{\lambda}_{ff} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{f} - \overline{\tau}_{f} - \gamma)^{2}\right) (\mathbf{a}_{f} \cdot \mathbf{n}) \mathbf{n} + \left(\overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{s} - \overline{\tau}_{s} + \gamma) (\alpha_{f} - \overline{\tau}_{f} - \gamma)\right) (\mathbf{a}_{s} \cdot \mathbf{n}) \mathbf{n} + \left(\overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{s} - \overline{\tau}_{s} + \gamma) (\alpha_{f} - \overline{\tau}_{f} - \gamma)\right) (\mathbf{a}_{s} \cdot \mathbf{n}) \mathbf{n} + \left(\overline{\lambda}_{sf} + \frac{\theta^{+}}{\overline{c}_{v}} (\alpha_{s} - \overline{\tau}_{s} + \gamma) (\alpha_{f} - \overline{\tau}_{f} - \gamma)\right) (\mathbf{a}_{s} \cdot \mathbf{n}) \mathbf{n}$$

$$= \rho_{f}^{+} u_{n}^{2} \mathbf{a}_{f}$$
(7.2.134)

These equations are formally identical to our previous results for acceleration waves (7.2.36) and (7.2.37) *providing* that we replace the coefficients according to the rules a), b) and c) above. The fact that shock waves in a linear theory have the same propagation as that for acceleration waves is typical. Sometimes such shock waves are called *weak shocks*. The growth and decay of these weak shocks in the longitudinal are given by equations formally identical to (7.2.125).

## References

- 1. TRUESDELL, C. and R. TOUPIN, *The Classical Field Theories*, Handbuch der Physik Vol III/1 (ed. S. Flugge), Spring Verlag, Berlin (1960).
- 2. FRIEDLANDER, R. G., Sound Pulses, Cambridge University Press (1958)
- 3. BOWEN, R. M., Theory of Mixtures, in *Continuum Physics*, Vol III, (ed A. C. Eringen), Academic Press, New York (1976).

- 4. BOWEN, R. M., Introduction to Continuum Mechanics for Engineers, *Plenum Press*, New York, N.Y., 1989.
- 5. BOWEN, R. M., AND P. J. CHEN, Shock Waves in a Mixture of Linear Elastic Materials, *Rend. Circ. Matem. Di Palermo*, **21**, 267-283, 1972.
- 6. BOWEN, R. M. and P. J. CHEN, Waves in a Binary Mixture of Linear Elastic Materials, *J. de Mecanique*, **14**, 237-266, 1975.
- 7. FARMER, I. W., Engineering Properties of Rocks, E. & F. N. Spon, London, 1958.
- 8. BIOT, M. A., Theory of Propagation of Elastic Waves in a Fluid-Saturated Porous Solid. I. Low Frequency Range, *J. Acoust. Soc. Amer.*, **28**, 179-191, 1956.

# **Plane Harmonic Waves**

In this chapter we shall build upon the results in Chapter 7 and look at a more interesting kind of wave propagation problem. The problem adopts one dimensional versions of the governing partial differential equations studied in Chapter 7 and examines a certain type of wave solution. The solution is that of a plane harmonic wave. The one dimensional assumption is made as a matter of convenience and can be avoided as was done in Chapter 6 by the assumption of a plane wave. For simplicity here, we shall just adopt the one dimensional formulation. The interest in studying harmonic waves is that it represents a nontrivial solution of the governing equations that carries a lot of interesting physical information about the features of the model. It is a solution that lends itself to analytical analysis as opposed to many other interesting problems that require numerical solutions. To a significant extent, the results in this chapter are a special case of the work of Bowen [Ref. 1] and Bowen and Reinicke [Ref. 2]. The earlier works of Biot [Ref. 3] and of Craine and Johnson [Ref. 4] are also important to the understanding of the material in this Chapter. This chapter is also influenced by the formalism used by Chadwick [Ref. 5] when he studied plane harmonic waves in thermoelasticity.

## 8.1 One Dimensional Governing Equations

If we again adopt the governing partial differential equations (7.2.1) through (7.2.3) and the restrictions on the properties (7.2.4) through (7.2.11), the one dimensional case corresponds to the assumptions

$$\mathbf{w}_s = \left(w_s(X,t), 0, 0\right) \tag{8.1.1}$$

$$\mathbf{w}_{f} = \left(w_{f}\left(X,t\right),0,0\right) \tag{8.1.2}$$

and

$$\theta = \theta(X, t) \tag{8.1.3}$$

These specializations reduce the governing partial differential equations (7.2.1), (7.2.2) and (7.2.3) to

$$\rho_{s_{R}} \frac{\partial^{2} w_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + 2\mu_{ss}) \frac{\partial^{2} w_{s}}{\partial X^{2}} + \overline{\lambda}_{sf} \frac{\partial^{2} w_{f}}{\partial X^{2}} + (\alpha_{s} - \overline{\tau}_{s} + \gamma) \frac{\partial \theta}{\partial X} - \xi \left(\frac{\partial w_{s}}{\partial t} - \frac{\partial w_{f}}{\partial t}\right)$$
(8.1.4)

# Chapter 8

$$\rho_{f}^{+} \frac{\partial^{2} w_{f}}{\partial t^{2}} = \overline{\lambda}_{sf} \frac{\partial^{2} w_{s}}{\partial X^{2}} + \overline{\lambda}_{ff} \frac{\partial^{2} w_{f}}{\partial X^{2}} + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \frac{\partial \theta}{\partial X} - \xi \left( \frac{\partial w_{f}}{\partial t} - \frac{\partial w_{s}}{\partial t} \right)$$
(8.1.5)

and

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+}\left((\overline{\tau}_{s} - \alpha_{s} + \zeta/\theta^{+})\frac{\partial^{2}w_{s}}{\partial X\partial t} + (\overline{\tau}_{f} - \alpha_{f} - \zeta/\theta^{+})\frac{\partial^{2}w_{f}}{\partial X\partial t}\right) = \kappa\frac{\partial^{2}\theta}{\partial X^{2}}$$
(8.1.6)

The manipulation of these equations is facilitated if we write them in matrix form as follows:

$$\mathbf{M}\frac{\partial^2 \mathbf{w}}{\partial t^2} = \mathbf{Q}\frac{\partial^2 \mathbf{w}}{\partial X^2} - (\mathbf{A} - \mathbf{B})\frac{\partial \theta}{\partial X} - \xi \mathbf{E}\frac{\partial \mathbf{w}}{\partial t}$$
(8.1.7)

and

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left(\mathbf{A}^{T} - \mathbf{D}^{T}\right) \frac{\partial^{2} \mathbf{w}}{\partial X \partial t} = \kappa \frac{\partial^{2}\theta}{\partial X^{2}}$$
(8.1.8)

where

$$\mathbf{w}(X,t) = \begin{bmatrix} w_f(X,t) \\ w_s(X,t) \end{bmatrix}$$
(8.1.9)

and, as in Chapter 7,

$$\mathbf{M} = \begin{bmatrix} \rho_f^+ & 0\\ 0 & \rho_{s_R} \end{bmatrix}$$
(8.1.10)

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix}$$
(8.1.11)

$$\mathbf{E} = \begin{bmatrix} 1 & -1 \\ -1 & 1 \end{bmatrix}$$
(8.1.12)

and

$$\mathbf{A} = \begin{bmatrix} \overline{\tau}_f - \alpha_f \\ \overline{\tau}_s - \alpha_s \end{bmatrix}$$
(8.1.13)

$$\mathbf{B} = \gamma \begin{bmatrix} -1\\1 \end{bmatrix} \tag{8.1.14}$$

$$\mathbf{D} = \frac{\varsigma}{\theta^+} \begin{bmatrix} 1\\ -1 \end{bmatrix}$$
(8.1.15)

It is perhaps useful to note in passing that one can derive from (8.1.7) and (8.1.8) a single uncoupled partial differential equation for each of the unknowns. In the case  $\kappa > 0$ , the result is three uncoupled partial differential equations of the form

$$Lw_f = Lw_s = L\theta = 0 \tag{8.1.16}$$

where L is the sixth order differential operator defined by

$$L = \frac{\partial}{\partial t} \left[ \left( \frac{\partial^2}{\partial t^2} - p_{n(1)}^2 \frac{\partial^2}{\partial X^2} \right) \left( \frac{\partial^2}{\partial t^2} - p_{n(2)}^2 \frac{\partial^2}{\partial X^2} \right) + \omega_{\xi} \frac{\partial}{\partial t} \left( \frac{\partial^2}{\partial t^2} - p_0^2 \frac{\partial^2}{\partial X^2} \right) \right] - \frac{\kappa}{\overline{c}_{\nu}} \frac{\partial^2}{\partial X^2} \left[ \left( \frac{\partial^2}{\partial t^2} - u_{n(1)}^2 \frac{\partial^2}{\partial X^2} \right) \left( \frac{\partial^2}{\partial t^2} - u_{n(2)}^2 \frac{\partial^2}{\partial X^2} \right) + \omega_0 \frac{\partial}{\partial t} \left( \frac{\partial^2}{\partial t^2} - u_0^2 \frac{\partial^2}{\partial X^2} \right) \right]$$
(8.1.17)

where  $\omega_0$  and  $\omega_1$  are characteristic frequencies defined by

$$\omega_{0} = \left(\xi - \frac{\theta^{+}}{4\kappa} \left(\gamma + \frac{\zeta}{\theta^{+}}\right)^{2}\right) \frac{\operatorname{tr}\left((\operatorname{adj}\mathbf{M})\mathbf{E}\right)}{\det\mathbf{M}} = \left(\xi - \frac{\theta^{+}}{4\kappa} \left(\gamma + \frac{\zeta}{\theta^{+}}\right)^{2}\right) \left(\frac{1}{\rho_{R}} + \frac{1}{\rho_{f}^{+}}\right)$$
(8.1.18)

and

$$\omega_{\xi} = \xi \frac{\operatorname{tr}((\operatorname{adj} \mathbf{M})\mathbf{E})}{\det \mathbf{M}} = \xi \left(\frac{1}{\rho_{R}} + \frac{1}{\rho_{f}^{+}}\right)$$
(8.1.19)

Note, in passing, that the inequalities (7.2.5) and (7.2.6) insure that the characteristic times  $\omega_0$  and  $\omega_{\xi}$  are non-negative. Also note that the definition (8.1.18) assumes a strengthened version of (7.2.4) in the form

$$\kappa > 0 \tag{8.1.20}$$

The quantities  $u_{n(1)}^2$  and  $u_{n(2)}^2$  in (8.1.17) are given by (7.2.50) and  $u_0^2$  is given by (7.2.103). The quantities  $p_{n(1)}^2$  and  $p_{n(2)}^2$ , as in equation (7.2.117), are the roots of

$$\det\left(\mathbf{Q} + \frac{\theta^{+}}{2c_{v}^{+}}\left(\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T}\right) - p^{2}\mathbf{M}\right) = 0$$
(8.1.21)

and  $p_0^2$  is given by

$$p_0^2 = \frac{\operatorname{tr}\left(\left(\mathbf{Q} + \frac{\theta^+}{2c_v^+} \left(\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^T\right)\right)\mathbf{E}\right)}{\operatorname{tr}\left((\operatorname{adj}\mathbf{M})\mathbf{E}\right)}$$
(8.1.22)

We shall say more about these roots in Section 8.2.

### 8.2 Plane Progressive Waves: Dispersion Relation

A plane progressive wave is a solution of (8.1.7) and (8.1.8) of the form

$$\mathbf{w}(X,t) = \mathbf{a}e^{i(kx-\omega t)}$$
(8.2.1)

and

$$\theta(X,t) - \theta^+ = b e^{i(kx - \omega t)}$$
(8.2.2)

where the 2×1 matrix **a** and the scalar *b* are the amplitudes of the wave. The quantity *k* is the *wave number* and  $\omega$  is the *frequency*. The point if view is that the frequency is given and we seek the relationship  $k(\omega)$  forced by requiring (8.2.1) and (8.2.2) to be solutions of the partial differential equations (8.1.7) and (8.1.8). The relationship  $k(\omega)$  is called the *dispersion relation*. Given this relationship, the *phase velocity* and the *attenuation coefficient* are defined by

$$q = \frac{\omega}{\mathscr{R}(k)} \tag{8.2.3}$$

and

$$s = \mathscr{I}(k) \tag{8.2.4}$$

where  $\Re(k)$  denotes the real part of  $k(\omega)$  and  $\mathscr{I}(k)$  its imaginary part.

If (8.2.1) and (8.2.2) are substituted into (8.1.7) and (8.1.8), respectively, the result is

$$\left(k^{2}\mathbf{Q}-\omega^{2}\mathbf{M}-i\omega\xi\mathbf{E}\right)\mathbf{a}+ik\left(\mathbf{A}-\mathbf{B}\right)b=0$$
(8.2.5)

and

$$\theta^{+} \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \omega k \mathbf{a} + \left( \kappa k^{2} - i \omega \overline{c}_{v} \right) b = 0$$
(8.2.6)

In order that equations (8.2.5) and (8.2.6) yield nonzero amplitudes, the  $3 \times 3$  determinant of coefficients must vanish, i.e.,<sup>1</sup>

$$(\kappa k^{2} - i\omega\overline{c}_{v}) \det(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}) -\theta^{+}i\omega k^{2} (\mathbf{A}^{T} - \mathbf{D}^{T}) \operatorname{adj}((k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}))(\mathbf{A} - \mathbf{B}) = 0$$

$$(8.2.7)$$

This expression can be written

$$i\omega \left( \det \left( k^{2} \mathbf{Q} - \omega^{2} \mathbf{M} - i\omega \xi \mathbf{E} \right) + \frac{\theta^{+}}{c_{v}} k^{2} \left( \mathbf{A}^{T} - \mathbf{D}^{T} \right) \operatorname{adj} \left( \left( k^{2} \mathbf{Q} - \omega^{2} \mathbf{M} - i\omega \xi \mathbf{E} \right) \right) \left( \mathbf{A} - \mathbf{B} \right) \right)$$

$$- \frac{\kappa}{c_{v}} k^{2} \det \left( k^{2} \mathbf{Q} - \omega^{2} \mathbf{M} - i\omega \xi \mathbf{E} \right) = 0$$

$$(8.2.8)$$

Next, we shall show that (8.2.8) can be written

$$i\omega\Big(\Big(p_{n(1)}^{2}k^{2}-\omega^{2}\Big)\Big(p_{n(2)}^{2}k^{2}-\omega^{2}\Big)-i\omega\omega_{\xi}\Big(p_{0}^{2}k^{2}-\omega^{2}\Big)\Big) \\ -\frac{\kappa}{\overline{c}_{v}}k^{2}\Big(\Big(u_{n(1)}^{2}k^{2}-\omega^{2}\Big)\Big(u_{n(2)}^{2}k^{2}-\omega^{2}\Big)-i\omega\omega_{0}\Big(u_{0}^{2}k^{2}-\omega^{2}\Big)\Big)=0$$
(8.2.9)

The first formula we need is

$$\det \left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)$$
  
= 
$$\det \mathbf{M}\left(\left(u_{n(1)}^{2}k^{2} - \omega^{2}\right)\left(u_{n(2)}^{2}k^{2} - \omega^{2}\right) - i\omega\xi\frac{\operatorname{tr}\left((\operatorname{adj}\mathbf{M})\mathbf{E}\right)}{\det\mathbf{M}}\left(u_{0}^{2}k^{2} - \omega^{2}\right)\right) \quad (8.2.10)$$

This formula follows from the elementary formula for the determinant of the sum of  $2 \times 2$  matrices, det $(\mathbf{A} + \mathbf{B}) = \det \mathbf{A} + tr((adj\mathbf{A})\mathbf{B}) + \det \mathbf{B}$  and the formulas

$$\det\left(k^{2}\mathbf{Q}-\omega^{2}\mathbf{M}\right)=\det\mathbf{M}\left(k^{2}u_{1}^{2}-\omega^{2}\right)\left(k^{2}u_{2}^{2}-\omega^{2}\right)$$
(8.2.11)

<sup>1</sup> For the 3×3 matrix  $\mathbf{F} = \begin{bmatrix} A_{11} & A_{12} & B_1 \\ A_{21} & A_{22} & B_2 \\ C_1 & C_2 & D \end{bmatrix}$ , it is not difficult to show that det  $\mathbf{F} = \mathbf{D} \det \mathbf{A} - \mathbf{C} (\operatorname{adj} \mathbf{A}) \mathbf{B}$ , where  $\mathbf{A} = \begin{bmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{bmatrix}$ ,  $\mathbf{B} = \begin{bmatrix} B_1 \\ B_2 \end{bmatrix}$  and  $\mathbf{C} = \begin{bmatrix} C_1 & C_2 \end{bmatrix}$ .

$$u_0^2 = \frac{\mathrm{tr}(\mathrm{adj}(\mathbf{Q})\mathbf{E})}{\mathrm{tr}(\mathrm{adj}(\mathbf{M})\mathbf{E})}$$
(8.2.12)

$$\det \mathbf{E} = 0 \tag{8.2.13}$$

Equation (8.2.11) follows from (7.2.92), equation (8.2.12) is the definition (7.2.103) repeated, and (8.2.13) follows from the definition (7.2.87).

The next formula we need involves a rather complicated rearrangement of the term  $(\mathbf{A}^T - \mathbf{D}^T) \operatorname{adj}((k^2\mathbf{Q} - \omega^2\mathbf{M} - i\omega\xi\mathbf{E}))(\mathbf{A} - \mathbf{B})$ . First, because the matrix  $\operatorname{adj}((k^2\mathbf{Q} - \omega^2\mathbf{M} - i\omega\xi\mathbf{E}))$  is symmetric, we can use (7.2.96) and write

$$\begin{aligned} \left(\mathbf{A}^{T} - \mathbf{D}^{T}\right) \operatorname{adj}\left(\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)\right)\left(\mathbf{A} - \mathbf{B}\right) \\ &= \operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)\right)\left(\mathbf{A} - \mathbf{B}\right)\left(\mathbf{A}^{T} - \mathbf{D}^{T}\right)\right) \\ &= \operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)\right)\left(\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T} - \frac{1}{4}(\mathbf{B} - \mathbf{D})(\mathbf{B} - \mathbf{D})^{T}\right)\right) \\ &= \operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)\right)\left(\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T}\right)\right) \\ &- \frac{1}{4}\operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)\right)\left((\mathbf{B} - \mathbf{D})(\mathbf{B} - \mathbf{D})^{T}\right)\right) \end{aligned}$$
(8.2.14)

Equation (7.2.98), repeated here, tells us that

$$(\mathbf{B} - \mathbf{D}) (\mathbf{B} - \mathbf{D})^{T} = \left(\gamma + \frac{\varsigma}{\theta^{+}}\right)^{2} \begin{bmatrix} -1\\1 \end{bmatrix} \begin{bmatrix} -1 & 1 \end{bmatrix} = \left(\gamma + \frac{\varsigma}{\theta^{+}}\right)^{2} \mathbf{E}$$
 (8.2.15)

This identity along with the definition (8.2.12) allows the last term in (8.2.14) to be written

$$-\frac{1}{4}\operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q}-\omega^{2}\mathbf{M}-i\omega\xi\mathbf{E}\right)\right)\left(\left(\mathbf{B}-\mathbf{D}\right)\left(\mathbf{B}-\mathbf{D}\right)^{T}\right)\right)=-\frac{1}{4}\left(\gamma+\frac{\zeta}{\theta^{+}}\right)^{2}\operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q}-\omega^{2}\mathbf{M}-i\omega\xi\mathbf{E}\right)\right)\mathbf{E}\right)$$
$$=-\frac{1}{4}\left(\gamma+\frac{\zeta}{\theta^{+}}\right)^{2}\operatorname{tr}\left(\left(\operatorname{adj}\left(k^{2}\mathbf{Q}-\omega^{2}\mathbf{M}\right)\right)\mathbf{E}\right)$$
$$=-\frac{1}{4}\left(\gamma+\frac{\zeta}{\theta^{+}}\right)^{2}\operatorname{tr}\left(\operatorname{adj}\mathbf{M}\right)\left(u_{0}^{2}k^{2}-\omega^{2}\right)$$
(8.2.16)

Given (8.2.14) and (8.2.16), it follows that

$$det(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}) + \frac{\theta^{+}}{c_{v}}k^{2}(\mathbf{A}^{T} - \mathbf{D}^{T})adj((k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}))(\mathbf{A} - \mathbf{B})$$

$$= det(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E})$$

$$+ \frac{\theta^{+}}{c_{v}}k^{2} tr\left((adj(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}))\left((\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D}))(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D}))^{T}\right)\right)$$

$$- \frac{\theta^{+}}{4c_{v}}k^{2} tr\left((adj(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}))((\mathbf{B} - \mathbf{D})(\mathbf{B} - \mathbf{D})^{T})\right)$$

$$= det\left(k^{2}\left(\mathbf{Q} + \frac{\theta^{+}}{c_{v}}\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T}\right) - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)$$

$$- \frac{\theta^{+}}{4c_{v}}k^{2}\left(\gamma + \frac{\xi}{\theta^{+}}\right)^{2} tr(adj\mathbf{M})(u_{0}^{2}k^{2} - \omega^{2})$$
(8.2.17)

This formula utilized, again, the identity  $det(\mathbf{A} + \mathbf{B}) = det \mathbf{A} + tr((adj\mathbf{A})\mathbf{B}) + det \mathbf{B}$  except this time to combine terms. This formula also made use of the result

$$\det\left(\left(\mathbf{A}-\frac{1}{2}(\mathbf{B}+\mathbf{D})\right)\left(\mathbf{A}-\frac{1}{2}(\mathbf{B}+\mathbf{D})\right)^{T}\right)=0.$$

Given (8.2.17), the result (8.2.8) can be written

$$i\omega \det\left(k^{2}\left(\mathbf{Q} + \frac{\theta^{+}}{c_{v}}\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)\left(\mathbf{A} - \frac{1}{2}(\mathbf{B} + \mathbf{D})\right)^{T}\right) - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right)$$

$$-\frac{\kappa}{c_{v}}k^{2}\left(\det\left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right) + i\omega\frac{\theta^{+}}{4\kappa}\left(\gamma + \frac{\zeta}{\theta^{+}}\right)^{2}\operatorname{tr}\left(\operatorname{adj}\mathbf{M}\right)\left(u_{0}^{2}k^{2} - \omega^{2}\right)\right) = 0$$
(8.2.18)

Given (8.2.10), the last factor in (8.2.18) can be written

$$\det \left(k^{2}\mathbf{Q} - \omega^{2}\mathbf{M} - i\omega\xi\mathbf{E}\right) + i\omega\frac{\theta^{+}}{4\kappa}\left(\gamma + \frac{\zeta}{\theta^{+}}\right)^{2}\operatorname{tr}\left(\operatorname{adj}\mathbf{M}\right)\left(u_{0}^{2}k^{2} - \omega^{2}\right)$$
  
= 
$$\det \mathbf{M}\left(\left(u_{n(1)}^{2}k^{2} - \omega^{2}\right)\left(u_{n(2)}^{2}k^{2} - \omega^{2}\right) - i\omega\omega_{0}\left(u_{0}^{2}k^{2} - \omega^{2}\right)\right)$$
(8.2.19)

where the definition of the characteristic frequency  $\omega_0$ , given by (8.1.18), has been used. Equation (8.2.18) reduces to the result (8.2.9) if we use the identity

$$\det\left(k^{2}\left(\mathbf{Q}+\frac{\theta^{+}}{c_{v}}\left(\mathbf{A}-\frac{1}{2}\left(\mathbf{B}+\mathbf{D}\right)\right)\left(\mathbf{A}-\frac{1}{2}\left(\mathbf{B}+\mathbf{D}\right)\right)^{T}\right)-\omega^{2}\mathbf{M}-i\omega\xi\mathbf{E}\right)$$

$$=\det\mathbf{M}\left(\left(p_{n(1)}^{2}k^{2}-\omega^{2}\right)\left(p_{n(2)}^{2}k^{2}-\omega^{2}\right)-i\omega\omega_{\xi}\left(p_{0}^{2}k^{2}-\omega^{2}\right)\right)$$
(8.2.20)

Equation (8.2.20) is derived by the same calculation that produced (8.2.10) with the exception that now the symmetric matrix  $\mathbf{Q} + \frac{\theta^+}{c_v} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right) \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^T$  is in the formula. The definition (8.1.19) has also been used.

A few words or explanation need to be given about the parameters  $p_{n(1)}^2$ ,  $p_{n(2)}^2$  and  $p_0^2$  defined by (8.1.21) and (8.1.22). Because the symmetric matrix  $\mathbf{Q} + \frac{\theta^+}{c_v} \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right) \left( \mathbf{A} - \frac{1}{2} (\mathbf{B} + \mathbf{D}) \right)^T$  is also positive definite, the roots of (8.1.21) are positive. Thus,  $p_{n(1)}$  and  $p_{n(2)}$ , like  $u_{n(1)}$  and  $u_{n(2)}$ , are characteristic speeds. The same is true for  $p_0$ . The same kind of argument that yielded (7.2.114) can be used to establish

$$p_{n(1)}^2 \ge p_0^2 \ge p_{n(2)}^2 \tag{8.2.21}$$

As was mentioned in Section 7.2 in equation (7.2.118), another set of inequalities on can establish are

$$p_{n(1)}^2 \ge u_{n(1)}^2 \ge p_{n(2)}^2 \ge u_{n(2)}^2$$
(8.2.22)

and

$$p_0^2 \ge u_0^2 \tag{8.2.23}$$

These inequalities follow from the definitions (7.2.50), (7.2.51), (8.1.21) and (8.1.22). It is useful to note in passing that the definition (8.1.22) of  $p_0^2$  can be replaced by

$$p_0^2 = \frac{\operatorname{tr}\left(\left(\mathbf{Q} + \frac{\theta^+}{c_v^+} \mathbf{A} \mathbf{A}^T\right) \mathbf{E}\right)}{\operatorname{tr}\left((\operatorname{adj} \mathbf{M}) \mathbf{E}\right)}$$
(8.2.24)

The dispersion relation (8.2.9) gives the wave number  $k(\omega)$  in terms of the frequency,  $\omega$ , six characteristic wave speeds squared,  $p_{n(1)}^2, p_{n(2)}^2, p_{n(2)}^2, u_{n(1)}^2, u_{n(2)}^2$  and  $u_0^2$ , two characteristic frequencies  $\omega_0$  and  $\omega_{\xi}$  and the dimensional ratio  $\frac{\kappa}{\overline{c_v}}$ . It is useful to isolate a third characteristic frequency  $\omega_{\kappa}$  defined by

$$\frac{1}{\omega_{\kappa}} = \frac{\kappa}{\overline{c}_{\nu} u_{n(2)}^2}$$
(8.2.25)

The velocity squared  $u_{n(2)}^2$  in (8.2.25) can be replaced by any of the other characteristic velocities. Given the definition (8.2.25), the dispersion relation (8.2.9) takes the form

$$u_{n(2)}^{2}k^{2}\left(\left(u_{n(1)}^{2}k^{2}-\omega^{2}\right)\left(u_{n(2)}^{2}k^{2}-\omega^{2}\right)-i\omega\omega_{0}\left(u_{0}^{2}k^{2}-\omega^{2}\right)\right) -i\omega\omega_{\kappa}\left(\left(p_{n(1)}^{2}k^{2}-\omega^{2}\right)\left(p_{n(2)}^{2}k^{2}-\omega^{2}\right)-i\omega\omega_{\xi}\left(p_{0}^{2}k^{2}-\omega^{2}\right)\right)=0$$
(8.2.26)

It is this third order polynomial in the wave number squared. This equation determines the wave number  $k(\omega)$  in terms of the frequency,  $\omega$ , the six characteristic wave speeds squared,  $p_{n(1)}^2, p_{n(2)}^2, p_0^2, u_{n(1)}^2, u_{n(2)}^2$  and  $u_0^2$  and the three characteristic frequencies  $\omega_0$ ,  $\omega_{\xi}$  and  $\omega_{\kappa}$ .

# 8.3 High Frequency Approximation

It is possible to gain insight into the behavior of the plane harmonic waves as defined by (8.2.1) and (8.2.2) by examining high and low frequency approximations to the dispersion relation. In the high frequency case, we look for approximations of the form

$$\frac{k^2}{\omega^2} = \alpha_0 + \alpha_1 \frac{1}{\omega} + \alpha_2 \frac{1}{\omega^2} + O\left(\frac{1}{\omega^3}\right)$$
(8.3.1)

where the coefficients  $\alpha_0, \alpha_1$  and  $\alpha_3$  are to be determined. The calculation of these coefficients can be tedious if done by hand.<sup>2</sup> Essentially what one does is to write the dispersion relationship (8.2.26) in the form

<sup>&</sup>lt;sup>2</sup> The calculations of the coefficients  $\alpha_0, \alpha_1$  and  $\alpha_3$  are greatly facilitated by use of a symbolic manipulator such as that found in Matlab or in Maple.

$$i\frac{\omega_{\kappa}}{\omega} \left( \left( p_{n(1)}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) \left( p_{n(2)}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) - i\frac{\omega_{\xi}}{\omega} \left( p_{0}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) \right) - u_{n(2)}^{2} \frac{k^{2}}{\omega^{2}} \left( \left( u_{n(1)}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) \left( u_{n(2)}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) - i\frac{\omega_{0}}{\omega} \left( u_{0}^{2} \frac{k^{2}}{\omega^{2}} - 1 \right) \right) = 0$$
(8.3.2)

and then utilize the approximation (8.3.1) to derive the equations satisfied by  $\alpha_0, \alpha_1$  and  $\alpha_3$ . The results of this calculation turn out to be

$$\frac{k_{(1)}^{2}}{\omega^{2}} = \frac{1}{u_{n(1)}^{2}} + \frac{i}{u_{n(1)}^{2}} \left( \frac{\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right) \left(u_{n(1)}^{2} - p_{n(2)}^{2}\right)}{u_{n(2)}^{2} \left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)} \right) \frac{\omega_{\kappa}}{\omega} + \frac{i}{u_{n(1)}^{2}} \left( \frac{u_{n(1)}^{2} - u_{0}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} \right) \frac{\omega_{0}}{\omega} + \frac{i}{u_{n(1)}^{2} \left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)} \left(\frac{\omega_{0}}{\omega}\right)^{2} + \frac{i}{\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} \left( \frac{\omega_{0}}{\omega}\right)^{2} - \frac{\left(-\left(p_{n(2)}^{2} + p_{n(1)}^{2}\right)u_{n(1)}^{2}u_{0}^{2} - \left(p_{n(1)}^{2} + p_{n(2)}^{2}\right)\right)\left(u_{n(1)}^{2} + u_{n(2)}^{2} - 2u_{0}^{2}\right)}{u_{n(2)}^{2} \left(u_{n(1)}^{2} - u_{n(1)}^{2}\right)u_{n(1)}^{2}u_{0}^{2} - \left(p_{n(1)}^{2} + p_{n(2)}^{2}\right)u_{n(2)}^{2}u_{0}^{2} + u_{n(1)}^{6} + 2u_{n(1)}^{2}u_{n(2)}^{2}u_{0}^{2} - 3u_{n(1)}^{4}u_{n(2)}^{2}\right)\left(\frac{\omega_{0}\omega_{\kappa}}{\omega^{2}}\right) - \frac{\left(\frac{p_{0}^{2} - u_{n(1)}^{2}}{u_{n(2)}^{2}\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)}\right)\left(\frac{\omega_{\xi}\omega_{\kappa}}{\omega^{2}}\right)}{u_{n(2)}^{2}\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)^{2} - \left(p_{n(1)}^{2} + p_{n(2)}^{2}\right)u_{n(2)}^{2} + 2u_{n(1)}^{2}u_{n(2)}^{2} - u_{n(1)}^{4}\right)\left(\frac{\omega_{0}\omega_{\kappa}}{\omega^{2}}\right) + O\left(\frac{1}{\omega^{3}}\right) - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}}{u_{n(2)}^{4}\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} - \frac{\left(2\left(p_{n(2)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(1)}^{2}\right)\left$$

$$\begin{aligned} \frac{k_{(2)}^{2}}{\omega^{2}} &= \frac{1}{u_{n(2)}^{2}} + \frac{i}{u_{n(2)}^{2}} \left( \frac{\left( p_{n(1)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(2)}^{2} - u_{n(2)}^{2} \right)}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)} \right) \frac{\omega_{\kappa}}{\omega} + \frac{i}{u_{n(2)}^{2}} \left( \frac{u_{0}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} - u_{n(2)}^{2}} \right) \frac{\omega_{0}}{\omega} \\ &- \frac{\left( u_{n(1)}^{2} - u_{0}^{2} \right) \left( u_{0}^{2} - u_{n(2)}^{2} \right)}{\left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{0}}{\omega} \right)^{2} \\ &+ \frac{\left( - \left( p_{n(2)}^{2} + p_{n(1)}^{2} \right) u_{n(1)}^{2} u_{0}^{2} - \left( p_{n(1)}^{2} + p_{n(2)}^{2} \right) \left( u_{n(1)}^{2} + u_{n(2)}^{2} - 2u_{0}^{2} \right)}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{0} \omega_{\kappa}}{\omega^{2}} \right) \\ &+ \frac{\left( \frac{p_{0}^{2} - u_{n(2)}^{2}}{u_{n(1)}^{2} \left( u_{n(1)}^{2} - \left( p_{n(1)}^{2} + p_{n(2)}^{2} \right) u_{n(1)}^{2} u_{n(2)}^{2} - \left( p_{n(2)}^{2} + p_{n(2)}^{2} \right) u_{n(1)}^{2} u_{n(2)}^{2} - 3u_{n(1)}^{2} u_{n(2)}^{4} \right)}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{0} \omega_{\kappa}}{\omega^{2}} \right) \\ &+ \frac{\left( \frac{p_{0}^{2} - u_{n(2)}^{2}}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)} \right) \left( \frac{\omega_{\xi} \omega_{\kappa}}}{\omega^{2}} \right)}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{u_{n(2)}^{2} - \left( p_{n(2)}^{2} + p_{n(2)}^{2} \right) u_{n(1)}^{2} + 2u_{n(1)}^{2} u_{n(2)}^{2} - u_{n(2)}^{4} \right)} \right) \left( \frac{\omega_{\xi} \omega_{\kappa}}}{\omega^{2}} \right) \\ &+ \frac{\left( 2 \left( p_{n(2)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(1)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(1)}^{2} p_{n(2)}^{2} - \left( p_{n(1)}^{2} + p_{n(2)}^{2} \right) u_{n(1)}^{2} + 2u_{n(1)}^{2} u_{n(2)}^{2} - u_{n(2)}^{4} \right)} \right) \left( \frac{\omega_{\xi} \omega_{\kappa}}}{\omega^{2}} \right) \\ &+ \frac{\left( 2 \left( p_{n(2)}^{2} - u_{n(2)}^{2} \right) \left( p_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{\xi} \omega_{\kappa}}}{u_{n(2)}^{2} + 2u_{n(1)}^{2} u_{n(2)}^{2} - u_{n(2)}^{4} \right) \right)} \right) \left( \frac{\omega_{\xi} \omega_{\kappa}}}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{\xi} \omega_{\kappa}}}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{\xi} \omega_{\kappa}}}{u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \right) \left( \frac{\omega_{\xi} \omega_{\kappa}}}{u_{n(2)}^{2}$$

$$\frac{k_{(3)}^2}{\omega^2} = \frac{i}{u_{n(2)}^2} \frac{\omega_{\kappa}}{\omega} + \left(\frac{\omega_{\kappa}\omega_0}{u_{n(2)}^2} - \frac{\omega_{\kappa}\omega_{\xi}}{u_{n(2)}^2} + \frac{\left(p_{n(1)}^2 - u_{n(1)}^2\right) + \left(p_{n(2)}^2 - u_{n(2)}^2\right)}{u_{n(2)}^2} \frac{\omega_{\kappa}}{u_{n(2)}^2}\right) \frac{1}{\omega^2} + O\left(\frac{1}{\omega^3}\right) \quad (8.3.5)$$

for the three squared wave numbers. For simplicity, we shall confine our discussion to the positive roots of these squared wave numbers. The negative roots simply correspond to waves propagating in the negative *x* direction. Each of these three roots will be referred to as a *mode of propagation*. The derivations of (8.3.3) and (8.3.4) have adopted the inequality  $u_{n(1)}^2 > u_{n(2)}^2$  as a strengthened version of (8.2.22).

It follows from (8.3.3) that the phase velocity and the attenuation coefficient for the first mode are
$$q_{(1)} = u_{n(1)} \begin{pmatrix} 1 - u_{n(1)}^{2} - u_{n(1)}^{2} - u_{n(2)}^{2} \end{pmatrix} \begin{pmatrix} \omega_{0} \\ \omega \end{pmatrix}^{2} \\ - \left( p_{n(2)}^{2} + p_{n(1)}^{2} \right) u_{n(1)}^{2} u_{0}^{2} - \left( p_{n(2)}^{2} + p_{n(1)}^{2} \right) u_{n(1)}^{2} u_{0}^{2} - \left( p_{n(2)}^{2} + p_{n(1)}^{2} \right) u_{n(1)}^{2} u_{0}^{2} - \left( p_{n(2)}^{2} + p_{n(2)}^{2} \right) u_{n(2)}^{2} u_{0}^{2} + u_{n(1)}^{6} + 2u_{n(1)}^{2} u_{n(2)}^{2} u_{0}^{2} - 3u_{n(1)}^{4} u_{n(2)}^{2} \right) \\ + u_{n(1)}^{2} \frac{\left( \frac{p_{0}^{2} - u_{n(1)}^{2}}{2u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right) u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}}{2u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{\omega_{0} \omega_{\kappa}}{\omega^{2}} \right) \\ + u_{n(1)}^{2} \frac{\left( \frac{p_{0}^{2} - u_{n(1)}^{2}}{2u_{n(2)}^{2} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)} \right) \left( \frac{\omega_{\varepsilon} \omega_{\kappa}}{\omega^{2}} \right)}{2u_{n(2)}^{4} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} \left( \frac{2\left( p_{n(2)}^{2} - u_{n(1)}^{2} \right) \left( p_{n(1)}^{2} - u_{n(1)}^{2} \right) \left( p_{n(1)}^{2} - p_{n(2)}^{2} - p_{n(1)}^{2} + p_{n(2)}^{2} \right) u_{n(2)}^{2} + 2u_{n(1)}^{2} u_{n(2)}^{2} - u_{n(1)}^{4} \right) \right)}{2u_{n(2)}^{4} \left( u_{n(1)}^{2} - u_{n(2)}^{2} \right)^{3}} + O\left( \frac{1}{\omega_{0}^{3}} \right)$$

$$(8.3.6)$$

$$s_{(1)} = \frac{1}{2u_{n(1)}} \left( \frac{\left(p_{n(1)}^2 - u_{n(1)}^2\right) \left(u_{n(1)}^2 - p_{n(2)}^2\right)}{u_{n(2)}^2 \left(u_{n(1)}^2 - u_{n(2)}^2\right)} \omega_{\kappa} + \frac{\left(u_{n(1)}^2 - u_{0}^2\right)}{\left(u_{n(1)}^2 - u_{n(2)}^2\right)} \omega_{0} \right) + O\left(\frac{1}{\omega}\right)$$
(8.3.7)

Likewise, it follows from (8.3.4) that for the second mode

$$q_{(2)} = u_{n(2)} \begin{pmatrix} \left(u_{n(1)}^{2} - u_{0}^{2}\right)\left(u_{0}^{2} - u_{n(2)}^{2}\right) \\ - \left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3} \\ - \left(\frac{\left(-\left(p_{n(2)}^{2} + p_{n(1)}^{2}\right)u_{n(1)}^{2}u_{0}^{2} - \left(p_{n(1)}^{2}p_{n(2)}^{2}\right)\left(u_{n(1)}^{2} + u_{n(2)}^{2} - 2u_{0}^{2}\right) \\ + 2\left(p_{n(2)}^{2} + p_{n(1)}^{2}\right)u_{n(1)}^{2}u_{n(2)}^{2} - \left(p_{n(1)}^{2} + p_{n(2)}^{2}\right)u_{n(2)}^{2}u_{0}^{2} + u_{n(2)}^{6} + 2u_{n(1)}^{2}u_{n(2)}^{2}u_{0}^{2} - 3u_{n(1)}^{2}u_{n(2)}^{4}\right) \\ - \frac{\left(\frac{p_{0}^{2} - u_{n(2)}^{2}}{2\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)}\right)\left(\frac{\omega_{\xi}\omega_{\kappa}}{\omega^{2}}\right)}{2\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} \\ - \left(\frac{2\left(p_{n(2)}^{2} - u_{n(2)}^{2}\right)\left(p_{n(1)}^{2} - u_{n(2)}^{2}\right)\left(p_{n(1)}^{2}p_{n(2)}^{2} - \left(p_{n(1)}^{2} + p_{n(2)}^{2}\right)u_{n(1)}^{2} + 2u_{n(1)}^{2}u_{n(2)}^{2} - u_{n(2)}^{4}\right)\right)}{2u_{n(2)}^{2}\left(u_{n(1)}^{2} - u_{n(2)}^{2}\right)^{3}} \\ + O\left(\frac{1}{\omega_{0}^{3}}\right)$$

$$(8.3.8)$$

$$s_{(2)} = \frac{1}{2u_{n(2)}} \left( \frac{\left(p_{n(1)}^2 - u_{n(2)}^2\right) \left(p_{n(2)}^2 - u_{n(2)}^2\right)}{u_{n(2)}^2 \left(u_{n(1)}^2 - u_{n(2)}^2\right)} \omega_{\kappa} + \frac{\left(u_0^2 - u_{n(2)}^2\right)}{\left(u_{n(1)}^2 - u_{n(2)}^2\right)} \omega_0 \right) + O\left(\frac{1}{\omega}\right)$$
(8.3.9)

Equations (8.3.6) through (8.3.9) show that the in the high frequency limit, the two phase velocities approach constants and the corresponding attenuation coefficients are positive numbers. The two phase velocities in this limit correspond to the acceleration wave speeds  $u_{n(1)}$  and  $u_{n(2)}$  defined by equation (7.2.50). The attenuation coefficients are positive because of the inequalities (7.2.114) and (8.2.22) along with the thermodynamic results (7.2.4) through (7.2.6). These latter results tell us that  $\omega_{\kappa}$  and  $\omega_0$  are positive numbers. If the attenuation coefficient (8.3.7) is compared to the one in (7.2.119), we see that the term in the bracket of (8.3.7) is the same factor in (7.2.119). A similar correspondence exists between (8.3.9) and (7.2.120). The high frequency correspondence acceleration wave speeds and phase velocities and attenuation coefficients is a standard result. The usual explanation is that the acceleration wave calculation yields a short time or, equivalently, high frequency solution in the neighborhood of the boundary represented by the acceleration wave.

The high frequency approximations for the phase velocity and the attenuation coefficient for the third mode turn out to be

$$q_{(3)} = u_{n(2)} \sqrt{\frac{2\omega}{\omega_{\kappa}}} \left( 1 - \frac{u_{n(2)}^2}{2\omega_{\kappa}} \left( \frac{\omega_{\kappa}\omega_0}{u_{n(2)}^2} - \frac{\omega_{\kappa}\omega_{\xi}}{u_{n(2)}^2} + \frac{\left(p_{n(1)}^2 - u_{n(1)}^2\right) + \left(p_{n(2)}^2 - u_{n(2)}^2\right)}{u_{n(2)}^2} \frac{\omega_{\kappa}}{u_{n(2)}^2} \right) + O\left(\frac{1}{\omega_{\kappa}^{3/2}}\right) (8.3.10)$$

$$s_{(3)} = \frac{1}{u_{n(2)}} \sqrt{\frac{\omega\omega_{\kappa}}{2}} \left( 1 - \frac{u_{n(2)}^2}{2\omega_{\kappa}} \left( \frac{\omega_{\kappa}\omega_0}{u_{n(2)}^2} - \frac{\omega_{\kappa}\omega_{\xi}}{u_{n(2)}^2} + \frac{\left(p_{n(1)}^2 - u_{n(1)}^2\right) + \left(p_{n(2)}^2 - u_{n(2)}^2\right)}{u_{n(2)}^2} \frac{\omega_{\kappa}^2}{u_{n(2)}^2} \right) + O\left(\frac{1}{\frac{\omega_{\kappa}\omega_0}{\omega_{\kappa}^2}}\right) (8.3.11)$$

The fact that the phase velocity depends upon the frequency, even in the high frequency limit, illustrates that the third mode is dispersive.

#### 8.4 Low Frequency Approximation

In the low frequency case, we look for approximations of the form

$$k^{2} = \alpha_{0} + \alpha_{1}\omega + \alpha_{2}\omega^{2} + \alpha_{3}\omega^{3} + O(\omega^{4})$$
(8.4.1)

where the coefficients  $\alpha_0, \alpha_1, \alpha_2$  and  $\alpha_3$  are to be determined. Like the high frequency case, the calculation details are tedious. They calculations begin with the substitution of (8.4.1) into (8.2.26). The next step is to solve for the unknown coefficients. The results for the three wave numbers squared turn out to be<sup>3</sup>

$$k_{(1)}^{2} = \frac{1}{p_{0}^{2}}\omega^{2} + \left(\frac{\omega_{0}}{\omega_{\kappa}\omega_{\xi}}u_{n(2)}^{2}\left(p_{0}^{2} - u_{0}^{2}\right) + \frac{1}{\omega_{\xi}}\left(p_{n(1)}^{2} - p_{0}^{2}\right)\left(p_{0}^{2} - p_{n(2)}^{2}\right)\right)\frac{i}{p_{0}^{6}}\omega^{3} + O\left(\omega^{4}\right) \quad (8.4.2)$$

$$k_{(1)}^{2} = \frac{1}{u_{0}^{2}}\omega^{2} + \left(u_{n(1)}^{2} - u_{0}^{2}\right)\left(u_{0}^{2} - u_{n(2)}^{2}\right)\frac{i}{\omega_{\xi}u_{0}^{6}}\omega^{3} + O\left(\omega^{4}\right)$$

and

$$k_{(2)}^{2} = i \frac{u_{0}^{2}}{u_{n(1)}^{2} u_{n(2)}^{2}} \omega_{\xi} \omega + \frac{\left(u_{0}^{2} - u_{n(2)}^{2}\right) u_{n(1)}^{2} + u_{0}^{2} u_{n(2)}^{2}}{u_{0}^{2} u_{n(2)}^{2} u_{n(2)}^{2}} \omega^{2} + i \frac{\left(u_{n(1)}^{2} - u_{0}^{2}\right) \left(u_{0}^{2} - u_{n(2)}^{2}\right)}{\omega_{\xi} u_{0}^{6}} \omega^{3} + O\left(\omega^{4}\right),$$

Respectively.

<sup>&</sup>lt;sup>3</sup> It is important to note that the result (8.4.2) assumes that  $\omega_{\kappa}$  and  $\omega_{\xi}$  are nonzero. The case where  $\omega_{\kappa}$  is zero, as shown in the definition (8.2.25), corresponds to an infinite conductivity  $\kappa$ . This case, as equation (8.1.8) shows, forces the temperature to obey  $\frac{\partial^2 \theta}{\partial X^2} = 0$ . With suitable boundary conditions, we can conclude that the case  $\omega_{\kappa} = 0$  corresponds to isothermal poroelasticity. In any case, for the case  $\omega_{\kappa} = 0$ , the results (8.4.2) through (8.4.3) are replaced by

$$k_{(2)}^{2} = \frac{i}{2u_{n(1)}^{2}u_{n(2)}^{4}} \bigg( \omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} - \sqrt{(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2})^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}} \bigg) \omega_{(8.4.3)} + O(\omega^{2})$$

$$k_{(3)}^{2} = \frac{i}{2u_{n(1)}^{2}u_{n(2)}^{4}} \left( \omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} + \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2}\right)^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}} \right) \omega_{(8.4.4)} + O\left(\omega^{2}\right)$$

where, in order to avoid long and uninteresting formulas, we have truncated the expansions for the second and third modes at terms linear in the frequency  $\omega$ . The first mode has been taken to higher order terms because it is nondispersive in the low frequency limit. It readily follows from (8.4.2) that

$$q_{(1)} = p_0 + O(\omega^2)$$
(8.4.5)

and the corresponding attenuation coefficient is

$$s_{(1)} = \frac{\omega^2}{2p_0} \left( \frac{\omega_0}{\omega_\kappa \omega_\xi} \frac{u_{n(2)}^2 \left( p_0^2 - u_0^2 \right)}{p_0^4} + \frac{1}{\omega_\xi} \frac{\left( p_{n(1)}^2 - p_0^2 \right) \left( p_0^2 - p_{n(2)}^2 \right)}{p_0^4} \right) + O(\omega^3)$$
(8.4.6)

The corresponding quantities for the second and third mode are

$$q_{(2)} = \frac{2u_{n(1)}u_{n(2)}^{2}\sqrt{\omega}}{\sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} - \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2}\right)^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}\right)}} + O(\omega) \quad (8.4.7)$$

$$s_{(2)} = \frac{\sqrt{\omega}}{2u_{n(1)}u_{n(2)}^{2}} \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} - \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2}\right)^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}}\right)} + O(\omega)$$
(8.4.8)

$$q_{(3)} = \frac{2u_{n(1)}u_{n(2)}^{2}\sqrt{\omega}}{\sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} + \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2}\right)^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}}\right)} + O(\omega) \quad (8.4.9)$$

and

\_ \

$$s_{(3)} = \frac{\sqrt{\omega}}{2u_{n(1)}u_{n(2)}^{2}}\sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2} + \sqrt{\left(\omega_{0}u_{0}^{2}u_{n(2)}^{2} + \omega_{\kappa}p_{n(1)}^{2}p_{n(2)}^{2}\right)^{2} - 4\omega_{\xi}\omega_{\kappa}u_{n(1)}^{2}u_{n(2)}^{4}p_{0}^{2}}\right)} + O(\omega)$$
(8.4.10)

## 8.5 High and Low Frequency Approximation for Nonconductors

The results summarized in Section 8.3 and 8.4 assume the result (8.1.20), i.e., the thermal conductivity is positive. In this Section, we shall briefly record the results for the dispersion relation in the case of a nonconductor, i.e., when

$$\kappa = 0 \tag{8.5.1}$$

In terms of the characteristic frequencies, the nonconductor case corresponds to

$$\frac{1}{\omega_{\kappa}} = 0 \tag{8.5.2}$$

and

$$\omega_0 = \omega_{\xi} \tag{8.5.3}$$

The result (8.5.3) follows from the definitions (8.1.18) and (8.1.19) along with the thermodynamic result (7.2.121). It follows from (8.2.26) that in this limiting case the dispersion relation  $k(\omega)$  is a solution of

$$\left(p_{n(1)}^{2}k^{2}-\omega^{2}\right)\left(p_{n(2)}^{2}k^{2}-\omega^{2}\right)-i\omega\omega_{\xi}\left(p_{0}^{2}k^{2}-\omega^{2}\right)=0$$
(8.5.4)

The high and low frequency approximations to this quadratic in  $k^2$  are as follows:

High Frequency

$$\frac{k_{(1)}^2}{\omega^2} = \frac{1}{p_{n(1)}^2} + \frac{i}{p_{n(1)}^2} \left( \frac{p_{n(1)}^2 - p_0^2}{p_{n(1)}^2 - p_{n(2)}^2} \right) \frac{\omega_0}{\omega} + \frac{\left(p_{n(1)}^2 - p_0^2\right) \left(p_0^2 - p_{n(2)}^2\right)}{\left(p_{n(1)}^2 - p_{n(2)}^2\right)^3} \left(\frac{\omega_0}{\omega}\right)^2 + O\left(\frac{1}{\omega^3}\right)$$
(8.5.5)

$$\frac{k_{(2)}^2}{\omega^2} = \frac{1}{p_{n(2)}^2} + \frac{i}{p_{n(2)}^2} \left( \frac{p_0^2 - p_{n(2)}^2}{p_{n(1)}^2 - p_{n(2)}^2} \right) \frac{\omega_0}{\omega} - \frac{\left(p_{n(1)}^2 - p_0^2\right) \left(p_0^2 - p_{n(2)}^2\right)}{\left(p_{n(1)}^2 - p_{n(2)}^2\right)^3} \left(\frac{\omega_0}{\omega}\right)^2 + O\left(\frac{1}{\omega^3}\right)$$
(8.5.6)

Low Frequency

$$k_{(1)}^{2} = \frac{1}{p_{0}^{2}}\omega^{2} + \left(p_{n(1)}^{2} - p_{0}^{2}\right)\left(p_{0}^{2} - p_{n(2)}^{2}\right)\frac{i}{\omega_{\xi}p_{0}^{6}}\omega^{3} + O\left(\omega^{4}\right)$$
(8.5.7)

$$k_{(2)}^{2} = i \frac{p_{0}^{2}}{p_{n(1)}^{2} p_{n(2)}^{2}} \omega_{\xi} \omega + \frac{\left(p_{0}^{2} - p_{n(2)}^{2}\right) p_{n(1)}^{2} + p_{0}^{2} p_{n(2)}^{2}}{p_{0}^{2} p_{n(1)}^{2} p_{n(2)}^{2}} \omega^{2} + i \frac{\left(p_{n(1)}^{2} - p_{0}^{2}\right) \left(p_{0}^{2} - p_{n(2)}^{2}\right)}{\omega_{\xi} p_{0}^{6}} \omega^{3} + O\left(\omega^{4}\right) (8.5.8)$$

Equations (8.5.5), (8.5.6) and (8.5.7) are nondispersive modes in the high and low frequency range respectively. The phase velocities are  $p_{n(1)}$ ,  $p_{n(2)}$  and  $p_0$ , respectively. It is possible to confirm that the attenuation coefficients in these cases are positive.

# **8.6 Phase Velocities**

The results in Sections 8.3 through 8.5 allow us to identify the phase velocities associated with the various high and low frequency approximations. In this way, we gain physical insight into the meaning of the six constants  $u_0^2$ ,  $u_{n(1)}^2$ ,  $p_{0,2}^2$ ,  $p_{0,1}^2$  and  $p_{n(2)}^2$  that appear in all of our formulas. The following table shows when these constants correspond to phase velocities for non dispersive modes of plane progressive waves.

### **Table of Phase Velocities**

	High Frequency	Low Frequency
	Approximation	Approximation
$0 < \omega_{\kappa}, \omega_{\xi}, \omega_0 < \infty$	$u_{n(1)}$ and $u_{n(2)}$	$p_0$
$\frac{1}{\omega_{\kappa}} = 0, 0 < \omega_{\xi}, \omega_0 < \infty$	$p_{n(1)}$ and $p_{n(2)}$	$P_0$
$\omega_{\kappa} = 0, 0 < \omega_{\xi}, \omega_0 < \infty$	$u_{n(1)}$ and $u_{n(2)}$	<i>u</i> <sub>0</sub>

## **8.7 Numerical Example**

It is interesting to examine the solutions to (8.2.26) for a hypothetical porous elastic material defined by specific numerical values. In this example, we shall make the choices

$$p_{n(1)} = 4500 \text{ m/sec}$$
  
 $u_{n(1)} = 4000 \text{ m/sec}$   
 $p_0 = 3500 \text{ m/sec}$   
 $u_0 = 3000 \text{ m/sec}$   
 $p_{n(2)} = 2500 \text{ m/sec}$   
 $u_{n(2)} = 2000 \text{ m/sec}$ 

These numbers have been selected such that the inequalities (7.2.114), (8.2.21), (8.2.22) and (8.2.23) are obeyed. In addition, the numbers selected are in the range of values displayed in Tables 1 and 2 of Chapter 7.

We also need typical numbers for the three characteristic frequencies  $\omega_0, \omega_{\xi}$  and  $\omega_{\kappa}$ . It follows from the definitions (8.1.18) and (8.1.19) that

$$\omega_{\xi} \ge \omega_0 \tag{8.7.2}$$

Consistent with this inequality, in our numerical example, we shall initially make the choices

$$\omega_0 = 10(10)^7 \text{ rad/sec}$$

$$\omega_{\xi} = 15(10)^7 \text{ rad/sec} \qquad (8.7.3)$$

$$\omega_{\kappa} = 40(10)^7 \text{ rad/sec}$$

These numbers reflect the relationship (8.7.2) and are of the order of magnitude that one would get for the materials shown in Tables 1 and 2 of Section 6.2. We shall also allow the characteristic frequency  $\omega_{\kappa}$  to be zero (infinite conductivity  $\kappa$ ) and infinite (zero conductivity  $\kappa$ ).

We shall first look at plots of the phase velocity for the hypothetical poroelastic material defined by (8.7.1) and (8.7.3). The three modes displayed in the figure below correspond to the three modes ordered as in Section 8.3. The resulting plot is

Phase Velocity vs. Frequency for Hypothetical Poroelastic Material



As the approximate results in Sections 8.3 and 8.4 illustrated, Figure 1 shows that the first mode, for low frequency, has a phase velocity  $p_0$  which grows, for high frequencies, to the phase velocity  $u_1$ . The second mode is dispersive for low frequencies but approaches the nondispersive phase velocity  $u_2$  for high frequencies.

The attenuation coefficients for these three modes plot as follows



Attenuation Coefficient vs. Frequency for Hypothetical Poroelastic Material

Figure 2 illustrates the analytical results contained in (8.3.7) and (8.3.9) that the attenuation coefficients for modes 1 and 2 approach constants for large frequencies.

The two limiting cases mentioned in Sections 8.4 and 8.5, namely, the cases of infinite and zero thermal conductivity, yield the following two results for the phase velocities.



# Phase Velocity vs. Frequency for Hypothetical Poroelastic Material (Zero Thermal Conductivity)

and

Phase Velocity vs. Frequency for Hypothetical Poroelastic Material (Infinite Thermal Conductivity)



These curves reflect the phase velocity results summarized in Section 8.6.

# References

- 1. Bowen, R. M., Plane Progressive Waves in a Heat Conducting Fluid Saturated Porous Material with Relaxing Porosity, *Acta Mechanica*, **46**, 189-206, 1983.
- 2. BOWEN, R. M., AND K. REINICKE, Plane Progressive Waves in a Binary Mixture of Linear Elastic Materials, J. Applied Mech., 45, 493-499, 1978.
- 3. BIOT, M. A., Theory of Propagation of Elastic Waves in a Fluid Saturated Porous Solid. I. Low Frequency Range, *J. Acoust. Soc. Amer.*, **28**, 168-178, 1956.
- 4. CRAINE, R. E., AND A. F. JOHNSON, Acoustical Wave Propagation in a Binary Mixture of Inviscid Fluids, J. Sound Vibrations, **20**, 191-207, 1972.
- 5. CHADWICK, P., Thermoelasticity, the Dynamic Theory in Progress in Solid Mechanics, Vol. 1, (Sneddon, I., Hill, R., eds.), *North Holland*, Amsterdam, 1960.

# **Boundary Initial Value Problems: Inertia Neglected**

In Chapter 5, we examined the field and constitutive equations in the special case where it was acceptable to neglect the inertia of the fluid and, later, the inertia of the solid. In this chapter, we shall adopt the field equations where both of these inertias are neglected and, in addition, we assume the porous elastic material is isothermal. Given these assumptions, we shall formulate rather simple boundary initial value problems and indicate how they might be solved. One of the historically important problems is one first solved by Biot [Ref. 1]. We shall generate this solution and some of its generalizations. The formalism for this chapter is drawn, in part, from the formulation given by Bowen [Ref. 2].

## 9.1 Governing Partial Differential Equations

We begin this discussion with the constitutive equations and field equations given in Section 4.2. Equations (4.2.1) through (4.2.5, repeated here, are

$$\frac{\varphi_f - \varphi_f^+}{\varphi_f^+} = \frac{\lambda_{sb}}{\varphi_f^+ K_f} \operatorname{tr} \mathbf{E}_s + \left(1 - \frac{\lambda_{ff}}{\varphi_f^+ K_f}\right) ((\rho_f - \rho_f^+) / \rho_f^+) - \frac{\overline{\tau}_b}{\varphi_f^+ K_f} (\theta - \theta^+)$$
(9.1.1)

$$\Psi_{I} = \alpha (\theta - \theta^{+}) - \frac{1}{2} \frac{\overline{c}_{v}}{\theta^{+}} (\theta - \theta^{+})^{2} - \overline{\tau}_{s} (\theta - \theta^{+}) (\operatorname{tr} \mathbf{E}_{s}) + \overline{\tau}_{f} (\theta - \theta^{+}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \overline{\lambda}_{sf} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ff} \left( (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} \right)^{2} + \mu_{ss} \operatorname{tr} (\mathbf{E}_{s} \mathbf{E}_{s}),$$

$$(9.1.2)$$

$$\rho\eta = -\alpha + \frac{\overline{c}_{v}}{\theta^{+}}(\theta - \theta^{+}) + \overline{\tau}_{s}(\operatorname{tr} \mathbf{E}_{s}) - \overline{\tau}_{f}(\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+}$$
(9.1.3)

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\mathbf{T}_{I} + \varphi_{f}^{+}P_{f}\mathbf{I}) + \alpha(\theta - \theta^{+})\mathbf{I} = -\overline{\lambda}_{ss}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \overline{\lambda}_{sf}((\rho_{f} - \rho_{f}^{+})/\rho_{f}^{+})\mathbf{I} + \overline{\tau}_{s}(\theta - \theta^{+})\mathbf{I}$$
(9.1.4)

and

$$\varphi_f^+ P_f = \overline{\tau}_f \left(\theta - \theta^+\right) - \overline{\lambda}_{sf} \left(\operatorname{tr} \mathbf{E}_s\right) + \overline{\lambda}_{ff} \left(\rho_f - \rho_f^+\right) / \rho_f^+$$
(9.1.5)

Equation (4.2.9) is

$$\mathbf{T}_{\mathrm{I}} + \varphi_{f}^{+} P_{f} \mathbf{I} = \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s}) \mathbf{I} + 2\mu_{ss} \mathbf{E}_{s} - \overline{\lambda}_{sf} ((\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+}) \mathbf{I} - (\overline{\tau}_{s} - \alpha)(\theta - \theta^{+}) \mathbf{I}$$
(9.1.6)

The field equations which follow from these constitutive equations are equations (4.2.14), (4.2.15) and (4.2.16). These equations, repeated here, are

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{s} - \overline{\tau}_{s} + \gamma) \operatorname{GRAD} \theta - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right)$$
(9.1.7)

$$\rho_{f}^{+} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} = \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) + (\alpha_{f} - \overline{\tau}_{f} - \gamma) \operatorname{GRAD} \theta$$

$$-\xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right)$$
(9.1.8)

and

$$\overline{c}_{v}\frac{\partial\theta}{\partial t} + \theta^{+} \left( (\overline{\tau}_{s} - \alpha_{s} + \varsigma/\theta^{+}) \operatorname{Div}\frac{\partial \mathbf{w}_{s}}{\partial t} + (\overline{\tau}_{f} - \alpha_{f} - \varsigma/\theta^{+}) \operatorname{Div}\frac{\partial \mathbf{w}_{f}}{\partial t} \right)$$

$$= \kappa \operatorname{Div}(\operatorname{GRAD}\theta)$$
(9.1.9)

The material constants  $\kappa$ ,  $\xi$ ,  $\gamma$  and  $\zeta$  obey (4.2.20), (4.2.21) and (4.2.22). These equations, repeated here, are

$$\kappa \ge 0 \tag{9.1.10}$$

and

$$\kappa \xi / \theta^+ \ge \frac{1}{4} (\gamma + \varsigma / \theta^+)^2 \tag{9.1.11}$$

It follows from (9.1.10) and (9.1.11) that

$$\xi \ge 0 \tag{9.1.12}$$

As has been mentioned, the coefficient  $\kappa$  is the *thermal conductivity*. The coefficient  $\xi$  is called the *drag coefficient*.

The special case of the above equations we wish to study in this chapter are those that neglect the inertia of the fluid and the solid and for which the temperature is a constant. The

isothermal assumption can be formally obtained from (9.1.9) by allowing  $\kappa \to \infty$ . In this limit, the temperature is a solution of

$$Div(GRAD\theta) = 0 \tag{9.1.13}$$

With suitable boundary conditions, the solution of this partial differential equation is

$$\theta = \Theta(\mathbf{X}, t) = \theta^+ \tag{9.1.14}$$

Given (9.1.14) and our assumption to neglect the inertia of the fluid and the solid, the two equations of motion (9.1.7) and (9.1.8) reduce to

$$(\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right) = \mathbf{0} (9.1.15)$$

$$\overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) - \xi \left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) = \mathbf{0}$$
(9.1.16)

The constitutive equations appropriate to this isothermal special case are special cases of (9.1.5) and (9.1.6). We shall write these special cases as

$$\varphi_f^+ P_f = -\overline{\lambda}_{sf} \operatorname{tr} \mathbf{E}_s - \overline{\lambda}_{ff} \operatorname{tr} \mathbf{E}_f$$
(9.1.17)

and

$$\mathbf{T}_{\mathrm{I}} + \varphi_{f}^{+} P_{f} \mathbf{I} = \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s}) \mathbf{I} + 2\mu_{ss} \mathbf{E}_{s} + \overline{\lambda}_{sf} (\operatorname{tr} \mathbf{E}_{f}) \mathbf{I}$$
(9.1.18)

where (3.39) and the identity

$$\operatorname{tr} \mathbf{E}_f = \operatorname{Div} \mathbf{w}_f \tag{9.1.19}$$

have been used.

# 9.2 Some Properties of the Space Part Of The Operator

It is helpful when we talk about solutions to boundary initial value problems based upon the partial differential equations (9.1.15) and (9.1.16) to introduce certain properties of the space part of the operator that defines these equations. In particular, we can define a formal linear operator L by the equation

.

$$L\mathbf{w} = -\begin{bmatrix} \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) \\ (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) \end{bmatrix}$$
(9.2.1)

where

$$\mathbf{w} = \begin{bmatrix} \mathbf{w}_f \\ \mathbf{w}_s \end{bmatrix}$$
(9.2.2)

The operator L is defined on the set of twice differentiable functions defined on an open subset of a spatial region  $\mathscr{V}$ . The definitions (9.2.1) and (9.2.2) adopt a mixed vector-matrix notation. If it aids ones thinking, the vectors in these two equations can be thought of as column matrices. This view allows (9.2.2), for example, to be seen as a column matrix of dimension  $6 \times 1$ . The way this notation has been adopted will be more or less obvious in the manipulation below.

If we define the inner produce by

$$\langle \mathbf{w}, \mathbf{v} \rangle = \int_{\boldsymbol{v}} (\mathbf{w}_f \cdot \mathbf{v}_f + \mathbf{w}_s \cdot \mathbf{v}_s) d\upsilon$$
 (9.2.3)

Then it is possible to show from (9.2.1) that

$$\langle \mathbf{w}, L\mathbf{v} \rangle = \oint_{\partial \mathbf{v}} \left( \left( -\overline{\lambda}_{ff} \operatorname{Div} \mathbf{v}_{f} - \overline{\lambda}_{sf} \operatorname{Div} \mathbf{v}_{s} \right) \mathbf{w}_{f} + \left( -\overline{\lambda}_{sf} \left( \operatorname{Div} \mathbf{v}_{f} \right) \mathbf{I} - \overline{\lambda}_{ss} \left( \operatorname{Div} \mathbf{v}_{s} \right) \mathbf{I} - \mu_{ss} \left( \operatorname{GRAD} \mathbf{v}_{s} + \left( \operatorname{GRAD} \mathbf{v}_{s} \right)^{T} \right) \right) \mathbf{w}_{s} \right) \cdot d\mathbf{s} + \int_{\mathbf{v}} \left( \overline{\lambda}_{ff} \operatorname{Div} \mathbf{w}_{f} \operatorname{Div} \mathbf{v}_{f} + \overline{\lambda}_{sf} \left( \operatorname{Div} \mathbf{w}_{f} \operatorname{Div} \mathbf{v}_{s} + \operatorname{Div} \mathbf{w}_{s} \operatorname{Div} \mathbf{v}_{f} \right) + \overline{\lambda}_{ss} \operatorname{Div} \mathbf{w}_{s} \operatorname{Div} \mathbf{v}_{s} \right)^{T} \right) (9.2.4) + \frac{1}{2} \mu_{ss} \operatorname{tr} \left( \left( \operatorname{GRAD} \mathbf{w}_{s} + \left( \operatorname{GRAD} \mathbf{w}_{s} \right)^{T} \right) \left( \operatorname{GRAD} \mathbf{v}_{s} + \left( \operatorname{GRAD} \mathbf{v}_{s} \right)^{T} \right) \right) \right)$$

Equation (9.2.4) is the first Green's identity for the operator L. A more convenient form of this equation is

$$\langle \mathbf{w}, L\mathbf{v} \rangle = \oint_{\partial \mathbf{v}} \left( \varphi_f^* P_f(\mathbf{v}) \mathbf{w}_f - \left( \mathbf{T}_I(\mathbf{v}) + \varphi_f^* P_f(\mathbf{v}) \mathbf{I} \right) \mathbf{w}_s \right) \cdot d\mathbf{s}$$

$$+ \int_{\mathbf{v}} \left( \overline{\lambda}_{ff} \operatorname{Div} \mathbf{w}_f \operatorname{Div} \mathbf{v}_f + \overline{\lambda}_{sf} \left( \operatorname{Div} \mathbf{w}_f \operatorname{Div} \mathbf{v}_s + \operatorname{Div} \mathbf{w}_s \operatorname{Div} \mathbf{v}_f \right) + \overline{\lambda}_{ss} \operatorname{Div} \mathbf{w}_s \operatorname{Div} \mathbf{v}_s \right)$$

$$+ \int_{\mathbf{v}} \left( \frac{1}{2} \mu_{ss} \operatorname{tr} \left( \left( \operatorname{GRAD} \mathbf{w}_s + \left( \operatorname{GRAD} \mathbf{w}_s \right)^T \right) \left( \operatorname{GRAD} \mathbf{v}_s + \left( \operatorname{GRAD} \mathbf{v}_s \right)^T \right) \right) \right)$$

$$(9.2.5)$$

where, from (9.1.17),  $P_f(\mathbf{v})$  is the pore pressure calculated from the pair  $\mathbf{v} = \begin{bmatrix} \mathbf{v}_f \\ \mathbf{v}_s \end{bmatrix}$  and , from

(9.1.18),  $\mathbf{T}_{I}(\mathbf{v})$  is the stress calculated from the same pair. It follows from (9.2.5) that the second Green's identity for the operator *L* is

$$\langle \mathbf{w}, L\mathbf{v} \rangle - \langle L\mathbf{w}, \mathbf{v} \rangle = \oint_{\partial \mathscr{V}} \begin{pmatrix} \varphi_f^+ (P_f(\mathbf{v})\mathbf{w}_f - P_f(\mathbf{w})\mathbf{v}_f) \\ - (\mathbf{T}_I(\mathbf{v}) + \varphi_f^+ P_f(\mathbf{v})\mathbf{I})\mathbf{w}_s + (\mathbf{T}_I(\mathbf{w}) + \varphi_f^+ P_f(\mathbf{w})\mathbf{I})\mathbf{v}_s \end{pmatrix} \cdot d\mathbf{s} \qquad (9.2.6)$$

or, equivalently,

$$\langle \mathbf{w}, L\mathbf{v} \rangle - \langle L\mathbf{w}, \mathbf{v} \rangle = \oint_{\partial \mathbf{v}} \begin{pmatrix} \varphi_f^+ (P_f(\mathbf{v}) (\mathbf{w}_f - \mathbf{w}_s) - P_f(\mathbf{w}) (\mathbf{v}_f - \mathbf{v}_s) \\ -\mathbf{T}_I(\mathbf{v}) \mathbf{w}_s + \mathbf{T}_I(\mathbf{w}) \mathbf{v}_s \end{pmatrix} \cdot d\mathbf{s}$$
(9.2.7)

Equation (9.2.7) allow us to characterize those circumstances where the operator L is self adjoint, i.e., when

$$\langle \mathbf{w}, L\mathbf{v} \rangle = \langle L\mathbf{w}, \mathbf{v} \rangle$$
 (9.2.8)

The following fives cases yield this result:

- *i)* The fluid and solid displacements vanish on  $\partial \mathscr{V}$ .
- *ii)* The stress vector  $\mathbf{T}_{\mathbf{l}}\mathbf{n}$  vanishes on  $\partial \mathscr{V}$  and the fluid and solid displacements are equal.
- *iii)* The stress vector  $\mathbf{T}_{\mathbf{I}}\mathbf{n}$  vanishes on  $\partial \mathscr{V}$  and the pore pressure  $P_f$  vanishes on  $\partial \mathscr{V}$ .
- *iv*) The solid displacement vanishes on  $\partial \mathscr{V}$  and the pore pressure  $P_f$  vanishes on  $\partial \mathscr{V}$ .
- v) Combinations of *i*) through *iv*), where the cases are prescribed on different parts of  $\partial \mathscr{V}$ .

where **n** is the outward drawn unit normal to the surface  $\partial \mathscr{V}$ .

From a formal stand point, each of these five choices defines a different operator. The operator is defined formally by (9.2.1) and by its domain, i.e. by the set of functions in its domain. Each of the above five choices represent possible different choices of the domain of L.

As explained in Section 4.3, the above list contains the usual traction and displacement boundary conditions from elasticity. The conditions on the pore pressure and the relative displacement of the fluid and solid correspond to whether or not the boundary is *pervious* or *impervious* to the flow of fluid across the boundary. If one prescribes the relative displacement to be zero, as in *ii*), then the fluid is not allowed to flow across the boundary. If instead, one prescribes the pore pressure, relative motion is allowed and, thus, fluid can flow across the boundary.

Given boundary conditions taken from the list *i*) through *v*) above, it follows from (9.2.5) that

$$\langle \mathbf{w}, L\mathbf{w} \rangle = \int_{\mathbf{v}} \left( \frac{\overline{\lambda}_{ff} \left( \text{Div} \, \mathbf{w}_{f} \right)^{2} + 2\overline{\lambda}_{sf} \, \text{Div} \, \mathbf{w}_{f} \, \text{Div} \, \mathbf{w}_{s} + \overline{\lambda}_{ss} \left( \text{Div} \, \mathbf{w}_{s} \right)^{2} \right) + \frac{1}{2} \mu_{ss} \, \text{tr} \left( \left( \text{GRAD} \, \mathbf{w}_{s} + \left( \text{GRAD} \, \mathbf{w}_{s} \right)^{T} \right)^{2} \right) \right)$$
(9.2.9)

The inequalities (4.3.5), (4.2.6) and (4.3.7) allow us to conclude that for the operators defined by the categories of boundary conditions i) through v), the operator L is positive semi definite, i.e.,

$$\langle \mathbf{w}, L\mathbf{w} \rangle \ge 0 \tag{9.2.10}$$

As usual, an eigenvalue of the self adjoint positive semi definite operator L is a nonzero function **u** that obeys

$$L\mathbf{u} = \lambda \mathbf{u} \tag{9.2.11}$$

The quantity  $\lambda$  is the eigenvalue associated with the eigenfunction **u**. Because the operator is self adjoint, the eigenvalues are necessarily real numbers. Because the operator is positive semi definite, the eigenvalues are nonnegative real numbers. It is also a standard result that when one has two different eigenvalues, the corresponding eigenvectors are orthogonal with respect to the inner product defined by (9.2.3).

### 9.3 Boundary Initial Value Problems

Consider next, a boundary initial value problem for the partial differential equations (9.1.15) and (9.1.16) as follows:

**Boundary Conditions:** 

$$\mathbf{T}_{I}(\mathbf{w})\mathbf{n} = -\mathbf{s}(\mathbf{X},t) \quad \text{for} \quad (\mathbf{X},t) \in \partial \mathscr{V}_{1} \times (0,\infty)$$
(9.3.1)

$$P_{f}(\mathbf{w}) = r(\mathbf{X}, t) \qquad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{1} \times (0, \infty)$$
(9.3.2)

$$\mathbf{w} = \mathbf{k} (\mathbf{X}, t) \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_2 \times (0, \infty)$$
(9.3.3)

Initial Condition:

$$\mathbf{w}(\mathbf{X},0) = \mathbf{f}(\mathbf{X})$$
 for  $\mathbf{X} \in \mathscr{V}$  (9.3.4)

where the boundary  $\partial \mathscr{V}$  of the volume  $\mathscr{V}$  consists of two parts  $\partial \mathscr{V}_1$  and  $\partial \mathscr{V}_2$ . On the part  $\partial \mathscr{V}_1$  the stress vector  $\mathbf{T}_I \mathbf{n}$  and the pore pressure  $P_f$  are prescribed by the functions  $\mathbf{s}(\mathbf{X},t)$  and  $r(\mathbf{X},t)$ , respectively. On the remainder,  $\partial \mathscr{V}_2$ , the displacement is prescribed by the function  $\mathbf{k}(\mathbf{X},t)$ . The initial condition is given by the function  $\mathbf{f}(\mathbf{X})$ . The four functions  $\mathbf{s}(\mathbf{X},t)$ ,  $r(\mathbf{X},t)$ ,  $\mathbf{k}(\mathbf{X},t)$  and  $\mathbf{f}(\mathbf{X})$  represent the data for this problem.

Other categories of boundary initial value problems can be constructed where, instead of prescribing the pore pressure, one prescribes the relative displacement  $\mathbf{w}_f - \mathbf{w}_s$  on the boundary  $\partial \mathscr{V}_1$ . An example of this kind of boundary condition would be one where the boundary  $\partial \mathscr{V}_1$  was impervious to the flow of the fluid. For the moment, we shall focus our discussion on the boundary initial value problem defined by (9.3.1) through (9.3.4).

There are conditions on the data that are required by the governing partial differential equations. It is easily established by summing (9.1.15) and (9.1.16) and making use of (9.1.17) and (9.1.18) that

$$\operatorname{Div} \mathbf{T}_{I} \left( \mathbf{X}, t \right) = 0 \tag{9.3.5}$$

This result, when forced to hold at t = 0 shows that the initial condition (9.3.4) is not fully arbitrary. The stress calculated from the initial displacement  $\mathbf{w}(\mathbf{X},0) = \mathbf{f}(\mathbf{X})$  from the constitutive equations (9.1.17) and (9.1.18) must obey (9.3.5). From a physical standpoint, this restriction on the initial data simply states that the porous elastic material is initially in a state that obeys the governing balance of linear momentum equations.

For certain boundary conditions, the result (9.3.5) implies a restriction on the surface traction  $\mathbf{s}(\mathbf{X},t)$  in (9.3.1). If (9.3.5) is integrated over the region  $\mathscr{V}$  we obtain the overall equilibrium result

$$\oint_{\partial \mathbf{Y}} \mathbf{T}_{I} d\mathbf{s} = \mathbf{0} \tag{9.3.6}$$

For boundary initial value problems like above, but in the special case where  $\partial \mathscr{V}_1 = \partial \mathscr{V}$ , the result (9.3.6) and with the boundary condition (9.3.1) force the requirement

$$\oint_{\partial \mathscr{V}} \mathbf{s}(\mathbf{X}, t) ds = \mathbf{0} \tag{9.3.7}$$

on the data. The condition (9.3.7) is no more than the requirement that the data be prescribed such that. for all *t* the porous elastic material is in overall elastic equilibrium.

#### 9.4 A One Dimensional Example

It is instructive to illustrate some of the features of problems governed by (9.1.15) and (9.1.16) by working a one dimensional example. The example is motivated by one worked by Biot in a fundamental paper published in 1941 [Ref. 2]. The particular problem is one where the two displacements are given by the one dimensional representations

$$\mathbf{w}_f = \left(w_f\left(X,t\right), 0, 0\right) \tag{9.4.1}$$

and

$$\mathbf{w}_{s} = \left(w_{s}\left(X,t\right),0,0\right) \tag{9.4.2}$$

In this special case, we can write the partial differential equations (9.1.15) and (9.1.16) as

$$\xi \mathbf{E} \frac{\partial \mathbf{w}}{\partial t} - \mathbf{Q} \frac{\partial^2 \mathbf{w}}{\partial X^2} = \mathbf{0}$$
(9.4.3)

where,

$$\mathbf{w} = \mathbf{w}(X,t) = \begin{bmatrix} w_f(X,t) \\ w_s(X,t) \end{bmatrix}$$
(9.4.4)

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix}$$
(9.4.5)

and

$$\mathbf{E} = \begin{bmatrix} 1 & -1 \\ -1 & 1 \end{bmatrix} \tag{9.4.6}$$

Of course, the definitions (9.4.4) through (9.4.6) have appeared in earlier chapters. The symbol **Q** was introduced in equations (7.2.46) and (8.1.11). The symbol **E** was first introduced in equation (7.2.87) and, again, in (8.1.12). The specialization of (9.1.15) and (9.1.16) to the one dimensional equation is a special case of the derivation of (8.1.7) and (8.1.8) of Chapter 8. The matrix of displacements (9.4.4) was also introduced in Chapter 8. It should be evident that it is the one dimensional counterpart of equation (9.2.2) above.

For this one dimensional problem, the constitutive equation (9.1.17) and (9.1.18) can be written in the matrix form

$$\begin{bmatrix} -\varphi_f^+ P_f \\ T_I + \varphi_f^+ P_f \end{bmatrix} = \mathbf{Q} \frac{\partial \mathbf{w}}{\partial X}$$
(9.4.7)

The boundary conditions we shall first apply to this one dimensional problem are one dimensional versions of (9.3.1) through (9.3.4). We are interested in solving a problem in the interval  $X \in (0,h)$  where the boundary conditions and initial condition are

$$T_{I}(0,t) = -s(t)$$
 (9.4.8)

$$P_f\left(0,t\right) = r\left(t\right) \tag{9.4.9}$$

$$\mathbf{w}(h,t) = \mathbf{k}(t) \tag{9.4.10}$$

and

$$\mathbf{w}(X,0) = \mathbf{f}(X) \tag{9.4.11}$$

If we adopt the symbol  $\mathbf{q}(t)$  defined by

$$\mathbf{q}(t) = \begin{bmatrix} -\varphi_f^+ r(t) \\ -s(t) + \varphi_f^+ r(t) \end{bmatrix}$$
(9.4.12)

then the two boundary conditions (9.4.8) and (9.4.9) can be combined with (9.4.7) to write the boundary condition at X = 0 as

$$\mathbf{q}(t) = \mathbf{Q} \frac{\partial \mathbf{w}(0,t)}{\partial X}$$
(9.4.13)

It is instructive to display the partial differential equation, the boundary conditions and the initial condition on the following schematic

$$\mathbf{q}(t) = \mathbf{Q} \frac{\partial \mathbf{w}(0,t)}{\partial X} \left| \begin{array}{c} \xi \mathbf{E} \frac{\partial \mathbf{w}}{\partial t} - \mathbf{Q} \frac{\partial^2 \mathbf{w}}{\partial X^2} = \mathbf{0} \\ 0 \\ \mathbf{w}(X,0) = \mathbf{f}(X) \end{array} \right| \mathbf{w}(h,t) = \mathbf{k}(t)$$

The one dimensional form of (9.3.5) is

$$\frac{\partial T_I}{\partial X} = 0 \tag{9.4.14}$$

Thus, the stress cannot depend upon X. This fact can be represented in an equivalent way by multiplying (9.4.3) by adjE to obtain

$$(adj\mathbf{E})\mathbf{Q}\frac{\partial^2 \mathbf{w}}{\partial X^2} = \mathbf{0}$$
 (9.4.15)

This simple formula can be integrated twice to yield

$$(\mathrm{adj}\mathbf{E})\mathbf{Qw}(X,t) = (\mathrm{adj}\mathbf{E})\mathbf{Qk}(t) + (\mathrm{adj}\mathbf{E})\mathbf{q}(t)(X-h)$$
(9.4.16)

after the boundary conditions (9.4.10) and (9.4.13) are used. If this formula is evaluated at t = 0, we see that the functions  $\mathbf{f}(X)$ ,  $\mathbf{k}(t)$  and  $\mathbf{q}(t)$  are linked by the formula

$$(adj\mathbf{E})\mathbf{Qf}(X) = (adj\mathbf{E})\mathbf{Qk}(0) + (adj\mathbf{E})\mathbf{q}(0)(X-h)$$
 (9.4.17)

Equation is the one dimensional version of the data compatibility issue mentioned in Section 9.3. In the following, we shall assume that (9.4.17) is obeyed by our choices of the three functions f(X), k(t) and q(t).

In order to exploit eigenfunction techniques to solve this boundary initial value problem, we need to change the dependent variable in a manner sufficient to produce homogeneous boundary conditions. Because the problem is one dimensional, it is not difficult to see that the function  $\mathbf{v}(X,t)$  defined by

$$\mathbf{v}(X,t) = \mathbf{w}(X,t) - \mathbf{Q}^{-1}\mathbf{q}(t)(X-h) - \mathbf{k}(t)$$
(9.4.18)

obeys the partial differential equation

$$\xi \mathbf{E} \frac{\partial \mathbf{v}}{\partial t} = \mathbf{Q} \frac{\partial^2 \mathbf{v}}{\partial X^2} + \mathbf{p}(t)$$
(9.4.19)

where

$$\mathbf{p}(t) = -\xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) (X - h) - \xi \mathbf{E} \dot{\mathbf{k}}(t)$$
(9.4.20)

The boundary conditions and the initial value that the function  $\mathbf{v}(X,t)$  must obey follow from (9.4.18) and the corresponding conditions on  $\mathbf{w}(X,t)$ . These results are

$$\frac{\partial \mathbf{v}(0,t)}{\partial X} = \mathbf{0} \tag{9.4.21}$$

$$\mathbf{v}(h,t) = \mathbf{0} \tag{9.4.22}$$

$$\mathbf{v}(X,0) = \mathbf{f}(X) - \mathbf{Q}^{-1}\mathbf{q}(0)(X-h) - \mathbf{k}(0)$$
(9.4.23)

The gain with the transformation (9.4.18) is that the function  $\mathbf{v}(X,t)$  obeys homogeneous boundary conditions. The trade off is that the partial differential equation obeyed by  $\mathbf{v}(X,t)$  is inhomogeneous and its initial condition is more complicated than that required of  $\mathbf{w}(X,t)$ .

It is useful at this point to note that the definition (9.4.18) and the result (9.4.16) combine to yield

$$(\operatorname{adj} \mathbf{E})\mathbf{Q}\mathbf{v}(X,t) = (\operatorname{adj} \mathbf{E})\mathbf{Q}(\mathbf{w}(X,t) - \mathbf{Q}^{-1}\mathbf{q}(t)(X-h) - \mathbf{k}(t)) = \mathbf{0}$$
(9.4.24)

Of course, this result also follows from (9.4.19) and the two boundary conditions, (9.4.21) and (9.4.22).

The solution technique we shall utilize is to represent the solution as an expansion in the eigenfunctions of the space part of the operator in (9.4.19). As explained in Section 9.2, the eigenfunctions are nonzero functions  $\mathbf{u}(X)$  that obey

$$-\mathbf{Q}\frac{d^{2}\mathbf{u}}{dX^{2}} = \lambda \mathbf{u} \quad \text{on} \ (0,h) \tag{9.4.25}$$

and the boundary conditions

$$\frac{d\mathbf{u}(0)}{dX} = 0 \qquad \text{and} \quad \mathbf{u}(h) = \mathbf{0} \tag{9.4.26}$$

It is elementary to show that solutions of (9.4.25) that obey (9.4.26) are of the form

$$\mathbf{u}_{n} = \mathbf{b}_{n} \cos \frac{(2n-1)\pi X}{2h}$$
 for  $n = 1, 2, ..., \infty$  (9.4.27)

where, for each n,

$$\left(\mathbf{Q} - \frac{4h^2}{\left(2n-1\right)^2} \lambda_n \mathbf{I}\right) \mathbf{b}_n = \mathbf{0}$$
(9.4.28)

The symbol I in (9.4.28) denotes the  $2 \times 2$  identity matrix

$$\mathbf{I} = \begin{bmatrix} 1 & 0\\ 0 & 1 \end{bmatrix} \tag{9.4.29}$$

Equation (9.4.28), which arises from the substitution of (9.4.27) into (9.4.25), shows that the eigenvalues  $\lambda_1, \lambda_2, ..., \lambda_n, ...$  and the column matrices  $\mathbf{b}_1, \mathbf{b}_2, ..., \mathbf{b}_n, ...$  are determined by the algebraic eigenvalue problem for the 2×2 symmetric matrix  $\mathbf{Q}$ . If  $\alpha_{(1)}$  and  $\alpha_{(2)}$  are the eigenvalues  $\mathbf{Q}$ , then they are the two roots of

$$\det\left(\mathbf{Q} - \alpha \mathbf{I}\right) = 0 \tag{9.4.30}$$

Given these two roots, it follows from (9.4.28) that the eigenvalues  $\lambda_1, \lambda_2, ..., \lambda_n, ...$  are

$$\lambda_{n(1)} = \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \alpha_{(1)} \tag{9.4.31}$$

and

$$\lambda_{n(2)} = \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \alpha_{(2)} \tag{9.4.32}$$

The form of (9.4.31) and (9.4.32) show that the column matrices  $\mathbf{b}_1, \mathbf{b}_2, \dots, \mathbf{b}_n, \dots$  actually do not depend upon the index *n*. They are the two nonzero column matrices that obey

$$\left(\mathbf{Q} - \boldsymbol{\alpha}_{(1)}\mathbf{I}\right)\mathbf{b}_{(1)} = \mathbf{0} \tag{9.4.33}$$

and

$$\left(\mathbf{Q} - \boldsymbol{\alpha}_{(2)}\mathbf{I}\right)\mathbf{b}_{(2)} = \mathbf{0} \tag{9.4.34}$$

Given this construction, the solution of our boundary initial value problem for  $\mathbf{v}(X,t)$  will be of the form

$$\mathbf{v}(X,t) = \sum_{n=1}^{\infty} T_{n(1)}(t) \mathbf{b}_{(1)} \cos\frac{(2n-1)\pi X}{2h} + \sum_{n=1}^{\infty} T_{n(2)}(t) \mathbf{b}_{(2)} \cos\frac{(2n-1)\pi X}{2h}$$
(9.4.35)

It simplifies the analysis somewhat if we rewrite (9.4.35) as

$$\mathbf{v}(X,t) = \sum_{n=1}^{\infty} \mathbf{T}_n(t) \cos \frac{(2n-1)\pi X}{2h}$$
(9.4.36)

where  $\mathbf{T}_{n}(t)$  is a column matrix whose representation with respect to the basis of eigenvectors of  $\mathbf{Q}$  is

$$\mathbf{T}_{n}(t) = T_{n(1)}(t)\mathbf{b}_{(1)} + T_{n(2)}(t)\mathbf{b}_{(2)}$$
(9.4.37)

As constructed, the proposed solution (9.4.36) obeys the boundary conditions (9.4.21) and (9.4.22). Our next step is to determine the time dependence of the solution. If (9.4.36) is substituted into (9.4.19) the result is

$$\sum_{n=1}^{\infty} \left( \xi \mathbf{E} \frac{d\mathbf{T}_n(t)}{dt} + \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \mathbf{Q} \mathbf{T}_n(t) \right) \cos \frac{\left(2n-1\right) \pi X}{2h} = \mathbf{p}(t)$$
(9.4.38)

The orthogonality of the functions  $\left\{\cos\frac{(2n-1)\pi X}{2h}\right\}$  allow us to derive from (9.4.38) the result

$$\xi \mathbf{E} \frac{d\mathbf{T}_n(t)}{dt} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q} \mathbf{T}_n(t) = \mathbf{p}_n(t)$$
(9.4.39)

where

$$\mathbf{p}_{n}(t) = \frac{2}{h} \xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) \int_{0}^{h} (h - X) \cos \frac{(2n - 1)\pi X}{2h} dX - \frac{2}{h} \xi \mathbf{E} \dot{\mathbf{k}}(t) \int_{0}^{h} \cos \frac{(2n - 1)\pi X}{2h} dX$$
(9.4.40)

After the elementary integrals in (9.4.40) are evaluated, we obtain

$$\mathbf{p}_{n}(t) = \frac{8h}{(2n-1)^{2}\pi^{2}} \xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) + \frac{4}{(2n-1)\pi} (-1)^{n} \xi \mathbf{E} \dot{\mathbf{k}}(t)$$
(9.4.41)

The solution of (9.4.39) requires an initial condition. This condition comes from (9.4.23) and (9.4.36). These two equations yield

$$\sum_{n=1}^{\infty} \mathbf{T}_{n}(0) \cos \frac{(2n-1)\pi X}{2h} = \mathbf{f}(X) - \mathbf{Q}^{-1}\mathbf{q}(0)(X-h) - \mathbf{k}(0)$$
(9.4.42)

The orthogonality of the functions  $\left\{\cos\frac{(2n-1)\pi X}{2h}\right\}$  can be used again to derive from (9.4.42) the result

$$\mathbf{\Gamma}_{n}(0) = \frac{2}{h} \int_{0}^{h} \mathbf{f}(X) \cos \frac{(2n-1)\pi X}{2h} dX + \frac{2}{h} \mathbf{Q}^{-1} \mathbf{q}(0) \int_{0}^{h} (h-X) \cos \frac{(2n-1)\pi X}{2h} dX - \frac{2}{h} \mathbf{k}(0) \int_{0}^{h} \cos \frac{(2n-1)\pi X}{2h} dX$$
(9.4.43)

The integrals in the second and third terms of (9.4.43) can be evaluated to yield the result

$$\mathbf{T}_{n}(0) = \mathbf{f}_{n} + \frac{8h}{(2n-1)^{2} \pi^{2}} \mathbf{Q}^{-1} \mathbf{q}(0) + \frac{4}{(2n-1)\pi} (-1)^{n} \mathbf{k}(0)$$
(9.4.44)

where, we have introduced the symbol  $\mathbf{f}_n$  defined by

$$\mathbf{f}_{n} = \frac{2}{h} \int_{0}^{h} \mathbf{f}(X) \cos \frac{(2n-1)\pi X}{2h} dX$$
(9.4.45)

In the manipulations that follow, it is convenient to simply refer to the initial condition as  $\mathbf{T}_n(0)$  and to utilize the explicit formula (9.4.44) at the end of the calculation.

In the way of summary, at this point in the solution we must find the solution of the matrix ordinary differential equation (9.4.39), repeated here,

$$\xi \mathbf{E} \frac{d\mathbf{T}_n(t)}{dt} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q} \mathbf{T}_n(t) = \mathbf{p}_n(t)$$
(9.4.46)

where  $\mathbf{p}_n(t)$  is given by (9.4.41), repeated here,

$$\mathbf{p}_{n}(t) = \frac{8h}{(2n-1)^{2} \pi^{2}} \xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) + \frac{4}{(2n-1)\pi} (-1)^{n} \xi \mathbf{E} \dot{\mathbf{k}}(t)$$
(9.4.47)

subject to the initial condition (9.4.44). Because the  $2 \times 2$  matrix **E** defined by (9.4.6) is singular, solutions of (9.4.46) necessarily reflect a certain indeterminacy. If we multiply (9.4.46) by adjE and use the formula

$$(adj\mathbf{E})\mathbf{E} = (det\mathbf{E})\mathbf{I} = \mathbf{0}$$
 (9.4.48)

equation (9.4.46) yields

$$(adj\mathbf{E})\mathbf{QT}_{n}(t) = \mathbf{0} \tag{9.4.49}$$

after one uses the definition (9.4.47). This result also follows from (9.4.36) and (9.4.24).

There are various ways to solve (9.4.46). Probably the most direct is to calculate its Laplace transform and use standard methods to solve and invert the result. The Laplace transform of (9.4.46) is

$$\left(s\boldsymbol{\xi}\mathbf{E} + \frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right)\overline{\mathbf{T}}_{n}\left(s\right) = \boldsymbol{\xi}\mathbf{E}\mathbf{T}_{n}\left(0\right) + \overline{\mathbf{p}}_{n}\left(s\right)$$
(9.4.50)

where  $\overline{\mathbf{T}}_{n}(s)$  is the Laplace transform of  $\mathbf{T}_{n}(t)$  and  $\overline{\mathbf{p}}_{n}(s)$  is the Laplace transform of  $\mathbf{p}_{n}(t)$ . Equation (9.4.50) yields

$$\overline{\mathbf{T}}_{n}(s) = \xi \left( s\xi \mathbf{E} + \frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}} \mathbf{Q} \right)^{-1} \mathbf{ET}_{n}(0) + \left( s\xi \mathbf{E} + \frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}} \mathbf{Q} \right)^{-1} \overline{\mathbf{p}}_{n}(s) \qquad (9.4.51)$$

The inverse matrix that appears in (9.4.51) can be written

$$\left(s\xi\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right)^{-1} = \frac{\operatorname{adj}\left(s\xi\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right)}{\operatorname{det}\left(s\xi\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right)}$$
(9.4.52)

Because we are dealing with  $2 \times 2$  matrices, it is true that

$$\operatorname{adj}\left(s\xi\mathbf{E} + \frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right) = s\xi\operatorname{adj}\mathbf{E} + \frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}}\operatorname{adj}\mathbf{Q}$$
(9.4.53)

It is also true that

$$\det\left(s\xi\mathbf{E} + \frac{(2n-1)^2 \pi^2}{4h^2}\mathbf{Q}\right) = s\xi\frac{(2n-1)^2 \pi^2}{4h^2} \operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right) + \frac{(2n-1)^4 \pi^4}{16h^4} \det\mathbf{Q} \quad (9.4.54)$$

where det  $\mathbf{E} = \mathbf{0}$  has been used. Given (9.4.53) and (9.4.54), we can write (9.4.52) as

$$\begin{split} \left[s\xi\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right]^{-1} &= \frac{s\xi\,\mathrm{adj}\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{adj}\mathbf{Q}}{s\xi\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right) + \frac{(2n-1)^{4}\pi^{4}}{16h^{4}}\,\mathrm{det}\mathbf{Q}} \\ &= \frac{1}{s\xi\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)} \frac{s\xi\,\mathrm{adj}\mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{adj}\mathbf{Q}}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\frac{\mathrm{det}\mathbf{Q}}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\right)} \\ &= \frac{1}{\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)} \mathrm{adj}\mathbf{E}} \\ &+ \left(\frac{1}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)} \left(\mathrm{adj}\mathbf{Q} - \frac{\mathrm{det}\mathbf{Q}}{\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\,\mathrm{adj}\mathbf{E}\right)\right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\frac{\mathrm{det}\mathbf{Q}}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\right)} \\ &= \frac{1}{\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}} \mathrm{adj}\mathbf{E} \\ &+ \left(\frac{1}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)} \left(\mathrm{adj}\mathbf{Q} - \frac{\mathrm{det}\mathbf{Q}}{\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\,\mathrm{adj}\mathbf{E}\right) \right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\frac{\mathrm{det}\mathbf{Q}}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\right)} \\ &+ \left(\frac{1}{\xi\,\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)} \left(\mathrm{adj}\mathbf{Q} - \frac{\mathrm{det}\mathbf{Q}}{\mathrm{tr}\left((\mathrm{adj}\mathbf{Q})\mathbf{E}\right)}\,\mathrm{adj}\mathbf{E}\right) \right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\,\mathrm{c}\right)} \end{aligned} \tag{9.4.55}$$

where c is the consolidation coefficient defined by equation (5.1.30), repeated here as

$$c = \frac{1}{\xi} \frac{\overline{\lambda}_{ff} (\overline{\lambda}_{ss} + \frac{2}{3} \mu_{ss}) - \overline{\lambda}_{sf}^2}{\overline{\lambda}_{ss} + \frac{2}{3} \mu_{ss} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ff}} = \frac{\det \mathbf{Q}}{\xi \operatorname{tr} ((\operatorname{adj} \mathbf{Q}) \mathbf{E})}$$
(9.4.56)

If this long formula is substituted into (9.4.51), we must invert

$$\begin{aligned} \mathbf{\overline{T}}_{n}(s) &= \xi \begin{cases} \frac{1}{\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}} \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}} \\ &+ \left(\frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \left(\operatorname{adj}\mathbf{Q} - \frac{\operatorname{det}\mathbf{Q}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}}\right) \right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \\ &+ \left(\frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}} + \left(\frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}} \left(\operatorname{adj}\mathbf{Q} - \frac{\operatorname{det}\mathbf{Q}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}}\right) \right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \\ &+ \left(\frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \left(\operatorname{adj}\mathbf{Q} - \frac{\operatorname{det}\mathbf{Q}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{\operatorname{adj}\mathbf{E}}\right) \right) \frac{1}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \\ &= \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \frac{(\operatorname{adj}\mathbf{Q})\mathbf{E}}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \mathbf{T}_{n}(0) \\ &+ \frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \frac{\operatorname{adj}\mathbf{Q}}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \\ &= \frac{1}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \frac{\operatorname{adj}\mathbf{Q}}{\left(s + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c\right)} \mathbf{\overline{P}}_{n}(s) \end{aligned}$$
(9.4.57)

where (9.4.48) has been used. We have also used  $(adj\mathbf{E})\overline{\mathbf{p}}_n(s) = \mathbf{0}$  which follows from (9.4.47) and (9.4.48). The inverse Laplace transform of (9.4.57) is<sup>1</sup>

$$\mathbf{T}_{n}(t) = \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{T}_{n}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}ct} + \frac{1}{\xi \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \operatorname{adj}\mathbf{Q} \int_{0}^{t} \mathbf{p}_{n}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau$$
(9.4.58)

Next, we shall utilize (9.4.44) and (9.4.47) to write (9.4.58) as

$$\mathbf{T}_{n}(t) = \mathbf{T}_{n}(0)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}ct} + \frac{1}{\xi \operatorname{tr}((\operatorname{adj} \mathbf{Q})\mathbf{E})}\operatorname{adj}\mathbf{Q}\int_{0}^{t}\mathbf{p}_{n}(\tau)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)}d\tau$$

This form of the solution illustrates that the initial condition is, in fact, satisfied.

<sup>&</sup>lt;sup>1</sup> While we do not need the formula here, it is useful to note that the identity (adj Q)E + (adj E)Q = tr((adj Q)E)Iand the formula (9.4.49) allow the solution (9.4.58) to be written

$$\begin{split} \mathbf{T}_{n}(t) &= \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{pmatrix} \mathbf{f}_{n} + \frac{8h}{(2n-1)^{2}\pi^{2}} \mathbf{Q}^{-1} \mathbf{q}(0) \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \mathbf{k}(0) \end{pmatrix} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} ct} \\ &+ \frac{1}{\xi \operatorname{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} \mathrm{adj} \mathbf{Q} \int_{0}^{t} \begin{pmatrix} \frac{8h}{(2n-1)^{2}\pi^{2}} \xi^{2} \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(\tau) \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \xi^{2} \mathbf{E} \dot{\mathbf{k}}(\tau) \end{pmatrix} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} c(t-\tau)} d\tau \\ &= \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{pmatrix} \mathbf{f}_{n} + \frac{8h}{(2n-1)^{2}\pi^{2}} \mathbf{Q}^{-1} \mathbf{q}(0) \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \mathbf{k}(0) \end{pmatrix} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} c(t-\tau)} d\tau \\ &+ \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \frac{8h}{(2n-1)^{2}\pi^{2}} \int_{0}^{t} \dot{\mathbf{q}}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} c(t-\tau)} d\tau \\ &+ \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \frac{8h}{(2n-1)^{2}\pi^{2}} \int_{0}^{t} \dot{\mathbf{q}}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} c(t-\tau)} d\tau \\ &+ \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \frac{8h}{(2n-1)\pi} (-1)^{n} \int_{0}^{t} \dot{\mathbf{k}}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} c(t-\tau)} d\tau \\ &+ \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{f}_{n} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{8h}{(2n-1)^{2}\pi^{2}} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{pmatrix} \mathbf{q}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{8h}{(2n-1)^{2}\pi^{2}} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{pmatrix} \mathbf{q}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{8h}{(2n-1)^{2}\pi^{2}} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{pmatrix} \mathbf{q}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{4}{(2n-1)^{2}\pi^{2}} (-1)^{n} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{bmatrix} \mathbf{k}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{bmatrix} \mathbf{k}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{bmatrix} \mathbf{k}(0) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} - c(t-\tau)} d\tau \\ &+ \frac{4}{(2n-1)\pi} (-1)^{n} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \begin{bmatrix} \mathbf{k}($$

If we utilize integration by parts to write

$$\mathbf{q}(0)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})^{t}}} + \int_{0}^{t}\dot{\mathbf{q}}(\tau)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)}d\tau$$

$$= \mathbf{q}(t) - \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}\int_{0}^{t}\mathbf{q}(\tau)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)}d\tau$$
(9.4.60)

$$\mathbf{k}(0)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})^{t}}} + \int_{0}^{t}\dot{\mathbf{k}}(\tau)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(t-\tau)}}d\tau$$

$$= \mathbf{k}(t) - \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}\int_{0}^{t}\mathbf{k}(\tau)e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\frac{\det\mathbf{Q}}{\xi\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(t-\tau)}}d\tau$$
(9.4.61)

the solution (9.4.59) can be written

$$\mathbf{T}_{n}(t) = \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{f}_{n} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}-ct} + \frac{8h}{(2n-1)^{2}\pi^{2}} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{pmatrix} \mathbf{q}(t) \\ -\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} \frac{\det \mathbf{Q}}{\xi \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \int_{0}^{t} \mathbf{q}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}-c(t-\tau)} d\tau \end{pmatrix}$$

$$+ \frac{4}{(2n-1)\pi} (-1)^{n} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \begin{pmatrix} \mathbf{k}(t) \\ -\frac{(2n-1)^{2}\pi^{2}}{4h^{2}} \frac{\det \mathbf{Q}}{\xi \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \int_{0}^{t} \mathbf{k}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}-c(t-\tau)} d\tau \end{pmatrix}$$

$$(9.4.62)$$

If this result is combined with (9.4.36), the solution for  $\mathbf{v}(X,t)$  can be written

$$\mathbf{v}(X,t) = \sum_{n=1}^{\infty} \mathbf{T}_{n}(t) \cos \frac{(2n-1)\pi X}{2h}$$

$$= \sum_{n=1}^{\infty} \left\{ \begin{array}{l} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{f}_{n} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c^{t}} \\ + \frac{8h}{(2n-1)^{2}\pi^{2}} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \\ \times \left( \frac{\mathbf{q}(t)}{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}} \frac{\det \mathbf{Q}}{\xi \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \int_{0}^{t} \mathbf{q}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \\ + \frac{4}{(2n-1)\pi} (-1)^{n} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(\operatorname{adj}\mathbf{Q}) \mathbf{E} \\ \times \left( \frac{\mathbf{k}(t)}{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}} \frac{\det \mathbf{Q}}{\xi \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \int_{0}^{t} \mathbf{k}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \right\}$$
(9.4.63)

Because

$$\frac{8h}{\pi^2} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \cos \frac{(2n-1)\pi X}{2h} = h - X$$
(9.4.64)

and

$$\frac{4}{\pi} \sum_{n=1}^{\infty} \left( \frac{1}{(2n-1)} (-1)^n \right) \cos \frac{(2n-1)\pi X}{2h} = -1$$
(9.4.65)

(9.4.63) reduces to

$$\mathbf{v}(X,t) = \sum_{n=1}^{\infty} \mathbf{T}_{n}(t) \cos \frac{(2n-1)\pi X}{2h}$$

$$= \sum_{n=1}^{\infty} \left( \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{f}_{n} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}ct}}{\left(\frac{-2}{h}\frac{1}{\xi} (\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E}))^{2}} (\operatorname{adj}\mathbf{Q}) \mathbf{E} (\operatorname{adj}\mathbf{Q}) \left( \int_{0}^{t} \mathbf{q}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \right) - (-1)^{n} \frac{(2n-1)\pi}{h^{2}} \frac{\operatorname{det}\mathbf{Q}}{\xi(\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E}))^{2}} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \left( \int_{0}^{t} \mathbf{k}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \right) \right) \cos \frac{(2n-1)\pi X}{2h} (9.4.66)$$

$$+ \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \mathbf{q}(t) (h-X)$$

$$- \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} (\mathbf{k}(t))$$

This result combines with (9.4.18) to yield

$$\begin{split} \mathbf{w}(X,t) &= \mathbf{v}(X,t) + \mathbf{Q}^{-1}\mathbf{q}(t)(X-h) + \mathbf{k}(t) \\ &= \sum_{n=1}^{\infty} \left( \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{f}_{n} e^{\frac{-(2n-1)^{2}\pi^{2}}{4h^{2}}ct} \\ &= \sum_{n=1}^{\infty} \left( -\frac{2}{h} \frac{1}{\xi(\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E}))^{2}} (\operatorname{adj}\mathbf{Q}) \mathbf{E} (\operatorname{adj}\mathbf{Q}) \left( \int_{0}^{t} \mathbf{q}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \\ &- (-1)^{n} \frac{(2n-1)\pi}{h^{2}} \frac{\operatorname{det}\mathbf{Q}}{\xi(\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E}))^{2}} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \left( \int_{0}^{t} \mathbf{k}(\tau) e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-\tau)} d\tau \right) \\ &+ \left( \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} - \mathbf{I} \right) \mathbf{Q}^{-1} \mathbf{q}(t) (h-X) - \left( \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} - \mathbf{I} \right) \mathbf{k}(t) \\ &= \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \sum_{n=1}^{\infty} \mathbf{f}_{n} e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}-ct} \cos\left(\frac{2n-1}{2h}\right) \frac{\pi^{2}}{2h}} \\ &- \frac{2}{h} \frac{\operatorname{det}\mathbf{Q}}{\xi(\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{2} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \sum_{n=1}^{\infty} \left( \sum_{n=1}^{(2n-1)^{2}\pi^{2}} - \cos\left(\frac{2n-1}{4h^{2}}\right) \frac{\pi^{2}}{2h}} \\ &- \frac{2}{h} \frac{\operatorname{det}\mathbf{Q}}{\xi(\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}^{2} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \sum_{n=1}^{\infty} \left( \sum_{n=1}^{(2n-1)^{2}\pi^{2}} - \cos\left(\frac{2n-1}{2h}\right) \frac{\pi^{2}}{2h}} \\ &\left( -1 \right)^{n} \frac{(2n-1)^{2}\pi^{2}}{2h} \left( \sum_{n=1}^{(2n-1)^{2}\pi^{2}} - \cos\left(\frac{2n-1}{4h^{2}}\right) \right) \right) \cos\left(\frac{(2n-1)\pi X}{2h}} \\ &+ \frac{\operatorname{adj}\mathbf{E}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \mathbf{q}(t) (X-h) + \frac{(\operatorname{adj}\mathbf{E})\mathbf{Q}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \mathbf{k}(t) \end{aligned} \right)$$

where the algebraic identity

$$(adj\mathbf{Q})\mathbf{E} + (adj\mathbf{E})\mathbf{Q} = tr((adj\mathbf{Q})\mathbf{E})\mathbf{I}$$
 (9.4.68)

has been used to simplify the last two terms.

Equation (9.4.67) is the analytical solution to the boundary initial value problem we have posed. Because of the relationship (9.4.17) and various identities one can write among the  $2 \times 2$  matrices that appear in the answer, one can write the solution in various other equivalent forms.

It is straight forward to show that the solution (9.4.67) obeys the boundary and initial conditions (9.4.8) through (9.4.11). Also, we note in passing that the solution (9.4.67) yields the following formula for the initial velocity.

$$\frac{\partial \mathbf{w}(X,0)}{\partial t} = \frac{\det \mathbf{Q}}{\xi \operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} \frac{\partial^2 \mathbf{f}(X)}{\partial X^2} + \frac{\operatorname{adj}\mathbf{E}}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} \dot{\mathbf{q}}(0)(X-h) + \frac{\operatorname{adj}\mathbf{E}}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} \mathbf{Q}\dot{\mathbf{k}}(0)(9.4.69)$$

This result, which can be shown to be consistent with (9.4.16), will be useful in Section 9.5

### 9.5 Biot Problem

The original problem solved by Biot is the special case of the solution (9.4.67) corresponding to the choices [Ref. 1]

$$\mathbf{q}(t) = P_0 \begin{bmatrix} 0\\-1 \end{bmatrix} \tag{9.5.1}$$

$$\mathbf{k}(t) = \mathbf{0} \tag{9.5.2}$$

and

$$\mathbf{f}(X) = \frac{P_0}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}(h-X)\begin{bmatrix}1\\1\end{bmatrix}$$
(9.5.3)

It follows from (9.4.12) that the choice (9.5.1) corresponds to the assumption that the pore pressure at the point x = 0 is zero and the stress  $T_i$  at x = 0 is the pressure  $P_0$ . As explained earlier, the choice  $P_f = 0$  at a boundary means that the boundary is pervious to the flow of the fluid out of and into the solid. The choice (9.5.2) corresponds to the assumption that the displacement vanishes at x = h.

The initial condition (9.5.3) arises from the physical assumption that the initial stress  $T_I(0)$  equals the imposed pressure  $P_0$  and that the change in fluid content at t = 0 is zero. The change in content is given by (4.1.46). In terms of the notation we are using, the assumption that at t = 0 the change in fluid content is zero reduces (4.1.46) to

$$-\frac{\partial w_f(X,0)}{\partial X} + \frac{\partial w_s(X,0)}{\partial X} = 0$$
(9.5.4)

It follows from the one dimensional forms of (9.1.17) and (9.1.18) that

$$T_{I}(0) = -P_{0} = \left(\overline{\lambda}_{ff} + \overline{\lambda}_{sf}\right) \frac{\partial w_{f}(X,0)}{\partial X} + \left(\overline{\lambda}_{ss} + 2\mu_{ss} + \overline{\lambda}_{sf}\right) \frac{\partial w_{s}(X,0)}{\partial X}$$
(9.5.5)

When we use (9.5.4), equation (9.5.5) simplifies to

$$-P_{0} = \left(\overline{\lambda}_{ff} + 2\overline{\lambda}_{sf} + \overline{\lambda}_{ss} + 2\mu_{ss}\right) \frac{\partial w_{f}(X,0)}{\partial X} = \operatorname{tr}\left(\left(\operatorname{adj}\mathbf{Q}\right)\mathbf{E}\right) \frac{\partial w_{f}(X,0)}{\partial X}$$
(9.5.6)

We can integrate (9.5.6) and obtain

$$w_f(X,0) = -\frac{P_0}{\operatorname{tr}((\operatorname{adj} \mathbf{Q})\mathbf{E})}(X-h)$$
(9.5.7)

where we have selected the constant of integration such that  $w_f(X,0)$  vanishes at X = h. This choice makes the initial displacement at X = h agree with the displacement at X = h initially. Given (9.5.7), we obtain from (9.5.4)

$$w_{s}(X,0) = -\frac{P_{0}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})}(X-h)$$
(9.5.8)

Equations (9.5.7) and (9.5.8) give the specified initial condition (9.5.3). We mentioned in Sections 3 and 4 that the initial condition and the boundary conditions must obey a compatibility condition. For one dimensional problems, we derived equation (9.4.17), repeated here,

$$(adj\mathbf{E})\mathbf{Qf}(X) = (adj\mathbf{E})\mathbf{Qk}(0) + (adj\mathbf{E})\mathbf{q}(0)(X-h)$$
 (9.5.9)

It is easily established that this result is obeyed for our choices (9.5.1) through (9.5.3). In fact, the initial condition (9.5.3) was constructed such that it obeys (9.5.9).

Given (9.5.1) through (9.5.3), the solution (9.4.67) reduces to

$$\mathbf{w}(X,t) = \frac{2P_0c}{h} \frac{1}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \sum_{n=1}^{\infty} \left( \int_0^t e^{-\frac{(2n-1)^2 \pi^2}{4h^2} c(t-\tau)} d\tau \right) \cos \frac{(2n-1)\pi X}{2h} + \frac{P_0}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} (\operatorname{adj}\mathbf{E}) \begin{bmatrix} 0\\-1 \end{bmatrix} (X-h)$$
(9.5.10)

The integral that appears in (9.5.10) can be evaluated to yield the final form of the answer

$$\mathbf{w}(X,t) = \frac{8P_0h}{\pi^2} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix}$$

$$\times \sum_{n=1}^{\infty} \left( \frac{1}{(2n-1)^2} \left( 1 - e^{-\frac{(2n-1)^2 \pi^2}{4h^2} ct} \right) \right) \cos \frac{(2n-1)\pi X}{2h}$$

$$+ \frac{P_0}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} \begin{bmatrix} 1\\1 \end{bmatrix} (h-X)$$
(9.5.11)

where the simple formula 
$$(adj \mathbf{E}) \begin{bmatrix} 0 \\ -1 \end{bmatrix} = -\begin{bmatrix} 1 \\ 1 \end{bmatrix}$$
 has also been used.<sup>2</sup>

It is instructive to plot the solution (9.5.11) for a typical poroelastic solid. If we adopt the material properties for Ruhr Sandstone as tabulated in Tables 1 and 2 of Chapter 6, then we shall use the following numerical values:

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix} = \begin{bmatrix} .165 & 5.21 \\ 5.21 & 206.41 + 2(133) \end{bmatrix} = \begin{bmatrix} .165 & 5.21 \\ 5.21 & 472.41 \end{bmatrix} \text{ kbar}$$
(9.5.12)

$$\varphi_f^+ = .02 \tag{9.5.13}$$

and

$$\xi = 1.9(10)^{-3}$$
 kbar-sec/cm<sup>2</sup> (9.5.14)

For purposes of illustration, we shall take

\_\_\_\_\_

$$P_0 = 10^2 \text{ kbar}$$
 (9.5.15)

We shall plot the solution (9.5.11) by first writing it the equivalent dimensionless form

$$\frac{\mathbf{w}(X,t)}{h} = \frac{8}{\pi^2} \frac{1}{\mathrm{tr}\left(\left(\operatorname{adj}\mathbf{Q}_{P_0}\right)\mathbf{E}\right)} \left(\operatorname{adj}\mathbf{Q}_{P_0}\right) \mathbf{E}\left(\mathbf{Q}_{P_0}\right)^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \\ \times \sum_{n=1}^{\infty} \left(\frac{1}{(2n-1)^2} \left(1 - e^{-\frac{(2n-1)^2\pi^2}{4} \left(\frac{tc}{h^2}\right)}\right)\right) \cos\left(\frac{(2n-1)\pi}{2} \frac{X}{h}\right)$$
(9.5.16)  
$$+ \frac{1}{\mathrm{tr}\left(\left(\operatorname{adj}\mathbf{Q}_{P_0}\right)\mathbf{E}\right)} \begin{bmatrix} 1\\1 \end{bmatrix} \left(1 - \frac{X}{h}\right)$$

$$\mathbf{w}_{\infty} = P_0 \left( h - X \right) \mathbf{Q}^{-1} \begin{bmatrix} 0 \\ 1 \end{bmatrix}$$

<sup>&</sup>lt;sup>2</sup> It is easily shown from (9.5.11) that as  $t \to \infty$ , the displacement  $\mathbf{w}(X, t)$  approaches
and forming a plot of 
$$\frac{\mathbf{w}(X,t)}{h}$$
 verses the dimensionless distance  $\frac{X}{h}$  and the dimensionless time  $\frac{tc}{h^2}$ .

The resulting plots of the fluid and solid displacements are



and



An examination of these two figures shows that roughly speaking the displacement of the fluid is negative and that of the solid is positive. Thus, the fluid tends to move out of the solid. This observation is somewhat easier to see if we plot the displacement of the fluid relative to that of the solid. The figure in this case is



Except very close to  $\frac{tc}{h^2} = 0$ , this figure shows that the relative displacement of the fluid is negative.

The plots we have displayed do not need numerical values of h and c. However, it is useful to note that (9.5.12), (9.5.14) and the definition (9.4.56) yield

$$c = .0055 \text{ m}^2 / \text{sec}$$
 (9.5.17)

for Ruhr Sandstone. Thus, a dimensionless time in the range  $0 < \frac{tc}{h^2} < 1$  corresponds to actual times of 0 < t < 505 hours for a poroelastic solid of length h = 100 m. This observation reveals the slow nature of the diffusion process governed by the poroelasticity equations. The characteristic time  $\frac{h^2}{c}$  is typically large resulting in slow displacement changes over time.

In Section 9.5, the formula (9.4.69) for the initial velocity was given. For the example being examined in this section, (9.4.69) yields

$$\frac{\partial \mathbf{w}(X,0)}{\partial t} = \mathbf{0} \tag{9.5.18}$$

While it is a little difficult to see, the result (9.5.18) is shown in the above figures.

#### 9.6 Modified Biot Problem: Time Dependent External Pressure

It is interesting to modify the problem discussed in Section 9.5 by allowing the externally applied pressure to be time dependent. In order to illustrate this point, we replace (9.5.1) through (9.5.3) by

$$\mathbf{q}(t) = P_0 \cos \omega t \begin{bmatrix} 0\\-1 \end{bmatrix}$$
(9.6.1)

$$\mathbf{k}(t) = \mathbf{0} \tag{9.6.2}$$

and

$$\mathbf{f}(X) = \frac{P_0 \cos \omega t}{\operatorname{tr}((\operatorname{adj} \mathbf{Q})\mathbf{E})} (h - X) \begin{bmatrix} 1\\1 \end{bmatrix}$$
(9.6.3)

where  $\omega$  is a prescribed external frequency. The choice (9.6.1) corresponds to the assumption that the pore pressure at the point x = 0 is zero and the stress  $T_i$  at x = 0 is the pressure  $P_0 \cos \omega t$ . The choice (9.6.2) retains the assumption used in Section 9.5 that vanishes at x = h. The compatibility requirement between the initial condition and the boundary conditions that we have mentioned is satisfied by the choices (9.6.1) through (9.6.3).

Given (9.6.1) through (9.6.3), the solution (9.4.67) reduces to

$$\mathbf{w}(X,t) = \frac{2P_0c}{h} \frac{1}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \sum_{n=1}^{\infty} \left( \int_0^t \cos \omega \tau e^{-\frac{(2n-1)^2 \pi^2}{4h^2} c(t-\tau)} d\tau \right) \cos \frac{(2n-1)\pi X}{2h} + \frac{P_0 \cos \omega t}{\operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} (\operatorname{adj}\mathbf{E}) \begin{bmatrix} 0\\-1 \end{bmatrix} (X-h)$$
(9.6.4)

After the integral in (9.6.4) is evaluated, the solution (9.6.4) becomes

$$\mathbf{w}(X,t) = \frac{2P_0c}{h} \frac{1}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} (\mathrm{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \sum_{n=1}^{\infty} \begin{pmatrix} \frac{1}{\left(\omega^2 + \frac{(2n-1)^4 \pi^4}{16h^4} c^2\right)} \\ \times \left(\frac{(2n-1)^2 \pi^2}{4h^2} c\left(\cos \omega t - e^{-\frac{(2n-1)^2 \pi^2}{4h^2} ct}\right) \\ + \omega \sin \omega t \end{pmatrix} \right) \right) \cos \frac{(2n-1)\pi X}{2h} + \frac{P_0 \cos \omega t}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})} \begin{bmatrix} 1\\1 \end{bmatrix} (h-X)$$

$$(9.6.5)$$

where, again, the simple formula  $(adj \mathbf{E}) \begin{bmatrix} 0 \\ -1 \end{bmatrix} = -\begin{bmatrix} 1 \\ 1 \end{bmatrix}$  has also been used. This solution, unlike the solution (9.5.11), displays a long time oscillation with the imposed frequency  $\omega$ . A more compact version of the solution (9.6.5) is

$$\mathbf{w}(X,t) = \frac{8hP_0}{\pi^2} \frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} (\operatorname{adj}\mathbf{Q}) \mathbf{E} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \sum_{n=1}^{\infty} \frac{\cos\varphi_n}{(2n-1)^2} \left( \cos\left(\omega t - \varphi_n\right) - \cos\varphi_n e^{-\frac{(2n-1)^2\pi^2}{4h^2}ct} \right) \cos\frac{(2n-1)\pi X}{2h} + \frac{P_0 \cos\omega t}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \begin{bmatrix} 1\\1 \end{bmatrix} (h-X)$$
(9.6.6)

where  $\varphi_n$  is the angle defined by

$$\cos\varphi_{n} = \frac{\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c}{\sqrt{\omega^{2} + \frac{(2n-1)^{4}\pi^{4}}{16h^{4}}c^{2}}}$$
(9.6.7)

and

$$\sin \varphi_n = \frac{\omega}{\sqrt{\omega^2 + \frac{(2n-1)^4 \pi^4}{16h^4}c^2}}$$
(9.6.8)

We shall repeat the plots of Section 9.5 but for the solution (9.6.5) or, equivalently, (9.6.6). We shall again adopt the numerical values for Ruhr Sandstone displayed in equations (9.5.12)

through (9.5.15). We shall need to prescribe the external frequency  $\omega$ . However, we shall first display the solution (9.6.5) in dimensionless form as follows:

$$\frac{\mathbf{w}(X,t)}{h} = \frac{2}{\mathrm{tr}\left(\left(\mathrm{adj}\mathbf{Q}_{P_{0}}\right)\mathbf{E}\right)} \left(\mathrm{adj}\mathbf{Q}_{P_{0}}\right) \mathbf{E}\left(\mathbf{Q}_{P_{0}}\right)^{-1} \begin{bmatrix} 0\\1 \end{bmatrix} \sum_{n=1}^{\infty} \left(\frac{(2n-1)^{2} \pi^{2}}{4} \left(\frac{\cos\left(\left(\frac{\omega h^{2}}{c}\right)\left(\frac{tc}{h^{2}}\right)\right)}{-e^{-\frac{(2n-1)^{2} \pi^{2}}{4}}\left(\frac{c}{c}\right)\left(\frac{tc}{h^{2}}\right)}\right)\right) \right)} \cos\left(\frac{(2n-1)\pi X}{2}h\right)$$

$$+ \frac{\cos\left(\left(\frac{\omega h^{2}}{c}\right)\left(\frac{tc}{h^{2}}\right)\right)}{\mathrm{tr}\left(\left(\mathrm{adj}\mathbf{Q}_{P_{0}}\right)\mathbf{E}\right)} \left[\frac{1}{1}\right] \left(1-\frac{X}{h}\right)}$$

$$(9.6.9)$$

The form of this equation suggests plots of  $\frac{\mathbf{w}(X,t)}{h}$  verses the dimensionless time  $\frac{tc}{h^2}$  and the dimensionless distance  $\frac{X}{h}$  for various values of the dimensionless ratio  $\frac{\omega h^2}{c}$ . This last ratio, written,

$$\frac{\omega h^2}{c} = \frac{\frac{h^2}{c}}{\frac{1}{\omega}}$$
(9.6.10)

is the ratio of the characteristic time of diffusion to the characteristic time of the oscillation of the external load. As we pointed out in Section 9.5, the characteristic time of diffusion is typically large. Because it is externally controlled, the characteristic time of the load oscillation can be small or large relative to the characteristic time of diffusion. We shall form plots of  $\frac{\mathbf{w}(X,t)}{h}$  verses the dimensionless distance  $\frac{X}{h}$  and the dimensionless time  $\frac{tc}{h^2}$  for various values of the ratio  $\frac{\omega h^2}{c}$ .

The resulting plots of the fluid and solid displacements for three different choices of the ratio (9.6.10) are



This series of figures suggests that for small values of the dimensionless frequency  $\frac{\omega h^2}{c}$ , the solution is not unlike that of the zero frequency case. However, when this ratio is large, as illustrated in the last figure, the solution is fundamentally different as one would expect.

As with the example in Section 9.5, the formula (9.4.69) yields

$$\frac{\partial \mathbf{w}(X,0)}{\partial t} = \mathbf{0} \tag{9.6.11}$$

for the initial conditions (9.6.1) through (9.6.3).

#### 9.7 The Use of Green's Functions

When one encounters inhomogeneous boundary conditions as in Section 9.3, it is often advantageous to formulate the solution in terms of Green's Functions. The representation of the solution to the boundary initial value problem stated in Section 9.3 in terms of Green's functions is given in Ref. 2. Basically, one defines the Green's function as the  $6\times6$  matrix of functions. We shall use the notation  $G(\mathbf{X}, t; \mathbf{X}_0, t_0)$ . As the notation suggests, the Green's function depends upon pairs of points and pairs of times. It is helpful, when dealing with the notation, to represent the elements of the  $6\times6$  matrix by six  $6\times1$  column matrices that we can write

$$\mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \begin{bmatrix} \mathbf{g}_{f(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ \mathbf{g}_{s(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{s(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{s(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{s(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{s(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{s(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ (9.7.1) \end{bmatrix}$$

where, for example,  $\mathbf{g}_{f(\alpha)}(\mathbf{X}, t; \mathbf{X}_0, t_0)$ , is a three dimensional vector that we have represented by a  $3 \times 1$  column vector.

The Green's function is the  $6 \times 6$  matrix of functions introduced in (9.7.1) defined such that, for every pair  $(\mathbf{X}_0, t_0) \in \mathscr{V} \times (-\infty, \infty)$ , it obeys the boundary initial value problem

$$\mathscr{D}\mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \delta(t-t_{0})\delta(\mathbf{X}-\mathbf{X}_{0})\mathbf{I} \quad \text{for} \quad (\mathbf{X},t) \in \mathscr{V} \times (-\infty,\infty) \quad (9.7.2)$$

$$\mathbf{T}_{I}(\mathbf{G})\mathbf{n} = \mathbf{0} \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{I} \times (-\infty, \infty)$$
(9.7.3)

$$P_{f}(\mathbf{G}) = 0 \qquad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{1} \times (-\infty, \infty)$$
(9.7.4)

$$\mathbf{G} = 0 \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_2 \times (-\infty, \infty) \tag{9.7.5}$$

and

$$\mathbf{G}\left(\mathbf{X}, t, \mathbf{X}_{0}, t_{0}\right) = \mathbf{0} \qquad \text{for} \quad \mathbf{X} \in \mathscr{V} \text{ and } t < t_{0}$$

$$(9.7.6)$$

where  $\mathcal{D}$  is the formal linear operator defined by

$$\mathcal{D}\mathbf{w} = \begin{bmatrix} \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) - \bar{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) - \bar{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) \\ -\xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) - (\bar{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) - \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) - \bar{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) \end{bmatrix} \\ = \begin{bmatrix} \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) \\ -\xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) \end{bmatrix} + L \mathbf{w}$$

$$(9.7.7)$$

The notation in (9.7.7) is the same as used in Section 9.2. In particular, the quantity **w** is defined by (9.2.2) and can be thought of as a column matrix of dimension  $6 \times 1$ . Given this

interpretation, the symbol  $\mathscr{L}\mathbf{G}$  is calculated by the rule (9.7.7) applied to each column of the  $6 \times 6$  matrix  $\mathbf{G}$ . The symbol  $\mathbf{I}$  in (9.7.2) represents the  $6 \times 6$  identity matrix. The symbol  $\delta(t-t_0)$  represents the one dimensional delta function, and the symbol  $\delta(\mathbf{X}-\mathbf{X}_0)$  denotes the three dimensional delta function.

While it is questionable whether or not our notation should be exploited excessively, there is a benefit of going the next step and writing

$$\begin{bmatrix} \left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) \\ -\left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) \end{bmatrix} = \mathbf{E} \frac{\partial \mathbf{w}}{\partial t}$$
(9.7.8)

where **E** is the symmetric  $6 \times 6$  matrix

$$\mathbf{E} = \begin{bmatrix} 1 & 0 & 0 & -1 & 0 & 0 \\ 0 & 1 & 0 & 0 & -1 & 0 \\ 0 & 0 & 1 & 0 & 0 & -1 \\ -1 & 0 & 0 & 1 & 0 & 0 \\ 0 & -1 & 0 & 0 & 1 & 0 \\ 0 & 0 & -1 & 0 & 0 & 1 \end{bmatrix}$$
(9.7.9)

This formalism allows the operator  $\mathscr{L}$  to be defined by

$$\mathscr{D}\mathbf{w} = \xi \mathbf{E} \frac{\partial \mathbf{w}}{\partial t} + L \,\mathbf{w} \tag{9.7.10}$$

rather than by (9.7.7).

Given the definition of the Green's function  $G(\mathbf{X}, t; \mathbf{X}_0, t_0)$ , the theory of Green's

functions can be used to write the solution of the boundary initial value problem defined by (9.3.1) through (9.3.4) in terms of its initial and boundary data. The details of this formalism can be found in Ref. 2, and the result is

$$\mathbf{w}(\mathbf{X},t) = \xi \int_{\mathbf{Y}} \mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},0) \mathbf{E}\mathbf{f}(\mathbf{X}_{0}) d\upsilon_{0} - \int_{0}^{t} \int_{\partial \mathbf{Y}_{1}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \cdot \mathbf{s}(\mathbf{X}_{0},t_{0}) \right) \right) d\upsilon_{0} dt_{0} - \varphi_{f}^{+} \int_{0}^{t} \int_{\partial \mathbf{Y}_{1}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \left( \mathbf{g}_{f(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) - \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \cdot \mathbf{n}r(\mathbf{X}_{0},t_{0}) \right) \right) d\upsilon_{0} dt_{0} - \int_{0}^{t} \int_{\partial \mathbf{Y}_{2}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \mathbf{T}_{I} \left( \mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \mathbf{k}_{s}(\mathbf{X}_{0},t_{0}) \right) \right) \cdot d\mathbf{s}_{0} dt_{0}$$

$$+ \varphi_{f}^{+} \int_{0}^{t} \int_{\partial \mathbf{Y}_{2}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( P_{f} \left( \mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \right) \left( \mathbf{k}_{f}(\mathbf{X}_{0},t_{0}) - \mathbf{k}_{s}(\mathbf{X}_{0},t_{0}) \right) \right) \right) \cdot d\mathbf{s}_{0} dt_{0}$$

$$(9.7.11)$$

where  $\mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_0,t_0)$  is the column matrix defined by

$$\mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \begin{bmatrix} \mathbf{g}_{f(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \end{bmatrix}, \qquad (9.7.12)$$

 $\mathbf{k}_{f}(\mathbf{X}_{0},t_{0})$  and  $\mathbf{k}_{s}(\mathbf{X}_{0},t_{0})$  are obtained from the boundary condition (9.3.3) expressed as the column matrix

$$\mathbf{k}(\mathbf{X},t) = \begin{bmatrix} \mathbf{k}_{f}(\mathbf{X},t) \\ \mathbf{k}_{s}(\mathbf{X},t) \end{bmatrix}$$
(9.7.13)

and  $\mathbf{e}_{\alpha}$ , for  $\alpha = 1, 2, ..., 6$ , is the 6×1 column matrix with entries equal to zero except in the  $\alpha$  position where the value is 1

The techniques that can be used to actually calculate the Green's function are complicated. One method is, however, straight forward. The method is to represent the solution of the initial boundary value problem (9.7.2) through (9.7.6) by an expansion in the eigenfunctions of the space part of the operator  $\mathcal{L}$ , i.e., in terms of the eigenfunctions of the operator L. These eigenfunctions, as defined in Section 9.2 allow us to seek solutions for the Green's function of the form

$$\mathbf{G}\left(\mathbf{X},t;\mathbf{X}_{0},t_{0}\right) = \sum_{n=1}^{\infty} T_{n}\left(t;t_{0}\right) \mathbf{u}_{n}\left(\mathbf{X}\right) \mathbf{u}_{n}\left(\mathbf{X}_{0}\right)^{T}$$
(9.7.14)

The solution in terms of eigenfunctions of L will also be complicated. It is helpful at this point in the discussion to illustrate this kind of solution for the class of one dimensional problems discussed in Section 9.4. In this case, the initial boundary value problem we are discussing is defined by (9.4.3) and (9.4.8) through (9.4.11). The Green's function is the  $2 \times 2$  matrix of functions

Chapter 9

$$\mathbf{G}(X,t;X_{0},t_{0}) = \begin{bmatrix} g_{f(1)}(X,t;X_{0},t_{0}) & g_{f(2)}(X,t;X_{0},t_{0}) \\ g_{s(1)}(X,t;X_{0},t_{0}) & g_{s(2)}(X,t;X_{0},t_{0}) \end{bmatrix}$$
(9.7.15)

that obeys the one dimensional forms of (9.7.2) through (9.7.6). Therefore, the Green's function is the solution of

$$\xi \mathbf{E} \frac{\partial \mathbf{G}}{\partial t} - \mathbf{Q} \frac{\partial^2 \mathbf{G}}{\partial X^2} = \delta(t - t_0) \delta(X - X_0) \qquad \text{for} \qquad (X, t) \in (0, h) \times (-\infty, \infty) \qquad (9.7.16)$$

$$\frac{\partial \mathbf{G}(0,t;X_0,t_0)}{\partial X} = \mathbf{0}$$
(9.7.17)

$$\mathbf{G}(h,t;X_0,t_0) = 0 \tag{9.7.18}$$

and

$$\mathbf{G}(x,t;x_0,t_0) = \mathbf{0} \qquad \text{for} \quad t < t_0 \tag{9.7.19}$$

As in Section (9.4), the eigenfunctions of the space part of the one dimensional operator are defined by (9.4.25) and (9.4.26) are given by (9.4.27) where the eigenvalues are given by (9.4.31) and (9.4.32). In the notation adopted in Section 9.4, the representation of the solution (9.7.14) becomes

$$\mathbf{G}(X,t;X_{0},t_{0}) = \sum_{n=1}^{\infty} \begin{pmatrix} T_{n(11)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(1)}^{T} + T_{n(22)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(2)}^{T} \\ + T_{n(21)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(1)}^{T} + T_{n(22)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(2)}^{T} \end{pmatrix} \cos\frac{(2n-1)\pi X}{2h} \cos\frac{(2n-1)\pi X}{2h} (9.7.20)$$

Just as when we expressed (9.4.35) in the form (9.4.36), we can simplify the notation somewhat if we write (9.7.20) as

$$\mathbf{G}(X,t;X_0,t_0) = \sum_{n=1}^{\infty} \mathbf{K}_n(t;t_0) \cos\frac{(2n-1)\pi X}{2h} \cos\frac{(2n-1)\pi X_0}{2h}$$
(9.7.21)

where, for each *n*,  $\mathbf{K}_n(t;t_0)$  is a 2×2 matrix whose representation with respect to the basis of eigenvectors of **Q** is

$$\mathbf{K}_{n}(t;t_{0}) = T_{n(11)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(1)}^{T} + T_{n(12)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(2)}^{T} + T_{n(21)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(1)}^{T} + T_{n(22)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(2)}^{T}$$
(9.7.22)

If (9.7.21) is substituted into (9.7.16), it is easily shown that  $\mathbf{K}_n(t;t_0)$  must obey

$$\xi \mathbf{E} \frac{d\mathbf{K}_{n}(t;t_{0})}{dt} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \mathbf{Q} \mathbf{K}_{n}(t;t_{0}) = \frac{2}{h} \delta(t-t_{0})$$
(9.7.23)

It follows from (9.7.19) and (9.7.21) that the matrix  $\mathbf{K}_n(t;t_0)$  must obey

$$\mathbf{K}_n(t;t_0) = \mathbf{0} \qquad \text{for} \quad t < t_0 \qquad (9.7.24)$$

The same technique used to solve (9.4.46) can be applied to finding the solution of (9.7.23). The result is

$$\mathbf{K}_{n}(t;t_{0}) = H(t-t_{0})\frac{2c}{h}\frac{(\mathrm{adj}\mathbf{Q})\mathbf{E}\mathbf{Q}^{-1}}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-t_{0})} + \delta(t-t_{0})\frac{8h}{\pi^{2}}\frac{\mathrm{adj}\mathbf{E}}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}\frac{1}{(2n-1)^{2}}(9.7.25)$$

where  $H(t-t_0)$  is the Heaviside step function and, as above,  $\delta(t-t_0)$  is the delta function. Given (9.7.25), equation (9.7.21) can be written<sup>3</sup>

$$\mathbf{G}(X,t;X_{0},t_{0}) = H(t-t_{0})\frac{2c}{h}\frac{(\mathrm{adj}\mathbf{Q})\mathbf{E}\mathbf{Q}^{-1}}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}\sum_{n=1}^{\infty}e^{-\frac{(2n-1)^{2}\pi^{2}}{4h^{2}}c(t-t_{0})}\cos\frac{(2n-1)\pi X}{2h}\cos\frac{(2n-1)\pi X}{2h} + \delta(t-t_{0})\frac{8h}{\pi^{2}}\frac{\mathrm{adj}\mathbf{E}}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}\sum_{n=1}^{\infty}\frac{1}{(2n-1)^{2}}\cos\frac{(2n-1)\pi X}{2h}\cos\frac{(2n-1)\pi X}{2h}\cos\frac{(2n-1)\pi X}{2h}$$
(9.7.26)

For the one dimensional problem we are discussing, the solution (9.7.11) can be written

$$\mathbf{w}(X,t) = \xi \int_0^h \mathbf{G}(X,t;X_0,0) \mathbf{E} \mathbf{f}(X_0) dX_0 - \int_0^t \mathbf{G}(X,t;0,t_0) \mathbf{q}(t_0) dt_0$$
  
- 
$$\int_0^t \frac{\partial \mathbf{G}(X,t;h,t_0)}{\partial X_0} \mathbf{Q} \mathbf{k}(t_0) dt_0$$
(9.7.27)

where  $\mathbf{q}(t)$  is defined by (9.4.12). If (9.7.26) were to be substituted into (9.7.27), a long and somewhat tedious calculation will again yield the solution (9.4.67).

#### References

$$\sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \cos \frac{(2n-1)\pi X}{2h} \cos \frac{(2n-1)\pi X_0}{2h} = \begin{cases} h-X_0 & \text{for } X < X_0 \\ h-X & \text{for } X > X_0 \end{cases}$$

can be used to simplify further (9.7.26)

<sup>&</sup>lt;sup>3</sup> If convenient, the identity

- 1. BIOT, M. A., General Theory of Three Dimensional Consolidation, *J. Appl. Phys*, **12**, 155-164, 1941.
- 2. BOWEN, R. M., Green's Functions for Consolidation Problems, *Lett. Appl. Engng. Sci.*, **19**, 455-466, 1981.

# **Boundary Initial Value Problems: Inertia Included**

In this chapter we shall discuss the solution of a certain class of porous elasticity problems but, unlike Chapter 9, retain the inertia terms. The formalism for this chapter is drawn, in part, from the formulation given by Bowen and Lockett [Ref. 1].

### **10.1 Governing Partial Differential Equations**

We shall adopt the isothermal constitutive equations listed in Section 9.1 except that we shall restore the neglected inertia terms. As in Section 9.1, we shall make assumptions sufficient to treat the temperature as a constant. Thus, the constitutive equations and field equations we shall utilize are as follows:

$$\frac{\varphi_f - \varphi_f^+}{\varphi_f^+} = \frac{\lambda_{sb}}{\varphi_f^+ K_f} \operatorname{tr} \mathbf{E}_s + \left(1 - \frac{\lambda_{ff}}{\varphi_f^+ K_f}\right) ((\rho_f - \rho_f^+) / \rho_f^+)$$
(10.1.1)

$$\Psi_{I} = \frac{1}{2} \overline{\lambda}_{ss} (\operatorname{tr} \mathbf{E}_{s})^{2} - \overline{\lambda}_{sf} (\operatorname{tr} \mathbf{E}_{s}) (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} + \frac{1}{2} \overline{\lambda}_{ff} \left( (\rho_{f} - \rho_{f}^{+}) / \rho_{f}^{+} \right)^{2} + \mu_{ss} \operatorname{tr}(\mathbf{E}_{s} \mathbf{E}_{s}),$$
(10.1.2)

$$\rho \eta = -\alpha + \overline{\tau}_s (\operatorname{tr} \mathbf{E}_s) - \overline{\tau}_f (\rho_f - \rho_f^+) / \rho_f^+$$
(10.1.3)

$$\rho_{s_{R}}\mathbf{K}_{s} = -(\mathbf{T}_{I} + \varphi_{f}^{+}P_{f}\mathbf{I}) = -\lambda_{ss}(\operatorname{tr}\mathbf{E}_{s})\mathbf{I} - 2\mu_{ss}\mathbf{E}_{s} + \overline{\lambda}_{sf}((\rho_{f} - \rho_{f}^{+})/\rho_{f}^{+})\mathbf{I}$$
(10.1.4)

and

$$\varphi_f^+ P_f = -\overline{\lambda}_{sf} \left( \operatorname{tr} \mathbf{E}_s \right) + \overline{\lambda}_{ff} \left( \rho_f - \rho_f^+ \right) / \rho_f^+$$
(10.1.5)

The field equations which follow from these constitutive equations are the isothermal versions of equations (4.2.14) and (4.2.15). These equations, repeated here, are

$$\rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} = (\overline{\lambda}_{ss} + \mu_{ss}) \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{s}) + \mu_{ss} \operatorname{Div}(\operatorname{GRAD} \mathbf{w}_{s}) + \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_{f}) - \xi \left(\frac{\partial \mathbf{w}_{s}}{\partial t} - \frac{\partial \mathbf{w}_{f}}{\partial t}\right)$$
(10.1.6)

and

$$\rho_f^+ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} = \overline{\lambda}_{sf} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_s) + \overline{\lambda}_{ff} \operatorname{GRAD}(\operatorname{Div} \mathbf{w}_f) - \xi \left(\frac{\partial \mathbf{w}_f}{\partial t} - \frac{\partial \mathbf{w}_s}{\partial t}\right)$$
(10.1.7)

Given the form of (10.1.6) and (10.1.7), the results in Section 9.2 remain valid when the inertia terms are restored.

#### **10.2 Boundary Initial Value Problems**

As a generalization of the boundary initial value problem considered in Chapter 9, we shall consider the following:

**Boundary Conditions:** 

$$\mathbf{T}_{I}(\mathbf{w})\mathbf{n} = -\mathbf{s}(\mathbf{X},t) \quad \text{for} \quad (\mathbf{X},t) \in \partial \mathscr{V}_{1} \times (0,\infty)$$
(10.2.1)

$$P_{f}(\mathbf{w}) = r(\mathbf{X}, t) \qquad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{1} \times (0, \infty)$$
(10.2.2)

$$\mathbf{w} = \mathbf{k} (\mathbf{X}, t) \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_2 \times (0, \infty)$$
(10.2.3)

Initial Conditions:

$$\mathbf{w}(\mathbf{X},0) = \mathbf{f}(\mathbf{X}) \qquad \text{for} \quad \mathbf{X} \in \mathscr{V}$$
(10.2.4)

$$\frac{\partial \mathbf{w}(\mathbf{X},0)}{\partial t} = \mathbf{g}(\mathbf{X}) \qquad \text{for} \quad \mathbf{X} \in \mathscr{V}$$
(10.2.5)

where, as in Section 9.3, the boundary  $\partial \mathscr{V}$  of the volume  $\mathscr{V}$  consists of two parts  $\partial \mathscr{V}_1$  and  $\partial \mathscr{V}_2$ . On the part  $\partial \mathscr{V}_1$  the stress vector  $\mathbf{T}_I \mathbf{n}$  and the pore pressure  $P_f$  are prescribed by the functions  $\mathbf{s}(\mathbf{X},t)$  and  $r(\mathbf{X},t)$ , respectively. On the remainder,  $\partial \mathscr{V}_2$ , the displacement is prescribed by the function  $\mathbf{k}(\mathbf{X},t)$ . The initial displacement given by the function  $\mathbf{f}(\mathbf{X})$ , and the initial velocity is given by the function  $\mathbf{g}(\mathbf{X})$ . The five functions  $\mathbf{s}(\mathbf{X},t)$ ,  $r(\mathbf{X},t)$ ,  $\mathbf{k}(\mathbf{X},t)$ ,  $\mathbf{f}(\mathbf{X})$  and  $\mathbf{g}(\mathbf{X})$  represent the data for this problem. It is interesting to note that when inertia is restored, we do not have the data compatibility that originated with (9.3.5). As illustrated in the examples worked in Sections 9.4 through 9.6, this compatibility condition links the data at t = 0.

#### **10.3 A One Dimensional Example**

If we make the same simplification as adopted in Section 9.4, the resulting one dimensional problem is to find the solution of

$$\mathbf{M}\frac{\partial^2 \mathbf{w}(X,t)}{\partial t^2} + \xi \mathbf{E}\frac{\partial \mathbf{w}(X,t)}{\partial t} = \mathbf{Q}\frac{\partial^2 \mathbf{w}(X,t)}{\partial X^2}$$
(10.3.1)

where,

$$\mathbf{w} = \mathbf{w}(X,t) = \begin{bmatrix} w_f(X,t) \\ w_s(X,t) \end{bmatrix}$$
(10.3.2)

$$\mathbf{M} = \begin{bmatrix} \rho_f^+ & 0\\ 0 & \rho_{s_R} \end{bmatrix}$$
(10.3.3)

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix}$$
(10.3.4)

and

$$\mathbf{E} = \begin{bmatrix} 1 & -1 \\ -1 & 1 \end{bmatrix} \tag{10.3.5}$$

As mentioned in Chapter 9, the above definitions have appeared in earlier chapters. Equation (10.3.1) is the isothermal version of (8.1.7).

For this one dimensional problem, the constitutive equation (10.1.4) and (10.1.5) can again be written in the matrix form

$$\begin{bmatrix} -\varphi_f^* P_f \\ T_I + \varphi_f^* P_f \end{bmatrix} = \mathbf{Q} \frac{\partial \mathbf{w}}{\partial X}$$
(10.3.6)

The one dimensional form of the boundary initial value problem introduced in Section 9.2 is easily seen to be

$$\mathbf{q}(t) \equiv \begin{bmatrix} -\varphi_f^* r(t) \\ -s(t) + \varphi_f^* r(t) \end{bmatrix} = \mathbf{Q} \frac{\partial \mathbf{w}(0, t)}{\partial X} \quad \text{for} \quad t \in (0, \infty) \quad (10.3.7)$$

$$\mathbf{w}(h,t) = \mathbf{k}(t) \qquad \text{for} \qquad t \in (0,\infty) \qquad (10.3.8)$$

$$\mathbf{w}(X,0) = \mathbf{f}(X) \qquad \text{for} \qquad X \in (0,h) \tag{10.3.9}$$

and

$$\frac{\partial \mathbf{w}(X,0)}{\partial t} = \mathbf{g}(X) \quad \text{for} \quad X \in (0,h)$$
(10.3.10)

We can display the partial differential equation, the boundary conditions and the initial condition on the following schematic

$$\mathbf{q}(t) = \mathbf{Q} \frac{\partial \mathbf{w}(0,t)}{\partial X} \begin{bmatrix} \mathbf{M} \frac{\partial^2 \mathbf{w}(X,t)}{\partial t^2} \\ +\xi \mathbf{E} \frac{\partial \mathbf{w}(X,t)}{\partial t} \\ = \mathbf{Q} \frac{\partial^2 \mathbf{w}(X,t)}{\partial X^2} \end{bmatrix} \mathbf{w}(h,t) = \mathbf{k}(t)$$
$$\mathbf{w}(X,0) = \mathbf{f}(X) \qquad h = X$$
$$\frac{\partial \mathbf{w}(X,0)}{\partial t} = \mathbf{g}(X)$$

As in Section 9.4, in order to exploit eigenfunction techniques to solve this boundary initial value problem, we need to change the dependent variable in a manner sufficient to produce homogeneous boundary conditions. The transformation that achieves this objective is again achieved by defining a function v(X,t) by

$$\mathbf{v}(X,t) = \mathbf{w}(X,t) - \mathbf{Q}^{-1}\mathbf{q}(t)(X-h) - \mathbf{k}(t)$$
(10.3.11)

The partial differential equation that v(X,t) obeys follows from (10.3.1) and (10.3.11). This result is

$$\mathbf{M}\frac{\partial^2 \mathbf{v}}{\partial t^2} + \xi \mathbf{E}\frac{\partial \mathbf{v}}{\partial t} = \mathbf{Q}\frac{\partial^2 \mathbf{v}}{\partial X^2} + \mathbf{p}(t)$$
(10.3.12)

where

$$\mathbf{p}(X,t) = -\mathbf{M}\left(\mathbf{Q}^{-1}\ddot{\mathbf{q}}(t)(X-h) + \ddot{\mathbf{k}}(t)\right) - \xi \mathbf{E}\left(\mathbf{Q}^{-1}\dot{\mathbf{q}}(t)(X-h) + \dot{\mathbf{k}}(t)\right)$$
(10.3.13)

The boundary conditions and the initial value that the function  $\mathbf{v}(X,t)$  must obey follow from (10.3.11) and the corresponding conditions on  $\mathbf{w}(X,t)$ . These results are

$$\frac{\partial \mathbf{v}(0,t)}{\partial X} = \mathbf{0} \tag{10.3.14}$$

$$\mathbf{v}(h,t) = \mathbf{0} \tag{10.3.15}$$

$$\mathbf{v}(X,0) = \mathbf{f}(X) - \mathbf{Q}^{-1}\mathbf{q}(0)(X-h) - \mathbf{k}(0)$$
(10.3.16)

and

$$\frac{\partial \mathbf{v}(X,0)}{\partial t} = \mathbf{g}(X) - \mathbf{Q}^{-1} \dot{\mathbf{q}}(0) (X-h) - \dot{\mathbf{k}}(0)$$
(10.3.17)

As in Section 9.4, we again seek a solution of our boundary initial value problem for v(X,t) in the form

$$\mathbf{v}(X,t) = \sum_{n=1}^{\infty} \mathbf{T}_{n}(t) \cos \frac{(2n-1)\pi X}{2h}$$
(10.3.18)

where  $\mathbf{T}_n(t)$  is a 2×1 column matrix that determines the time dependence of  $\mathbf{v}(X,t)$ . As constructed, the proposed solution (10.3.18) obeys the boundary conditions (10.3.14) and (10.3.15)

. Our next step is to determine the time dependence of the solution. If (10.3.18) is substituted into (10.3.12) the result is

$$\sum_{n=1}^{\infty} \left( \mathbf{M} \frac{d^2 \mathbf{T}_n(t)}{dt^2} + \xi \mathbf{E} \frac{d \mathbf{T}_n(t)}{dt} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q} \mathbf{T}_n(t) \right) \cos \frac{(2n-1)\pi X}{2h} = \mathbf{p}(X,t) \quad (10.3.19)$$

The orthogonality of the functions  $\left\{\cos\frac{(2n-1)\pi X}{2h}\right\}$  allow us to derive from (10.3.19) the result

$$\mathbf{M}\frac{d^{2}\mathbf{T}_{n}(t)}{dt^{2}} + \xi \mathbf{E}\frac{d\mathbf{T}_{n}(t)}{dt} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\mathbf{T}_{n}(t) = \mathbf{p}_{n}(t)$$
(10.3.20)

where, from (10.3.13),

$$\begin{aligned} \mathbf{p}_{n}(t) &= \frac{2}{h} \int_{0}^{h} \mathbf{p}(X,t) \cos \frac{(2n-1)\pi X}{2h} dX \\ &= \frac{2}{h} \int_{0}^{h} \left( -\mathbf{M} \Big( \mathbf{Q}^{-1} \ddot{\mathbf{q}}(t) (X-h) + \ddot{\mathbf{k}}(t) \Big) - \xi \mathbf{E} \Big( \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) (X-h) + \dot{\mathbf{k}}(t) \Big) \Big) \cos \frac{(2n-1)\pi X}{2h} dX \\ &= \Big( \mathbf{M} \mathbf{Q}^{-1} \ddot{\mathbf{q}}(t) + \xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) \Big) \frac{2}{h} \int_{0}^{h} (h-X) \cos \frac{(2n-1)\pi X}{2h} dX \\ &- \Big( \mathbf{M} \ddot{\mathbf{k}}(t) + \xi \mathbf{E} \dot{\mathbf{k}}(t) \Big) \frac{2}{h} \int_{0}^{h} \cos \frac{(2n-1)\pi X}{2h} dX \\ &= \Big( \mathbf{M} \mathbf{Q}^{-1} \ddot{\mathbf{q}}(t) + \xi \mathbf{E} \mathbf{Q}^{-1} \dot{\mathbf{q}}(t) \Big) \frac{8h}{\pi^{2}} \frac{1}{(2n-1)^{2}} + \frac{4}{(2n-1)} \Big( -1 \Big)^{n} \Big( \mathbf{M} \ddot{\mathbf{k}}(t) + \xi \mathbf{E} \dot{\mathbf{k}}(t) \Big) \end{aligned}$$
(10.3.21)

after the elementary integrals are evaluated.

The solution of (10.3.21) requires an initial condition. These conditions follow from (10.3.11), (10.3.16) and (10.3.17). These three equations yield

$$\sum_{n=1}^{\infty} \mathbf{T}_{n}(0) \cos \frac{(2n-1)\pi X}{2h} = \mathbf{f}(X) - \mathbf{Q}^{-1} \mathbf{q}(0) (X-h) - \mathbf{k}(0)$$
(10.3.22)

and

$$\sum_{n=1}^{\infty} \frac{d\mathbf{T}_{n}(0)}{dt} \cos \frac{(2n-1)\pi X}{2h} = \mathbf{g}(X) - \mathbf{Q}^{-1} \dot{\mathbf{q}}(0) (X-h) - \dot{\mathbf{k}}(0)$$
(10.3.23)

The orthogonality of the functions  $\left\{\cos\frac{(2n-1)\pi X}{2h}\right\}$  can be used again to derive from (10.3.22) and (10.3.23) the results

$$\mathbf{T}_{n}(0) = \frac{2}{h} \int_{0}^{h} \mathbf{f}(X) \cos \frac{(2n-1)\pi X}{2h} dX + \frac{2}{h} \mathbf{Q}^{-1} \mathbf{q}(0) \int_{0}^{h} (h-X) \cos \frac{(2n-1)\pi X}{2h} dX - \frac{2}{h} \mathbf{k}(0) \int_{0}^{h} \cos \frac{(2n-1)\pi X}{2h} dX$$
(10.3.24)

and

$$\frac{d\mathbf{T}_{n}(0)}{dt} = \frac{2}{h} \int_{0}^{h} \mathbf{g}(X) \cos \frac{(2n-1)\pi X}{2h} dX + \frac{2}{h} \mathbf{Q}^{-1} \dot{\mathbf{q}}(0) \int_{0}^{h} (h-X) \cos \frac{(2n-1)\pi X}{2h} dX - \frac{2}{h} \dot{\mathbf{k}}(0) \int_{0}^{h} \cos \frac{(2n-1)\pi X}{2h} dX$$
(10.3.25)

The integrals in the second and third terms of (10.3.24) can be evaluated to yield the result

$$\mathbf{T}_{n}(0) = \mathbf{f}_{n} + \frac{8h}{(2n-1)^{2} \pi^{2}} \mathbf{Q}^{-1} \mathbf{q}(0) + \frac{4}{(2n-1)\pi} (-1)^{n} \mathbf{k}(0)$$
(10.3.26)

where, we have introduced the symbol  $\mathbf{f}_n$  defined by

$$\mathbf{f}_{n} = \frac{2}{h} \int_{0}^{h} \mathbf{f}(X) \cos \frac{(2n-1)\pi X}{2h} dX$$
(10.3.27)

Likewise, the integrals in the second and third terms of (10.3.25) can be evaluated to yield

$$\frac{d\mathbf{T}_{n}(0)}{dt} = \mathbf{g}_{n} + \frac{8h}{\left(2n-1\right)^{2}\pi^{2}}\mathbf{Q}^{-1}\dot{\mathbf{q}}(0) + \frac{4}{\left(2n-1\right)\pi}\left(-1\right)^{n}\dot{\mathbf{k}}(0)$$
(10.3.28)

where, we have introduced the symbol  $\mathbf{g}_n$  defined by

$$\mathbf{g}_{n} = \frac{2}{h} \int_{0}^{h} \mathbf{g}(X) \cos \frac{(2n-1)\pi X}{2h} dX$$
(10.3.29)

Following the approach in Section 9.4, we shall utilize the Laplace transform method to solve (10.3.20). The Laplace transform of this equation yields

$$\left(s^{2}\mathbf{M}+s\xi\mathbf{E}+\frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\right)\overline{\mathbf{T}}_{n}\left(s\right)=\left(s\mathbf{M}+\xi\mathbf{E}\right)\mathbf{T}_{n}\left(0\right)+\mathbf{Q}\frac{d\mathbf{T}_{n}\left(0\right)}{dt}+\overline{\mathbf{p}}_{n}\left(s\right) \quad (10.3.30)$$

It is convenient at this point to define the  $2 \times 2$  matrix

$$\overline{\mathbf{K}}_{n}(s) = \frac{2}{h} \left( s^{2}\mathbf{M} + s\xi \mathbf{E} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q} \right)^{-1}$$
(10.3.31)

Because we are dealing with  $2 \times 2$  matrices it is true that

$$\overline{\mathbf{K}}_{n}(s) = \frac{2}{h} \frac{s^{2} \operatorname{adj} \mathbf{M} + s\xi \operatorname{adj} \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}}{\operatorname{det} \left( s^{2} \mathbf{M} + s\xi \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \mathbf{Q} \right)}$$
(10.3.32)

Therefore, we need to calculate the inverse Laplace transform of

Chapter 10

$$\overline{\mathbf{T}}_{n}(s) = \frac{h}{2}\overline{\mathbf{K}}_{n}(s)\left(\left(s\mathbf{M} + \xi\mathbf{E}\right)\mathbf{T}_{n}(0) + \mathbf{M}\frac{d\mathbf{T}_{n}(0)}{dt} + \overline{\mathbf{p}}_{n}(s)\right)$$
(10.3.33)

If we now use (10.3.21), (10.3.26) and (10.3.28), equation (10.3.33) reduces to

$$\overline{\mathbf{T}}_{n}(s) = \frac{h}{2}\overline{\mathbf{K}}_{n}(s) \begin{pmatrix} (s\mathbf{M} + \xi\mathbf{E})\mathbf{f}_{n} + \mathbf{M}\mathbf{g}_{n} + \frac{8h}{(2n-1)^{2}\pi^{2}}(\mathbf{M}s^{2} + \xi s\mathbf{E})\mathbf{Q}^{-1}\overline{\mathbf{q}}(s) \\ + \frac{4}{(2n-1)\pi}(-1)^{n}(\mathbf{M}s^{2} + \xi s\mathbf{E})\overline{\mathbf{k}}(s)() \end{pmatrix}$$
(10.3.34)

This expression can be simplified further by use of (10.3.31) written in the form

$$\frac{h}{2}\overline{\mathbf{K}}_{n}(s)\left(s^{2}\mathbf{M}+s\boldsymbol{\xi}\mathbf{E}\right)=\mathbf{I}-\frac{h}{2}\frac{\left(2n-1\right)^{2}\pi^{2}}{4h^{2}}\overline{\mathbf{K}}_{n}(s)\mathbf{Q}$$
(10.3.35)

The result of this substitution is

$$\overline{\mathbf{T}}_{n}(s) = \frac{h}{2}\overline{\mathbf{K}}_{n}(s)(s\mathbf{M} + \xi\mathbf{E})\mathbf{f}_{n} + \frac{h}{2}\overline{\mathbf{K}}_{n}(s)\mathbf{M}\mathbf{g}_{n} - \overline{\mathbf{K}}_{n}(s)\overline{\mathbf{q}}(s) - \frac{(2n-1)\pi}{2h}(-1)^{n}\overline{\mathbf{K}}_{n}(s)\mathbf{Q}\overline{\mathbf{k}}(s) + \frac{8h}{(2n-1)^{2}\pi^{2}}\mathbf{Q}^{-1}\overline{\mathbf{q}}(s) + \frac{4}{(2n-1)\pi}(-1)^{n}\overline{\mathbf{k}}(s)$$
(10.3.36)

If  $\mathbf{K}_{n}(t)$  is the inverse Laplace transform of  $\overline{\mathbf{K}}_{n}(s)$ , then the inverse of (10.3.36) is

$$\mathbf{T}_{n}(t) = \frac{h}{2} \left( \frac{\partial \mathbf{K}_{n}(t)}{\partial t} \mathbf{M} + \xi \mathbf{K}_{n}(t) \mathbf{E} \right) \mathbf{f}_{n} + \frac{h}{2} \mathbf{K}_{n}(t) \mathbf{M} \mathbf{g}_{n} - \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \mathbf{q}(\tau) d\tau - \frac{(2n-1)\pi}{2h} (-1)^{n} \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \mathbf{Q} \mathbf{k}(\tau) d\tau + \frac{8h}{(2n-1)^{2} \pi^{2}} \mathbf{Q}^{-1} \mathbf{q}(t) + \frac{4}{(2n-1)\pi} (-1)^{n} \mathbf{k}(t)$$

$$(10.3.37)$$

Among the elementary properties of the Laplace transform that were used to derive (10.3.37) from (10.3.36) is the formula

$$\lim_{s \to \infty} \left( s \overline{\mathbf{K}}_n(s) \right) = \mathbf{K}_n(0) = \mathbf{0}$$
(10.3.38)

The first part of (10.3.38) is the initial value theorem for Laplace transforms. The fact that  $\lim_{s\to\infty} (s\overline{\mathbf{K}}_n(s)) = \mathbf{0}$  follows from the definition (10.3.32). Another result that follows from(10.3.32) and the initial value theorem is

$$\lim_{s \to \infty} \left( s^2 \overline{\mathbf{K}}_n(s) \right) = \lim_{s \to \infty} \left( s \left( s \overline{\mathbf{K}}_n(s) \right) \right) = \frac{\partial \mathbf{K}_n(0)}{\partial t} = \frac{2}{h} \mathbf{M}^{-1}$$
(10.3.39)

Given the formula (10.3.37), we can use (10.3.18) and (10.3.11) to obtain the following solution for  $\mathbf{w}(X,t)$ 

$$\mathbf{w}(X,t) = \sum_{n=1}^{\infty} \mathbf{T}_{n}(t) \cos \frac{(2n-1)\pi X}{2h} + \mathbf{Q}^{-1} \mathbf{q}(t)(X-h) + \mathbf{k}(t)$$

$$= \frac{h}{2} \sum_{n=1}^{\infty} \mathbf{K}_{n}(t) \mathbf{M} \mathbf{g}_{n} \cos \frac{(2n-1)\pi X}{2h} + \frac{h}{2} \sum_{n=1}^{\infty} \left( \frac{\partial \mathbf{K}_{n}(t)}{\partial t} \mathbf{M} + \xi \mathbf{K}_{n}(t) \mathbf{E} \right) \mathbf{f}_{n} \cos \frac{(2n-1)\pi X}{2h}$$

$$- \sum_{n=1}^{\infty} \left( \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \mathbf{q}(\tau) d\tau \right) \cos \frac{(2n-1)\pi X}{2h}$$

$$- \frac{\pi}{2h} \sum_{n=1}^{\infty} \left( (2n-1)(-1)^{n} \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \mathbf{Q} \mathbf{k}(\tau) d\tau \right) \cos \frac{(2n-1)\pi X}{2h}$$
(10.3.40)

In order to utilize the solution (10.3.40), we must use (10.3.32) to invert

$$\overline{\mathbf{K}}_{n}(s) = \frac{2}{h} \frac{s^{2} \operatorname{adj} \mathbf{M} + s\xi \operatorname{adj} \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}}{\operatorname{det} \left( s^{2} \mathbf{M} + s\xi \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \mathbf{Q} \right)}$$
(10.3.41)

Ultimately, the properties of this inverse depend upon the zeros of the fourth order polynomial

$$\det\left(\beta_n^2 \mathbf{M} + \beta_n \xi \mathbf{E} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q}\right) = 0$$
(10.3.42)

We shall examine the nature of these zeros in the next Section.

**10.4 Properties of the Roots of** det 
$$\left(\beta_n^2 \mathbf{M} + \beta_n \xi \mathbf{E} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q}\right) = 0$$

The fourth order polynomial (10.3.42) will have either four real roots, two real and one complex conjugate pair or two complex conjugate pairs. If we expand the determinant in (10.3.42), the polynomial is

Chapter 10

$$(\det \mathbf{M})\beta_{n}^{4} + \xi \operatorname{tr}((\operatorname{adj}\mathbf{M})\mathbf{E})\beta_{n}^{3} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\operatorname{tr}((\operatorname{adj}\mathbf{M})\mathbf{Q})\beta_{n}^{2} + \xi \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})\beta_{n} + \frac{(2n-1)^{4}\pi^{4}}{16h^{4}}\det \mathbf{Q} = 0$$
(10.4.1)

In parts of this section, it will be convenient to utilize the definitions (7.2.50), (7.2.103) and (8.1.19) to write the polynomial (10.4.1) in the form

$$\beta_{n}^{4} + \omega_{\xi}\beta_{n}^{3} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\left(u_{n(1)}^{2} + u_{n(2)}^{2}\right)\beta_{n}^{2} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\omega_{\xi}u_{0}^{2}\beta_{n} + \frac{(2n-1)^{4}\pi^{4}}{16h^{4}}u_{n(1)}^{2}u_{n(2)}^{2} = 0$$
(10.4.2)

Equation (10.4.2) is only valid in the case where det  $\mathbf{M} \neq 0$ . The case  $\mathbf{M} = \mathbf{0}$  formally corresponds to the inertia free approximation discussed in Chapter 9 and elsewhere in these Lectures.

Because  $\xi \ge 0$ , and because the symmetric matrix **Q** defined by (7.2.46) is positive definite and the symmetric matrix **E** defined by (7.2.87) is positive semi definite, it is a theorem that the polynomial (10.3.42) has no roots with positive real parts [Ref. 2, page 246]. In our case, unless we wish to study the case  $\mathbf{M} = \mathbf{0}$  mentioned above, the symmetric matrix **M** is also positive definite.

It is interesting to characterize the case where the real parts of the roots of (10.4.2) are zero. In the special case  $\xi = 0$ , it readily follows from (10.4.2) that the four roots are the complex numbers

$$\beta_{n}|_{\xi=0} = \begin{cases} \pm i \frac{(2n-1)\pi}{2h} u_{n(1)} \\ \pm i \frac{(2n-1)\pi}{2h} u_{n(2)} \end{cases}$$
(10.4.3)

Conversely, if one forces the roots of (10.4.2) to have zero real parts, i.e., to be of the form

$$\beta_n = i\alpha \tag{10.4.4}$$

where  $\alpha$  is real, it is readily established from (10.4.2) that

$$\alpha = \begin{cases} \pm \frac{(2n-1)\pi}{2h} u_{n(1)} \\ \pm \frac{(2n-1)\pi}{2h} u_{n(2)} \end{cases}$$
(10.4.5)

and

$$\omega_{\xi} \left( \alpha^2 - \frac{\left(2n-1\right)^2 \pi^2}{4h^2} u_0^2 \right) = 0$$
 (10.4.6)

Equation (10.4.6) is obeyed when  $\xi = 0$  or when  $u_0$  equals  $u_{n(1)}$  or  $u_{n(2)}$ . As was pointed out in Section 7.2, it is a theoretical result that

$$u_{n(1)}^2 \ge u_0^2 \ge u_{n(2)}^2 \tag{10.4.7}$$

The numerical values in Table 1 of Section 6.2 illustrate that for these example materials  $u_0$  and  $u_{n(1)}$  are close in their values. To the accuracy of the numerical approximation used in this table,  $u_0 = u_{n(1)}$  for Weber Sandstone. The summary of the discussion just completed is that when the real part of the roots of (10.4.8) are zero, then either  $\xi = 0$  or  $u_0$  equals either  $u_{n(1)}$  or  $u_{n(2)}$ . In the following, we shall always make the assumption that

$$\xi > 0$$
 (10.4.9)

We shall not, however, rule out the case where  $u_0$  equals  $u_{n(2)}$ 

Given what we know about the roots of (10.4.2), in those cases where the real parts of the roots are less than zero, we can use the final value theorem from Laplace transform theory and conclude from (10.3.41) that

$$\lim_{t \to \infty} \mathbf{K}_{n}(t) = \lim_{s \to 0} \left( s \overline{\mathbf{K}}_{n}(s) \right) = \mathbf{0}$$
(10.4.10)

$$\lim_{t \to \infty} \frac{\partial \mathbf{K}_n(t)}{\partial t} = \lim_{s \to 0} \left( s \left( s \overline{\mathbf{K}}_n(s) \right) \right) = \lim_{s \to 0} \left( s^2 \overline{\mathbf{K}}_n(s) \right) = \mathbf{0}$$
(10.4.11)

and

$$\lim_{t \to \infty} \int_{0}^{t} \mathbf{K}_{n} (t - \tau) d\tau = \lim_{s \to 0} \overline{\mathbf{K}}_{n} (s) = \frac{8h}{\pi^{2}} \frac{1}{(2n - 1)^{2}} \mathbf{Q}^{-1}$$
(10.4.12)

These results are available without our actually computing the inverse of (10.3.41) and, from (10.4.9) and the above argument, they hold when  $u_0$  does not equal either  $u_{n(1)}$  or  $u_{n(2)}$ 

Analytical solutions to fourth order polynomials such as (10.4.2) are difficult to interpret. Because of this fact, it is convenient to look at certain analytical approximations. The first is based upon the observation just made, namely that  $u_0$  and  $u_{n(1)}$  are close. We shall measure their closeness by a dimensionless parameter defined by

$$\varepsilon = \frac{u_1^2 - u_0^2}{u_0^2} \tag{10.4.13}$$

and look for roots of the form of an asymptotic expansion

$$\beta_n = a_0 + a_1 \varepsilon + O(\varepsilon^2) \tag{10.4.14}$$

While algebraically tedious, it is possible to derive from (10.4.2) the following approximations for the four roots<sup>1,2</sup>

$$\beta_n = \omega_{\xi} f_n \left( \frac{u_0}{h\omega_{\xi}}, \frac{u_{n(1)}}{h\omega_{\xi}}, \frac{u_{n(2)}}{h\omega_{\xi}}, \frac{(2n-1)\pi}{2} \right)$$

<sup>&</sup>lt;sup>1</sup> The derivation of the solutions (10.4.15) and (10.4.16) is facilitated if one uses a symbolic manipulator such as found in Maple or Matlab.

<sup>&</sup>lt;sup>2</sup> Equations (10.4.15) and (10.4.16) display a scaling that can also be established from (10.4.2), namely, that the roots of (10.4.2) will always be expressible as function of the dimensionless ratios of the form

$$\begin{split} \beta_{n(1)} \\ \beta_{n(2)} \\ \end{array} &= -\frac{1}{2} \omega_{\xi} \pm \frac{1}{2} \sqrt{\omega_{\xi}^{2} - \frac{(2n-1)^{2} \pi^{2}}{h^{2}}} u_{n(2)}^{2} \\ &+ \begin{pmatrix} \omega_{\xi} \left( -\frac{1}{2} \omega_{\xi} \pm \frac{1}{2} \sqrt{\omega_{\xi}^{2} - \frac{(2n-1)^{2} \pi^{2}}{h^{2}}} u_{n(2)}^{2} \right) u_{0}^{2} \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \\ \left( \frac{\omega_{\xi} \left( u_{0}^{2} + u_{n(2)}^{2} \right) \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} + 2 \left( -\frac{1}{2} \omega_{\xi} \pm \frac{1}{2} \sqrt{\omega_{\xi}^{2} - \frac{(2n-1)^{2} \pi^{2}}{h^{2}}} u_{n(2)}^{2} \right) \left( u_{0}^{2} - u_{n(2)}^{2} \right) \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \\ &+ \omega_{\xi}^{2} \left( -\frac{1}{2} \omega_{\xi} \pm \frac{1}{2} \sqrt{\omega_{\xi}^{2} - \frac{(2n-1)^{2} \pi^{2}}{h^{2}}} u_{n(2)}^{2} \right) \\ &+ O(\varepsilon^{2}) \end{split}$$

$$(10.4.15)$$

and

$$\begin{split} & \left. \beta_{n(3)} \atop \beta_{n(4)} \right\} = -\frac{1}{2} \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \left( u_0^2 - u_{n(2)}^2 \right) \frac{\omega_{\xi}}{\left( \omega_{\xi}^2 + \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \frac{\left(u_0^2 - u_{n(2)}^2\right)^2}{u_0^2} \right)^{\xi}} \varepsilon \\ & \pm i \frac{\left(2n-1\right)\pi}{2h} u_0 \pm i \frac{1}{2} u_0 \frac{\left(2n-1\right)^3 \pi^3}{8h^3} \frac{\left(u_0^2 - u_{n(2)}^2\right)^2}{\left(\omega_{\xi}^2 u_0^2 + \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \left(u_0^2 - u_{n(2)}^2\right)^2\right)} \varepsilon \quad (10.4.16) \\ & + O\left(\varepsilon^2\right) \end{split}$$

These approximate expressions obey the theoretical conclusions summarized above. Namely, since we have assumed (10.4.9), the real part of the roots are negative unless  $u_0$  equals either  $u_{n(1)}$  or  $u_{n(2)}$ . Equation (10.4.15) shows that the two roots  $\beta_{n(1)}$  and  $\beta_{n(2)}$  are real for all n such that

$$\omega_{\xi}^{2} - \frac{\left(2n-1\right)^{2} \pi^{2}}{h^{2}} u_{n(2)}^{2} \ge 0$$
(10.4.17)

It follows from (10.4.17) that the *n*'s that cause  $\beta_{n(1)}$  and  $\beta_{n(2)}$  to be real obey

$$n \le \frac{1}{2} \left( 1 + \frac{\omega_{\xi} h}{\pi u_{n(2)}} \right) \tag{10.4.18}$$

As we shall see next, the largest integer allowed by (10.4.18) for six materials introduced in Chapter 6 is large.

Because we are going to display numerical values for the roots of (10.4.2), it is helpful to repeat some of the numerical results contained in earlier Chapters. From Table 2 of Section 6.2 and Tables 1 and 2 of Section 7.2

Property	Ruhr	Tennessee	Charcoal	Berea	Westerly	Weber	
1100000	Sandstone	Marble	Granite	Sandstone	Granite	Sandstone	
$\gamma_{s_R}$ (g/cm <sup>3</sup> )	2.6	2.7	2.7	2.6	2.7	2.6	
$arphi_{f}^{+}$	.02	.02	.02	.19	.01	.06	
$\overline{\lambda}_{ff}$ (kbar)	.165	.486	.326	4.44	.075	1.03	
$\overline{\lambda}_{sf}$ (kbar)	5.21	4.02	4.13	13.990	3.48	9.91	
$\overline{\lambda}_{ss}$ (kbar)	206.41	273.21	271.91	84.05	311.71	147.62	
$\mu_{ss}$ (kbar)	133	240	187	60	160	122	
$u_{n(1)}$ (m/sec)	4340	5346	4955	3210	4805	4059	
$u_{n(2)}$ (m/sec)	728	1522	1220	1313	740	1124	
$u_{n(3)}$ (m/sec)	2285	3012	2658	1688	2369	2234	
$u_0$ (m/sec)	4337	5345	4954	3209	4802	4059	
$\xi$ (kbar-sec/cm <sup>2</sup> )	$1.9(10)^{-3}$	3.92	3.92	$1.77(10)^{-4}$	.245	.353(10) <sup>-3</sup>	
$\omega_{\xi}(1/\text{sec})$	9.58(10) <sup>7</sup>	1.98(10) <sup>11</sup>	1.98(10) <sup>11</sup>	$1.02(10)^{6}$	2.46(10) <sup>11</sup>	6.06(10) <sup>7</sup>	
$c (\mathrm{m^{2}/sec})$	.005550	.000011	.000008	1.698365	.000022	.208487	

Table 1 Rock Properties-Fluid Mixture Properties

where numerical values for  $\varpi_{\!\xi}$  and the consolidation coefficient, c , have been added to the table.

If we utilize values from Table 1, we can calculate the factor  $\frac{1}{2} \left( 1 + \frac{\omega_{\xi} h}{\pi u_{n(2)}} \right)$  in equation (10.4.18). The results are

Table 2 The factor $\frac{1}{2} \left( 1 + \frac{\omega_{\xi} n}{\pi u_{n(2)}} \right)$ for $h = 100$ m								
	Ruhr Sandston e	Tennessee Marble	Charcoal Granite	Berea Sandston e	Westerly Granite	Weber Sandston e		
$\frac{1}{2} \left( 1 + \frac{\omega_{\xi} h}{\pi u_{n(2)}} \right)$	2,091,883	2,065,612,19 3	2,576,551,743	12,310	528,983,52 1	85,833		

1(ah)

The numbers show that , unless we adopt an h that is especially small, the integer n that would cause the roots cause  $\beta_{n(1)}$  and  $\beta_{n(2)}$  to be complex is large. The smallest *n* arises for Berea Sandstone and, in that case, it would take 12,000 terms in our solution (10.3.40) before the two real roots would need to be replaced by a complex conjugate pair.

In the following, we shall only consider the case where (10.4.2) has two real roots and a single pair of complex conjugate roots. When we need actual numbers for these roots, we shall take them from the appendices to this Chapter. These appendices provide the first fifty roots of (10.4.2) for the six poroelastic materials whose properties are given in the above table. For our purposes here, it is useful to display, from the information in the appendices, the first ten roots for Ruhr Sandstone. These values are

Table	3 The	First '	Ten I	Roots	of	(10.4)	12)	for	Ruhr	Sand	lstone	for	h =	100	m
raute	JIIIC	Inst	IUIII	NUOLS	UI (	10.4	t. <i>L</i> )	101	Num	San	astone	101	n -	100	/ 111

n	First Root: $oldsymbol{eta}_{n(1)}$	Second Root: $oldsymbol{eta}_{\scriptscriptstyle n(2)}$	Third Root: $oldsymbol{eta}_{n(3)}$	Fourth Root: $oldsymbol{eta}_{\scriptscriptstyle n(4)}$
1	-0.000001	-95745682.888539	68.12334i	-68.12334i
2	-0.000012	-95745682.888527	204.37002i	-204.37002i
3	-0.000034	-95745682.888504	-0.000001+340.6167i	-0.000001-340.6167i
4	-0.000067	-95745682.88847	-0.000002+476.86338i	-0.000002-476.86338i
5	-0.000111	-95745682.888423	-0.000003+613.11006i	-0.000003-613.11006i
6	-0.000166	-95745682.888366	-0.000004+749.35674i	-0.000004-749.35674i
7	-0.000231	-95745682.888297	-0.000006+885.60342i	-0.000006-885.60342i
8	-0.000308	-95745682.888216	-0.000008+1021.8501i	-0.000008-1021.8501i
9	-0.000396	-95745682.888124	-0.00001+1158.09678i	-0.00001-1158.09678i
10	-0.000494	-95745682.88802	-0.000013+1294.34346i	-0.000013-1294.34346i

Another useful approximation to the roots of (10.4.2) is when we look for solutions of the form

$$\beta_n = a_0 + a_1 \varepsilon_n + a_2 \varepsilon_n^2 + O\left(\varepsilon_n^3\right)$$
(10.4.19)

where  $\varepsilon_n$  is a dimensionless parameter defined by

$$\varepsilon_n = \frac{(2n-1)\pi}{2h} \frac{u_0}{\omega_{\varepsilon}}$$
(10.4.20)

Unlike the approximations (10.4.15) and (10.4.16), because  $\varepsilon_n$  increases with *n*, the approximation given in the following are not useful for all *n*. However, when we make the choice h = 100 m,  $\varepsilon_n$  is small for rather large *n* for the example materials shown in the above table. The approximations to the roots of the form (10.4.19) turn out to be

$$\beta_{n(1)} \\ \beta_{n(2)} \\ = \begin{cases} -\frac{u_{n(1)}^{2}u_{n(2)}^{2}}{u_{0}^{4}}\omega_{\xi}\varepsilon_{n}^{2} + O(\varepsilon_{n}^{4}) \\ -\frac{u_{n(1)}^{2} + u_{n(2)}^{2} - u_{0}^{2}}{u_{0}^{2}}\omega_{\xi}\varepsilon_{n}^{2} + O(\varepsilon_{n}^{4}) \end{cases}$$
(10.4.21)

and

$$\beta_{n(3)} \\ \beta_{n(4)} \\ = \begin{cases} -\frac{\left(u_{n(1)}^{2} - u_{0}^{2}\right)\left(u_{0}^{2} - u_{n(2)}^{2}\right)}{2u_{0}^{4}}\omega_{\xi}\varepsilon_{n}^{2} + i\omega_{\xi}\varepsilon_{n} + O\left(\varepsilon_{n}^{3}\right) \\ -\frac{\left(u_{n(1)}^{2} - u_{0}^{2}\right)\left(u_{0}^{2} - u_{n(2)}^{2}\right)}{2u_{0}^{4}}\omega_{\xi}\varepsilon_{n}^{2} - i\omega_{\xi}\varepsilon_{n} + O\left(\varepsilon_{n}^{3}\right) \end{cases}$$
(10.4.22)

The two sets of approximations, (10.4.15) and (10.4.16) and (10.4.21) and (10.4.22), are self consistent in the sense that if we assume  $\varepsilon_n = \frac{(2n-1)\pi}{2h} \frac{u_0}{\omega_{\xi}}$  is small in (10.4.15) and (10.4.16) and,

likewise, assume  $\varepsilon = \frac{u_1^2 - u_0^2}{u_0^2}$  is small in (10.4.21) and (10.4.22) the resulting two sets of new

approximations yield the same results. Also, if (10.4.21) and (10.4.22) are utilized to calculate the first ten roots for Ruhr Sandstone, the results are excellent approximations to the numbers in Table 3 above.

## **10.5** Inversion of $\overline{\mathbf{K}}_n(s)$

Given what we have learned about the roots of (10.3.42), we shall always assume a factorization of the form

$$\det\left(\beta_n^2 \mathbf{M} + \beta_n \xi \mathbf{E} + \frac{(2n-1)^2 \pi^2}{4h^2} \mathbf{Q}\right) = \det \mathbf{M} (\beta_n + \alpha_n) (\beta_n + \gamma_n) ((\beta_n + \zeta_n)^2 + \omega_n^2) \quad (10.5.1)$$

which allows to identify the four roots by<sup>3</sup>

$$\begin{cases} \beta_{n(1)} \\ \beta_{n(2)} \\ \beta_{n(3)} \\ \beta_{n(4)} \end{cases} = \begin{cases} -\alpha_n \\ -\gamma_n \\ -\zeta_n + i\omega_n \\ -\zeta_n - i\omega_n \end{cases}$$
(10.5.2)

Given (10.5.1), it follows from (10.3.41) that we must compute the inverse Laplace transform of

$$\overline{\mathbf{K}}_{n}(s) = \frac{2}{h} \frac{s^{2} \operatorname{adj} \mathbf{M} + s\xi \operatorname{adj} \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}}{\det \mathbf{M}(s+\alpha_{n})(s+\gamma_{n})((s+\zeta_{n})^{2}+\omega_{n}^{2})}$$
(10.5.3)

As the ratio of two polynomials in s, the inverse of (10.5.3) can be shown to be

$$\mathbf{K}_{n}(t) = \frac{2}{h} \begin{pmatrix} e^{-\alpha_{n}t} \mathbf{A}_{n} + e^{-\gamma_{n}t} \mathbf{B}_{n} - e^{-\varsigma_{n}t} \cos \omega_{n}t \left(\mathbf{A}_{n} + \mathbf{B}_{n}\right) \\ + e^{-\varsigma_{n}t} \sin \omega_{n}t \left(\frac{\alpha_{n} - \varsigma_{n}}{\omega_{n}} \mathbf{A}_{n} + \frac{\gamma_{n} - \varsigma_{n}}{\omega_{n}} \mathbf{B}_{n} + \frac{1}{\omega_{n}} \mathbf{M}^{-1} \right) \end{pmatrix}$$
(10.5.4)

where

$$\begin{split} \omega_{\xi} &= \beta_{n(1)} + \beta_{n(2)} + \beta_{n(3)} + \beta_{n(4)} \\ &= -(\alpha_n + \gamma_n + 2\zeta_n) \\ \frac{(2n-1)^2 \pi^2}{4h^2} \left( u_{n(1)}^2 + u_{n(2)}^2 \right) &= \beta_{n(1)} \beta_{n(2)} + \left( \beta_{n(1)} + \beta_{n(2)} \right) \left( \beta_{n(3)} + \beta_{n(4)} \right) + \beta_{n(3)} \beta_{n(4)} \\ &= \alpha_n \gamma_n + 2\zeta_n \left( \alpha_n + \gamma_n \right) + \zeta_n^2 + \omega_n^2 \\ \frac{(2n-1)^2 \pi^2}{4h^2} \omega_{\xi} u_0^2 &= \left( \beta_{n(1)} + \beta_{n(2)} \right) \beta_{n(3)} \beta_{n(4)} + \left( \beta_{n(3)} + \beta_{n(4)} \right) \beta_{n(1)} \beta_{n(2)} \\ &= -\left( \left( \alpha_n + \gamma_n \right) \left( \zeta_n^2 + \omega_n^2 \right) + 2\zeta_n \alpha_n \gamma_n \right) \\ \frac{(2n-1)^4 \pi^4}{16h^4} u_{n(1)}^2 u_{n(2)}^2 &= \beta_{n(1)} \beta_{n(2)} \beta_{n(3)} \beta_{n(4)} \\ &= \alpha_n \gamma_n \left( \zeta_n^2 + \omega_n^2 \right) \end{split}$$

 $<sup>^{3}</sup>$  The coefficients of the polynomial (10.4.2) and its roots (10.5.2) can be shown to be related by the formulas

$$\mathbf{A}_{n} = \frac{1}{\det \mathbf{M}(\gamma_{n} - \alpha_{n})\left(\left(\alpha_{n} - \zeta_{n}\right)^{2} + \omega_{n}^{2}\right)} \left(\alpha_{n}^{2} \operatorname{adj} \mathbf{M} - \alpha_{n} \xi \operatorname{adj} \mathbf{E} + \frac{\left(2n - 1\right)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}\right) (10.5.5)$$

and

$$\mathbf{B}_{n} = \frac{1}{\det \mathbf{M} \left(\alpha_{n} - \gamma_{n}\right) \left(\left(\gamma_{n} - \varsigma_{n}\right)^{2} + \omega_{n}^{2}\right)} \left(\gamma_{n}^{2} \operatorname{adj} \mathbf{M} - \gamma_{n} \xi \operatorname{adj} \mathbf{E} + \frac{\left(2n - 1\right)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}\right) (10.5.6)$$

Given (10.5.4), the solution (10.3.40) is complete. It is helpful to note in passing that (10.5.4) obeys the conditions

$$\mathbf{K}_n(0) = \mathbf{0} \tag{10.5.7}$$

$$\dot{\mathbf{K}}_{n}(0) = \frac{2}{h} \mathbf{M}^{-1} \tag{10.5.8}$$

and, if  $\zeta_n \neq 0$ ,

$$\lim_{t \to \infty} \mathbf{K}_n(t) = \mathbf{0} \tag{10.5.9}$$

The results (10.5.7), (10.5.8) and (10.5.9) were stated previously in equation (10.3.38), (10.3.39) and (10.4.10), respectively.

#### 10.6 The Use of Green's Functions

As in Section 9.7 there are some advantages when dealing with boundary initial value problems with inhomogeneous boundary conditions to formulate the solution in terms of Green's Functions. The formulation is similar to that given in Section 9.7 except that the boundary value problem is the one formulated in Section 10.2 and the underlying differential operator is the one defined by (10.1.6) and (10.1.7).

As in Section 9.7, we shall use the notation  $G(\mathbf{X},t;\mathbf{X}_0,t_0)$  for the 6×6 matrix of functions that constitute the Green's function. As the notation suggests, the Green's function still depends upon pairs of points and pairs of times. It is helpful, when we need to display the components of this matrix to write

$$\mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \begin{bmatrix} \mathbf{g}_{f(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{f(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(6)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(5)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(4)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(3)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ = \begin{bmatrix} \mathbf{g}_{g(1)}(\mathbf{X},t;\mathbf{X},t;\mathbf{X}_{0},t_{0}) & \mathbf{g}_{g(2)}(\mathbf{X},t;\mathbf{X}_{0},t_{0$$

where, again each entry is a three dimensional vector that we have represented by a  $3 \times 1$  column vector.

The Green's function is the  $6 \times 6$  matrix of functions introduced in (10.6.1) defined such that, for every pair  $(\mathbf{X}_0, t_0) \in \mathscr{V} \times (-\infty, \infty)$ , it obeys the boundary initial value problem

$$\mathscr{D}\mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \delta(t-t_{0})\delta(\mathbf{X}-\mathbf{X}_{0})\mathbf{I} \quad \text{for} \quad (\mathbf{X},t) \in \mathscr{V} \times (-\infty,\infty) \quad (10.6.2)$$

$$\mathbf{T}_{I}(\mathbf{G})\mathbf{n} = \mathbf{0} \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{1} \times (-\infty, \infty)$$
(10.6.3)

$$P_{f}(\mathbf{G}) = 0 \qquad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_{1} \times (-\infty, \infty)$$
(10.6.4)

$$\mathbf{G} = 0 \quad \text{for} \quad (\mathbf{X}, t) \in \partial \mathscr{V}_2 \times (-\infty, \infty) \tag{10.6.5}$$

and

$$\mathbf{G}(\mathbf{X}, t, \mathbf{X}_0, t_0) = \mathbf{0} \qquad \text{for} \quad \mathbf{X} \in \mathscr{V} \text{ and } t < t_0$$
(10.6.6)

where  $\mathcal{D}$  is now formal linear operator defined by

$$\mathscr{D}\mathbf{w} = \begin{bmatrix} \rho_{f}^{+} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} + \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) - \bar{\lambda}_{sf} \text{ GRAD}(\text{Div } \mathbf{w}_{s}) - \bar{\lambda}_{ff} \text{ GRAD}(\text{Div } \mathbf{w}_{f}) \\ \rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} - \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) - (\bar{\lambda}_{ss} + \mu_{ss}) \text{ GRAD}(\text{Div } \mathbf{w}_{s}) - \mu_{ss} \text{ Div}(\text{GRAD } \mathbf{w}_{s}) - \bar{\lambda}_{sf} \text{ GRAD}(\text{Div } \mathbf{w}_{f}) \\ = \begin{bmatrix} \rho_{f}^{+} \frac{\partial^{2} \mathbf{w}_{f}}{\partial t^{2}} + \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) \\ \rho_{s_{R}} \frac{\partial^{2} \mathbf{w}_{s}}{\partial t^{2}} - \xi \left( \frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t} \right) \end{bmatrix} + L \mathbf{w} \\ \end{cases}$$
(10.6.7)

As in Section 9.7, we can write

$$\begin{bmatrix} \left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) \\ -\left(\frac{\partial \mathbf{w}_{f}}{\partial t} - \frac{\partial \mathbf{w}_{s}}{\partial t}\right) \end{bmatrix} = \mathbf{E} \frac{\partial \mathbf{w}}{\partial t}$$
(10.6.8)

where **E** is the symmetric  $6 \times 6$  matrix

$$\mathbf{E} = \begin{bmatrix} 1 & 0 & 0 & -1 & 0 & 0 \\ 0 & 1 & 0 & 0 & -1 & 0 \\ 0 & 0 & 1 & 0 & 0 & -1 \\ -1 & 0 & 0 & 1 & 0 & 0 \\ 0 & -1 & 0 & 0 & 1 & 0 \\ 0 & 0 & -1 & 0 & 0 & 1 \end{bmatrix}$$
(10.6.9)

In addition, we can write

$$\begin{bmatrix} \rho_f^+ \frac{\partial^2 \mathbf{w}_f}{\partial t^2} \\ \rho_{s_R} \frac{\partial^2 \mathbf{w}_s}{\partial t^2} \end{bmatrix} = \mathbf{M} \frac{\partial^2 \mathbf{w}}{\partial t^2}$$
(10.6.10)

\_

where **M** is the symmetric  $6 \times 6$  matrix

$$\mathbf{M} = \begin{bmatrix} \rho_{f}^{+} & 0 & 0 & 0 & 0 & 0 \\ 0 & \rho_{f}^{+} & 0 & 0 & 0 & 0 \\ 0 & 0 & \rho_{f}^{+} & 0 & 0 & 0 \\ 0 & 0 & 0 & \rho_{s_{R}} & 0 & 0 \\ 0 & 0 & 0 & 0 & \rho_{s_{R}} & 0 \\ 0 & 0 & 0 & 0 & 0 & \rho_{s_{R}} \end{bmatrix}$$
(10.6.11)

This formalism allows the operator  $\mathscr{L}$  to be defined by

$$\mathscr{D}\mathbf{w} = \mathbf{M} \frac{\partial^2 \mathbf{w}}{\partial t^2} + \xi \mathbf{E} \frac{\partial \mathbf{w}}{\partial t} + L \mathbf{w}$$
(10.6.12)

rather than by (10.6.7).

Given the definition of the Green's function  $G(X,t;X_0,t_0)$ , the theory of Green's functions can be used to write the solution of the boundary initial value problem defined by (10.2.1) through (10.2.5) in terms of its initial and boundary data. The result is

$$\mathbf{w}(\mathbf{X},t) = \int_{\mathbf{v}} \mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},0) \mathbf{M}\mathbf{g}(\mathbf{X}_{0}) d\upsilon_{0} + \frac{\partial}{\partial t} \int_{\mathbf{v}} \mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},0) \mathbf{M}\mathbf{f}(\mathbf{X}_{0}) d\upsilon_{0}$$

$$\xi \int_{\mathbf{v}} \mathbf{G}(\mathbf{X},t;\mathbf{X}_{0},0) \mathbf{E}\mathbf{f}(\mathbf{X}_{0}) d\upsilon_{0} - \int_{0}^{t} \int_{\partial\mathbf{v}_{f}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \cdot \mathbf{s}(\mathbf{X}_{0},t_{0}) \right) \right) d\upsilon_{0} dt_{0}$$

$$-\varphi_{f}^{+} \int_{0}^{t} \int_{\partial\mathbf{v}_{f}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \left( \mathbf{g}_{f(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) - \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \cdot \mathbf{n}r(\mathbf{X}_{0},t_{0}) \right) \right) d\upsilon_{0} dt_{0} \qquad (10.6.13)$$

$$- \int_{0}^{t} \int_{\partial\mathbf{v}_{2}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( \mathbf{T}_{I} \left( \mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \mathbf{k}_{s}(\mathbf{X}_{0},t_{0}) \right) \right) \cdot d\mathbf{s}_{0} dt_{0}$$

$$+ \varphi_{f}^{+} \int_{0}^{t} \int_{\partial\mathbf{v}_{2}} \left( \sum_{\alpha=1}^{6} \mathbf{e}_{\alpha} \left( P_{f} \left( \mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \right) \right) \left( \mathbf{k}_{f} \left( \mathbf{X}_{0},t_{0} \right) - \mathbf{k}_{s} \left( \mathbf{X}_{0},t_{0} \right) \right) \right) \cdot d\mathbf{s}_{0} dt_{0}$$

where  $\mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_0,t_0)$  is the column matrix defined by

$$\mathbf{g}_{(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) = \begin{bmatrix} \mathbf{g}_{f(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \\ \mathbf{g}_{s(\alpha)}(\mathbf{X},t;\mathbf{X}_{0},t_{0}) \end{bmatrix},$$
(10.6.14)

 $\mathbf{k}_{f}(\mathbf{X}_{0},t_{0})$  and  $\mathbf{k}_{s}(\mathbf{X}_{0},t_{0})$  are obtained from the boundary condition (10.2.3) expressed as the column matrix

$$\mathbf{k}(\mathbf{X},t) = \begin{bmatrix} \mathbf{k}_{f}(\mathbf{X},t) \\ \mathbf{k}_{s}(\mathbf{X},t) \end{bmatrix}$$
(10.6.15)

and  $\mathbf{e}_{\alpha}$ , for  $\alpha = 1, 2, ..., 6$ , is the 6×1 column matrix with entries equal to zero except in the  $\alpha$  position where the value is 1

The eigenfunction expansion technique mentioned in Section 9.7 also works in the case where the inertia terms are present. The eigenfunctions, as defined in Section 9.2 allow us to seek solutions for the Green's function of the form.

$$\mathbf{G}\left(\mathbf{X},t;\mathbf{X}_{0},t_{0}\right) = \sum_{n=1}^{\infty} T_{n}\left(t;t_{0}\right) \mathbf{u}_{n}\left(\mathbf{X}\right) \mathbf{u}_{n}\left(\mathbf{X}_{0}\right)^{T}$$
(10.6.16)

If we apply this formalism to the one dimensional problem considered in Section 10.3, the Green's function is the  $2 \times 2$  matrix of functions

$$\mathbf{G}(X,t;X_{0},t_{0}) = \begin{bmatrix} g_{f(1)}(X,t;X_{0},t_{0}) & g_{f(2)}(X,t;X_{0},t_{0}) \\ g_{s(1)}(X,t;X_{0},t_{0}) & g_{s(2)}(X,t;X_{0},t_{0}) \end{bmatrix}$$
(10.6.17)

that obeys the one dimensional forms of (10.6.2) through (10.6.6). Therefore, the Green's function is the solution of

$$\mathbf{M}\frac{\partial^{2}\mathbf{G}}{\partial t^{2}} + \xi \mathbf{E}\frac{\partial \mathbf{G}}{\partial t} - \mathbf{Q}\frac{\partial^{2}\mathbf{G}}{\partial X^{2}} = \delta(t - t_{0})\delta(X - X_{0}) \quad \text{for} \quad (X, t) \in (0, h) \times (-\infty, \infty)(10.6.18)$$
$$\frac{\partial \mathbf{G}(0, t; X_{0}, t_{0})}{\partial X} = \mathbf{0} \quad (10.6.19)$$
$$\mathbf{G}(h, t; X_{0}, t_{0}) = \mathbf{0} \quad (10.6.20)$$

and

$$\mathbf{G}(x,t;x_0,t_0) = \mathbf{0} \qquad \text{for} \quad t < t_0 \tag{10.6.21}$$

As in Section (9.4), the eigenfunctions of the space part of the one dimensional operator are defined by (9.4.25) and (9.4.26) are given by (9.4.27) where the eigenvalues are given by (9.4.31) and (9.4.32). In the notation adopted in Section 9.7, the representation of the solution (10.6.16) becomes

$$\mathbf{G}(X,t;X_0,t_0) = \sum_{n=1}^{\infty} \mathbf{K}_n(t;t_0) \cos\frac{(2n-1)\pi X}{2h} \cos\frac{(2n-1)\pi X_0}{2h}$$
(10.6.22)

where, for each *n*,  $\mathbf{K}_n(t;t_0)$  is a 2×2 matrix whose representation with respect to the basis of eigenvectors of **Q** is

$$\mathbf{K}_{n}(t;t_{0}) = T_{n(11)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(1)}^{T} + T_{n(12)}(t;t_{0})\mathbf{b}_{(1)}\mathbf{b}_{(2)}^{T} + T_{n(21)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(1)}^{T} + T_{n(22)}(t;t_{0})\mathbf{b}_{(2)}\mathbf{b}_{(2)}^{T}$$
(10.6.23)

If (10.6.22) is substituted into (10.6.18), it is easily shown that  $\mathbf{K}_n(t;t_0)$  must obey

$$\mathbf{M}\frac{d^{2}\mathbf{K}_{n}(t;t_{0})}{dt^{2}} + \xi \mathbf{E}\frac{d\mathbf{K}_{n}(t;t_{0})}{dt} + \frac{(2n-1)^{2}\pi^{2}}{4h^{2}}\mathbf{Q}\mathbf{K}_{n}(t;t_{0}) = \frac{2}{h}\delta(t-t_{0})$$
(10.6.24)

It follows from (10.6.21) and (10.6.22) that the matrix  $\mathbf{K}_n(t;t_0)$  must obey

$$\mathbf{K}_{n}(t;t_{0}) = \mathbf{0} \qquad \text{for} \quad t < t_{0} \qquad (10.6.25)$$

The same Laplace transform techniques that were used to solve (10.3.20) can be used to establish that

$$\mathbf{K}_{n}(t;t_{0}) = \frac{2}{h} \mathscr{L}^{-1} \left( e^{-st_{0}} \frac{s^{2} \operatorname{adj} \mathbf{M} + s\xi \operatorname{adj} \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \operatorname{adj} \mathbf{Q}}{\operatorname{det} \left( s^{2} \mathbf{M} + s\xi \mathbf{E} + \frac{(2n-1)^{2} \pi^{2}}{4h^{2}} \mathbf{Q} \right)} \right)$$
(10.6.26)

where  $\mathscr{L}^{-1}$  denotes the inverse Laplace transform operator. A comparison of (10.6.26) with (10.3.41) shows that

$$\mathbf{K}_{n}(t;t_{0}) = H(t-t_{0})\mathbf{K}_{n}(t-t_{0})$$
(10.6.27)

where  $H(t-t_0)$  is the Heaviside step function that was introduced in Section 9.7 and  $\mathbf{K}_n(t)$  is defined by (10.5.4). Thus, without explicit mention, our manipulations in Section 10.3 through 10.5 were, in effect, manipulations equivalent to finding the Green's function for our boundary initial value problem. The results in Section 10.3 are equivalent to those here. The one dimensional form of (10.6.13) is

$$\mathbf{w}(X,t) = \int_{0}^{h} \mathbf{G}(X,t;X_{0},0) \mathbf{Mg}(X_{0}) dX_{0} + \frac{\partial}{\partial t} \int_{0}^{h} \mathbf{G}(X,t;X_{0},0) \mathbf{Mf}(X_{0}) dX_{0}$$
$$+ \xi \int_{0}^{h} \mathbf{G}(X,t;X_{0},0) \mathbf{Ef}(X_{0}) dX_{0} - \int_{0}^{t} \mathbf{G}(X,t;0,t_{0}) \mathbf{q}(t_{0}) dt_{0} \qquad (10.6.28)$$
$$- \int_{0}^{t} \frac{\partial \mathbf{G}(X,t;h,t_{0})}{\partial X_{0}} \mathbf{Qk}(t_{0}) dt_{0}$$

where  $\mathbf{q}(t)$  is defined by (10.3.7). If (10.6.22) were to be substituted into (10.6.28) one would again derive (10.3.40).

#### 10.7 Biot Problem with Inertia

In this section we shall resolve the problem introduced in Section 9.5 except that this time we shall include the inertia terms for the fluid and the solid. As in Section 9.5, the boundary and initial conditions are

$$\mathbf{q}(t) = P_0 \begin{bmatrix} 0\\-1 \end{bmatrix} \tag{10.7.1}$$

$$\mathbf{k}(t) = \mathbf{0} \tag{10.7.2}$$

and
$$\mathbf{f}(X) = \frac{P_0}{\mathrm{tr}((\mathrm{adj}\mathbf{Q})\mathbf{E})}(h-X)\begin{bmatrix}1\\1\end{bmatrix}$$
(10.7.3)

The boundary initial value problem for the model with inertia requires that the initial velocity be prescribed. We shall make the choice

$$\mathbf{g}(X) = \mathbf{0} \tag{10.7.4}$$

The physical meaning of the boundary conditions (10.7.1) and (10.7.2) and the initial condition (10.7.3) are explained in Section 9.5. The second initial condition (10.7.4) forces this solution with inertia to obey the same initial condition, i.e., zero velocity, that is automatically obeyed by the inertia free solution when the other boundary and initial conditions (10.7.1) through (10.7.3) are obeyed. The proof of this assertion is given in Section 9.5 where equation (9.5.19) was established.

Given (10.7.1) through (10.7.4), the solution (10.3.40) reduces to

$$\mathbf{w}(X,t) = \frac{h}{2} \sum_{n=1}^{\infty} \left( \frac{\partial \mathbf{K}_{n}(t)}{\partial t} \mathbf{M} + \xi \mathbf{K}_{n}(t) \mathbf{E} \right) \mathbf{f}_{n} \cos \frac{(2n-1)\pi X}{2h} - \sum_{n=1}^{\infty} \left( \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \mathbf{q}(\tau) d\tau \right) \cos \frac{(2n-1)\pi X}{2h}$$
(10.7.5)

It follows from (10.3.27) and (10.7.3) that

$$\mathbf{f}_{n} = \frac{2}{h} \int_{0}^{h} \mathbf{f}(X) \cos \frac{(2n-1)\pi X}{2h} dX = \frac{P_{0}}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \begin{bmatrix} 1\\ 1 \end{bmatrix} \frac{2}{h} \int_{0}^{h} (h-X) \cos \frac{(2n-1)\pi X}{2h} dX = \frac{P_{0}}{\frac{1}{\operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \begin{bmatrix} 1\\ 1 \end{bmatrix} \frac{8h}{(2n-1)^{2} \pi^{2}} dX$$
(10.7.6)

This result along with (10.7.1) reduce the answer (10.7.5) to

$$\mathbf{w}(X,t) = \frac{4h^2 P_0}{\pi^2 \operatorname{tr}\left((\operatorname{adj}\mathbf{Q})\mathbf{E}\right)} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \frac{\partial \mathbf{K}_n(t)}{\partial t} \mathbf{M} \begin{bmatrix} 1\\1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h} + P_0 \sum_{n=1}^{\infty} \left( \int_0^t \mathbf{K}_n(t-\tau) d\tau \right) \begin{bmatrix} 0\\1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h}$$
(10.7.7)

where  $\mathbf{E}\begin{bmatrix}1\\1\end{bmatrix} = \begin{bmatrix}0\\0\end{bmatrix}$  has been used. The analytical solution is essentially complete when one substitutes (10.5.4) into (10.7.7). It follows from (10.5.4) that

$$\frac{h}{2} \frac{\partial \mathbf{K}_{n}(t)}{\partial t} = -\alpha_{n} e^{-\alpha_{n} t} \mathbf{A}_{n} - \gamma_{n} e^{-\gamma_{n} t} \mathbf{B}_{n} + \left(\varsigma_{n} e^{-\varsigma_{n} t} \cos \omega_{n} t + \omega_{n} e^{-\varsigma_{n} t} \sin \omega_{n} t\right) \left(\mathbf{A}_{n} + \mathbf{B}_{n}\right) \\ + \left(-\varsigma_{n} e^{-\varsigma_{n} t} \sin \omega_{n} t + \omega_{n} e^{-\varsigma_{n} t} \cos \omega_{n} t\right) \left(\frac{\alpha_{n} - \varsigma_{n}}{\omega_{n}} \mathbf{A}_{n} + \frac{\gamma_{n} - \varsigma_{n}}{\omega_{n}} \mathbf{B}_{n} + \frac{1}{\omega_{n}} \mathbf{M}^{-1}\right)$$
(10.7.8)

and

$$\frac{h}{2} \int_{0}^{t} \mathbf{K}_{n} (t-\tau) d\tau = \frac{1}{\alpha_{n}} (1-e^{-\alpha_{n}t}) \mathbf{A}_{n} + \frac{1}{\gamma_{n}} (1-e^{-\gamma_{n}t}) \mathbf{B}_{n}$$

$$-\frac{1}{\omega_{n}^{2}+\zeta_{n}^{2}} (\xi_{n} (1-e^{-\zeta_{n}t}\cos\omega_{n}t)+\omega_{n}e^{-\zeta_{n}t}\sin\omega_{n}t) (\mathbf{A}_{n}+\mathbf{B}_{n})$$

$$+\frac{1}{\omega_{n}^{2}+\zeta_{n}^{2}} (\omega_{n} (1-e^{-\zeta_{n}t}\cos\omega_{n}t)-\xi_{n}e^{-\zeta_{n}t}\sin\omega_{n}t) \left(\frac{\alpha_{n}-\zeta_{n}}{\omega_{n}}\mathbf{A}_{n} + \frac{\gamma_{n}-\zeta_{n}}{\omega_{n}}\mathbf{B}_{n} + \frac{1}{\omega_{n}}\mathbf{M}^{-1}\right)$$

$$(10.7.9)$$

In Section 9.5, it was pointed out that the long time solution of the inertia free Biot problem, i.e. the solution (9.5.12) is  $\mathbf{w}_{\infty} = P_0 \left( h - X \right) \mathbf{Q}^{-1} \begin{bmatrix} 0 \\ 1 \end{bmatrix}$ . It turns out that the solution (10.7.7) yields the same result. To see this, we note from (10.7.7) that

$$\mathbf{w}_{\infty} = \lim_{t \to \infty} \mathbf{w}(X,t) = \frac{4h^2 P_0}{\pi^2 \operatorname{tr}((\operatorname{adj} \mathbf{Q})\mathbf{E})} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \left( \lim_{t \to \infty} \frac{\partial \mathbf{K}_n(t)}{\partial t} \right) \mathbf{M} \begin{bmatrix} 1\\ 1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h} + P_0 \sum_{n=1}^{\infty} \left( \lim_{t \to \infty} \int_0^t \mathbf{K}_n(t-\tau) d\tau \right) \begin{bmatrix} 0\\ 1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h}$$
(10.7.10)

It follows from equations (10.4.11) and (10.4.12) that (10.7.10) reduces to

$$\mathbf{w}_{\infty} = \lim_{t \to \infty} \mathbf{w}(X, t) = P_0 \frac{8h}{\pi^2} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \cos \frac{(2n-1)\pi X}{2h} \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix}$$
(10.7.11)

Equation (9.4.64) allows us to write the last result as

$$\mathbf{w}_{\infty} = \lim_{t \to \infty} \mathbf{w}(X, t) = P_0(h - X) \mathbf{Q}^{-1} \begin{bmatrix} 0\\1 \end{bmatrix}$$
(10.7.12)

which is identical to the result for the inertia free case. It is important to keep in mind that the key results (10.4.11) and (10.4.12) assume we are not dealing with a case where  $u_0$  equals either  $u_{n(1)}$  or  $u_{n(2)}$ . In such a case, the real part of the roots  $\beta_{n(3)} = -\zeta_n + i\omega_n$  and  $\beta_{n(4)} = -\zeta_n - i\omega_n$  is

zero. As (10.7.8) and (10.7.9) show, if  $\zeta_n = 0$  it is no longer true that (10.4.10) through (10.4.12) hold.

As in the inertia free solution in Section 9.5, it is useful to plot the solution (10.7.7). We shall again adopt the material properties for Ruhr Sandstone as tabulated in Tables 1 and 2 of Chapter 6, and repeated in Table 1 of this chapter. The numerical values we shall adopt are again

$$\mathbf{Q} = \begin{bmatrix} \overline{\lambda}_{ff} & \overline{\lambda}_{sf} \\ \overline{\lambda}_{sf} & \overline{\lambda}_{ss} + 2\mu_{ss} \end{bmatrix} = \begin{bmatrix} .165 & 5.21 \\ 5.21 & 206.41 + 2(133) \end{bmatrix} = \begin{bmatrix} .165 & 5.21 \\ 5.21 & 472.41 \end{bmatrix} \text{ kbar} \quad (10.7.13)$$

$$\varphi_{f}^{+} = .02 \quad (10.7.14)$$

and

$$\xi = 1.9(10)^{-3}$$
 kbar-sec/cm<sup>2</sup> (10.7.15)

Also, it follows from the numbers in Table 1 that

$$\mathbf{M} = \begin{bmatrix} \varphi_{f}^{+} \gamma_{f} & 0\\ 0 & (1 - \varphi_{f}^{+}) \gamma_{s_{R}} \end{bmatrix} = \begin{bmatrix} .02 & 0\\ 0 & .98(2.6) \end{bmatrix} \text{ g/cm}^{3}$$
(10.7.16)

We shall again take

$$P_0 = 10^2 \text{ kbar}$$
 (10.7.17)

As in Section 9.5, we shall plot the solution (10.7.7) by plotting the dimensionless displacement  $\frac{\mathbf{w}(X,t)}{h}$  verses the dimensionless distance  $\frac{X}{h}$  and the dimensionless time  $\frac{tc}{h^2}$ . Unlike the inertia free solution (9.5.17), the solution (10.7.7) is not naturally represented in terms of these dimensionless groups. Each of the roots (10.5.2) has dimensions of inverse time and could be used to normalize the time. We have made the choice  $\frac{tc}{h^2}$  in order to allow comparisons to be made with the inertia free solution (9.5.17).

The resulting plots of the fluid and solid displacements are

and





## The displacement of the fluid relative to that of the solid is

The evident conclusion from these three figures is that they appear virtually identical to the corresponding three figures in Section 9.5. In effect, the problem we have posed for the poroelastic solid with inertia produces results that, at least to the scale of observation in the above three figures, are indistinguishable from the corresponding problem without inertia.

One issue that masks the differences in the two models is the initial condition (10.7.4) which we forced in the inertia case but was a theorem in the inertia free case. The partial differential equations which define the model without inertia are (9.1.15) and (9.1.16). Because they are first order in time, they represent singular perturbations of the equations used in this chapter, equations (10.1.6) and (10.1.7). As singular perturbations, they will not satisfy the same number of initial conditions as will the more general equations. The problem just solved does not exploit this difference because we forced the initial condition (10.7.4). A more significant display of the differences would arise if we selected a different initial velocity. As equation (9.4.68) displays, the model used in Section 9.5 is such that the initial condition is determined by the data, i.e., the values of the initial displacement, the imposed load and the displacement at x = h determines the initial velocity. For the equations with inertia, the initial velocity is a part of the data.

In the next section, we shall consider the more general problem where the external pressure oscillates with a prescribed frequency. We considered this problem in Section 9.6 for the case of zero inertia.

## 10.8 Biot Problem with Inertia: Time Dependent External Pressure

As in Section 9.6, we adopt the boundary and initial conditions

$$\mathbf{q}(t) = P_0 \cos \omega t \begin{bmatrix} 0\\-1 \end{bmatrix}$$
(10.8.1)

$$\mathbf{k}(t) = \mathbf{0} \tag{10.8.2}$$

and

$$\mathbf{f}(X) = \frac{P_0 \cos \omega t}{\operatorname{tr}((\operatorname{adj} \mathbf{Q})\mathbf{E})} (h - X) \begin{bmatrix} 1\\1 \end{bmatrix}$$
(10.8.3)

where  $\omega$  is a prescribed external frequency. We shall continue to adopt (10.7.4) as the initial condition. Thus, we shall take

$$\mathbf{g}(X) = \mathbf{0} \tag{10.8.4}$$

Given (10.8.1) through (10.8.4), the solution (10.3.40) reduces to

$$\mathbf{w}(X,t) = \frac{h}{2} \sum_{n=1}^{\infty} \left( \frac{\partial \mathbf{K}_n(t)}{\partial t} \mathbf{M} + \xi \mathbf{K}_n(t) \mathbf{E} \right) \mathbf{f}_n \cos \frac{(2n-1)\pi X}{2h} -\sum_{n=1}^{\infty} \left( \int_0^t \mathbf{K}_n(t-\tau) \mathbf{q}(\tau) d\tau \right) \cos \frac{(2n-1)\pi X}{2h}$$
(10.8.5)

where  $\mathbf{q}(t)$  is given by (10.8.1) and  $\mathbf{f}_n$  is again given by (10.7.6). If these two results are used, (10.8.5) reduces to

$$\mathbf{w}(X,t) = \frac{4h^2 P_0}{\pi^2 \operatorname{tr}\left((\operatorname{adj} \mathbf{Q})\mathbf{E}\right)} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^2} \left(\frac{\partial \mathbf{K}_n(t)}{\partial t}\mathbf{M}\right) \begin{bmatrix} 1\\1 \end{bmatrix} \cos\frac{(2n-1)\pi X}{2h} + P_0 \sum_{n=1}^{\infty} \left(\int_0^t \mathbf{K}_n(t-\tau)\cos\omega\tau d\tau\right) \begin{bmatrix} 0\\1 \end{bmatrix} \cos\frac{(2n-1)\pi X}{2h}$$

Given the formula (10.5.4), a series of straight forward, but lengthy, integrations yields

$$\frac{h}{2}\int_{0}^{t}\mathbf{K}_{n}(t-\tau)\cos\omega\tau d\tau = \left(\frac{1}{\alpha_{n}^{2}+\omega^{2}}\left(\alpha_{n}\left(\cos\omega t-e^{-\alpha_{n}t}\right)+\omega\sin\omega t\right)\right)\mathbf{A}_{n}$$

$$+\left(\frac{1}{\gamma_{n}^{2}+\omega^{2}}\left(\gamma_{n}\left(\cos\omega t-e^{-\gamma_{n}t}\right)+\omega\sin\omega t\right)\right)\mathbf{B}_{n}$$

$$-\frac{1}{\sqrt{\left(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2}\right)^{2}+\left(2\omega\zeta_{n}\right)^{2}}}\left(\sqrt{\frac{\xi_{n}^{2}+\omega^{2}}{\zeta_{n}^{2}}}\sin\left(\omega t+\psi_{n}\right)\right)\left(\mathbf{A}_{n}+\mathbf{B}_{n}\right)$$

$$+\frac{1}{\sqrt{\left(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2}\right)^{2}+\left(2\omega\zeta_{n}\right)^{2}}}\left(\sum_{-\sqrt{\omega_{n}^{2}+\zeta_{n}^{2}}}e^{-\zeta_{n}t}\sin\left(\omega_{n}t-\varphi_{n}\right)\right)\left(\frac{\alpha_{n}-\xi_{n}}{\omega_{n}}\mathbf{A}_{n}\right)$$

$$+\frac{1}{\sqrt{\left(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2}\right)^{2}+\left(2\omega\zeta_{n}\right)^{2}}}\left(\sum_{-\sqrt{\omega_{n}^{2}+\zeta_{n}^{2}}}e^{-\zeta_{n}t}\cos\left(\omega_{n}t-\varphi_{n}\right)\right)\left(\frac{\alpha_{n}-\xi_{n}}{\omega_{n}}\mathbf{A}_{n}\right)$$

$$+\frac{1}{\omega_{n}}\mathbf{M}^{-1}$$

where

$$\sin \varphi_{n} = \frac{\zeta_{n} \left(\omega^{2} + \omega_{n}^{2} + \zeta_{n}^{2}\right)}{\sqrt{\omega_{n}^{2} + \zeta_{n}^{2}} \sqrt{\left(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2}\right)^{2} + \left(2\omega\zeta_{n}\right)^{2}}}$$
(10.8.8)

$$\cos \varphi_{n} = \frac{\omega_{n} \left(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2}\right)}{\sqrt{\omega_{n}^{2} + \zeta_{n}^{2}} \sqrt{\left(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2}\right)^{2} + \left(2\omega\zeta_{n}\right)^{2}}}$$
(10.8.9)

$$\sin \psi_{n} = \frac{\zeta_{n} \left(\omega^{2} + \omega_{n}^{2} + \zeta_{n}^{2}\right)}{\sqrt{\xi_{n}^{2} + \omega^{2}} \sqrt{\left(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2}\right)^{2} + \left(2\omega\zeta_{n}\right)^{2}}}$$
(10.8.10)

$$\cos\psi_{n} = \frac{\omega(\omega^{2} - \omega_{n}^{2} + \zeta_{n}^{2})}{\sqrt{\xi_{n}^{2} + \omega^{2}}\sqrt{(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2})^{2} + (2\omega\zeta_{n})^{2}}}$$
(10.8.11)

$$\sin \sigma_n = \frac{2\omega\zeta_n}{\sqrt{\left(\omega_n^2 + \zeta_n^2 - \omega^2\right)^2 + \left(2\omega\zeta_n\right)^2}}$$
(10.8.12)

and

$$\cos \sigma_{n} = \frac{\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2}}{\sqrt{(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2})^{2} + (2\omega\zeta_{n})^{2}}}$$
(10.8.13)

This solution displays a transient solution,  $\mathbf{w}_t$ , defined by

$$\mathbf{w}_{t}(X,t) = \frac{4h^{2}P_{0}}{\pi^{2} \operatorname{tr}((\operatorname{adj}\mathbf{Q})\mathbf{E})} \sum_{n=1}^{\infty} \frac{1}{(2n-1)^{2}} \left( \frac{\partial \mathbf{K}_{n}(t)}{\partial t} \mathbf{M} \right) \begin{bmatrix} 1\\ 1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h} + P_{0} \sum_{n=1}^{\infty} \left( \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \cos \omega \tau d\tau \right) \Big|_{\operatorname{Transient}} \begin{bmatrix} 0\\ 1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h}$$
(10.8.14)

where 
$$\frac{\partial \mathbf{K}_{n}(t)}{\partial t}$$
 is given by (10.7.8) and  $\left(\int_{0}^{t} \mathbf{K}_{n}(t-\tau)\cos\omega\tau d\tau\right|_{\mathrm{Transient}}$ , from (10.8.7) is given by  
 $\frac{h}{2}\int_{0}^{t} \mathbf{K}_{n}(t-\tau)\cos\omega\tau d\tau\Big|_{\mathrm{Transient}} = e^{-\alpha_{n}t}\left(-\frac{\alpha_{n}}{\alpha_{n}^{2}+\omega^{2}}\right)\mathbf{A}_{n} + e^{-\gamma_{n}t}\left(-\frac{\gamma_{n}}{\gamma_{n}^{2}+\omega^{2}}\right)\mathbf{B}_{n}$   
 $-e^{-\zeta_{n}t}\frac{1}{\sqrt{(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2})^{2}+(2\omega\zeta_{n})^{2}}}\left(\sqrt{\omega_{n}^{2}+\zeta_{n}^{2}}\sin(\omega_{n}t-\varphi_{n})\right)(\mathbf{A}_{n}+\mathbf{B}_{n})$   
 $+e^{-\zeta_{n}t}\frac{1}{\sqrt{(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2})^{2}+(2\omega\zeta_{n})^{2}}}\left(-\sqrt{\omega_{n}^{2}+\zeta_{n}^{2}}\cos(\omega_{n}t-\varphi_{n})\right)\left(\frac{\alpha_{n}-\zeta_{n}}{\omega_{n}}\mathbf{A}_{n}\right)$   
 $+\frac{\gamma_{n}-\zeta_{n}}{\sqrt{(\omega_{n}^{2}+\zeta_{n}^{2}-\omega^{2})^{2}+(2\omega\zeta_{n})^{2}}}\left(-\sqrt{\omega_{n}^{2}+\zeta_{n}^{2}}\cos(\omega_{n}t-\varphi_{n})\right)\left(\frac{\gamma_{n}-\zeta_{n}}{\omega_{n}}\mathbf{A}_{n}\right)$   
 $+\frac{1}{\omega_{n}}\mathbf{M}^{-1}$ 
(10.8.15)

Likewise, the steady state solution is given by

$$\mathbf{w}_{s}(X,t) = P_{0} \sum_{n=1}^{\infty} \left( \int_{0}^{t} \mathbf{K}_{n}(t-\tau) \cos \omega \tau d\tau \Big|_{\text{Steady}} \right) \begin{bmatrix} 0\\1 \end{bmatrix} \cos \frac{(2n-1)\pi X}{2h}$$
(10.8.16)

where, from (10.8.7),

$$\int_{0}^{t} \mathbf{K}_{n}(t-\tau) \cos \omega \tau d\tau \Big|_{\text{Steady}} = \frac{1}{\alpha_{n}^{2} + \omega^{2}} (\alpha_{n} \cos \omega t + \omega \sin \omega t) \mathbf{A}_{n}$$

$$+ \frac{1}{\gamma_{n}^{2} + \omega^{2}} (\gamma_{n} \cos \omega t + \omega \sin \omega t) \mathbf{B}_{n}$$

$$- \frac{\sqrt{\xi_{n}^{2} + \omega^{2}}}{\sqrt{(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2})^{2} + (2\omega\zeta_{n})^{2}}} \sin(\omega t + \psi_{n}) (\mathbf{A}_{n} + \mathbf{B}_{n})$$

$$+ \frac{\omega_{n}}{\sqrt{(\omega_{n}^{2} + \zeta_{n}^{2} - \omega^{2})^{2} + (2\omega\zeta_{n})^{2}}} \cos(\omega t - \sigma_{n}) \begin{pmatrix} \frac{\alpha_{n} - \xi_{n}}{\omega_{n}} \mathbf{A}_{n} \\ + \frac{\gamma_{n} - \xi_{n}}{\omega_{n}} \mathbf{B}_{n} \\ + \frac{1}{\omega_{n}} \mathbf{M}^{-1} \end{pmatrix}$$

Next, we shall present a series of plots that attempt to show the similarities and the differences produced by the solution (10.8.6) and the inertia free solution of the same problem, equation (9.65). As we did in Section 10.7, we shall present the results in dimensionless forms using  $\frac{\mathbf{w}(X,t)}{h}$  as the dimensionless distance,  $\frac{tc}{h^2}$  as the dimensionless time and  $\frac{X}{h}$  as the dimensionless distance. As explained in Section 10.7, the choice of the dimensionless time is somewhat arbitrary given the many possible choices. The physical problem simply does not naturally scale with this choice. It is done here so that we can make comparisons of the results with inertia free results in Section 9.6. A similar issue arises with our choice of the ratio  $\frac{\omega h^2}{c}$  as the dimensionless frequency. Its choice is convenient when we compare our results to those in Section 9.6. Because the consolidation coefficient *c* for the poroelastic materials listed in Table 1 are small and because we are typically dealing with large values of the length *h* the dimensionless frequencies can be large for modest sized imposed frequencies  $\omega$ . For example, if we select  $\omega$  to be the frequency associated with the choice n = 1 for the third and fourth roots for Ruhr Sandstone shown in Table 3 above, then  $\omega = 68.12334$  yields a dimensionless frequency,  $\frac{\omega h^2}{c}$ , of approximately 1,228(10)<sup>5</sup>.

As in Section 9.6, we shall form plots of  $\frac{\mathbf{w}(X,t)}{h}$  verses the dimensionless distance  $\frac{X}{h}$ and the dimensionless time  $\frac{tc}{h^2}$  for various values of the ratio  $\frac{\omega h^2}{c}$ . The range of values selected for the dimensionless frequencies will be large enough to have actual frequencies near the smaller natural frequencies for Ruhr Sandstone that show in Table 3.











This series of figures suggests that for small values of the dimensionless frequency  $\frac{\omega h^2}{c}$ , the solution is not unlike that of the zero frequency cases shown in Section 9.6 when the inertia is zero

and in Section 10.8 when the imposed frequency is zero. However as the imposed frequency increases to near the first natural frequency,  $\omega = 68.12334$ , the two displacements become large. At frequencies this large, the imposed frequency is near one of the natural frequencies. This typical resonance phenomena shows a circumstance where the inertia terms in the governing partial differential equations significantly alter the solution.

In reference 2, Bowen and Lockett look at the zero mass case as an approximation of the case with inertia included. They establish analytical expressions which, when satisfied, justify neglecting the inertia. They also show analytically the large displacements illustrated in the figures above. We shall examine some of these results later. It is interesting to numerically illustrate the effects of inertia by constructing plots of the solutions with inertia minus the same solutions without inertia. These difference plots illustrate differences that are not apparent in the figures above as well as those in Sections 9.6 and 10.7. Because of the physical significance of the relative displacements between the fluid and the solid, we shall plot these displacements with inertia, without inertia and then for the difference between these two plots. We shall display these plots for various external frequencies. The resulting plots are the following:











# References

- 1. BOWEN, R. M., AND R. R. LOCKETT, Inertial Effects in Poroelasticity, J. Appl. Mech., 50, 334-342, 1983.
- 2. BELLMAN, R., Introduction to Matrix Analysis, McGraw-Hill, New York, 1960.

# Appendix A

**Roots of** det 
$$\left(\beta_n^2 \mathbf{M} + \beta_n \xi \mathbf{E} + \frac{\left(2n-1\right)^2 \pi^2}{4h^2} \mathbf{Q}\right) = 0$$

## **Ruhr Sandstone**

	First Root	Second Root	Third Root	Fourth Boot:
n	$\beta = -\alpha$	$\beta = -\gamma$	$\beta = -c + i\rho$	$\beta = -c - i\omega$
	$\boldsymbol{\rho}_{n(1)} - \boldsymbol{\alpha}_n$	$\mathcal{P}_{n(2)} - \mathcal{I}_n$	$\mathcal{P}_{n(3)} = \mathcal{S}_n + \mathcal{W}_n$	$\mathcal{P}_{n(4)} - \mathcal{G}_n - \mathcal{U}_n$
1	-0.000001	-95745682.888539	68.12334i	-68.12334i
2	-0.000012	-95745682.888527	204.37002i	-204.37002i
3	-0.000034	-95745682.888504	-0.000001+340.6167i	-0.000001-340.6167i
4	-0.000067	-95745682.88847	-0.000002+476.86338i	-0.000002-476.86338i
5	-0.000111	-95745682.888423	-0.000003+613.11006i	-0.000003-613.11006i
6	-0.000166	-95745682.888366	-0.000004+749.35674i	-0.000004-749.35674i
7	-0.000231	-95745682.888297	-0.000006+885.60342i	-0.000006-885.60342i
8	-0.000308	-95745682.888216	-0.000008+1021.8501i	-0.000008-1021.8501i
9	-0.000396	-95745682.888124	-0.00001+1158.09678i	-0.00001-1158.09678i
10	-0.000494	-95745682.88802	-0.000013+1294.34346i	-0.000013-1294.34346i
11	-0.000604	-95745682.887905	-0.000016+1430.59014i	-0.000016-1430.59014i
12	-0.000724	-95745682.887778	-0.000019+1566.83682i	-0.000019-1566.83682i
13	-0.000856	-95745682.88764	-0.000022+1703.0835i	-0.000022-1703.0835i
14	-0.000998	-95745682.88749	-0.000026+1839.33018i	-0.000026-1839.33018i
15	-0.001152	-95745682.887329	-0.00003+1975.57686i	-0.00003-1975.57686i
16	-0.001316	-95745682.887156	-0.000034+2111.82354i	-0.000034-2111.82354i
17	-0.001491	-95745682.886972	-0.000038+2248.07022i	-0.000038-2248.07022i
18	-0.001678	-95745682.886776	-0.000043+2384.3169i	-0.000043-2384.3169i
19	-0.001875	-95745682.886569	-0.000048+2520.56358i	-0.000048-2520.56358i
20	-0.002083	-95745682.88635	-0.000054+2656.81026i	-0.000054-2656.81026i
21	-0.002302	-95745682.88612	-0.000059+2793.05694i	-0.000059-2793.05694i
22	-0.002532	-95745682.885878	-0.000065+2929.30362i	-0.000065-2929.30362i
23	-0.002773	-95745682.885624	-0.000071+3065.5503i	-0.000071-3065.5503i
24	-0.003025	-95745682.885359	-0.000078+3201.79698i	-0.000078-3201.79698i
25	-0.003288	-95745682.885083	-0.000085+3338.04366i	-0.000085-3338.04366i
26	-0.003562	-95745682.884795	-0.000092+3474.29034i	-0.000092-3474.29034i
27	-0.003847	-95745682.884495	-0.000099+3610.53702i	Appendix A
28	-0.004143	-95745682.884184	-0.000106+3746.7837i	-0.000106-3746.7837i
29	-0.00445	-95745682.883862	-0.000114+3883.03038i	-0.000114-3883.03038i
30	-0.004767	-95745682.883528	-0.000123+4019.27706i	-0.000123-4019.27706i
31	-0.005096	-95745682.883182	-0.000131+4155.52374i	-0.000131-4155.52374i
32	-0.005436	-95745682.882825	-0.00014+4291.77042i	-0.00014-4291.77042i
33	-0.005786	-95745682.882456	-0.000149+4428.0171i	-0.000149-4428.0171i
34	-0.006148	-95745682.882076	-0.000158+4564.26378i	-0.000158-4564.26378i
35	-0.00652	-95745682.881685	-0.000168+4700.51046i	-0.000168-4700.51046i
36	-0.006904	-95745682.881281	-0.000177+4836.75714i	-0.000177-4836.75714i
37	-0.007298	-95745682.880867	-0.000188+4973.00382i	-0.000188-4973.00382i
38	-0.007704	-95745682.88044	-0.000198+5109.2505i	-0.000198-5109.2505i
39	-0.00812	-95745682.880003	-0.000209+5245.49718i	-0.000209-5245.49718i
40	-0.008547	-95745682.879554	-0.00022+5381.74386i	-0.00022-5381.74386i
41	-0.008986	-95745682.879093	-0.000231+5517.99054i	-0.000231-5517.99054i
42	-0.009435	-95745682.87862	-0.000242+5654.23722i	-0.000242-5654.23722i
43	-0.009895	-95745682.878137	-0.000254+5790.4839i	-0.000254-5790.4839i
44	-0.010366	-95745682.877641	-0.000266+5926.73058i	-0.000266-5926.73058i
45	-0.010848	-95745682.877134	-0.000279+6062.97726i	-0.000279-6062.97726i
46	-0.011341	-95745682.876616	-0.000291+6199.22394i	-0.000291-6199.22394i
47	-0.011845	-95745682.876086	-0.000304+6335.47062i	-0.000304-6335.47062i
48	-0.01236	-95745682.875545	-0.000318+6471.7173i	-0.000318-6471.7173i
49	-0.012886	-95745682.874992	-0.000331+6607.96398i	-0.000331-6607.96398i
50	-0.013423	-95745682.874427	-0.000345+6744.21066i	-0.000345-6744.21066i

# **Tennessee Marble**

	First Root:	Second Root:	Third Root:	Fourth Root:
()	$\beta_{n(1)} = -\alpha_n$	$\beta_{n(2)} = -\gamma_n$	$\beta_{n(3)} = -\varsigma_n + i\omega_n$	$\beta_{n(4)} = -\varsigma_n - i\omega_n$
1	0	-197481481481.481	-83.964i	83.964i
2	-0.0000003	-197481481481.481	-251.8919i	251.8919i
3	-0.0000007	-197481481481.481	-419.8198i	419.8198i
4	-0.00000014	-197481481481.481	-587.7478i	587.7478i
5	-0.0000023	-197481481481.481	-755.6757i	755.6757i
6	-0.0000035	-197481481481.481	-923.6036i	923.6036i
7	-0.00000049	-197481481481.481	-1091.5315i	1091.5315i
8	-0.00000065	-197481481481.481	-1259.4595i	1259.4595i
9	-0.0000084	-197481481481.481	-1427.3874i	1427.3874i
10	-0.00000104	-197481481481.481	-1595.3153i	1595.3153i
11	-0.00000128	-197481481481.481	-1763.2433i	1763.2433i
12	-0.00000153	-197481481481.481	-1931.1712i	1931.1712i
13	-0.0000181	-197481481481.481	-2099.0991i	2099.0991i
14	-0.00000211	-197481481481.481	-2267.027i	2267.027i
15	-0.00000243	-197481481481.481	-2434.955i	2434.955i
16	-0.00000278	-197481481481.481	-2602.8829i	2602.8829i
17	-0.00000315	-197481481481.481	-2770.8108i	2770.8108i
18	-0.00000354	-197481481481.481	-2938.7388i	2938.7388i
19	-0.00000396	-197481481481.481	-3106.6667i	3106.6667i
20	-0.0000044	-197481481481.481	-3274.5946i	3274.5946i
21	-0.00000486	-197481481481.481	-3442.5226i	3442.5226i
22	-0.00000535	-197481481481.481	-3610.4505i	3610.4505i
23	-0.00000586	-197481481481.481	-3778.3784i	3778.3784i
24	-0.00000639	-197481481481.481	-3946.3063i	3946.3063i
25	-0.00000695	-197481481481.481	-4114.2343i	4114.2343i
26	-0.00000753	-197481481481.481	-4282.1622i	4282.1622i
27	-0.00000813	-197481481481.481	-4450.0901i	4450.0901i
28	-0.00000875	-197481481481.481	-4618.0181i	4618.0181i
29	-0.0000094	-197481481481.481	-4785.946i	4785.946i
30	-0.00001007	-197481481481.481	-4953.8739i	4953.8739i
31	-0.00001077	-197481481481.481	-5121.8018i	5121.8018i
32	-0.00001149	-197481481481.481	-5289.72981	5289.72981
33	-0.00001223	-197481481481.481	-5457.65771	5457.65771
34	-0.00001299	-197481481481.481		5025.5850
35	-0.00001378	-197481481481.481		
36	-0.00001439	-197401401401.401	-5901.44151	5901.44151
37	0.00001542	-197401401401.401	-0129.30941	6207 2074
38	-0.00001028	-197401401401.401	-0297.29741	6465 2252
39	-0.00001710	-197401401401.401	-0403.22331	6622 1522
40	-0.00001800	-197401401401.401	-6801 0811	6801 0811
41	-0.00001899	-197481481481.481	-6060.0011	6060 0001;
42	-0.00001994	-197/81/81/81 /81	-7136 937i	7136 937i
43	-0.00002091	-197/81/81/81 /81	-7304 8649i	7304 8649i
44 /C	-0 0000219	-197481481481 481	-7472 7929i	7472 7929i
40	-0.00002292	-197481481481	-7640 7208i	7640 7208i
40	-0.00002503	-197481481481 481	-7808 6487i	7808 6487i
47	-0.00002505	-197481481481 481	-7976 5766i	7976 5766i
40	-0.00002723	-197481481481 481	-8144 5046i	8144 5046i
50	-0.00002836	-197481481481 481	-8312,4325i	8312.4325i
50	5.00002000			

# **Charcoal Granite**

	First Root:	Second Root:	Third Root:	Fourth Root:
11	$\beta_{n(1)} = -\alpha_n$	$\beta_{n(2)} = -\gamma_n$	$\beta_{n(3)} = -\varsigma_n + i\omega_n$	$\beta_{n(4)} = -\varsigma_n - i\omega_n$
1	0	-197481481481.481	-77.8296i	77.8296i
2	-0.0000002	-197481481481.481	-233.4887i	233.4887i
3	-0.0000005	-197481481481.481	-389.1478i	389.1478i
4	-0.0000009	-197481481481.481	-544.807i	544.807i
5	-0.0000015	-197481481481.481	-700.4661i	700.4661i
6	-0.0000022	-197481481481.481	-856.1253i	856.1253i
7	-0.0000031	-197481481481.481	-1011.7844i	1011.7844i
8	-0.00000042	-197481481481.481	-1167.4435i	1167.4435i
9	-0.00000054	-197481481481.481	-1323.1027i	1323.1027i
10	-0.00000067	-197481481481.481	-1478.7618i	1478.7618i
11	-0.0000082	-197481481481.481	-1634.421i	1634.421i
12	-0.00000098	-197481481481.481	-1790.0801i	1790.0801i
13	-0.00000116	-197481481481.481	-1945.7392i	1945.7392i
14	-0.00000136	-197481481481.481	-2101.3984i	2101.3984i
15	-0.00000156	-197481481481.481	-2257.0575i	2257.0575i
16	-0.00000179	-197481481481.481	-2412.7167i	2412.7167i
17	-0.0000202	-197481481481.481	-2568.3758i	2568.3758i
18	-0.00000228	-197481481481.481	-2724.0349i	2724.0349i
19	-0.0000255	-197481481481.481	-2879.6941i	2879.6941i
20	-0.0000283	-197481481481.481	-3035.3532i	3035.3532i
21	-0.00000313	-197481481481.481	-3191.0124i	3191.0124i
22	-0.00000344	-197481481481.481	-3346.6715i	3346.6715i
23	-0.00000377	-197481481481.481	-3502.3306i	3502.3306i
24	-0.00000411	-197481481481.481	-3657.9898i	3657.9898i
25	-0.00000446	-197481481481.481	-3813.6489i	3813.6489i
26	-0.00000484	-197481481481.481	-3969.3081i	3969.3081i
27	-0.00000522	-197481481481.481	-4124.9672i	4124.9672i
28	-0.0000562	-197481481481.481	-4280.6263i	4280.6263i
29	-0.00000604	-197481481481.481	-4436.2855i	4436.2855i
30	-0.00000647	-197481481481.481	-4591.9446i	4591.9446i
31	-0.00000692	-197481481481.481	-4747.6038i	4747.6038i
32	-0.00000738	-197481481481.481	-4903.2629i	4903.2629i
33	-0.00000786	-197481481481.481	-5058.922i	5058.922i
34	-0.00000835	-197481481481.481	-5214.5812i	5214.5812i
35	-0.00000885	-197481481481.481	-5370.2403i	5370.2403i
36	-0.00000937	-197481481481.481	-5525.8995i	5525.8995i
37	-0.00000991	-197481481481.481	-5681.5586i	5681.5586i
38	-0.00001046	-197481481481.481	-5837.2177i	5837.2177i
39	-0.00001102	-197481481481.481	-5992.8769i	5992.8769i
40	-0.0000116	-197481481481.481	-6148.536i	6148.536i
41	-0.0000122	-197481481481.481	-6304.1952i	6304.1952i
42	-0.00001281	-197481481481.481	-6459.8543i	6459.8543i
43	-0.00001343	-197481481481.481	-6615.5134i	6615.5134i
44	-0.00001407	-197481481481.481	-6771.1726i	6771.1726i
45	-0.00001473	-197481481481.481	-6926.8317i	6926.8317i
46	-0.0000154	-197481481481.481	-7082.4909i	7082.4909i
47	-0.00001608	-197481481481.481	-7238.15i	7238.15i
48	-0.00001678	-197481481481.481	-7393.8091i	7393.8091i
49	-0.00001749	-197481481481.481	-7549.4683i	7549.4683i
50	-0.00001822	-197481481481.481	-7705.1274i	7705.1274i

#### **Berea Sandstone**

5	First Root:	Second Root:	Third Root:	Fourth Root:
11	$\beta_{n(1)} = -\alpha_n$	$\beta_{n(2)} = -\gamma_n$	$\beta_{n(3)} = -\varsigma_n + i\omega_n$	$\beta_{n(4)} = -\varsigma_n - i\omega_n$
1	-0.00041905	-1015624.531	-50.4107i	50.4107i
2	-0.00377149	-1015624.5276	-151.232i	151.232i
3	-0.01047637	-1015624.5209	-252.0534i	252.0534i
4	-0.02053369	-1015624.5108	-352.8747i	352.8747i
5	-0.03394345	-1015624.4974	-453.6961i	453.6961i
6	-0.05070564	-1015624.4806	-554.5174i	554.5174i
7	-0.07082028	-1015624.4605	-0.0001-655.3388i	-0.0001+655.3388i
8	-0.09428736	-1015624.437	-0.0001-756.1601i	-0.0001+756.1601i
9	-0.12110688	-1015624.4101	-0.0001-856.9815i	-0.0001+856.9815i
10	-0.15127884	-1015624.3799	-0.0001-957.8028i	-0.0001+957.8028i
11	-0.18480324	-1015624.3463	-0.0002-1058.6241i	-0.0002+1058.6241i
12	-0.22168009	-1015624.3093	-0.0002-1159.4455i	-0.0002+1159.4455i
13	-0.26190938	-1015624.269	-0.0002-1260.2668i	-0.0002+1260.2668i
14	-0.30549111	-1015624.2254	-0.0003-1361.0882i	-0.0003+1361.0882i
15	-0.35242529	-1015624.1783	-0.0003-1461.9095i	-0.0003+1461.9095i
16	-0.40271191	-1015624.128	-0.0004-1562.7309i	-0.0004+1562.7309i
17	-0.45635099	-1015624.0742	-0.0004-1663.5522i	-0.0004+1663.5522i
18	-0.51334251	-1015624.0171	-0.0005-1764.3736i	-0.0005+1764.3736i
19	-0.57368648	-1015623.9567	-0.0005-1865.1949i	-0.0005+1865.1949i
20	-0.6373829	-1015623.8929	-0.0006-1966.0163i	-0.0006+1966.0163i
21	-0.70443177	-1015623.8257	-0.0006-2066.8376i	-0.0006+2066.8376i
22	-0.77483309	-1015623.7552	-0.0007-2167.659i	-0.0007+2167.659i
23	-0.84858687	-1015623.6813	-0.0008-2268.4803i	-0.0008+2268.4803i
24	-0.92569311	-1015623.604	-0.0008-2369.3017i	-0.0008+2369.3017i
25	-1.0061518	-1015623.5234	-0.0009-2470.123i	-0.0009+2470.123i
26	-1.08996295	-1015623.4395	-0.001-2570.9444i	-0.001+2570.9444i
27	-1.17712657	-1015623.3521	-0.0011-2671.7657i	-0.0011+2671.7657i
28	-1.26764264	-1015623.2615	-0.0012-2772.5871i	-0.0012+2772.5871i
29	-1.36151118	-1015623.1674	-0.0012-2873.4084i	-0.0012+2873.4084i
30	-1.45873219	-1015623.07	-0.0013-2974.2297i	-0.0013+2974.2297i
31	-1.55930566	-1015622.9693	-0.0014-3075.0511i	-0.0014+3075.0511i
32	-1.66323161	-1015622.8652	-0.0015-3175.8724i	-0.0015+3175.8724i
33	-1.77051002	-1015622.7577	-0.0016-3276.6938i	-0.0016+3276.6938i
34	-1.88114091	-1015622.6469	-0.001/-33//.5151	-0.001/+33/7.5151
35	-1.99512428	-1015622.5327	-0.0018-3478.33651	-0.0018+34/8.33651
36	-2.11246013	-1015622.4151	-0.0019-3579.15781	-0.0019+35/9.15/81
37	-2.23314845	-1015622.2942	-0.002-30/9.9/921	-0.002+36/9.9/921
38	-2.35/18926	-1015622.1699		
39	-2.48458256	1015622.0423		
40	-2.01032834	-1015621.9113	-0.0024-3982.44321	-0.0024+3982.44321
41	-2./4942002	-1015621.///		
42	2 02769065	1015621.0393	0.0020-4184.08591	-0.0020+4184.08591
43	-3.02/08005	-1015621.498Z	-0.0027-4284.90731	-0.002/+4284.90/31
44	2 2102//20	1015621.3338	-0.0029-4303.72801	-0.0029+4363.72801
45	-3.31334400	-1015621.200	-0.003-4400.331	-0.00374400.331
40	-3.47020343	-1015620 0004	-0.0032-4307.37131	-0.0032+4307.37131
47	-3.024410/3	-1015620.3004	-0.0033-4000.19271	-0.0033+4000.19271
40	-3 94290282	-1015620 5813	-0 0036-4880 8354	-0 0036+4880 8354i
49	-4 10717365	-1015620 /162	-0 0037-4990 6567i	-0 0037+4990 6567i
50	-4.10/1/202	-1010020.4100	-0.0037-4330.03071	-0.0037+4330.03071

# Westerly Granite

	First Root:	Second Root:	Third Boot:	Fourth Root:
n	$\beta_{(1)} = -\alpha$	$\beta_{r}(2) = -\gamma$	$\beta_{r(3)} = -\zeta_r + i\omega_r$	$\beta_{i} = -c_{i} - i\omega$
1	$\mathcal{P}_n(1)$	n(2) $n$	7 (3) 7 n n	$\mathcal{P}_{n(4)}$ $\mathfrak{S}_{n}$ $\mathfrak{too}_{n}$
1	-0.00000001	-2459105/313.8/95	-/5.4341	75.4341
2	-0.00000003	-24591057515.0795	-220.3021	220.3021
3	-0.00000014	-24591057515.0795	-3/7.171	577.171
4	-0.00000027	-24591057515.0795	-520.0501	520.050
5	0.00000045	24591057515.6795	-078.9001 920 774;	078.9001 920 77 <i>4</i> ;
6	0.00000007	24391037313.0793	029.7741	029.7741
/	-0.00000093	-24391037313.8793	-960.0421	1121 51
8	0.00000124	24391037313.0793	1202 270;	1101.011
9	0.00000139	24391037313.0793	1/22 2/6	1/22 2/6
10	0.00000199	24391037313.0793	150/ 11/	1455.2401
11	-0.00000243	-24391037313.8793	-173/ 022i	173/ 022
12	-0.00000291	-24591057313.8795	-1985 85i	1734.3821
13	0.00000344	24391037313.0793	2026 719;	2026 719;
14	0.00000401	24391037313.0793	2030.710	2030.7101
15	0.00000402	24391037313.0793	-2107.3001 2220 /E/i	2107.300
16	0.00000528	24591057515.6795	2350.4541	2556.4541
17	-0.00000599	-24591057515.0795	-2409.52211	2409.52211
18	-0.00000674	-24591057515.0795	2040.1901	2040.19011
19	-0.00000753	-2459105/313.8/95	-2791.05811	2/91.05811
20	-0.00000836		-2941.92011	2941.92011
21	-0.00000924		-3092.79411	3092.79411
22	-0.00001017		-3243.00211	3243.00211
23	-0.00001114		-3394.53011	3394.33011
24	-0.00001215		-3545.39811	3545.39811
25	-0.0000132	-24591057515.0795	-2090.20011	2047 1241;
26	-0.0000145	-24591057515.0795	-3047.13411	2008 0021;
27	0.00001343	24391037313.0793	-3336.00211 4149.9701;	11/0 0701;
28	-0.00001003	24591057515.6795	-4146.67011	4140.07011
29	0.00001787	24391037313.0793	4299.73011	4299.75011
30	0.00001914	24391037313.0793	4430.00011	4430.00011
31	0.00002040	24391037313.0793	4001.4741	4001.47411
32	-0.00002183	-24591057313.8795	-4752.54211	4752.54211
24	-0.00002323	-24551057515.8755	-505/ 0781i	505/ 0781i
25	-0.00002408	-24551057515.8755	-5204 9/61i	520/ 9/61i
26	-0.00002010	-24591657313.8795	-5355 8141i	5255 8141i
27	-0.00002772	-24591657313.8795	-5506 6821i	5506 6821i
27 20	-0.0000255	-24551057515.8755	-5657 5501i	5657 5501i
20	-0.00003055	-24551057515.8755	-5808 /181i	5808 /181i
39	-0.0000320	-24551057515.8755	-5050.41011	5959 2861i
40	-0.00003432	-24551057515.8755	-6110 15/11	6110 15/11
41	-0.00003008	-24591057313.8795	-6261 0221i	6261 0221
42	-0.00003788	-24591057313.8795	-6201.02211	6/11 8001;
43	-0.00003973	-24591657212 9705	-6562 7581	6562 7581i
44	-0.00004102	-24591057313.0795	-6713 6261	6713 6261i
45	-0.00004330	-24591057313.0795	-6864 4941	686/ /9/11
40	-0.00004334	-24591057313.0795	-0004.43411	7015 3621
47	-0.00004730	-2433103/313.0/33	-7013.30211	7166 22021
48	0.00004903	2433103/313.8/33	7217 0092	7217 0023
49	0.00005280	-2433103/313.8/95	-1311.09821	7317.03021
50	-0.00005389	-2459105/313.8/95	-7467.96621	7407.96621

## Weber Sandstone

	First Root:	Second Root:	Third Root:	Fourth Root:
n	$\beta_{n(1)} = -\alpha_n$	$\beta_{n(2)} = -\gamma_n$	$\beta_{n(3)} = -\zeta_n + i\omega_n$	$\beta_{n(4)} = -\zeta_n - i\omega_n$
1	-0.00005144	-6061920.3491	-63.7532i	63.7532i
2	-0.00046298	-6061920.3487	-191.2597i	191.2597i
3	-0.00128605	-6061920.3479	-318.7661i	318.7661i
4	-0.00252066	-6061920.3466	-446.2726i	446.2726i
5	-0.00416681	-6061920.345	-573.779i	573.779i
6	-0.0062245	-6061920.3429	-701.2855i	701.2855i
7	-0.00869372	-6061920.3404	-828.7919i	828.7919i
8	-0.01157448	-6061920.3375	-956.2984i	956.2984i
9	-0.01486677	-6061920.3342	-1083.8048i	1083.8048i
10	-0.01857061	-6061920.3305	-1211.3113i	1211.3113i
11	-0.02268598	-6061920.3264	-1338.8177i	1338.8177i
12	-0.02721288	-6061920.3218	-1466.3242i	1466.3242i
13	-0.03215133	-6061920.3169	-0.0001-1593.8306i	-0.0001+1593.8306i
14	-0.03750131	-6061920.3115	-0.0001-1721.3371i	-0.0001+1721.3371i
15	-0.04326283	-6061920.3057	-0.0001-1848.8435i	-0.0001+1848.8435i
16	-0.04943588	-6061920.2995	-0.0001-1976.35i	-0.0001+1976.35i
17	-0.05602047	-6061920.2929	-0.0001-2103.8564i	-0.0001+2103.8564i
18	-0.0630166	-6061920.2859	-0.0001-2231.3629i	-0.0001+2231.3629i
19	-0.07042427	-6061920.2785	-0.0001-2358.8693i	-0.0001+2358.8693i
20	-0.07824347	-6061920.2706	-0.0001-2486.3758i	-0.0001+2486.3758i
21	-0.08647421	-6061920.2624	-0.0002-2613.8822i	-0.0002+2613.8822i
22	-0.09511649	-6061920.2537	-0.0002-2741.3887i	-0.0002+2741.3887i
23	-0.10417031	-6061920.2446	-0.0002-2868.8951i	-0.0002+2868.8951i
24	-0.11363566	-6061920.2351	-0.0002-2996.4016i	-0.0002+2996.4016i
25	-0.12351254	-6061920.2252	-0.0002-3123.908i	-0.0002+3123.908i
26	-0.13380097	-6061920.2149	-0.0002-3251.4145i	-0.0002+3251.4145i
27	-0.14450093	-6061920.2041	-0.0003-3378.9209i	-0.0003+3378.9209i
28	-0.15561243	-6061920.193	-0.0003-3506.4274i	-0.0003+3506.4274i
29	-0.16713547	-6061920.1814	-0.0003-3633.9338i	-0.0003+3633.9338i
30	-0.17907004	-6061920.1694	-0.0003-3761.4403i	-0.0003+3761.4403i
31	-0.19141615	-6061920.157	-0.0004-3888.9467i	-0.0004+3888.9467i
32	-0.2041738	-6061920.1442	-0.0004-4016.4532i	-0.0004+4016.4532i
33	-0.21734299	-6061920.131	-0.0004-4143.9596i	-0.0004+4143.9596i
34	-0.23092371	-6061920.1174	-0.0004-4271.4661i	-0.0004+4271.4661i
35	-0.24491597	-6061920.1033	-0.0004-4398.9725i	-0.0004+4398.9725i
36	-0.25931976	-6061920.0889	-0.0005-4526.479i	-0.0005+4526.479i
37	-0.2741351	-6061920.074	-0.0005-4653.9854i	-0.0005+4653.9854i
38	-0.28936197	-6061920.0587	-0.0005-4781.4919i	-0.0005+4781.4919i
39	-0.30500037	-6061920.043	-0.0006-4908.9983i	-0.0006+4908.9983i
40	-0.32105032	-6061920.0269	-0.0006-5036.5048i	-0.0006+5036.5048i
41	-0.3375118	-6061920.0104	-0.0006-5164.0112i	-0.0006+5164.0112i
42	-0.35438482	-6061919.9935	-0.0006-5291.5177i	-0.0006+5291.5177i
43	-0.37166938	-6061919.9761	-0.0007-5419.0241i	-0.0007+5419.0241i
44	-0.38936547	-6061919.9584	-0.0007-5546.5306i	-0.0007+5546.5306i
45	-0.40/4/31	-6061919.9402	-0.0007-5674.037i	-0.0007+5674.0371
46	-0.42599227	-6061919.9216	-0.0008-5801.5435i	-0.0008+5801.5435i
47	-0.44492297	-6061919.9026	-0.0008-5929.04991	-0.0008+5929.04991
48	-0.46426521	-6061919.8832	-0.0009-6056.55641	-0.0009+6056.55641
49	-0.48401899	-6061919.8634	-0.0009-6184.06281	-0.0009+6184.06281
50	-0.50418431	-6061919.8431	-0.0009-6311.5693i	-0.0009+6311.5693i

- 1. ADKINS, J. E., Nonlinear Diffusion II Constitutive Equations for Mixtures of Isotropic Fluids, *Phil. Trans. Roy. Soc. London A*, **255**, 635-648, 1963.
- 2. ADKINS, J. E., Diffusion of Fluids Through Highly Elastic Solids, *Arch. Rational Mech. Anal.*, **15**, 222-234, 1964.
- 3. ADKINS, J. E., Nonlinear Diffusion Through Isotropic Highly Elastic Solids, *Phil. Trans. Roy. Soc. London A*, **256**, 301-316, 1964.
- 4. ADKINS, J. E., Nonlinear Diffusion I Diffusion and Flow of Mixtures of Fluids, *Phil. Trans. Roy. Soc. London A*, **255**, 607-722, 1969.
- 5. AIFANTIS, E. C., On the Problem of Diffusion in Solids, *Acta Mech.*, **37**, 265-296, 1980.
- 6. AIFANTIS, E. C., On Barenblatt's Problem, Lett. Appl. Engr. Sci., 18, 857-867, 1980.
- 7. AIZICOVICI, S., AND M. ARON, A Variational Theorem in the Linear Theory of Mixtures of Two Elastic Solids. The Quasi Static Case, *Acta Mech.*, **27**, 275-280, 1977.
- 8. ALLEN, N., R. RICHART JR., AND R. WOODS, Fluid Wave Propagation in Saturated and Nearly Saturated Sands, *J. of the Geotechnical Engineering Division ASCE*, **106**, 235-254, 1989.
- 9. ARON, M. AND V. SAVA, Weak Solutions in the Theory of Mixtures, *Acta Mech.*, **16**, 297-302, 1973.
- 10. ARON, M., On the Existence and Uniqueness of Solutions in the Linear Theory of Mixtures of Two Elastic Solids Mixtures of Two Elastic Solids, *Arch. of Mech.*, **26**, 717-728, 1974.
- 11. ARON, M., Bending of Beams of Two interacting Elastic Solids, *Acta Mech.*, **20**, 309-313, 1974.
- 12. ARON, M., An Application of a Theorem of Piola to the Theory of Mixtures, *Acta Mech.*, **23**, 285-289, 1975.
- 13. ARON, M., On the Plane Strain Problem for an Isotropic Mixture, *Arch. of Mech.*, **27**, 679-684, 1975.
- 14. ARON, M., Variational Principles in the Linear Theory of Mixtures, *Arch. of Mech.*, **26**, 31-40, 1976.
- 15. ASAY, J. R., Shock and Release Behavior in Porous 1100 Aluminum, *J. Appl. Phys.*, **46**, 197-203, 1975.
- 16. ATKIN, R. J., AND R. E. CRAINE, Continuum Theories of Mixtures: Basic Theory and Historical Development, *Q. J. Mech and Appl. Math.*, **24**, 209-244, 1976.
- 17. ATKIN, R. J., AND R. E. CRAINE, Continuum Theories of Mixtures: Applications, J. *Inst. Maths Applics*, **17**, 153-207, 1976.
- 18. ATKIN, R. J., Completeness Theorems for Linearized Theories of Interacting Continua, *Quart. J. Mech. Appl. Math.*, **21**, 171-193, 1968.
- 19. ATKIN, R. J., CHADWICK, P., AND STEEL, T. R., Uniqueness Theorems for Linearized Theories of Interacting Continua, *Mathematika*, **14**, 27-42, 1967.
- 20. BANERJEE, S., AND A. M. C. CHAN, Separated Flow Models-I. Analysis of the Averaged and Local Instantaneous Formulations, *Intl. J. Multiphase Flow*, **6**, 38376, 1980.
- 21. BARDET, J. P., A Viscoelastic Model for the Dynamic Behavior of Saturated Poroelastic Solids, *J. Appl. Mech.*, **59**, 128-135, 1992.

- 22. BARENBLATT, G. I., On Some Unsteady Motions of a Liquid and Gas in a Porous Media (In Russian), *Prikl. Mat. Mekh.*, **16**, 67-68, 1952.
- 23. BARENBLATT, G. I., On Certain Boundary Value Problems for the Equations of Seepage of a Liquid quid in Fissured Rocks, *PMM*, **27**, 348-350, 1963.
- 24. BARENBLATT, G. I., IU. P. ZHELTOV AND I. N. KOCHINA, Basic Concepts in the Theory of Seepage of Homogeneous Liquids in Fissured Rocks(Strata), *PMM*, **24**, 852-864, 1960.
- 25. BARZAM, V. A., The Apparent Additional Mass Effect and Waves in Poroelastic Layered Media, *Lett. Appl. Engr. Sci.*, **18**, 641-649, 1980.
- 26. BAZANT, Z. Mechanics of Geomaterials, John Wiley & Sons Inc. New York, , .
- 27. BAZANT, Z. P., AND R. J. KRIZEK, Saturated Sand as an Inelastic Two-Phase Medium, *J. Engr. Mech. Div. ASCE*, **101**, 317-332, 1975.
- 28. BEAR, J., AND CAROL BRAESTER, On the Flow of Two Immiscible Fluids in Fractured Porous Media in Fundamentals of Transport Phenomena in Porous Media, *Elsevier*, 1972.
- 29. BEAR, JACOB, Dynamics of Fluids in Porous Media, American Elsevier, New York, 1972.
- 30. BEAR, JACOB, AND M. YAVUZ CORAPICIOGLU (ED), Fundamentals of Transport Phenomena in Porous Media, Proceedings of the NATO Advanced Study Institute on Mechanics of Fluids in Porous Media, Newark, Deleware, *Martinus Nijhoff, Dordrecht*, 1984.
- 31. BECKER, R., AND I. K. MCIVOR, A Nonlinear Mixture Theory Representation of Saturated Sand, *Intl. J. Engr. Sci.*, **16**, 931-942, 1978.
- 32. BEDFORD, A., AND DOUGLAS S. DRUMHELLER, A Variational Theory of Immiscible Mixtures, *Arch. Rational Mech. Anal.*, **68**, 37-51, 1978.
- 33. BEDFORD, A., AND DOUGLAS S. DRUMHELLER, Theory of Immiscible and Structured Mixtures, *Intl. J. Engr. Sci.*, **21**, 873-960, 1978.
- 34. BEDFORD, A., AND DOUGLAS S. DRUMHELLER, A Theory of Bubbly Liquids, J. Acoust. Soc. Am., 66, 172-208, 1979.
- 35. BEDFORD, A., COSTLEY, R. D., AND M. STERN, On the Drag and Virtual Mass Coefficients in Biot's Equations, *J. Acoust. Soc. Am.*, **76**, 1804-1809, 1984.
- 36. BERRYMAN, J. G., Confirmation of Biot's Theory, *Appl. Phys. Lett.*, **37**, 382-384, 1980.
- 37. BERRYMAN, J. G., Single-Scattering Approximations for Coefficients in Biot's Equations of Poroelasticity, J. Acoust. Soc. Am., **91**, 551-571, 1992.
- 38. BERRYMAN, J. G., AND G. W. MILTON, Exact Results for Generalized Gassman's Equations in Space with Compressible Constituents, *J. Appl. Mech.*, **56**, 1950-1960, 1991.
- 39. BESKOS, D. E., AND E. C. AIFANTIS On the Theory of Consolidation with Double Porosity-II, *Intl. J. Engr. Sci.*, **24**, 1697-1716, 1986.
- 40. BHUTANI, O. P., AND P. CHANDRAN, Weak Waves in Dusty Gas, *Intl. J. Engr. Sci.*, **15**, 537-544, 1977.
- 41. BIOT, MAURICE A. AND D. G. WILLIS, The Elastic Coefficients of the Theory of Consolidation, *J. Appl. Mech.*, **24**, 594-601, 1957.
- 42. BIOT, MAURICE A. AND F. M. CLINGAN, Consolidation Settlement of a Soil with an Impervious Top Surface, *J. Appl. Phys.*, **12**, 578-581, 1941.

- 43. BIOT, MAURICE A. AND F. M. CLINGAN, Bending Settlement of a Slab Resting on a Consolidating Foundation, *J. Appl. Phys.*, **13**, 35-40, 1942.
- 44. BIOT, MAURICE A., Consolidation Settlement Under a Rectangular Load Distribution, *J. Appl. Phys.*, **12**, 426-430, 1941.
- 45. BIOT, MAURICE A., General Theory of Three-Dimensional Consolidation, J. Appl. *Phys.*, **12**, 155-164, 1941.
- 46. BIOT, MAURICE A., Theory of Elasticity and Consolidation for a Porous Anisotropic Solid, *J. Appl. Mech.*, **23**, 91-96, 1956.
- 47. BIOT, MAURICE A., Theory of Propagation of Elastic Waves in a Fluid-Saturated Porous Solid. I. Low-Frequency Range., *J. Acoust. Soc. Am.*, **28**, 168-178, 1956.
- 48. BIOT, MAURICE A., Theory of Propagation of Elastic Waves in a Fluid-Saturated Porous Solid. II. High Frequency Range, J. Acoust. Soc. Am., 28, 179-191, 1956.
- 49. BIOT, MAURICE A., General Solutions of the Equations of Elasticity and Consolidation for a Porous Material, *J. Appl. Mech.*, **23**, 91-96, 1956.
- 50. BIOT, MAURICE A., Thermoelasticity and Irreversible Thermodynamics, *J. Appl. Phys.*, **27**, 240-253, 1956.
- 51. BIOT, MAURICE A., Theory of Deformation of a Porous Viscoelastic Anisotropic Solid, *J. Appl. Phys.*, **27**, 459-467, 1956.
- 52. BIOT, MAURICE A., Generalized Theory of Acoustic Propagation in Porous Dissipative Media, *J. Acoust. Soc. Am.*, **34**, 1254-1264, 1962.
- 53. BIOT, MAURICE A., Mechanics of Deformation and Acoustic Propagation in Porous Media, *J. Appl. Phys.*, **33**, 1482-1498, 1962.
- 54. BIOT, MAURICE A., Theory of Stability and Consolidation of a Porous Media Under Initial Stress, *J. Math. and Mech.*, **12**, 521-541, 1963.
- 55. BIOT, MAURICE A., Theory of Buckling of a Porous Slab and its Thermoelastic Analogy, *J. Appl. Mech.*, **31**, 194-198, 1964.
- 56. BIOT, MAURICE A., Theory of Finite Deformations of Porous Solids, *Indiana Univ. Math. Journal*, **21**, 597-620, 1972.
- 57. BIOT, MAURICE A., Nonlinear and Semilinear Rheology of Porous Solids, *J. Geophysical Res.*, **78**, 4924-4937, 1973.
- 58. BIOT, MAURICE A., New Chemical Thermodynamics of Open Systems. Thermobaric Potential a New Concept, *Bull. Roy. Acad. Belguim(Class des Sciences)*, **62**, 239-258 678, 1976.
- 59. BIOT, MAURICE A., Variational-Lagrangian Irreversible Thermodynamics of Initially-Stressed Solids with Thermomolecular Diffusion and Chemical Reactions, *J. Mech. Phys. Solids.*, **25**, 289-307, 1977.
- 60. BIOT, MAURICE A., Variational Lagrangian-Thermodynamics of Nonisothermal Finite Strain Mechanics of Porous Solids and Thermolecular Diffusion, *Intl. J. Solids & Structures*, **13**, 579-597, 1977.
- 61. BIOT, MAURICE A., New Fundamental Concepts and Results in Thermodynamics with Chemical Applications, *J. Chem. Phys.*, **22**, 183-198, 1977.
- 62. BIOT, MAURICE A., Variational Irreversible Thermodynamics of Heat and Mass Transfer in Porous Solids: New Concepts and Methods, *Q. Appl. Math.*, **36**, 19-38, 1978.

- 63. BIOT, MAURICE A., Thermodynamic Principle of Virtual Dissipation and the Dynamics of Physical-Chemical Fluid Mixtures Including Radiation Pressure, *Quart. Appl. Math.*, **29**, 517-540, 1982.
- 64. BISHOP, A. W., The Influence of an Undrained Change in Stress on the Pore Pressure in Porous Media of Low Compressibility, *Geotechnique*, **23**, 435-442, 1973.
- 65. BISHOP, A. W., AND G. E. BLIGHT, Some Aspects of Effective Stress in Saturated and Partly Saturated Soils, *Geotechnique*, **13**, 177-197, 1963.
- 66. BOOKER, J. R., AND C. SAVVIDOU, Consolidation around a Spherical Heat Source, *Intl. J. Solids & Structures*, **20**, 1079-1090, 1984.
- 67. BOOKER, JOHN R., Time Dependent Strain Following Faulting of a Porous Medium, J. Geophysical Res., **79**, 2037-2044, 1974.
- 68. BOWEN, R. M., Thermochemistry of Reacting Materials, J. Chem. Phys., 49, 1625-1637, 1968.
- 69. BOWEN, R. M., Thermochemistry of a Reacting Mixture of Elastic Materials with Diffusion, *Arch. Rational Mech. Anal.*, **34**, 97-127, 1969.
- 70. BOWEN, R. M., Erratum: Thermochemistry of Reacting Materials, J. Chem. Phys., **50**, 4601-4602, 1969.
- 71. BOWEN, R. M., Theory of Mixtures, in Continuum Physics, Vol. III (ed. A. C. Eringen), *Academic Press*, **New York**, N.Y., 1976.
- 72. BOWEN, R. M., Incompressible Porous Media Models by Use of the Theory of Mixtures, *Int. J. Engr. Sci.*, **18**, 1129-1148, 1980.
- 73. BOWEN, R. M., Compressible Porous Media Models by Use of the Theory of Mixtures, *Int. J. Engr. Sci.*, **20**, 697-736, 1982.
- 74. BOWEN, R. M., Diffusion Models Implied by the Theory of Mixtures, in Rational Thermodynamics, by C. Truesdell, 2nd Ed., *Springer-Verlag*, New York, N.Y., 1984.
- 75. BOWEN, R. M., Introduction to Continuum Mechanics for Engineers, *Plenum Press*, New York, N.Y., 1989.
- 76. BOWEN, R. M., AND J. C. WIESE Diffusion in Mixtures of Elastic Materials, *Intl. J. Engr. Sci.*, **7**, 689-722, 1969.
- 77. BOWEN, R. M., AND R. R. LOCKETT, Inertial Effects in Poroelasticity, *J. Appl. Mech.*, **50**, 334-342, 1983.
- 78. BRENNER, HOWARD, Rheology of Two-Phase Systems, *Rev. of Fluid Mechanics*, **2**, 137-176, 1970.
- 79. BRUTSAERT, WILFRIED, The Propagation of Elastic Waves in Unconsolidated Unsaturated Granular Mediums, *J. Geophysical Res.*, **69**, 243-257, 1964.
- 80. BURKE, MICHAEL, AND HERBERT B. KINGSBURY Response of Poroelastic Layers to Moving Loads, *Intl. J. Solids & Structures*, **20**, 499-511, 1984.
- 81. BURRIDGE, R., AND C. A. VARGAS, The Fundamental Solution in Dynamic Poroelasticity, *Geophys. J. R. Astr. Soc.*, **58**, 61-90, 1979.
- 82. BURRIDGE, R., AND KELLER, J. B., Poro-elasticity Equations Derived from Micro-Structure, J. Acoust. Soc. Am., **70**, 1140-1146, 1981.
- BUTCHER, B. M., The Description of Strain-Rate Effects in Shocked Porous Materials in Shock Waves and the Mechanical Properties of Solids (ed. J. J. Burke & V. Weiss), *Syracuse University Press*, Syracuse, N.Y., 1970.

- 84. BUYEVICH, YU. A., Statistical Hydromechanics of Dispersed Systems Part 1. Physical Background and General Equations, *J. Fluid Mech.*, **49**, 489-507, 1971.
- 85. CARLSON, D. E., Linear Thermoelasticity Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), *Springer-Verlag*, Berlin, Germany, 1972.
- 86. CARROLL, M. M. AND N. KATSUBE The Role of Terzaghi Effective Stress in Linearly Elastic Deformation, *J. of Energy Res. Tech.*, **105**, 509-511, 1983.
- 87. CARROLL, M. M., AND A. C. HOLT, Static and Dynamic Pore-Collapse Relations for Ductile Porous Materials, J. Appl. Phys., 43, 1626-1636, 1972.
- 88. CARTER, J. P., SMALL, J. C., AND J. R. BOOKER, A Theory of Finite Elastic Consolidation, *Intl. J. Solids & Structures*, **13**, 467-478, 1977.
- 89. CHANDLER, R. N., Transient Streaming Potential Measurements on Fluid-Saturated Porous Structures: An Experimental Verification of Biot's Slow Wave in the Quasi-Static Limit, *J. Acoust. Soc. Am.*, **70**, 116-121, 1981.
- 90. CHANDLER, R. N., AND D. L. JOHNSON, The Equivalence of Quasi Static Flow in Fluid-Saturated Porous Media and Biot's Slow Wave in the Limit of Zero Frequency, *J. Appl. Phys.*, **52**, 3391-3395, 1981.
- 91. CHANG, ALEXANDER H-D., AND JAMES A. LIGGETT, Boundary Integral equation Method for Linear Porous-Elasticity with Applications to Soil Consolidation, *Intl. J. Num. Methods in Engr.*, **20**, 255-278, 1984.
- 92. CHANG, ALEXANDER H-D., AND JAMES A. LIGGETT, Boundary Integral equation Method for Linear Porous-Elasticity with Applications to Fracture Propagation, *Intl. J. Num. Methods in Engr.*, **20**, 279-296, 1984.
- 93. CHIARELLA, C., AND J. R. BOOKER, The Time-Settlement Behavior of a Rigid Die Resting on a Deep Clay Layer, *Quart. J. Mech. Appl. Math.*, **28**, 317-328, 1975.
- 94. CLEARY, MICHAEL P., Fundamental Solutions for Fluid-Saturated Porous Media and Application to Localised Rupture Phenomena Division of Engineering Brown University, *ProvidenceR. I.*, 1-207, 1975.
- 95. CLEARY, MICHAEL P., Fundamental Solutions for a Fluid-Saturated Porous Solid, *Intl. J. Solids & Structures*, **13**, 785-806, 1977.
- 96. CLEARY, MICHAEL P., Elastic and Dynamic Response Regimes of Fluid-Impregnated Solids with Diverse Microstructure, *Intl. J. Solids & Structures*, **14**, 795-819, 1978.
- 97. CLEARY, MICHAEL P., Moving Singularities in Elasto-Diffusive Solids with Applications to Fracture Propagation, *Intl. J. Solids & Structures*, **14**, 81-97, 1978.
- 98. CLEARY, MICHAEL P., Rate and Structure Sensitivity in Hydraulic Fracturing of Fluid-Saturated Porous Formations, 20th U. S. Symposium on Rock Mechanics, , 127-142, 1979.
- 99. CORNET, F. H., AND C. FAIRHURST, Influence of Pore Pressure on the Deformation Behavior of Saturated Rocks, *Proc. 3rd Int. Congress of the Int. Soc. for Rock Mech. Denver Colo. (Nat. Acad. Of Sci.)*, **2A**, 638-644, 1974.
- 100. CRAINE, R. E., Oscillations of a Plate in a Binary Mixture of Incompressible Newtonian Fluids, *Intl. J. Engr. Sci.*, **9**, 1177-1192, 1971.
- 101. CRAINE, R. E., An Oscillating Plate in a Binary Mixture of Hemihedral Fluids, *ZAMP*, **24**, 365-376, 1973.
- 102. CRYER, C. W., A Comparison of the Three-Dimensional Consolidation Theories of Biot and Terzaghi, *Quart. J. Mech. Appl. Math.*, **16**, 401-412, 1963.

- 103. DAKSHANAMURTHY, V., AND D. G. FREDLUND, A Mathematical Model for Predicting Moisture Flow in an Unsaturated Soil Under hydraulic and Temperature Gradients, *Water Resources Research*, **17**, 714-722, 1981.
- 104. DARGUSH, G. F., AND P. K. BANERJEE, A New Boundary Element Method for Three-Dimensional Coupled Problems of Consolidation and Thermoelasticity, *J. Appl. Mech.*, 58, 28-35, 1991.
- 105. DAVIDSON, L., Shock-Wave Structure in Porous Solids, *J. Appl. Phys.*, **42**, 5503-5512, 1975.
- 106. DAVIS, E. H., AND G. P. RAYMOND, A Non-Linear Theory of Consolidation, *Geotechnique*, **15**, 161-173, 1977.
- 107. DE BOER, R., AND W. EHLERS A Historical View of Porous Media Theories, *Acta Mech.*, **74**, 38360, 1988.
- 108. DE BOER, R., AND W. EHLERS Uplift Friction and Capillarity: Three Fundamental Effects for Liquid-Saturated Porous Solids, *Intl. J. Solids & Structures*, **26**, 43-57, 1990.
- 109. DE BOER, R., W. EHLERS, AND Z. LIU, One-Dimensional Transient Wave Propagation in Fluid-Saturated Incompressible Porous Media, Arch. Of Appl. Mech., 63, 59-72, 1993.
- 110. DE JOSSELING DE JONG, G., A Capacitive Cell Apparatus, *Proceedings of the Fourth International Conference on Soil Mechanics and Foundation Engineering*, **1**, 52-55, 1957.
- 111. DERESIEWICA, H., The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: I. Reflection of Plane Waves at a Free Plane Boundary (Non-Dissipative Case), *Bull. Seismological Soc. America*, **50**, 599-607, 1960.
- 112. DERESIEWICA, H., The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid II. Love Waves in a Porous Layer, *Bull. Seismological Soc. America*, 51, 51-59, 1961.
- 113. DERESIEWICA, H., The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: IV. Surface Waves in a Half-Space, *Bull. Seismological Soc. America*, 52, 627-638, 1962.
- 114. DERESIEWICA, H., AND A. LEVY, The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: X: Transmission Through a Stratified Medium, *Bull. Seismological Soc. America*, **57**, 381-391, 1967.
- 115. DERESIEWICA, H., AND J. R. RICE, The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: III. Reflection of Plane Waves at a Free Plane Boundary (General Case), *Bull. Seismological Soc. America*, **52**, 595-625, 1962.
- 116. DERESIEWICA, H., AND J. R. RICE, The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: V. Transmission Across a Plane Interface, *Bull. Seismological Soc. America*, **54**, 409-416, 1964.
- 117. DERESIEWICA, H., AND J. R. RICE, The Effect of Boundaries on Wave Propagation in a Liquid-Filled Porous Solid: VIII. Reflection of Plane Waves at an Irregular Boundary, *Bull. Seismological Soc. America*, **54**, 1537-1561, 1964.
- 118. DERESIEWICA, H., AND R. SKALAK, On Uniqueness in Dynamic Poroelasticity, *Bull. Seismological Soc. America*, **53**, 783-788, 1963.
- 119. DERSKI, W., Equations of Motion for a Fluid-saturated Porous Solid, *Bulletin de L'Academie Polonaise des Sciences Serie des Sciences*, **26**, 38672, 1978.

- 120. DERSKI, W., AND S. J. KOWALSKI, Equations of Linear Thermoconsolidation, *Arch.* of Mech., **31**, 303-316, 1979.
- 121. DETOURNAY, E., AND A. H-D. CHENG, Poroelastic Solution of a Plane Strain Point Displacement Discontinuity, J. Appl. Mech., 54, 783-787, 1987.
- 122. DETOURNAY, E., AND A. H-D. CHENG, Fundamentals of Poroelasticity in Comprehensive Rock Engineering: Principles Practice and Projects Vol. II Analysis and Design Method ed. D. Fairhurst, *Pergman Press*, , , 1993.
- 123. DEWIEST, ROGER J. M. (ED.) Flow Through Porous Media, *Academic Press*, New York, N.Y., 1969.
- 124. DIGBY, P. J., The Effective Elastic Moduli of Porous Granular Rocks, J. Appl. Mech., 48, 803-808, 1981.
- 125. DINULESCU, H. A., AND E. R. G. ECKERT, Analysis of the One-Dimensional Moisture Migration Caused by Temperature Gradients in a Porous Medium, *Intl. J. Heat and Mass Transfer*, **23**, 1069-1078, 1980.
- 126. DOMINGUIEZ, JOSE, An Integral Formulation for Dynamic Poroelasticity, *J. Appl. Mech.*, **58**, 588-591, 1991.
- 127. DREW, D. A., Mathematical Modeling of Two-Phase Flow, *Ann. Rev. Fluid Mech.*, **15**, 261-291, 1983.
- 128. DRUMHELLER, DOUGLAS S., The Theoretical Treatment of a Porous Solid Using a Mixture Theory, *Intl. J. Solids & Structures*, **14**, 441-456, 1978.
- 129. DULLIEN, R. A. L., Porous Media, Fluid Transport and Pore Structure, *Academic Press*, **New York**, N.Y., 1979.
- 130. DZIECIELAK, R., On Determination of the Constants of a Consolidating Medium, *Acta Technica Academiac Scientarium Hungaricae*, **73**, 433-439, 1972.
- 131. ECKERT, E. R. G., AND M. FAGHRI, A General Analysis of Moisture Migration Caused by Temperature Differences in an Unsaturated Porous Medium, *Intl. J. Heat and Mass Transfer*, **23**, 1613-1623, 1980.
- 132. EHLERS, W. Zur Entwicklung Konstitutive Gleichungen fur Porose Medien, *ZAMM*, **68**, 170-171, 1988.
- 133. EHLERS, W. On Thermodynamics of Elasto-Plastic Porous Media, *Arch. of Mech.*, **???**, ???, 1989.
- 134. EHLERS, W. Ein Elastizitatsgesetz fur Porose Medien, ZAMM, 69, 472-473, 1989.
- 135. FATT, I., The Biot-Willis Elastic Coefficients for a Sandstone, J. Appl. Mech., 26, 296-297, 1959.
- 136. FICHERA, G., Uniqueness Existence and Estimate of the Solution in the Dynamical Problem of Thermodiffusion in an Elastic Solid, *Arch. of Mech.*, **26**, 903-920, 1974.
- 137. FRENKEL, J., On the Theory of Seismic and Seismoelectric Phenomena in a Moist Soil, *J. of Physics*, **8**, 230-241, 1944.
- 138. FREUDENTHAL, A. M., AND W. R. SPILLERS, Solutions for the infinite Layer and the Half-Space for Quasi-Static Consolidating Elastic and Viscoelastic Media, *J. Appl. Phys.*, **33**, 2661-2668, 1962.
- 139. FUKUO, Y., Visco-Elastic Theory of the Deformation of a Confined Aquifer, Proceedings of the Tokyo Symposium on Land Subsidence, , , 1969.
- 140. GANDHI, M. V., AND USMAN, M., Exact Solutions for the Uniaxial Extension of a Mixture Slab, *Arch. of Mech.*, **23**, 1988.

- 141. GANDHI, M. V., M. USMAN, A. S. WINEMAN, AND K. R. RAJAGOPAL Combined Extension and Torsion of a Swollen Cylinder Within the Context of Mixture Theory, *Acta Mech.*, **79**, 81-93, 1989.
- 142. GANDHI, M. V., RAJAGOPAL, K. R., AND WINEMAN, A. S. A Universal Relation in Torsion for a Mixture of Solid and Fluid, *Journal of Elasticity*, **15**, 155-166, 1985.
- 143. GARG, S. K., D. H. BROWNELL, JR., J. W. PRITCHETT, AND R. G. HERMANN, Shock-Wave Propagation in Fluid-Saturated Porous Media, J. Appl. Phys., 46, 702-713, 1975.
- 144. GARG. S. K., Wave Propagation Effects in a Fluid-Saturated Porous Solid., J. *Geophysical Res.*, **76**, 7947-7962, 1971.
- 145. GARG. S. K., A. H. NAYFEH, AND A. J. GOOD, Compressional Waves in Fluid-Saturated Elastic Porous Media, J. Appl. Phys., 45, 1968-1974, 1974.
- 146. GEERTSMA, J., A Remark on the Analogy Between Thermoelasticity and the Elasticity of Saturated Porous Media, *J. Mech. Phys. Solids*, **6**, 13-16, 1957.
- 147. GEERTSMA, J., The Effect of Fluid Pressure Decline on Volumetric Changes of Porous Rocks, *Trans. A. S. M. E.*, **210**, 331-340, 1957.
- 148. GEERTSMA, J., AND D. C. SMIT, Some Aspects of Elastic Wave Propagation in Fluid-Saturated Porous Solids, *Geophysics*, **26**, 169-181, 1961.
- 149. GHABOUSSI, J., AND E. L. WILSON, Variational Formulation of Dynamics of Fluid-Saturated Porous Elastic Solids, J. Engr. Mech. Div. ASCE, **98**, 947-962, 1972.
- 150. GHABOUSSI, J., AND E. L. WILSON, Flow of Compressible Fluid in Porous Elastic Media, *Intl J. Numerical Methods in Engr.*, **5**, 419-442, 1973.
- 151. GHABOUSSI, J., AND E. L. WILSON, Seismic Analysis of Earth Dam-Reservoir Systems, J. Engr. Mech. Div. ASCE, 99, 849-862, 1973.
- 152. GIBSON, R. E., AND JOHN MCNAMEE, A Three-Dimensional Problem of the Consolidation of a Semi-Infinite Clay Stratum, *Quart. J. Mech. Appl. Math.*, **16**, 115-127, 1963.
- 153. GIBSON, R. E., G. L. ENGLAND AND M. J. L. HUSSEY, The Theory of One Dimensional Consolidation of Saturated Clays, *Geotechnique*, **17**, 261-273, 1967.
- 154. GILDING, B. H., Properties of Solutions of an Equation in the Theory of Infiltration, *Arch. Rational Mech. Anal.*, **65**, 203-225, 1977.
- 155. GILDING, B. H., AND L. A. PELETIER, The Cauchy Problem for an Equation in the Theory of Infiltration, *Arch. Rational Mech. Anal.*, **61**, 127-140, 1976.
- 156. GRAY, W. G., AND K. O'NEILL, On the General Equations for Flow in Porous Media and Their Reduction to Darcy's Law, *Water Resources Research*, **12**, 148-154, 1976.
- 157. GRAY, W. G., AND P. C. Y. LEE, On the Theorems for Local Volume Averaging of Multiphase Systems, *Intl. J. Multiphase Flow*, **3**, 333-340, 1977.
- 158. GREEN, A. E. AND P. M. NAGHDI, A Dynamical Theory of Interacting Continua, *Intl. J. Engr. Sci.*, **3**, 231-241, 1965.
- 159. GREEN, A. E. AND P. M. NAGHDI, On the Basic Equations of Mixtures, Q. J. Mech and Appl. Math., 22, 427-438, 1969.
- 160. GREEN, A. E. AND T. STEEL, Constitutive Equations for Interactive Continua, *Intl. J. Engr. Sci.*, **4**, 483-500, 1966.
- 161. GURTIN, M. E., The Linear Theory of Elasticity Handbuch der Physik Vol VIa/2 (ed. C. Truesdell), *Springer Verlag*, **Berlin**, Germany, 1972.

- 162. HALPERN, MARC R. Dynamic Response of a Saturated Poroelastic Halfspace to Imposed Surface Loadings, *PhD Dissertation*, **Carnegie Mellon University**, 1982.
- 163. HASSANIZADEH, A. MAJID, Derivation of Basic Equation of Mass Transport in Porous Media, Part 1, Macroscopic Balance Laws, *Adv. Water Resources*, **9**, 196-206, 1986.
- HASSANIZADEH, A. MAJID, Derivation of Basic Equations of Mass Transport in Porous Media, Part 2, Generalized Darcy's and Fick's Law, *Adv. Water Resources*, 9, 209-222, 1986.
- 165. HELFFERICH, FRIEDRICH G., Theory of Multicomponent Multiphase Displacement in Porous Media, *Soc. Pet. Engr. J.*, **271**, 51-62, 1981.
- 166. HERMANN, W., Constitutive Equation for the Dynamic Compaction of Ductile Porous Materials, *J. Appl. Mech.*, **40**, 2490-2499, 1969.
- 167. HOVEM, J. M., AND G. D. INGRAM, Viscous Attenuation of Sound in Saturated Sand, J. Acoust. Soc. Am., 66, 1807-1812, 1979.
- 168. HUBBERT, M. KING, The Theory of Ground-Water Motion, J. of Geology, 48, 785-944, 1940.
- 169. HUBBERT, M. KING, Darcy's Law and the Field Equations of the Flow of Underground Fluids, *Petroleum Trans.AIME*, **207**, 222-239, 1956.
- 170. ISHIHARA, KENJI, Propagation of Compressional Waves in a Saturated Soil International Symposium on Wave Propagation and Dynamics of Earth Materials Albuquerque NM, , , , 1967.
- 171. JANA, R. N., Deformation In An Infinite Poroelastic Medium With A Long Cylindrical Hole, *Quart. J. Mech. Appl. Math.*, **16**, 137-148, 1963.
- 172. JENKINS, J. T., Static Equilibrium of a Fluid-Saturated Porous Solid, *J. Appl. Mech.*, **47**, 493-495, 1980.
- 173. JOHNSON, D. L., Equivalence Between Fourth Sound in Liquid He II at Low Temperatures and the Biot Slow Wave in Consolidated Porous Media, *Appl. Phys. Lett.*, **37**, 1065-1067, 1980.
- 174. JOHNSON, D. L., AND T. J. PLONA, Acoustical Slow Waves and the Consolidation Transition, *J. Acoust. Soc. Am.*, **72**, 556-565, 1982.
- 175. JONES, JOHN P., Raleigh Waves in a Porous Elastic Saturated Solid, J. Acoust. Soc. Am., **33**, 959-962, 1961.
- 176. JONES, JOHN P., Pulse Propagation in a Poroelastic Solid, J. Appl. Mech, **91**, 878-880, 1969.
- 177. KAASSCHIETER, E. F., AND A. J. H. FRIJNS, Squeezing a sponge: a Three-Dimensional Solution in Poroelastity, *Computational Geosci.*, **7**, 49-59, 2003.
- 178. KARAKIN, A. V., AND L. I. LOBKOVSKII, Mechanics of Porous Two-Phase Visco-Deformed Medium and its Geophysical Applications, *Lett. Appl. Engr. Sci.*, **17**, 797-805, 1979.
- 179. KASACHENSKII, L., On the Propagation of Elastic Waves in Two Phase Media, *PPM*, **23**, 1593-1604, 1959.
- 180. KASACHENSKII, L., On the Propagation of Elastic Waves in a Layered Two Phase Medium, J. Appl. Math. And Mech., 25, 1608-1617, 1961.
- 181. KATSUBE, N., The Constitutive Theory for Fluid-Filled Porous Materials, J. Appl. Mech., 52, 185-189, 1985.
- 182. KATSUBE, N., Purely Mechanical Constitutive Theory for Fluid-Filled Porous Materials with Broken Outer Boundaries, *Acta Mech.*, **70**, 83-96, 1987.
- 183. KATSUBE, N., The Anisotropic Thermomechanical Constitutive Theory for A Fluid-Filled Porous Material with Solid/Fluid Outer Boundary, *Intl. J. Solids & Structures*, **24**, 375-380, 1988.
- 184. KATSUBE, N., The Steady State Fluid Flow Through A Porous Material Between Two Infinite Plates Maintained at Different Temperatures, *Intl. J. Engr. Sci.*, **26**, 253-266, 1988.
- 185. KATSUBE, N., The One-Temperature Field Thermo-Mechanical Theory for Fluid-Filled Porous Materials, *Acta Mech.*, **80**, 287-298, 1989.
- 186. KATSUBE, N., AND M. M. CARROLL The Modified Mixture Theory for Fluid-Filled Porous Materials, Theory, J. Appl. Mech., 54, 35-40, 1987.
- 187. KATSUBE, N., AND M. M. CARROLL The Modified Mixture Theory for Fluid-Filled Porous Materials, Applications, J. Appl. Mech., **54**, 41-46, 1987.
- 188. KELLY, P., A Reacting Continuum, Intl. J. Engr. Sci., 2, 129-153, 1964.
- 189. KENYON, D. E., Thermostatics of Solid-Fluid Fluid Mixtures, *Arch. Rational Mech. Anal.*, **62**, 117-129, 1976.
- 190. KENYON, D. E., The Theory of an Incompressible Solid-Fluid Mixture, *Arch. Rational Mech. Anal.*, **62**, 131-147, 1976.
- 191. KENYON, D. E., Consolidation in Transversely Isotropic Solids, J. Appl. Mech., 46, 65-70, 1979.
- 192. KHALED, M. Y., D. E. BESKOS, AND E. C. AIFANTIS, On the Theory of Consolidation with Double Porosity-III A Finite Element Formulation, *Intl J. Numerical and Analytical Methods in Geomechanics*, **8**, 101-123, 1984.
- 193. KIM, Y. K., AND H. B. KINGSBURY, Dynamic Characterization of Poroelastic Materials, *Experimental Mechanics*, **19**, 252-258, 1979.
- 194. KNOPS, R. J., AND T. R. STEEL, Uniqueness in the Linear Theory of a Mixture of Two Elastic Solids, *Intl. J. Engr. Sci.*, **7**, 571-577, 1969.
- 195. KNOPS, R. J., AND T. R. STEEL, On the Stability of a Mixture of Two Elastic Solids, *J. Compos. Mater.*, **3**, 652-663, 1969.
- 196. KOWACKI, W., Certain Problems of Thermodiffusion in Solids, Arch. of Mech., 23, 731-755, 1971.
- 197. KOWACKI, W., Dynamic Problems of Diffusion in Solids, *Engr. Fracture Mech.*, **8**, 261-266, 1976.
- 198. KOWALSKI, S. J., Application of Fluid-Saturated Porous Solids to Vibration Isolation Problems, *J. of Sound and Vibrations*, **66**, 577-587, 1979.
- 199. KOWALSKI. S. J., Comparison of the Biot Equations of Motion for a Fluid-Saturated Porous Solid with Those of Derski, *Bull. de L'Academie Polonaise des Sciences*, 27, 455-461, 1979.
- 200. KOWALSKI. S. J., Boundary Conditions in the Theory of Consolidation, *Studia Geotechnica et Mechanica*, **2**, 25-34, 1980.
- 201. KUBIK, J., On Internal Coupling in Dynamic Equations of Fluid-Saturated Porous Solid, *Intl. J. Engr. Sci.*, **24**, 981-990, 1986.
- 202. KUBIK, J., A Macroscopic Description of Geometrical Pore Structure of Porous Solids, *Intl. J. Engr. Sci.*, **24**, 971-980, 1986.

- 203. LAWS, N., AND J. L. PARRY Mathematical Modeling of Heat and Mass Transfer in Agricultural Grain Drying, *Proc. R. Soc. Lond. A*, **385**, 169-187, 1983.
- 204. LEWIS, R. W., AND B. SCHREFLER, A Fully Coupled Consolidation Model of the Subsidence of Venice, *Water Resources Research*, **14**, 223-230, 1978.
- 205. LEWIS, R. W., K. MORGAN AND H. R. THOMAS, Drying-Induced Stresses in Porous Bodies-An Elastoviscoplastic Model, *Computational Methods in Appl. Mech. and Engr.*, **20**, 291-301, 1979.
- 206. LIU, I-SHIH, On Chemical Potential and Incompressible Porous Media, *J. de Mechanique*, **19**, 327-342, 1980.
- 207. LIU, I-SHIH, On Fluid Pressure and Buoyancy Force in Porous Media, *Revisita Brasileira Tecnologia*, **11**, 35-43, 1980.
- 208. LUBINSKI, ARTHUR, The Theory of Elasticity for Porous Bodies Displaying a Strong Pore Structure, *Proc. 2nd U. S. Nat. Congr. Appl. Mech.*, , 247-256, 1954.
- 209. MANDEL, J., Consolidation des Sols(Etude Mathematique), *Geotechnique*, **3**, 287-299, 1953.
- MANDEL, J., Consolidation des Couches d'Argiles Proc. 4th Intl. Conf. on Soil Mech. and Foundation Engr., *Butterworth Scientific Publications*, 1, 360-367, 1957.
- 211. MANINS, P. C., AND B. W. ROBERTS, Compression of an Elastoporous Medium, *Intl. J. Non-Linear Mech.*, **13**, 75-93, 1975.
- 212. MARBLE, FRANK E., Dynamics of Dusty Gases, *Rev. of Fluid Mechanics*, **2**, 397-446, 1970.
- 213. MCNABB, A., A Mathematical Treatment of One-Dimensional Soil Consolidation, *Quart. Appl. Math.*, **17**, 337-347, 1960.
- MCNAMEE, JOHN, AND R. E. GIBSON, Plane Strain and Axially Symmetric Problems of the Consolidation of a Semi-Infinite Clay Stratum, *Quart. J. Mech. Appl. Math.*, 13, 210-227, 1960.
- 215. MCNAMEE, JOHN, AND R. E. GIBSON, Displacement Functions and Linear Transforms Applied to Diffusion Through Porous Elastic Media, *Quart. J. Mech. Appl. Math.*, **13**, 89-111, 1960.
- 216. McTIGUE, R. K. WILSON, AND J. W. NUNZIATO, An Effective Stress Principle for Partially Saturated Media, *Sandia Report SAND82-1977*, , , 1984.
- 217. MEI, CHIANG C., AND MOSTAFA A. FODA, Wave-Induced Responses in a Fluid-Filled Poro-Elastic Solid with a Free Surface-a Boundary Layer Theory, *Geophys. J. R. Astr. Soc.*, **66**, 597-631, 1981.
- 218. MEI, CHIANG C., AND MOSTAFA A. FODA, Wave-Induced Stresses Around a Pipe Laid on a Poro-Elastic Sea Bed, *Geotechnique*, **31**, 509-517, 1981.
- 219. MICHAEL, D. H., AND D. A. MILLER, Plane Parallel Flow of a Dusty Gas, *Mathematika*, **13**, 97-100, 1966.
- 220. MILLS, N., Incompressible Mixtures of Newtonian Fluids, *Intl. J. Engr. Sci.*, **4**, 97-112, 1966.
- 221. MONTE, J. L., AND R. J. KRIZEK, One-Dimensional Mathematical Model for Large-Strain Consolidation, *Geotechnique*, **26**, 495-510, 1976.
- 222. MORLAND, L. W., A Simple Constitutive Theory for a Fluid-Saturated Porous Solid, *J. Geophysical Res.*, **74**, 890-900, 1972.
- 223. MORSE, R. W., Acoustical Propagation in Granular Media, J. Acoust. Soc. Am., 24, 696-700, 1952.

- 224. MULLER, I., A Thermodynamic Theory of Mixtures of Fluids, *Arch. Rational Mech. Anal.*, **28**, 1-39, 1968.
- 225. MURRAY, J. D., On the Mathematics of Fluidization Part. 2 Steady Motion of Fully Developed Bubbles, *J. Fluid Mech.*, **22**, 57-80, 1965.
- 226. MURRAY, J. D., On the Mathematics of Fluidization Part. 1 Fundamental Equations and Wave Propagation, *J. Fluid Mech.*, **21**, 465-493, 1965.
- 227. NIGMATULIN, R. I., Spatial Averaging in the Mechanics of Heterogeneous and Dispersed Systems, *Intl. J. Multiphase Flow*, **5**, 353-385, 1979.
- 228. NOWACKI, W., Certain Problems of Thermodiffusion in Solids, Arch. of Mech., 23, 731-755, 1971.
- 229. NOWACKI, W., Dynamic Problems of Diffusion in Solids, *Engr. Fracture Mech.*, **8**, 261-266, 1976.
- 230. NOWINSKI, J. L., A Model of the Human Skull as a Poroelastic Spherical Shell Subjected to a Sudden Applied Load, *Math. Biosci.*, **8**, 397-416, 1970.
- 231. NOWINSKI, J. L., Bone Articulations as Systems of Poroelastic Bodies in Contact, *AIAA J.*, **9**, 62-67, 1971.
- 232. NOWINSKI, J. L., Propagation of Longitudinal Waves in Circularly Cylindrical Bone Elements, *J. Appl. Mech.*, **38**, 578-584, 1971.
- 233. NOWINSKI, J. L., AND C. F. DAVIS, The Flexure and Torsion of Bones Viewed as Anisotropic Poroelastic Bodies, *Intl. J. Engr. Sci.*, **10**, 1063-1079, 1972.
- 234. NOWINSKI, J. L., The Flexure and Torsion of Cancellous Bones as Poroelastic MaterialsTech. Report No. 101 Dept of Mech. and Aerospace Eng., *U. of Delaware*, , , 1969.
- 235. NOWINSKI. J. L., Cylindrical Bones as Anisotropic Poroelastic Members Subjected to Hydrostatic Pressure Tech. Report No. 107 Dept of Mech. and Aerospace Engr., *U. of Delaware*, , , 1969.
- 236. NOWINSKI. J. L., Stress Concentration Around Holes in a Class of Rheological Materials Displaying a Poroelastic Structure in Develop. Mech., *Proc. Midwestern Conf. 12th*, , 445-458, 1971.
- 237. NUR, A., AND J. D. BYERLEE, An Exact Effective Stress Law for Elastic Deformation of Rock with Fluids, *J. Geophysical Res.*, **76**, 6414-6419, 1971.
- 238. PANDE, G. N., AND O. C. ZIENKIEWICZ, Soils Under Cyclic and Transient Loading, Proc. Int. Symp. Soils Under Cyclic and Transient Loading Swansea, , , 1980.
- 239. PARI, GUNADHAR, Axisymmetric Consolidation for a Porous Elastic Material Containing a Fluid, *J. Math. and Phys.*, **36**, 338-346, 1958.
- 240. PARI, GUNADHAR, Flow of Fluids through Porous Deformable Solids, *Appl. Mech. Reviews*, **16**, 421-423, 1963.
- 241. PARKER, K. H., MEHTA, R. V., AND C. G. CARO Steady Flow in Porous Elastically Deformable Materials, *J. Appl. Mech.*, **54**, 794-800, 1987.
- 242. PASCAL, H., Some Problems Related to the Quantitative Evaluation of Physical Properties of the Porous Medium from Flow Tests with Non-Newtonian Fluids, *Int. J. Engr. Sci.*, **23**, 307-317, 1985.
- 243. PASCAL, H., A Theoretical Analysis on Stability of a Moving Interface in a Porous Medium for Bingham Displacing Fluids and Its Application in Oil displacement Mechanism, *Canadian Journal of Chemical Engineering*, **64**, 375-379, 1986.

- 244. PASCAL, H., AND F. PASCAL Flow of Non-Newtonian Fluid Through Porous Media, *Intl. J. Engr. Sci.*, 23, 571-585, 1985.
- 245. PASSMAN, S. L., NUNZIATO, J. W., AND E. K. WALSH, A Theory of Multiphase Mixtures in Rational Thermodynamics by C. Truesdell 2nd Ed., *Springer-Verlag New York*, , 1984.
- 246. PAUL, S., On the Displacement Produced in a Porous Elastic Halfspace by an Impulsive Live Load (Non Dissipative Case), J. of Pure and Appl. Geophysics, 114, 605-614, 1976.
- 247. PAUL, S., On the Disturbance Produced in a Semi-Infinite Poroelastic Medium by a Surface Load, *J. of Pure and Appl. Geophysics*, **114**, 615-627, 1976.
- 248. PECKER, C., AND H. DERESIEWICZ, Thermal Effects on Wave Propagation in a Liquid-Filled Porous Media, *Acta Mech.*, **16**, 45-64, 1973.
- 249. PHILIP, J. R., Flow in Porous Media, Rev. of Fluid Mechanics, 2, 177-204, 1970.
- 250. PHILIP, J. R., AND D. A. DE VRIES, Moisture Movement in Porous Materials under Temperature Gradients, *Trans. Amer. Geophys. Union*, **38**, 222-232, 1957.
- 251. PIRSON, SYLVAIN J., Performance of Fractured Oil Reservoirs, Bull. Amer. Asso. Petroleum Geologists, **37**, 232-244, 1953.
- 252. PLONA, THOMAS J., Observation of a Second Bulk Compressional Wave in a Porous Medium at Ultrasonic Frequencies, *Appl. Phys. Lett.*, **36**, 259-261, 1980.
- 253. PLONA, THOMAS J., AND D. L. JOHNSON, Experimental Study of the Two Bulk Compression Modes in Water-Saturated Porous Structures, *Ultrasonic Symposium*, *IEEE*, New York, 868-872, 1980.
- 254. PODSTRIGACH, YA. S., Differential Equations of the Problem of Thermal Diffusion in a Solid Isotropic Body, *Dopovidi Adad. Nauk USUR*, **2**, 169-172, 1961.
- 255. PODSTRIGACH, YA. S., AND V. S. PAVLINA, Fundamental Equations of Plane Thermodiffusion Problems, *Prikladnaya Mekhanika*, **1**, 101-106, 1972.
- 256. POLUBARINOVA-KOCHINA, P. YA., Theory of Ground Water Movement, translated by J. M. Roger de Wiest, *Princeton Univ. Press*, **Princeton**, N. J., 1962.
- 257. PRANTIL, VINCENT C. AND PAUL R. DAWSON, Comparison of a Mixture Formulation with a Similarity Solution for Binary Alloy Solidification, *Intl. J. Engr. Sci.*, **25**, 1283-1304, 1987.
- 258. PREVOST, J. H., Consolidation of Anelastic Porous Media, J. Engr. Mech. Div. ASCE, 107, 169-186, 1981.
- 259. PREVOST, J. H., Nonlinear Transient Phenomena in Saturated Porous Media, *Computational Methods Appl. Mech. Engr.*, **20**, 38429, 1982.
- 260. RAICHENKO, L. M., The Problem of the influx of Liquid to a Perfect Well in a Layer of Fissured-Porous Rock in the Presence of an End-Face Zone (Trans. by Plenum Pub. Corp.), *Prikladnaya Mekhanika*, **9**, ?-95, 1973.
- 261. RAICHENKO, L. M., Flow Inflow to an Imperfect Borehole in an Inhomogeneous Media (Trans. By Plenum Pub. Corp.), *Prikladnaya Mekhanika*, **13**, 108-???, 1977.
- 262. RAJAGOPAL, K. R., WINEMAN, A. S., AND GANDHI, M. V., On Boundary Conditions for a Certain Class of Problems in Mixture Theory, *Intl. J. Engr. Sci.*, 24, 1453-1463, 1986.
- 263. RAJAPAKSE, R. K. N. D., AND T. SENJUNTICHAI, Fundamental Solutions for a Poroelastic Half-Space with Compressible Constituents, *J. Appl. Mech.*, **60**, 847-856, 1993.

- 264. RICE, JAMES R., On the Stability of Hardening for Saturated Rock Masses, J. *Geophysical Res.*, **80**, 1531-1536, 1975.
- 265. RICE, JAMES R., AND DONALD A. SIMONS, The Stabilization of Spreading Shear Faults by Coupled Deformation-Diffusion Effects in Fluid-Infiltrated Porous Materials, *J. Geophysical Res.*, **81**, 5322-5334, 1976.
- 266. RICE, JAMES R., AND MICHAEL P. CLEARY, Some Basic Stress Diffusion Solutions for Fluid-Filled Elastic Porous Media with Compressible Constituents, *Rev. of Geophysics and Space Physics*, **14**, 227-241, 1976.
- 267. RICE, JAMES R., J. W. RUDNICKI, AND D. A. SIMONS, Deformation of Spherical Cavities and Inclusions in Fluid-Infiltrated Elastic Materials, *Intl. J. Solids & Structures*, **14**, 289-303, 1978.
- 268. RUDNICKI, J. W., On "Fundamental Solutions for a Fluid-Saturated Porous Solid" By M. P. Cleary, *Int. J. Solids & Structures*, **17**, 855-857, 1981.
- 269. RUDNICKI, J. W., On Fundamental Solutions for a Fluid-Saturated Porous Solid, *Intl. J. Solids & Structures*, **17**, 855-857, 1981.
- 270. RUDNICKI, J. W., Plane Strain Dislocations in Linear Elastic Diffusive Solids, J. *Appl. Mech*, **54**, 545-552, 1987.
- 271. SANYAL, D. C., On the Two-Dimensional Deformation of a Semi-Infinite Porous Elastic Medium, *Applied Sci. Research*, **26**, 227-245, 1972.
- 272. SCHIFFMAN, R. L., The Stress Components of a Porous Medium, J. Geophysical Res., **75**, 4035-4038, 1970.
- 273. SCHIFFMAN, R. L., A Thermoelastic Theory of Consolidation, *Environmental and Geophys. Heat Transfer*, **4**, 78-81, 1971.
- 274. SCHIFFMAN, R. L., A Bibliography of Sedimentation and Consolidation, *NATO Conf.*, **U of Delaware**, June, 1984.
- 275. SCHOENBERG, M., Wave Propagation in a Finely Laminated Elastoacoustic Medium, *Appl. Phys. Lett.*, **42**, 350-352, 1982.
- 276. SCHOENBERG, M., Wave Propagation in Alternating Solid and Fluid Layers, *Wave Motion*, **6**, 303-320, 1984.
- 277. SCHOENBERG, M., AND P. N. SEN, Properties of a Periodically Stratified Acoustic Half-Space and its Relation to a Biot Fluid, *J. Acoust. Soc. Am.*, **73**, 61-67, 1983.
- 278. SETH, M. S., AND K. E. GRAY, Transient Stresses and Displacement Around a Wellbore Due to Fluid Flow in Transversely Isotropic Porous Media: I. Infinite Reservoirs, *Soc. Pet. Engr. J.*, **243**, 63-78, 1968.
- 279. SETH, M. S., AND K. E. GRAY, Transient Stresses and Displacement Around a Wellbore Due to Fluid Flow in Transversely Isotropic Porous Media:II. Finite Reservoirs, *Soc. Pet. Engr. J.*, **243**, 69-86, 1968.
- 280. SHI, J. J., WINEMAN, A. S., AND RAJAGOPAL, K. R., Applications of the Theory of Interacting Continua to the Diffusion of a Fluid Through a Nonlinear Elastic Medium, *Intl. J. Engr. Sci.*, 9, 871-889, 1981.
- 281. SKEMPTON, A. W., The Pore-Pressure Coefficients A and B, *Geotechnique*, **4**, 143-147, 1954.
- 282. STOLL, ROBERT D. AND GEORGE M. BRYAN, Wave Attenuation in Saturated Sediments, J. Acoust. Soc. Am., 47, 1440-1447, 1977.
- 283. STOLL, ROBERT D., Acoustic Waves in Ocean Sediments, *Geophysics*, **42**, 715-725, 1977.

- 284. STUHMILLER, J. H., The Influence of Interfacial Pressure Forces on the Character of Two-Phase Flow Model Equations, *Intl. J. Multiphase Flow*, **3**, 551-560, 1977.
- 285. TABADDOR, FARHAD, AND ROBERT W. LITTLE, Constitutive Equations for Mixtures of Newtonian Fluids and Viscoelastic Solids, *Dev. in Mech.*, **6**, 459-469, 1971.
- 286. TABADDOR, FARHAD, AND ROBERT W. LITTLE, Interacting Continuous Medium Composed of an Elastic Solid and an Incompressible Newtonian Fluid, *Intl. J. Solids & Structures*, **7**, 825-841, 1971.
- 287. TERZAGHI, K., Erdbaumechanik auf Bodenphysikalischer Grundlage, *Leipzig R. Deuticke*, , , 1925.
- 288. THOMPSON, M., AND J. R. WILLIS, A Reformation of the Equations of Anisotropic Poroelasticity, *J. Appl. Mech.*, **58**, 612-616, 1991.
- 289. TRUESDELL, C., Sulle Basi Della Thermomeccannica, *Rend. Lincei* (8), **22**, 33-38, 158-166, 1957.
- 290. TRUESDELL, C., Mechanical Basis of Diffusion, J. Chem. Phys., 37, 2336-2344, 1962.
- 291. TRUESDELL, C., AND R. TOUPIN The Classical Field Theories, Handbuch der Physik Vol III/1 (ed. S. Flugge), *Springer Verlag*, **Berlin**, Germany, 1962.
- 292. TWISS, ROBERT J. AND A. CEMAL ERINGEN, Theory of Mixtures for Micromorphic Materials-I Balance Laws, *Intl. J. Engr. Sci.*, **9**, 1019-1044, 1971.
- 293. TWISS, ROBERT J. AND A. CEMAL ERINGEN, Theory of Mixtures for Micromorphic Materials-II Elastic Constitutive Equations, *Intl. J. Engr. Sci.*, **10**, 437-465, 1972.
- 294. UNGER, DAVID J., AND ELIAS C. AIFANTIS, Completeness of Solutions in the Double Porosity Theory, *Acta Mech.*, **75**, 269-274, 1988.
- 295. VAN DUYN, AND L. A. PELETIER, Nonstationary Filtration in Partially Saturated Porous Media, *Arch. Rational Mech. Anal.*, **78**, 173-198, 1982.
- 296. VERRUIJT, A., The Completeness of Biot's Solution of the Coupled Thermoelasticity Problem, *Q. Appl. Math.*, **26**, 485-490, 1969.
- 297. WALTON, K., Elastic Wave Propagation in Model Sediments-I, *Geophysical Journal* of the Royal Astronomical Society, **48**, 461-478, 1977.
- 298. WALTON, K., The Effective Elastic Moduli of a Random Packing of Spheres, J. *Mech. Phys. Solids*, **35**, 213-226, 1987.
- 299. WALTON, K., Wave Propagation within Random Packings of Spheres, *Geophysical Journal of the Royal Astronomical Society*, **92**, 89-97, 1988.
- 300. WALTON, K., AND P. J. DIGBY Wave Propagation Through Fluid Saturated Porous Rocks, J. Appl. Mech., 54, 788-793, 1987.
- 301. WANG, HERBERT F., Theory of Linear Poroelasticity with Applications to Geomechanics and Hydrogeology, *Princeton Univ. Press*, **Princeton**, N. J., 2000.
- 302. WHITAKER, S., The Equations of Motion in Porous Media, *Chem. Engr. Sci.*, **21**, 291-300, 1966.
- 303. WHITAKER, S., Advances in Theory of Fluid Motion in Porous Media, *Ind. and Engr. Chemistry*, **61**, 14-28, 1969.
- 304. WHITAKER, S., The Transport Equations for Multi-Phase Systems, *Chem. Engr. Sci.*, **28**, 139-147, 1973.
- 305. WHITAKER, S., Simultaneous Heat Mass and Momentum Transfer in Porous Media, A Theory of Drying in Advances in Heat Transfer (ed. J. P. Hartnett and T.F. Irvine Jr.), *Academic Press New York*, **13**, 1977.

## INDEX

Acceleration, 7, 9, 18, 99, 101-107, 109, 111, 113-116, 118-121, 123, 125, 139 Acceleration wave, 104, 106-107, 109, 114, 116, 118, 120, 139 Acoustic, 85 Adjoint, 119 Amplitude, 107, 114-115, 118, 122-123 Appendix, 229-230, 232, 234, 236 Asymptotic, 198 Attenuation, 130, 137, 139, 143, 145 Attenuation coefficient, 130, 137, 139, 145 Balance, 4-6, 12-17, 19-21, 26, 103-104, 106, 155 Berea Sandstone, 96-97, 109, 129, 200, 233 Bibliography, 8 Binary, 45, 54, 68-69, 71-72, 87-88, 102, 125, 147 Biot, 17, 27, 52, 69, 85, 91-92, 94, 97-98, 120, 125, 127, 147, 149, 156, 171, 177, 186, 209, 211, 215 Body, 1, 9, 12, 14, 16, 18, 21, 26, 91 Boundary condition, 155, 157, 183, 207 Boundary layer, 71 Bulk, 1-3, 10, 12, 31, 55, 59 Bulk density, 1-3, 10, 59 Bulk modulus, 55 Bulk volume, 2-3 Bulk volume flux, 3 Capillary pressure, 6 Cartesian, 30, 111 Chapter, 1-2, 4, 6, 8-12, 14, 16-18, 20, 22, 24, 26, 28-30, 32, 34, 36, 40, 42, 44-46, 50, 52, 54, 57-58, 60, 62, 64, 66, 68, 70-72, 74, 76, 78, 80, 84, 86-88, 92, 94, 96, 98-100, 102-104, 106, 110, 112, 114, 116, 118, 120, 122, 126-128, 130, 132, 136, 138, 140, 144, 146, 148-149, 152, 156, 158, 160, 170, 172-174, 176, 178, 180, 182, 184, 186-190, 192, 196, 198, 206, 210, 214, 216, 218, 220, 222, 224, 226, 228 Characteristic, 40, 71, 119, 135, 144, 176, 179 Charcoal Granite, 96-97, 109, 120, 200-201, 232 Chemical potential, 20, 43, 45-46, 87

Components, 30, 107, 111 Compressible, 1, 5, 7, 16, 29, 43, 54, 58, 65, 68-69, 71, 85, 87, 97, 102 Concentration, 10 Conductivity, 36, 43, 57, 140, 144-146 Conductor, 42, 106, 122 Configuration, 1, 9, 16-17 Consolidation coefficient, 83, 218 Constants, 29, 32, 39-40, 43, 47, 63, 92, 95, 110, 112, 139, 143, 145 Constituent, 2-3, 7, 9-13, 15-17, 19-21, 26, 29-31, 43, 45, 55, 71 Constituents, 1, 11-12, 15-17, 21, 25, 29, 40, 43, 85, 97, 120 Constitutive, 4-7, 17, 19-20, 23, 25-26, 29, 31-32, 36, 40, 43, 45-46, 50, 62-65, 72, 87-88, 91, 93-94, 104, 122, 149, 151, 155-156, 187, 189 Content, 52, 171 Continuous, 9, 59, 101 Continuum, 8, 27, 32, 43, 103, 125 Darcy's law, 4-7, 17-18 Decay, 109, 123 Density of the mixture, 10 Derivation, 21, 49, 63, 72, 103, 113, 118, 120, 122, 156, 198 Derivative, 6-7, 10, 71-72, 76, 100-101 Derivatives, 10, 32, 53, 107, 111 Determinant, 109 Diagonal, 109 Differentiable, 152 Differential, 4-5, 78, 84, 99, 102, 111, 116, 122, 127, 130, 149, 151, 155-159, 187, 190, 214, 223 Differentiation, 17, 21, 101 Diffusion velocity, 11 Diffusion, 11-12, 25-27, 43, 69, 85, 109, 120, 176, 179 Diffusion force, 26 Dimension, 40, 152, 181 Dimensional, 69, 97, 99, 127, 135, 155-158, 171-172, 181-184, 186, 189, 205, 207, 209 Dimensionless, 173-174, 176, 179-180, 198, 218-219, 222 Dimensions, 3, 120

Discharge, 3 **Discontinuities**, 99 Discontinuity, 103 Dispersion, 130, 135 Dispersion relation, 130, 135 Dispersive, 140, 143, 145 Displacement, 6-7, 31, 60, 72, 74, 76, 78, 80, 84, 100-101, 155, 171-173, 175-176, 188, 214 Displacement derivative, 100-101 Distance, 174, 179, 218 Distribution, 5, 19 Divergence, 17, 72, 84 Drag, 18, 26, 36, 43, 57, 92, 97, 120 Drag coefficient, 18, 26, 57, 92, 97 Dusty, 8 Eigenfunction, 158, 190, 207 Eigenfunctions, 159, 183-184, 207 Eigenvalue, 109, 116, 118, 160 Eigenvalues, 109, 116, 160, 184 Eigenvector, 109, 116 Eigenvectors, 122-123, 161, 184 Elastic, 1, 6-7, 43, 57, 62, 97-98, 102, 125, 143, 147, 149, 155 Elasticity, 1, 6-7, 29, 31, 33, 35, 37, 39, 41, 43, 45, 47, 49-51, 53, 55, 57, 59-61, 63, 65, 69, 71, 73, 78, 83, 85, 87, 91-93, 95, 97, 187 Energy, 12-14, 19-20, 23, 34, 53, 73, 97, 106, 122 Energy supply, 13 Entropy, 19-23, 29, 36, 50, 54, 65, 71, 87-88 Entropy density, 19-20, 65 Equilibrium, 30-31, 40-41, 43, 57-59, 61-62, 155 Examples, 1, 16, 45, 104, 189 Experimental, 97-98 External body force density, 12, 14 Family of surfaces, 99 Filtration velocity, 3-4, 16, 18, 27 Finite, 27, 69 Flow, 8, 60, 155, 171 Fluid, 1-8, 16-19, 24-26, 40, 45, 47, 49-50, 52, 54-55, 57-62, 68, 71-74, 78, 84-85, 87-88, 92, 97, 102, 107, 109, 114, 125, 147,

149, 151, 155, 171, 174-176, 179, 209, 214, 219, 223 Flux, 3, 13, 17, 20, 30, 72 Force, 12, 14, 18-19, 26, 43, 155 Formulation, 4, 20, 60, 91, 94, 127, 149, 187 Fourier, 5 Frequencies, 135, 144-145, 218-219, 223 Frequency, 97, 125, 130, 135, 139-140, 143-147, 177-180, 214-215, 218, 222-223 Frozen, 40-41, 120 Frozen equilibrium, 41 Frozen isothermal wave speed, 120 Function, 2-3, 6, 9-10, 16, 25, 30, 45, 51, 100-101, 115, 155, 158-159, 180-184, 188, 190-191, 198, 205-207, 209 Fundamental, 29, 45, 95, 156 Gases, 8 General, 23, 29, 45, 69, 97, 186, 214 Geology, 8 Geomechanics, 98 Geometric, 1-2 Geometrical, 101 Geophysical, 27, 69 Geophysics, 85, 97 Governing, 15, 43, 54, 58, 71, 84, 87, 127, 149, 155, 187, 223 Gradient, 9-10, 30-31, 43, 122 Gradient of the deformation, 9 Gradients, 21, 104 Gravitational, 18, 96 Harmonic, 122, 127, 135, 137, 139, 143, 145, 147 Heat, 5, 13, 15, 20, 30, 37, 60, 72, 147 Heat fluxes, 15 Heat supply density, 13, 15 Heaviside, 209 Helmholtz, 19-20, 23 Homogeneous, 158-159, 190 Hydrogeology, 98 Identities, 170 Identity, 132, 151-152, 160, 170, 182 Immiscible, 11, 29, 31, 36, 45, 54, 60, 68 Impervious, 60, 155 Incompressibility, 12, 60, 62-63

Incompressible, 1, 4-6, 12, 15, 17, 21-23, 26-27, 29, 45, 60-66, 68-69, 80, 83-84, 87 Incompressible consolidation coefficient, 83 Incompressible misture, 21, 64, 87 Independent, 21, 43, 50, 53, 88 Indeterminacies, 19 Indeterminacy, 63 Indeterminant, 62 Indeterminate, 20, 63 Index, 29, 160 Indifference, 32 Inequalities, 103, 139, 144 Inequality, 19-23, 25-27, 29, 36, 39, 42, 55, 57-58, 73, 109, 122, 137, 144 Inertia, 71, 73-74, 78, 84, 149, 151, 155, 157, 159, 161, 163, 167, 169, 171, 173, 175, 177, 179, 181, 183, 187-189, 191, 193, 196, 199, 203, 205, 207, 209-211, 213-215, 217-219, 221-223, 225, 227 Infinite, 43, 140, 144-146 Infinitesimal, 6-8, 32 Infinitesimal strain tensor, 8, 32 Inhomogeneous, 159, 180 Initial, 4, 57-60, 69, 71, 74, 80, 109, 149, 151, 155, 157-161, 163, 167, 169-173, 175-177, 179-183, 187-193, 199, 203, 205-207, 209-211, 213-215, 217, 219, 221, 223, 225, 227 Inner part, 14-15, 23, 34, 53, 87 Inner part of the stress, 15, 87 Integrability, 78 Integrand, 59 Integrate, 16, 172 Inverse, 9-10, 36, 40, 163, 193, 198, 203, 209 Inversion, 31, 36, 46, 50 Invert, 163 Inverted, 31, 45, 100 Invertible, 49 Inviscid, 147 Irreversible, 27, 85 Isothermal, 20, 42, 78, 83-85, 91-92, 94, 120, 140, 149, 151, 187, 189 Isotropic, 4-7, 22, 26, 31-32, 43, 45, 102 Jump, 101-104, 106, 122

Kbar, 96-97, 173 Kinematic, 4, 22, 102, 112-113 Kinematic condition of compatibility, 102 Kinematics, 9, 99 Kronecker, 47 Lagrangian, 27 Lame' parameters, 7 Laplace, 163, 193, 203, 209 Limit, 40, 43, 60-61, 63-65, 68, 80, 83, 87, 139-140, 151 Linear, 1, 5-7, 9, 12-14, 17, 19, 21, 26, 43, 62, 85, 98-99, 102, 104, 125, 147, 151, 155, 181, 205 Linearization, 32, 34, 39 Linearized, 26, 31, 35-36, 39, 46-47, 52, 57, 62 Load, 179, 214 Longitudinal, 107, 115, 118, 120, 123 Magnitude, 116, 144 Magnitudes, 114 Maple, 135, 198 Mass, 2-6, 10-13, 15-17, 52, 223 Mass concentration, 10 Material forms, 16-17 Materials, 1, 5, 9, 19, 27, 43, 54, 84, 95, 97-98, 102, 123, 125, 144, 147, 218 Mathematical, 4-5, 21, 29, 60 Matlab, 135, 198 Matrix, 31, 39-40, 42, 49, 66, 109, 115-116, 119, 121, 128, 130, 132, 152, 156, 160-161, 163, 180-184, 189, 191, 193, 196, 205-207, 227 Maxwell's theorem, 102 Mean velocity, 10 Mechanical, 12, 97 Mechanics, 8, 27, 32, 43, 85, 97-98, 103, 125, 147 Media, 1, 3-5, 7-8, 22, 27, 43, 57, 69, 85, 97, 120 Medium, 27 Minor, 69 Miscible, 11 Mixture, 6-7, 9-16, 19-21, 23, 26, 29, 31, 34, 36-37, 40-43, 45, 47, 53, 60-61, 64, 87-88, 97, 102, 122, 125, 147

Mixtures, 8-9, 11-13, 15, 17, 19-23, 25, 27, 29, 31, 36, 43, 45, 60, 68-69, 91 Model, 4-7, 11, 20, 22-23, 29-30, 36, 40, 45-47, 57, 60, 63, 71, 74, 91-92, 127, 210, 214 Modes, 143-145 Modulus, 55, 92 Moment, 12-14, 71, 155 Momentum, 12-14, 17-19, 21, 26, 104, 155 Momentum supply, 12, 18, 21 Motion, 4, 6-8, 17, 19, 43, 60, 73, 76, 151 Muskat's law, 6 Negative, 26, 118, 137, 175-176, 199 Nonconductor, 43, 106, 123 Nondimensional, 71 Nondispersive, 143, 145 Nonlinear, 5, 27, 69 Nonsingular, 31 Nonsteady, 5 Nontrivial, 127 Nonzero, 27, 72, 140, 159-160 Normal, 60, 99-102, 107, 109, 114-115 Normal speed, 99-100, 109, 114 Normal trajectory, 101 Normalized, 116 Operator, 17, 151-152, 159, 181-184, 205-206, 209 Orthogonality, 161, 191-192 Oscillates, 214 Oscillation, 178-179 Parallel, 107, 109 Parameters, 7, 45, 97 Partial, 4-5, 12-13, 18, 32, 34, 78, 84, 99, 102, 127, 130, 149, 151, 155-159, 187, 190, 214, 223 Partial heat flux vector, 13 Partial internal energy density, 13 Partial stress tensor, 12 Permeability, 4-6, 92, 96 Perpendicular, 107 Perturbation, 71 Perturbations, 214 Pervious, 60, 171 Phase, 8, 27, 130, 137, 139-140, 143-147 Phase velocity, 130, 137, 139-140, 144-147 Plane acceleration wave, 109

Plane progressive wave, 130 Poisson, 92, 94 Poisson ratio, 92 Polynomial, 135, 196, 203 Polynomials, 198, 203 Pore, 4, 6-8, 18, 24, 45-47, 49-51, 53-55, 57-63, 65, 69, 71, 74, 78, 80, 84-85, 87-88, 92, 102, 155, 171, 177, 188 Pore pressure, 4, 6-7, 18, 45-47, 49, 51, 53-55, 57-63, 65, 69, 71, 74, 78, 84, 87-88, 92, 102, 155, 171, 177, 188 Pores, 5, 11, 19 Poroelastic, 97-98, 122, 144-146, 173, 176, 214, 218 Porosity, 2-3, 5, 7, 11, 16, 18-19, 22, 147 Porous, 1-5, 7-9, 22, 27, 29, 31, 33, 35, 37, 39, 41, 43, 45, 47, 49, 51, 53-55, 57, 59, 61, 63, 65, 69, 71, 73, 83, 85, 87, 91, 93, 95, 97, 102, 104, 120, 123, 125, 143, 147, 149, 155, 187 Porous elasticity, 1, 29, 31, 33, 35, 37, 39, 41, 43, 45, 47, 49, 51, 53, 55, 57, 59, 61, 63, 65, 69, 71, 73, 83, 85, 87, 91, 93, 95, 97, 187 Porous material, 1-3, 71, 102, 104, 123, 147 Potential, 18, 20, 43, 45-46, 51, 53, 87-88 Pressure, 4-7, 18, 45-47, 49, 51, 53-55, 57-63, 65, 69, 71, 74, 78, 84-85, 87-88, 92, 102, 155, 171, 177, 188, 214-215 Pressure head, 18 Progressive, 130, 143, 147 Propagates, 109 Properties, 6, 40-41, 87, 95-97, 99, 116, 125, 127, 151, 173 Property, 5, 16, 46, 92, 94, 96-97, 109, 120 Ouadratic, 42, 109 Range, 23, 29, 97, 125, 143-144, 147, 176, 218 Rate, 101 Ratio, 4, 18, 27, 92, 94, 135, 179-180, 203, 218-219 Reacting, 43 Reactions, 12, 27 Rectangular, 30 References, 9, 27, 29, 43, 69, 85, 87, 97, 147, 227

Relationships, 46, 51 Relative permeability, 6 Relaxation, 40, 46 Relaxing, 147 Representation, 118, 120, 161, 180, 184 Resonance, 223 Resources, 97 Response, 11, 29 Rheology, 8, 27, 69 Rigid, 1-6, 16-17, 19, 21-22, 26, 45, 60 Rock, 95-98 Root, 230-235 Roots, 121, 130, 137, 160, 196, 198-199, 203, 211, 218, 229, 231, 233, 235 Ruhr Sandstone, 96-97, 109, 120, 173, 176, 178, 200-202, 212, 218-219, 230 Sandstone, 96-98, 109, 120, 173, 176, 178, 218-219, 230, 233, 235 Sandstones, 97 Saturated, 85, 97, 125, 147 Scalar, 21, 116, 130 Scalars, 36 Scale, 214, 218 Scaling, 198 Semilinear, 27, 69 Shifting equilibrium, 40-41, 43, 61-62 Shock, 99, 123, 125 Shock Wave, 123 Singular, 71, 99, 101-105, 107, 109, 111, 113, 115, 119, 121, 123, 125, 214 Singular perturbation, 71 Skempton, 78, 85, 94 Solid, 1-8, 11-12, 16-19, 21-22, 26, 29, 31-33, 40, 43, 45, 49, 52-54, 57-60, 62, 68, 71, 73-74, 76, 78, 80, 84, 87-88, 97, 102, 107, 109, 125, 147, 149, 151, 171, 173-176, 179, 209, 214, 219, 223 Solids, 5, 8, 27, 69, 85, 109 Solution, 4, 31-34, 36, 43, 51, 60, 78, 83, 127, 130, 139, 149, 151, 159-161, 167, 170-173, 177-180, 182-184, 187, 189, 191-192, 206, 210-211, 215, 217-218, 222-223 Solutions, 5-6, 42, 59-60, 71, 74, 85, 97-99, 114, 127, 130, 143, 151, 159, 183, 198, 207, 223

Sound, 147 Spatial, 9-10, 12, 19, 152 Specific discharge, 3 Specific heat at constant volume, 37 Speed, 99-101, 109, 114, 116, 120 Speeds, 107, 109, 112, 120, 135, 139 Stability, 57-58 Static, 31-34, 36, 97 Static solution, 31-34, 36 Steady, 217 Strain, 7-8, 27, 32, 49, 59, 78, 84, 88 Stress, 12, 14-15, 18, 34, 40, 43, 49-50, 60, 71, 74, 76, 78, 84, 87, 155, 158, 171, 177, 188 Stressed, 27, 97 Surface, 12, 59-60, 99-104, 123, 155 Surfaces, 99-103, 105, 107, 109, 111, 113, 115, 119, 121, 123, 125 Symbol, 2-3, 10, 16, 93-94, 156-157, 160, 182, 193 Symbolic, 135, 198 Symbols, 93-94, 112 Symmetric, 14-15, 31, 39, 42, 49, 66, 109, 121, 132, 160, 182, 196, 206 Symmetry, 50, 116 Table, 95-97, 109, 120, 143, 218-219 Tables, 109, 120, 144, 173 Tabulated, 97, 173 Tangent, 101, 112 Tangential, 101, 115 Technology, 97 Temperature, 19-20, 29-31, 40, 43, 49, 59-60, 74, 78, 84, 88, 104, 107, 120, 122, 140, 151, 187 Temperature of the mixture, 20 Temperatures, 20 Tennessee Marble, 96-97, 109, 120, 200-201, 231 Terzaghi, 91, 97 Theorem, 39, 58, 60, 102, 196, 214 Thermal conductivity, 36, 43, 57, 145-146 Thermodynamic, 6, 26, 30-31, 33, 51, 63, 88, 115, 139 Thermoelasticity, 43, 85, 127, 147

xiii

Three Dimensional, 97, 99, 181-182, 186, 205 Traction, 60, 155 Trajectory, 101 Transfer, 60 Transformations, 87 Transient, 217 Transport, 8 Transverse, 107, 114-115, 123 Transverse wave, 107, 115 True density, 1-6, 11, 45, 96-97, 120 True volume, 3 True volume flux, 3 Unbounded, 63 Undeformed, 52-54, 87-88 Undeformed solid volume, 52-54, 88 Unique, 60, 74 Uniqueness, 39, 58, 60, 74 Unit normal, 60, 99-101, 107, 109, 114 Variational, 27 Vectors, 25, 36, 152 Velocities, 10-12, 21, 25, 31, 43, 135, 139, 143, 145 Velocity, 3-4, 6-7, 9-12, 16-18, 21-22, 25, 27, 59, 120, 130, 135, 137, 139-140, 144-147, 170, 176, 188, 210, 214 Velocity gradient, 10 Velocity of the mixture, 10 Vibrations, 147 Viscosity, 4-6, 92, 96-97 Visualized, 9 Volume, 1-3, 6-7, 11, 17-22, 29, 31, 37, 40, 52-54, 61-62, 87-88, 103, 155, 188 Volume fraction, 1-3, 6-7, 11, 18-19, 62 Volumes, 1-2, 11 Water, 8, 96-97, 109 Wave, 97, 104, 106-107, 109, 112, 114-116, 118, 120, 123, 127, 130, 135, 137, 139-140, 147 Wave number, 130, 135 Waves, 97, 99, 101-103, 105, 107, 109, 111, 113, 115, 119, 121-123, 125, 127, 130, 135, 137, 139, 143, 145, 147 Weber Sandstone, 96-97, 109, 120, 235 Weighted, 10

Well, 1, 63, 78, 101, 223 Westerly Granite, 96-97, 109, 120, 234 Young's modulus, 92 Zero mass, 223

- 306. WILSON, R. K., AND E. C. AIFANTIS, On the Theory of Consolidation with Double Porosity, *Intl. J. Engr. Sci.*, **20**, 1009-1035, 1982.
- 307. WYLLIE, M. R. J., G. H. F. GARDNER AND A. R. GREGORY, Studies of Elastic Wave Attenuation in Porous Media, *Geophysics*, 27, 569-589, 1962.
- 308. YAVARI, B., AND A. BEDFORD, The Biot Drag and Virtual Mass Coefficients for Face-centered Cubic Granular materials, *Int. J. Multiphase Flow*, **16**, 885-897, 1990.
- 309. YAVARI, B., AND A. BEDFORD, Comparison of Numerical Calculations of Two Biot Coefficients with Analytical Solutions, J. Acoust. Soc. Am., 90, 985-990, 1991.
- 310. YEW, C. H. AND P. N. JOGI, Study of Wave Motions in Fluid Saturated Porous Rocks, J. Acoust. Soc. Am., 60, 38391, 1976.
- 311. YEW, C. H. AND P. N. JOGI, The Determination of Biot's Parameters for Sandstones Part 1: Static Tests, *Experimental Mechanics*, **18**, 167-172, 1978.
- 312. YEW, C. H., P. N. JOGI, AND K. E. GRAY, Estimation of the Mechanical Properties of Fluid-Saturated Rocks Using the Measured Wave Motions, *J. Energy Resources Technology*, **101**, 112-116, 1979.
- 313. ZIENKIEWICZ, O. C., AND T. SHIOMI, Dynamic Behaviour of Saturated Porous Media; The Generalized Biot Formulation and its Numerical Solution, *Intl J. Numerical and Analytical Methods in Geomechanics*, **8**, 71-96, 1984.